EXPECTED UTILITY MAXIMIZATION FOR COMPETITIVE AGENTS

Zur Erlangung des akademischen Grades eines

DOKTORS DER NATURWISSENSCHAFTEN

von der KIT-Fakultät für Mathematik des Karlsruher Instituts für Technologie (KIT) genehmigte

DISSERTATION

von

Tamara Isabell Göll, M.Sc. aus Mosbach

Tag der mündlichen Prüfung: 17.11.2023

Referentin:

Prof. Dr. Nicole Bäuerle

Korreferent:

Prof. Dr. Frank Seifried

Meinen Eltern gewidmet – Für alles.

DANKSAGUNG

Es gibt einige Menschen, ohne deren Unterstützung die Umsetzung dieser Arbeit niemals möglich gewesen wäre. Ihnen möchte ich an dieser Stelle danken.

Allen voran möchte ich mich bei meiner Betreuerin, Prof. Dr. Nicole Bäuerle, bedanken. Bereits während meines Masterstudiums durfte ich sie als herausragende Dozentin und später als Betreuerin meiner Masterarbeit kennenlernen. Im Rahmen meiner Promotionszeit hat sie meine Erwartungen in jeglicher Hinsicht übertroffen. Auf fachlicher Ebene war sie jederzeit offen für Fragen und Diskussionen und hat sich immer viel Zeit genommen. Bei Problemen hat sie stets geduldig weitergeholfen und oft die richtigen Fragen gestellt. Auf persönlicher Ebene weiß ich ihre unkomplizierte Art sehr zu schätzen und wusste, dass ich bei Problemen immer auf sie zählen kann. Mir ist bewusst, dass all das nicht selbstverständlich ist. Danke, Nicole!

Zudem möchte ich mich bei Prof. Dr. Frank Seifried bedanken, der das Zweitgutachten für diese Arbeit übernommen hat.

In meiner Zeit am Institut für Stochastik habe ich schnell ein zweites Zuhause gefunden und viele der Menschen für immer ins Herz geschlossen. Es ist von unschätzbarem Wert, von Menschen umgeben zu sein, die gute Zeiten noch schöner und schlechte Zeiten leichter machen. Dafür bedanke ich mich bei allen aktuellen und ehemaligen Kolleginnen und Kollegen, die meine Promotionszeit zu etwas Unvergesslichem gemacht haben. Einigen möchte ich ganz persönlich danken. Lea danke ich dafür, dass sie mich bereits durch das Studium und nun auch durch die Promotion begleitet hat und stets ein offenes Ohr hatte – seien es fachliche Diskussionen oder private Angelegenheiten. Ebenso möchte ich mich bei Steffen bedanken, der mich einerseits unzählige Male zum Lachen bringen konnte, andererseits in schwierigen Zeiten immer da war – manchmal auch beides gleichzeitig. Insbesondere danke ich ihm dafür, dass er in den letzten Monaten vor der Abgabe nie müde wurde, meine Vielzahl an Fragen zu beantworten. Andy danke ich vor allem für unzählige Gespräche über mal mehr, mal weniger ernste Themen, die sich oftmals bis in die Dunkelheit gezogen haben und von denen ich keines missen möchte. Sebastian, meinem (großen) kleinen Bruder im akademischen Sinne, und Franz, meinem liebsten Kaffeebuddy, möchte ich vor allem für die gemeinsame Zeit danken. Für die hervorragende Gesellschaft an Wochenenden und Feiertagen im Büro bin ich außerdem Maximilian sehr dankbar. Bei Celeste möchte ich mich ganz besonders bedanken – für viele wichtige Ratschläge zu Beginn meiner Promotionszeit und dafür, dass sie mittlerweile eine wunderbare Freundin geworden ist.

Neben meiner Promotion habe ich große Freude an der Betreuung von Lehrveranstaltungen am Institut für Stochastik gefunden und möchte mich bei allen bedanken, mit denen ich in diesem Rahmen zusammenarbeiten durfte. Besonders hervorheben möchte ich die Zusammenarbeit mit Sebastian Lerch, dessen lobende Worte in der Anfangszeit eine große Stütze waren, und bei Benedikt, mit dem ich ein Semester voller Diskussionen über die Freuden und Leiden der Einführung in die Stochastik verbringen durfte.

Fernab der Stochastik gibt es noch einige Menschen, die mir immer unterstützend zur Seite standen. Julia möchte ich dafür danken, dass sie mich vom ersten Tag des Studiums an immer begleitet hat. Ohne sie wäre diese Zeit bei weitem nicht so schön gewesen. Außerdem möchte ich Lisa, meiner besten Freundin seit der ersten Klasse, danken – für alles. Besonders großer Dank gebührt nicht zuletzt meinem Freund Nicholas – dafür, dass er in allen guten und schlechten Phasen geduldig an meiner Seite stand und mich oftmals wieder zurück auf den Boden der Tatsachen geholt hat.

Und da das Beste bekanntlich zum Schluss kommt, stehen zwei besonders wichtige Menschen erst ganz am Ende – meine Eltern. Ohne sie wäre ich nicht der Mensch geworden, der ich heute bin. Sie haben immer an mich geglaubt, hatten stets ein offenes Ohr, haben Erfolge mit mir gefeiert, mir durch schwere Phasen geholfen und standen stets unterstützend an meiner Seite. Danke, Mama und Papa!

PRIOR PUBLICATIONS

Parts of Chapters 3, 4, 5, and 7 are direct quotes from the following prior publications:

- Bäuerle and Göll (2023a). Nash equilibria for relative investors via no-arbitrage arguments. Mathematical Methods of Operations Research 97: 1–23.
- Bäuerle and Göll (2023b). Nash equilibria for relative investors with (non)linear price impact. arxiv:2303.18161.

Contents

1.	Introduction		1
2.	Fundamentals		5
	2.1. Basic tools from stochastic calculus		5
	2.2. Definition of σ -martingales		7
	2.3. Basics from mathematical finance		9
	2.3.1. General semimartingale market		9
	2.3.2. A note on completeness	1	12
	2.3.3. Introduction to portfolio optimization	1	13
	2.4. Introduction to multi-objective optimization	1	15
2	Nach equilibria for relative investors	1	10
J.		1	19
	3.1. Relative performance problem under general utility	••• 4	23
	3.2. Solution method via problem reduction	2	25
4.	Application of the method from Section 3.2	2	29
	4.1. Lévy market	2	29
	4.2. Black-Scholes market	ŝ	31
	4.2.1. Optimization under exponential utility	🤅	31
	4.2.2. Optimization under power utility		36
	4.3. Binomial model in discrete time	🤅	37
	4.4. Market with stochastic volatility		38
	4.5. Cumulative prospect theory	3	39
5.	Mean field equilibria for relative investors	Δ	13
	5.1 Motivation and heuristic derivation of the mean field game	-	16
	5.2 Formal definition of the mean field game	•••	10
	5.2. Formai deminion of the mean neighbor reduction		±9
	5.5. Solution method via problem reduction	••••	50

6.	Par	eto optima for relative investors	55
	6.1.	$Problem \ formulation \ \ \ldots $	56
	6.2.	Scalarization of the objective function $\ldots \ldots \ldots$	58
	6.3.	Pareto optimum for the relative performance problem $\ldots \ldots \ldots \ldots \ldots \ldots$	65
		6.3.1. Maximization of the collective terminal wealth	65
		6.3.2. Pareto optimal terminal wealth of the individual investors \ldots \ldots \ldots	67
	6.4.	Comparison with the Nash equilibrium from Section 3.2	71
7.	\mathbf{Nas}	h equilibria for relative investors under (non)linear price impact	73
	7.1.	Introduction of the price impact market	76
	7.2.	Relative performance problem under general utility	79
	7.3.	Optimization under exponential utility	81
		7.3.1. Unique constant Nash equilibrium	81
		7.3.2. Influence of the price impact parameter	87
		7.3.3. Discussion of nonlinear price impact	92
	7.4.	Optimization under power utility	97
8.	Rela	ative performance via a VaR-type constraint	109
	8.1.	Problem formulation	111
	8.2.	Discussion of terminal wealth	112
	8.3.	Solution of the fixed point problem for two agents	115
	8.4.	Solution of the fixed point problem for n agents	123
Bi	bliog	graphy	133
Aj	ppen	ndix	142
A.	Add	litional material for Chapter 6	143
	A.1.	Application of the dominated convergence theorem in Lemma 6.6	143
	A.2.	Expectation of a normal and a lognormal random variable	144
в.	Add	litional material for Chapter 7	147
	B.1.	Supplements for the proof of Theorem 7.2	147
	B.2.	Supplements for the proof of Theorem 7.12	150
C.	Add	litional material for Chapter 8	155
	C.1.	Proofs of Lemma 8.8 and Proposition 8.9	155
	C.2.	Dimensional reduction of the fixed point problem	158
	C.3.	Solution of the fixed point problem for three agents	162
	C.4.	Proofs of lemmas from Section 8.4	170

CHAPTER 1

INTRODUCTION

The general idea behind portfolio optimization problems is the following: An investor, endowed with some fixed initial capital $x \in \mathbb{R}$, has to decide how many shares of which assets she should hold at which time in order to maximize her optimality criterion in terms of her wealth at some fixed time T > 0 (Korn, 1997, p. 1).

The most intuitive approach to defining such a problem is probably to maximize the expected value of the investor's terminal wealth. In general, however, this does not reflect an investor's attitude towards risk. Indeed, most investors prefer a sure return of 2 percent over some risky investment with an expected return of 2.2 percent (Eberlein and Kallsen, 2019, p. 461). In the early stages of portfolio optimization problems, there were two approaches to describe the preferences of investors. Von Neumann and Morgenstern (1947) proposed axioms of rational behavior and proved that an investor accepts and acts according to these axioms if, and only if, she measures her preferences in terms of the expectation of some utility function applied to her terminal wealth. A utility function is a strictly increasing, strictly concave function capturing that most investors are risk-averse and prefer larger amounts of money over smaller amounts. The second approach was established by Markowitz (1952). He used a mean-variance criterion, i.e., he considered investors trying to achieve a high expected return while keeping the variance of the portfolio small. More specifically, the expected return is maximized under the constraint that the variance is bounded by some constant. While Von Neumann and Morgenstern (1947) only considered gambles or lotteries, Markowitz (1952) was the first to consider a problem of optimal investment. At that time, the mean-variance approach was only suited for one-period problems (Korn, 1997, p. 1). The starting point for continuous-time portfolio optimization problems using expected utility was provided by Merton (1969). He considered stock prices in continuous time governed by geometric Brownian motions. He also showed that the problem of optimal investment (and consumption) can be reduced to solving the so-called Hamilton-Jacobi-Bellman equation, a nonlinear partial differential equation. Later, Karatzas et al. (1987) introduced a different method for solving continuous-time portfolio optimization problems – namely, the martingale approach.

Note that many authors, including Merton (1969) and Karatzas et al. (1987), consider optimal investment-consumption pairs and maximize the sum of cumulated (over time) utility from consumption and terminal utility of wealth. In this thesis, we are interested exclusively in optimal portfolios and do not consider the problem of optimal consumption any further.

Soon after its introduction, expected utility theory attracted substantial criticism. One famous example is the Allais paradox (Allais, 1953; Karmarkar, 1979). Further criticism followed from Kahneman and Tversky (1979) (see also Tversky and Kahneman, 1992), whose (cumulative) prospect theory presented an alternative to the classical terminal utility maximization. In spite of all criticism, expected utility maximization is still the most widely used objective for portfolio optimization. However, the method has since been further developed in many different directions. Let us mention a few examples of modifications of the original problem. One direction includes financial markets beyond geometric Brownian motion, including stochastic volatility, jumps, or more general semimartingale stock dynamics (see, for example, Kallsen and Muhle-Karbe, 2010, for utility maximization in a stochastic volatility model; Bäuerle and Blatter, 2011, for optimal investment and reinsurance in a jump-diffusion model; and Černy and Kallsen, 2007, for the mean-variance problem in a general semimartingale model). Other generalizations of the original problem include the use of adaptive utility functions (so-called forward utilities), additional constraints, or separate utility functions for gains and losses (see Musiela and Zariphopoulou, 2006, for the introduction of forward utilities; Basak and Shapiro, 2001, for an example of an additional constraint; and Tversky and Kahneman, 1992, for the cumulative prospect theory in which gains and losses are treated separately).

A relatively recent development in the field of portfolio optimization is the inclusion of competition between investors. This feature is motivated by the large variety of empirical evidence for competition between managers in the mutual and hedge fund industry. Indeed, Lacker and Soret (2020) state that a "particularly important and by now well-established point is that mutual fund choice is highly influenced by relative performance". Moreover, Bielagk et al. (2017) give the following intuitive justification of relative performance concerns: "Making a 1 Euro profit while everyone else made 2 Euro *feels* distinctly different had everyone lost 2 Euro". The empirical literature on relative performance is vast and we can only cover a fraction here. Some frequently mentioned articles are Chevalier and Ellison (1997) as well as Sirri and Tufano (1998) for competition between mutual fund managers, Agarwal et al. (2004) for competition between hedge fund managers, and Brown et al. (2001) as well as Kempf and Ruenzi (2008) for competition motivated by career motives. Further, we refer to the introductions of Basak and Makarov (2015) as well as Lacker and Zariphopoulou (2019) for a more detailed overview. Relative concerns can be motivated by different factors. On one hand, there are career concerns (Basak and Makarov, 2015) like promotion schedules for managers (Anthropelos et al., 2022) or peer-based underperformance penalties (Bielagk et al., 2017). On the other hand, companies are concerned with their reputation as well as their desire to attract new clients (Anthropelos et al., 2022) and to generate higher money inflows (Lacker and Zariphopoulou, 2019). Taking the argument a step further, Dos Reis and Platonov (2022) argue that "benchmarking is a feature of human nature".

The empirical evidence on competition between managers gave motivation to the new research area of *competitive portfolio optimization problems* in continuous time which has grown rapidly over the last decade. Adding to the introductory work of Espinosa (2010), Basak and Makarov (2014), and their many successors, this thesis is focused on portfolio optimization problems in continuous time for some (infinitely) large number of agents that base their decisions on relative performance concerns. In five main chapters, we take a look at such problems from various perspectives and on different levels of generality. The competitive feature is incorporated in three different ways – via the objective function, in a stochastic constraint, and via cumulative price impact. In each of these situations, we are able to find explicit solutions to the emerging multi-objective optimization problems applying two different notions of optimality.

Each of the main chapters (Chapters 3 and 5-8) includes a comprehensive introduction containing a literature overview of the corresponding research area and a general motivation of the problem. More precisely, this thesis is structured as follows. Chapter 2 contains basic tools and definitions from stochastic calculus, mathematical finance, and multi-objective optimization, which are used throughout this thesis. It also contains the specification of a very general semimartingale financial market which forms the basis of most problems considered in this thesis. In Chapter 3, we display a portfolio optimization problem for a finite number of competitive agents. We use the notion of competitive utility functions, which incorporate relative performance concerns into the classical portfolio optimization problem. Without any specific assumptions on the model and the utility functions, we explain a method for determining Nash equilibria for n-player problems by decomposing the multi-objective optimization into a single-agent problem and a system of linear equations, which turns out to be uniquely solvable. This method is applied to various examples in Chapter 4, including models with jumps or stochastic volatility, and the more generalized cumulative prospect theory. In the subsequent Chapter 5, we consider the corresponding mean field game to the *n*-agent game solved in Chapter 3. First, we motivate the problem by analyzing the Nash equilibrium in the limit as the number of agents tends to infinity. Afterwards, we define the mean field game properly and provide a solution method similar to the one displayed in Chapter 3. In Chapter 6, we take a look at a different notion of optimality, namely, Pareto optimality. In a general model, without specific restrictions on the financial market or the utility functions, we explain a suitable scalarization of the n-player game and use it to find a Pareto optimum in terms of terminal wealth. It turns out that the Pareto optimum coincides with the Nash equilibrium from Chapter 3. In the next Chapter 7, we consider a financial market which differs strongly from the previous chapters as it consists of agents whose investments influence the stock prices. Thus, the competition between the agents now originates from two different sources. One source remains the competitive utility function from Chapter 3. However, as the agents have a cumulative impact on the stock price, they undergo a different, more subtle, source of competition. In a linear price impact model, we are able to determine the unique constant Nash equilibrium for both exponential and power utility functions. In the case of exponential utility, we also consider nonlinear price impact and analyze whether the resulting optimization problem has a finite optimal solution. In the final Chapter 8, we apply the relative concerns through a stochastic constraint instead of the competitive utility function. The constraint is motivated by the value at risk based approach introduced by Basak and Shapiro (2001). In a general financial market, we find the optimal terminal wealth for a finite number of agents using general utility functions. Due to the complicated structure of the solution, we consider a different terminal wealth profile similar to the optimal solution found by Basak and Shapiro (2001). In this more restrictive class of wealth profiles, we find the unique Nash equilibrium in terms of terminal wealth and discuss some of its properties. Finally, Chapters A–C in the Appendix contain additional material for Chapters 6–8.

Last but not least, it should be noted that the terms *agent*, *player*, *investor*, and *trader* are used interchangeably throughout this thesis.

CHAPTER 2

FUNDAMENTALS

The following chapter contains some mathematical basics used throughout this thesis. We assume that the reader is familiar with basic topics in probability theory, mathematical finance, and stochastic optimization. More specifically, we expect a basic understanding of stochastic processes, Itô calculus, no-arbitrage pricing theory, and portfolio optimization. For the mentioned topics, we refer to Protter (2005) for a rigorous discussion of Itô calculus, to Karatzas and Shreve (1998) as well as Eberlein and Kallsen (2019) for a widespread overview of topics regarding mathematical finance, and to Pham (2009) for an introduction to stochastic optimization. Nevertheless, we state below some classical results which are used throughout this thesis.

The remainder of this chapter is based on a filtered probability space $(\Omega, \mathcal{F}, (\mathcal{F}_t)_{t \in [0,T]}, \mathbb{P})$, where T > 0 is some arbitrary but fixed finite time horizon. We assume that the filtration $(\mathcal{F}_t)_{t \in [0,T]}$ satisfies the usual conditions, i.e., it is right-continuous and complete.

2.1. Basic tools from stochastic calculus

This section contains important tools from stochastic calculus, which are used frequently throughout this thesis. The first and probably most important tool from stochastic calculus is the Itô-Doeblin formula.

Theorem 2.1 (Itô-Doeblin formula; Protter, 2005, pp. 81-82). Let $X = (X^1, \ldots, X^d)$ be a continuous d-dimensional semimartingale (i.e., each X^j , $j = 1, \ldots, d$, is a semimartingale) and $f \in C^2(\mathbb{R}^d, \mathbb{R})$, i.e., all second order partial derivatives $\frac{\partial^2 f}{\partial x_i \partial x_j}$, $i, j \in \{1, \ldots, d\}$, exist and are continuous. Then the process f(X) is also a semimartingale and the following formula holds

$$f(X_t) - f(X_0) = \sum_{i=1}^d \int_0^t \frac{\partial f}{\partial x_i}(X_s) \, \mathrm{d}X_s^i + \frac{1}{2} \sum_{1 \le i,j \le d} \int_0^t \frac{\partial^2 f}{\partial x_i \partial x_j}(X_s) \, \mathrm{d}\langle X^i, X^j \rangle_s, \quad t \in [0, T].$$

According to Protter (2005, p. 66), the quadratic (co)variation (or bracket) process $\langle \cdot, \cdot \rangle$ is defined by

$$\langle X, Y \rangle_t = X_t Y_t - \int_0^t X_s dY_s - \int_0^t Y_s dX_s, \quad \langle X \rangle_t = \langle X, X \rangle_t, \quad t \in [0, T],$$

where X and Y are continuous semimartingales. Protter (2005) provides a formula for the quadratic (co)variation of Itô integrals, which is often a helpful tool when dealing with stochastic differential equations. The formula is given in the following lemma.

Lemma 2.2 (Protter, 2005, p. 167). Let X and Y be semimartingales and let $H \in L(X)$, $G \in L(Y)$, where L(X) describes the set of predictable, X-integrable processes (see (2.3) in Subsection 2.3.1 below). Then

$$\langle H \cdot X, G \cdot Y \rangle_t = \int_0^t H_s G_s \, \mathrm{d} \langle X, Y \rangle_s, \quad t \in [0, T].$$
 (2.1)

In (2.1), we used the abbreviation $(H \cdot X)_t = \int_0^t H_s \, \mathrm{d}X_s, t \in [0, T].$

The Itô-Doeblin formula has many useful applications. Below are two results that can be deduced directly from the Itô-Doeblin formula.

Corollary 2.3 (Integration by Parts, Protter, 2005, p. 68). Let X and Y be continuous semimartingales. Then XY is a semimartingale and

$$X_t Y_t = \int_0^t X_s \mathrm{d}Y_s + \int_0^t Y_s \mathrm{d}X_s + \langle X, Y \rangle_t, \quad t \in [0, T].$$

Another important consequence of the Itô-Doeblin formula is the stochastic exponential displayed in Theorem 2.4 below. Note that Theorem 2.4 is stated for a general semimartingale which is not necessarily continuous. Thus, a generalized version of the Itô-Doeblin formula is necessary to prove the assertion of the theorem. The general Itô-Doeblin formula can be found in Protter (2005, pp. 81–82).

Theorem 2.4 (Stochastic Exponential; Protter, 2005, p. 84). Let X be a semimartingale with $X_0 = 0$. Then there exists a unique semimartingale Z that satisfies the equation $Z_t = 1 + \int_0^t Z_{s-} dX_s$, given by

$$Z_t = \exp\left(X_t - \frac{1}{2}\langle X \rangle_t\right) \prod_{0 < s \le t} (1 + \Delta X_s) \exp\left(-\Delta X_s + \frac{1}{2}(\Delta X_s)^2\right), \quad t \in [0, T],$$
(2.2)

where the infinite product converges.

If X is a continuous semimartingale, the representation (2.2) of Z_t simplifies to (see Protter, 2005, p. 85)

$$Z_t = \exp\left(X_t - \frac{1}{2}\langle X \rangle_t\right).$$

The stochastic exponential Z corresponding to a semimartingale X is sometimes denoted by $\mathcal{E}(X)$, i.e.,

$$\mathcal{E}(X) = \exp\left(X - \frac{1}{2}\langle X \rangle\right).$$

We conclude this section with a stochastic version of Fubini's theorem. The theorem uses the the *predictable* σ -algebra \mathcal{P} , which is generated by the class of adapted processes with càglàd paths

(i.e., the paths are left continuous with existing right limits). We refer to Definition 2.6 below for a formal definition and a brief discussion of predictable processes.

Theorem 2.5 (Stochastic version of Fubini's theorem; Protter, 2005, pp. 211–212). Let (A, \mathcal{A}, μ) be a measure space, where μ is a positive, finite measure on \mathcal{A} . Further, let X be a semimartingale on a probability space $(\Omega, \mathcal{F}, \mathbb{P})$ and let $(H_t^a)_{t\geq 0}$ be an $\mathcal{A} \otimes \mathcal{P}$ -measurable stochastic process. Further, assume that

$$\left(\int_A (H_t^a)^2 \mu(\mathrm{d}a)\right)^{\frac{1}{2}} \in L(X)$$

where L(X) describes the set of predictable processes that are integrable with respect to X. Finally, assume that $(a,t,\omega) \mapsto Z_t^a(\omega)$, where $Z_t^a \coloneqq \int_0^t H_s^a dX_s$, defines an $\mathcal{A} \otimes \mathcal{B}((0,\infty)) \otimes \mathcal{F}$ -measurable process and that Z^a is càdlàg (i.e., the paths are right continuous with existing left limits) for any $a \in A$. Then the integral

$$\int_{A} Z_t^a \mu(\mathrm{d}a) = \int_{A} \int_0^t H_s^a \mathrm{d}X_s \mu(\mathrm{d}a)$$

exists and

$$\int_{A} \int_{0}^{t} H_{s}^{a} \mathrm{d}X_{s} \mu(\mathrm{d}a) = \int_{0}^{t} \int_{A} H_{s}^{a} \mu(\mathrm{d}a) \mathrm{d}X_{s}$$

holds in the sense that the integral on the left-hand side is a càdlàg version of the integral on the right-hand side.

2.2. Definition of σ -martingales

In what follows, we explain a generalization of local martingales, so-called σ -martingales. Such processes play an important role in the general semimartingale financial market explained in Subsection 2.3.1 below. First, we need to introduce the notion of predictable processes.

As it is the case in most applications, we associate the filtration $(\mathcal{F}_t)_{t\in[0,T]}$ with the flow of information in our model. To be coherent with this interpretation, we need to make sure that all considered processes are in some sense in line with the flow of information. Apart from adapted and progressively measurable processes, we also need the definition of a predictable process. The following definition is taken from Protter (2005, pp. 56, 102) and Eberlein and Kallsen (2019, p. 98).

- **Definition 2.6.** a) The predictable σ -algebra \mathcal{P} is the smallest σ -algebra on $\Omega \times [0, \infty)$ such that all adapted processes with càglàd paths (i.e., paths that are left continuous with existing right limits) are measurable with respect to \mathcal{P} .
 - b) A random set $D \subset \Omega \times [0, \infty)$ is called *predictable* if $D \in \mathcal{P}$.
 - c) An \mathbb{R}^d -valued stochastic process X is called *predictable* if it is measurable with respect to \mathcal{P} (interpreted as a mapping $X : \Omega \times [0, \infty) \to \mathbb{R}^d$).

Remark 2.7. To understand the concept of predictable processes, it helps to consider the discretetime analogue. A discrete-time process $(X_n)_{n \in \mathbb{N}_0}$ is called *predictable* if X_0 is \mathcal{F}_0 -measurable and, for any $n \in \mathbb{N}$, X_n is \mathcal{F}_{n-1} -measurable (Jacod and Shiryaev, 2003, Definition I.2.36). This definition can be interpreted as follows: A decision made at time n can only use information available at previous times, i.e., up to time n-1. This justifies the term predictable. Measurability of processes with càglàd paths provides a continuous-time analogue to the definition in discrete time. \Diamond

According to Protter (2005, p. 103), a predictable stochastic process is also progressively measurable, but the reverse does not hold in general.

Now that we are familiar with predictable processes, we can generalize the concept of a local martingale. Recall that a local martingale is a stochastic process that behaves like a martingale if we truncate time with respect to an increasing sequence of stopping times (see, e.g., Jacod and Shiryaev, 2003, Definition I.1.45). If we not only truncate with respect to t, but also with respect to ω , we obtain a σ -martingale.

Definition 2.8 (Jacod and Shiryaev, 2003, Definition III.6.33). An \mathbb{R} -valued semimartingale X is called a σ -martingale if there exists an increasing sequence $(D_n)_{n \in \mathbb{N}}$ of predictable sets with $D_n \uparrow \Omega \times [0, \infty)$ (up to an evanescent¹ set) such that, for any $n \geq 1$, the process $\mathbb{1}_{D_n} X$ is a uniformly integrable martingale. An \mathbb{R}^d -valued semimartingale is called a σ -martingale if each component is an \mathbb{R} -valued σ -martingale.

Using Proposition III.6.34 in Jacod and Shiryaev (2003), this is in fact a generalization of a local martingale as any local martingale is a σ -martingale while the reverse does not hold in general (see, e.g., Jacod and Shiryaev, 2003, Example III.6.40). Further, Delbaen and Schachermayer (1998) state that the notion of σ -martingales "relates to martingales similarly as σ -finite measures relate to finite measures".

There is also a different definition of σ -martingales used throughout the literature (see, for example, Protter, 2005, p. 237; Delbaen and Schachermayer, 2006, Definition 14.2.1). There, an \mathbb{R}^d -valued semimartingale X is called σ -martingale if there exist an \mathbb{R}^d -valued martingale M and an Mintegrable, predictable, $(0, \infty)$ -valued process φ such that X can be written as² $X = X_0 + \varphi \cdot M$. However, it can be shown, using Theorem 89 in Protter (2005, pp. 237–238) and Theorem III.6.41 by Jacod and Shiryaev (2003), that the two definitions are equivalent.

The notion of σ -martingales is popular in the mathematical finance literature due to the paper by Delbaen and Schachermayer (1998). There, the equivalence of the "no free lunch with vanishing risk" condition and the existence of an equivalent σ -martingale measure is shown, which provides a generalization of the classical fundamental theorem of asset pricing. We also use the absence of arbitrage in the presence of an equivalent σ -martingale measure in Subsection 2.3.1. However, we understand arbitrage strategies in a slightly different sense. Delbaen and Schachermayer (2006, p. 142) also argue that the notion of a σ -martingale is "tailormade" for the purpose of excluding arbitrage (in some sense) in general semimartingale markets. Namely, it is first unavoidable to find a more general concept than local martingales if S should be allowed to have jumps of unbounded size. Second, they argue that "for the purposes of hedging contingent claims the notion of a

¹A set $D \subset \Omega \times [0, \infty)$ is called an *evanescent set* if its projection onto Ω is a \mathbb{P} -null set, i.e., if $\mathbb{P}(\{\omega \in \Omega | \exists t \in [0, \infty) \text{ with } (\omega, t) \in D\}) = 0$ (see, for example, Jacod and Shiryaev, 2003, p. 3).

²Note that the definition given by Protter (2005) does not include the initial value X_0 . However, at the beginning of Chapter IV, Protter (2005) assumes, without loss of generality, that all processes X start in 0.

 σ -martingale is just as useful as the notion of a local martingale" (see Delbaen and Schachermayer, 2006, p. 142).

2.3. Basics from mathematical finance

In this section, we explain some basic concepts from the general area of mathematical finance. In Subsection 2.3.1, we explain a very general semimartingale market which forms the foundation of most problems considered in this thesis. Afterwards, we discuss the issue of completeness in Subsection 2.3.2. Finally, we give a short introduction to portfolio optimization problems and common methods to solve such problems in Subsection 2.3.3.

2.3.1. General semimartingale market

In the following, we describe a general semimartingale financial market. The market is taken from Černỳ and Kallsen (2007) and Delbaen and Schachermayer (1996). We want to emphasize at this point that we do not need any special requirements for our later analysis. The financial market is therefore as general as possible, while all stochastic integrals that arise are well-defined. Thus, the stock prices are taken as semimartingales with càdlàg paths (i.e., the paths are right continuous with existing left limits). Delbaen and Schachermayer (2006, p. 130) argue that semimartingales are "precisely the class of processes allowing for a satisfactory integration theory".

The financial market is assumed to be frictionless and perfectly elastic³. It consists of $d \in \mathbb{N}$ stocks with price processes $S_k = (S_k(t))_{t \in [0,T]}$, $k = 1, \ldots, d$, (which are assumed to be not identical, i.e., there are no two distinct indices $j, k \in \{1, \ldots, d\}$ such that S_j and S_k are modifications of another) collected in the *d*-dimensional process $S = (S_1(t), \ldots, S_d(t))_{t \in [0,T]}$. Moreover, there exists a riskless bond which is, without loss of generality, assumed to be identical to 1. This is not a restriction since we could simply take the bond as numéraire⁴ (see Delbaen and Schachermayer, 1995). Moreover, we assume that S is an $L^2(\mathbb{P})$ -semimartingale with càdlàg paths (i.e., the paths are right continuous with existing left limits). Following Delbaen and Schachermayer (1996), an $L^2(\mathbb{P})$ -semimartingale is defined as follows.

Definition 2.9. An \mathbb{R}^d -valued semimartingale S that satisfies

$$\sup \left\{ \mathbb{E} \left[(S_k(\tau \wedge T))^2 \right] : \tau \text{ is an } (\mathcal{F}_t)_{t \in [0,T]} \text{-stopping time, } k = 1, \dots, d \right\} < \infty$$

is called $L^2(\mathbb{P})$ -semimartingale

To ensure that our financial market does not contain arbitrage opportunities, we require the existence of an equivalent probability measure under which the stock price processes become

³A market is called frictionless if there are no transaction costs and no limitations on short selling (Karatzas and Shreve, 1998, p. 8). Moreover, the term perfectly elastic expresses that the stock prices are not changed by the agents' orders, i.e., investors are assumed to be "small". We refer to Chapter 7 for a more detailed discussion of this topic.

⁴According to Bingham and Kiesel (2004), a numéraire is a price process $(X_t)_{t \in [0,T]}$ that is strictly positive almost surely for any $t \in [0,T]$. It can be interpreted as a benchmark unit used in the financial market. The prices of different assets in the market are given in terms of (relative to) the numéraire.

 σ -martingales. Thus, following Černỳ and Kallsen (2007), we make the following standing assumption.

Assumption 2.10. There exists an equivalent probability measure \mathbb{Q} with square-integrable density, i.e., $\mathbb{Q} \sim \mathbb{P}$ with $\mathbb{E}\left[\left(\frac{\mathrm{d}\mathbb{Q}}{\mathrm{d}\mathbb{P}}\right)^2\right] < \infty$, such that S is a σ -martingale with respect to \mathbb{Q} . Such a \mathbb{Q} is called σ -martingale measure (with respect to S) (S σ MM) with square integrable density.

For the aforementioned $S\sigma MM \mathbb{Q}$, we denote the associated density process by

$$Z_t^{\mathbb{Q}} \coloneqq \mathbb{E}\left[\frac{\mathrm{d}\mathbb{Q}}{\mathrm{d}\mathbb{P}}\Big|\mathcal{F}_t\right], \ t \in [0,T].$$

Černỳ and Kallsen (2007) argue that Assumption 2.10 "can be interpreted as a natural no-freelunch condition in the present quadratic context". We can explain this statement in more detail, but we have to define the set of admissible strategies first. Instead of simply stating the set of admissible strategies, we give a more illustrative derivation of the set similar to Černỳ and Kallsen (2007).

First, we need to ensure that a strategy φ , i.e., an \mathbb{R}^d -valued stochastic process, where $\varphi_k(t)$ represents the number of shares of stock k held at time t, is integrable with respect to the stock price process S. Hence, we require that $\varphi \in L(S)$, where

$$L(S) \coloneqq \left\{ \varphi : \varphi \text{ is } (\mathcal{F}_t) \text{-predictable and } S \text{-integrable} \right\}.$$
(2.3)

The term S-integrable" needs to be defined in more detail. Let

$$S = S_0 + M + A$$

describe the unique (up to modifications) semimartingale decomposition of S, i.e., S_0 is \mathcal{F}_0 measurable, M is a local martingale with $M_0 = 0$, and A is a finite variation process with $A_0 = 0$ (see, for example, Protter, 2005, Theorem 1 and 2, pp. 102–103). Then we say that φ is S-integrable if, and only if, φ satisfies (see Jacod and Shiryaev, 2003, Definition III.6.17)

$$\mathbb{P}\left(\int_0^T (\varphi_k(t))^2 \mathrm{d}\langle S_k \rangle_t < \infty\right) = 1, \qquad \mathbb{P}\left(\int_0^T \varphi_k(t) \mathrm{d}A_k(t) < \infty\right) = 1,$$

where the integrals with respect to $\langle S_k \rangle$ and A are defined as pathwise Lebesgue-Stieltjes integrals (see, for example, Protter, 2005, p. 39). For $\varphi \in L(S)$, we denote the stochastic integral of φ with respect to S by

$$(\varphi \cdot S)_t \coloneqq \sum_{k=1}^d \int_0^t \varphi_k(u) \mathrm{d}S_k(u), t \in [0, T].$$

The set \mathcal{A} of admissible strategies will be constructed from the $L^2(\mathbb{P})$ -closure (in some sense) of the set of simple strategies. More specifically, strategies are called admissible if their integral with respect to S can be approximated in $L^2(\mathbb{P})$ using a sequence of simple strategies. Let us give a more detailed explanation of this choice. A stochastic process φ is called *simple* (or *elementary*) if there exist finite stopping times $\tau_1 \leq \cdots \leq \tau_n$ and bounded, \mathcal{F}_{τ_k} -measurable random variables Y_k such that φ can be written as

$$\varphi_t(\omega) = \sum_{k=1}^n Y_k(\omega) \mathbb{1}_{(\tau_1(\omega), \tau_2(\omega)]}(t), t \in [0, T], \omega \in \Omega$$

(see, e.g., Černỳ and Kallsen, 2007; Delbaen and Schachermayer, 1996). Eberlein and Kallsen (2019, p. 584) argue that simple strategies, although not very interesting mathematically, are still important because they are the only ones that are feasible in real life.

Now a stochastic process $\varphi \in L(S)$ is called *admissible* if the stochastic integral of φ with respect to S can be approximated by a sequence of stochastic integrals of simple strategies $\varphi^{(n)}$ with respect to S. More specifically, $\varphi \in L(S)$ is called admissible if there exists a sequence $(\varphi^{(n)})_{n \in \mathbb{N}}$ of simple strategies such that⁵

$$(\varphi^{(n)} \cdot S)_t \xrightarrow{\mathbb{P}} (\varphi \cdot S)_t \text{ for any } t \in [0, T],$$
$$(\varphi^{(n)} \cdot S)_T \xrightarrow{L^2(\mathbb{P})} (\varphi \cdot S)_T, \ n \to \infty.$$

While this definition of admissible strategies allows for a straightforward interpretation as limits of strategies which are feasible in real life, the above condition is impractical in general. Therefore, Černỳ and Kallsen (2007) presented an equivalent characterization of this condition which is more convenient to use. They showed that some \mathbb{R}^d -valued process φ is admissible (regarding the above definition) if, and only if, φ is in L(S), $(\varphi \cdot S)_T \in L^2(\mathbb{P})$, and $(\varphi \cdot S)Z^{\mathbb{Q}}$ is a \mathbb{P} -martingale for any $S\sigma MM \mathbb{Q}$ with square integrable density. Thus, the set of admissible strategies reads as

$$\mathcal{A} = \Big\{ \varphi \in L(S) : (\varphi \cdot S)_T \in L^2(\mathbb{P}), \ (\varphi \cdot S)Z^{\mathbb{Q}} \text{ is a } \mathbb{P}\text{-martingale for all } S\sigma MM \ \mathbb{Q} \\ \text{with density process } Z^{\mathbb{Q}} \text{ and square integrable density} \Big\}.$$
(2.4)

The wealth process $(X_t^{\varphi})_{t \in [0,T]}$ associated to some $\varphi \in \mathcal{A}$ is given by

$$X_t^{\varphi} = x_0 + (\varphi \cdot S)_t, \, t \in [0, T],$$
(2.5)

for the initial capital $x_0 \in \mathbb{R}$. In the previously described setting, the time-zero price of any claim $X \in L^2(\mathbb{P})$ is given by (see, for example, Bingham and Kiesel, 2004, Theorem 6.1.4)

$$\mathbb{E}_{\mathbb{Q}}[X] = \mathbb{E}\Big[Z_T^{\mathbb{Q}}X\Big]$$
(2.6)

for all $S\sigma MM \mathbb{Q}$ with square integrable density. Note that, in general, the price depends on the choice of \mathbb{Q} . Nevertheless, for X_T^{φ} from (2.5), we obtain $\mathbb{E}_{\mathbb{Q}}[X_T^{\varphi}] = x_0$ for any choice of \mathbb{Q} (see Eberlein and Kallsen, 2019, p. 540).

Now that the set of admissible strategies is fixed, we can justify that Assumption 2.10 implies the absence of arbitrage strategies contained in \mathcal{A} . Černý and Kallsen (2007) explain that Assumption 2.10 implies

$$\overline{K_2^S(0) - L_+^2} \cap L_+^2 = \{0\},$$
(2.7)

⁵By $\xrightarrow{\mathbb{P}}$ we denote convergence in probability.

where L^2_+ is the set of non-negative, square integrable random variables. The closure needs to be understood in the $L^2(\mathbb{P})$ -sense. In their notation, the set $K^S_2(0)$ describes the $L^2(\mathbb{P})$ -closure of the set of all claims that are attainable from simple strategies with zero initial endowment, i.e., claims H that can be written as $H = (\varphi \cdot S)_T$ for a simple strategy φ . Those are exactly the wealth processes attainable from admissible strategies $\varphi \in \mathcal{A}$ with an initial capital of 0. Then (2.7) implies that the only non-negative, square integrable random variable obtainable by a reduction of some attainable claim is constantly equal to zero. Hence, \mathcal{A} does not contain arbitrage strategies.

2.3.2. A NOTE ON COMPLETENESS

In addition to the absence of arbitrage, an often desirable property of financial markets is its completeness. Without going into detail, a financial market is called *complete* if any claim H is attainable by some admissible strategy φ (see, for example, Eberlein and Kallsen, 2019, p. 540). In general, a *claim* is an \mathcal{F}_T -measurable random variable H. In most cases, H is required to satisfy additional properties like non-negativity, integrability or boundedness by a suitable random variable (see, e.g., Karatzas and Shreve, 1998, pp. 21-22; Eberlein and Kallsen, 2019, p. 587; Jeanblanc et al., 2009, p. 87). A claim is called *attainable* if there exist an admissible strategy φ and an initial capital x_0 such that $H = X_T^{\varphi} = x_0 + (\varphi \cdot S)_T$. Apart from requiring that φ is self financing, the conditions imposed on admissible strategies depend strongly on the specific model. If one is interested in defining the unique arbitrage free price of a claim H, being able to hedge this claim by some replicating strategy is of utmost importance.

In many cases, the so-called second fundamental theorem of asset pricing gives an equivalent characterization of completeness of an arbitrage free financial market. The statement of this theorem is (more or less) the same in different market models. While absence of arbitrage is associated with the existence of an equivalent (σ -/local) martingale measure, completeness is related to the uniqueness of such a measure. This type of assertion can be found in many cases varying from discrete-time models, considered, for example, by Föllmer and Schied (2016, Theorem 5.37), to general semimartingale models like the one used by Eberlein and Kallsen (2019, Theorem 11.54). Similar to the characterization of absence of arbitrage, there is neither the perfect model choice nor the optimal characterization of completeness. That is why we decided not to give a general characterization of completeness that fits the semimartingale market from Subsection 2.3.1. For the characterization that comes closest to the model considered above, we refer to Theorem 11.54 in Eberlein and Kallsen (2019).

Let us conclude this discussion with a popular example of a complete financial market. Assume that there exist a riskless bond with zero interest rate and d stocks. Let the d stock price processes be geometric Brownian motions, i.e., they take the form

$$S_k(t) = S_k(0) \exp\left(\left(\mu_k - \frac{1}{2}\sum_{\ell=1}^d \sigma_{k\ell}^2\right)t + \sum_{\ell=1}^d \sigma_{k\ell}W_\ell(t)\right), \ t \in [0, T],$$

k = 1, ..., d, where $W_1, ..., W_d$ are independent Brownian motions. The drift vector μ and the volatility matrix σ are assumed to be deterministic and constant in time. Then the market is

complete if, and only if, the volatility matrix σ is regular (see, for example, Karatzas and Shreve, 1998, Theorems 4.2 and 6.6 in Chapter 1). Here, completeness means that any \mathcal{F}_T -measurable random variable X, which is bounded from below, is attainable by a square integrable, progressively measurable portfolio process.

2.3.3. INTRODUCTION TO PORTFOLIO OPTIMIZATION

The basic idea behind portfolio optimization is the following. Given some initial capital x_0 and fixed time horizon T > 0, an investor aims to maximize her expected terminal wealth via optimal investment. Maximizing the expected terminal wealth usually results in an unbounded optimization problem. Thus, instead of maximizing expected wealth itself, it is usually assumed that investors measure their preferences by some (Inada) utility function applied to their terminal wealth. A utility function is defined as follows (see, for example, Korn, 1997, p. 38; Eberlein and Kallsen, 2019, pp. 462–463).

Definition 2.11. A strictly increasing, continuous function $U : \mathcal{D} \to \mathbb{R}$, $\mathcal{D} \in \{(0, \infty), \mathbb{R}\}$, is called *utility function*. Additionally, if U is strictly concave and continuously differentiable, and satisfies the *Inada conditions*

$$\lim_{x \to \inf \mathcal{D}} U'(x) = \infty, \quad \lim_{x \to \infty} U'(x) = 0,$$

U is called *Inada utility function*.

Some common examples of Inada utility functions are the natural logarithm, the power utility function

$$U: (0,\infty) \to \mathbb{R}, x \mapsto \left(1 - \frac{1}{\delta}\right)^{-1} x^{1 - \frac{1}{\delta}}$$
(2.8)

for a parameter $\delta > 0, \, \delta \neq 1$, or the exponential utility function

$$U: \mathbb{R} \to \mathbb{R}, x \mapsto -\exp\left(-\frac{1}{\delta}x\right)$$
 (2.9)

for a parameter $\delta > 0$ (see, for example, Eberlein and Kallsen, 2019, p. 463). Utility functions are often characterized in terms of their absolute or relative risk aversion. The notion was introduced by Pratt (1964) and Arrow (1974) and is defined as follows (see also Föllmer and Schied, 2016, pp. 82-83).

Definition 2.12. Let $U : \mathcal{D} \to \mathbb{R}$, $\mathcal{D} \in \{(0, \infty), \mathbb{R}\}$ be a twice continuously differentiable Inada utility function. Then

$$A(x) \coloneqq -\frac{U''(x)}{U'(x)}, x \in \mathcal{D},$$

defines the *absolute* and

$$R(x) \coloneqq -\frac{xU''(x)}{U'(x)}, \, x \in \mathcal{D},$$

the relative Arrow-Pratt risk aversion coefficient.

The absolute Arrow-Pratt risk aversion coefficient of the exponential utility function (2.9) is given by δ^{-1} . Thus, the exponential utility function belongs to the so-called CARA (constant absolute risk aversion) utility functions. Moreover, the relative risk aversion coefficient of the logarithm and

4

the power utility function (2.8) is given by δ^{-1} with $\delta = 1$ for the logarithm. Hence, power and logarithmic utility belong to the class of CRRA (constant relative risk aversion) utility functions. This explains why we refer to δ^{-1} as the risk aversion and to δ as the risk tolerance parameter.

For an investor with initial capital x_0 and utility function $U : \mathcal{D} \to \mathbb{R}$, the optimization problem of expected utility maximization reads as

$$\begin{cases} \max_{\varphi \in \mathcal{A}} \mathbb{E}[U(X_T^{\varphi})], \\ \text{s.t.} \quad X_T^{\varphi} = x_0 + (\varphi \cdot S)_T. \end{cases}$$
(2.10)

The set of admissible portfolio strategies \mathcal{A} depends on the specific financial market. It usually contains some measurability and integrability conditions imposed on the portfolio process φ : $[0,T] \times \Omega \to \mathbb{R}^d$, where d describes the number of stocks in the underlying financial market. As the control of an agent is a stochastic process, (2.10) describes a special *dynamic optimization problem.* For portfolio optimization problems like (2.10), there are two solution methods used in the literature - the martingale method and the dynamic programming approach. We do not explain these methods in detail, but we give a short idea on the methods as they are both used in this thesis.

The martingale method (see, for example, Section 3.4 in Korn, 1997) consists of two sub-problems. The first one is the static optimization problem in which the optimal terminal wealth is determined. The second one is the representation problem that comprises the search for a replicating strategy for the optimal terminal wealth from the first step. The martingale approach is usually applied in complete markets since, in that case, it can be guaranteed that the representation problem has a solution. The static optimization problem is often uniquely solvable due to the concave objective function. If the underlying financial market has a unique (σ -/local) martingale measure \mathbb{Q} with $\frac{d\mathbb{Q}}{d\mathbb{P}} = Z_T$ (see Subsection 2.3.1), the unique solution to the static optimization problem is given by

$$X^* = I(\lambda^* Z_T)$$

(see, for example, Kramkov and Schachermayer, 1999). The function $I = (U')^{-1}$ describes the inverse of the first order derivative of U and $\lambda^* > 0$ denotes the Lagrange multiplier which is determined by the budget constraint $\mathbb{E}[Z_T X^*] = x_0$. For a more in-depth explanation of the martingale method, we refer to Kramkov and Schachermayer (1999) as well as Section 3.4 in Korn (1997).

Let us now explain the second method. The idea behind the dynamic programming approach is to derive a partial differential equation for the value function

$$J(t,x) \coloneqq \sup_{\varphi \in \mathcal{A}} \mathbb{E}\left[U(X_T^{\varphi}) | X_t^{\varphi} = x\right], t \in [0,T], x \in \mathcal{D},$$

with the terminal constraint J(T, x) = U(x), $x \in \mathcal{D}$. The derivation of the partial differential equation is based on the so-called *Bellman principle* (or sometimes called *dynamic programming principle*). It states that

$$J(t,x) = \sup_{\varphi} \mathbb{E} \left[J(t', X_{t'}^{\varphi}) | X_t^{\varphi} = x \right]$$

holds for all $0 \le t \le t' \le T$ (see, for example, Equation (3.20) in Pham, 2009). Together with the Itô-Doeblin formula (see Theorem 2.1), one can derive a partial differential equation, the so-called Hamilton-Jacobi-Bellman (HJB) equation, which is then used to find a candidate for the value function J. From a solution G to the HJB equation, one can infer a candidate for the optimal control. Finally, a verification theorem is necessary to ensure that the candidate control is in fact optimal and that the solution to the HJB equation coincides with the value function. For a more detailed explanation of the method, we refer to Chapter 3 in Pham (2009).

2.4. INTRODUCTION TO MULTI-OBJECTIVE OPTIMIZATION

Generally speaking, a multi-objective optimization problem is an optimization problem that involves multiple (coupled) objective functions. We want to take a game theoretic approach to multi-objective optimization problems. Thus, we consider n agents with objective functions J_1, \ldots, J_n . Each agent is able to choose some control $\varphi^i \in \mathcal{A}_i, i = 1, \ldots, n$, where \mathcal{A}_i is the set of admissible strategies of agent i, i.e., the set of controls agent i is allowed to choose from. In the problems considered throughout this thesis, controls φ^i are \mathbb{R}^d -valued stochastic processes, defined on a finite time interval [0, T]. The set of admissible strategies usually contains some measurability and integrability conditions that depend on the specific model. Moreover, agent i aims to maximize $J_i(\varphi^1,\ldots,\varphi^n)$, i.e., each objective function depends on all *n* strategies. Thus, there are *n* objective functions that need to be maximized simultaneously. As the resulting objective function is \mathbb{R}^{n} valued, it is not clear a priori under which conditions a vector of admissible strategies should be called optimal. Therefore, we need to clarify when a vector $(\varphi^1, \ldots, \varphi^n)$ with $\varphi^i \in \mathcal{A}_i$ for all $i \in \{1, \ldots, n\}$ is considered optimal. Throughout this thesis, we apply two different notions of optimality. The first one is the so-called Nash equilibrium, in which agents act competitive whereas the second one, a so-called *Pareto optimum*, is obtained when the public good of all n players is optimized. In the following, we explain these two concepts in more depth. First, we consider Nash equilibria. The definition dates back to Nash (1951).

Definition 2.13. Let $J_i : X_{j=1}^n \mathcal{A}_j \to \mathbb{R}$ be the objective function of agent i, i = 1, ..., n. A vector $(\varphi^{1,*}, \ldots, \varphi^{n,*})$ of admissible strategies is called a *Nash equilibrium* if, for all admissible $\varphi^i \in \mathcal{A}_i$ and $i \in \{1, \ldots, n\}$,

$$J_i(\varphi^{1,*},\ldots,\varphi^{i,*},\ldots,\varphi^{n,*}) \ge J_i(\varphi^{1,*},\ldots,\varphi^{i-1,*},\varphi^i,\varphi^{i+1,*},\ldots,\varphi^{n,*}).$$

$$(2.11)$$

Further, $(\varphi^{1,*}, \ldots, \varphi^{n,*})$ is called a *constant Nash equilibrium*, if $\varphi_t^{i,*} = \varphi_0^{i,*}$ holds for each $t \in [0,T]$ and each $i \in \{1, \ldots, n\}$.

Remark 2.14. For a vector $(\varphi^{1,*}, \ldots, \varphi^{n,*})$ of constant strategies, i.e., $\varphi^{i,*}(t) = \varphi^{i,*}(0)$ for all $i \in \{1, \ldots, n\}$, to be a constant Nash equilibrium, condition (2.11) still needs to be satisfied for any $\varphi^i \in \mathcal{A}_i$, not just for constant strategies.

Definition 2.13 shows that a Nash equilibrium can be interpreted as a vector of admissible strategies, chosen by the n players, such that none of the agents would benefit from deviating from the equilibrium strategy unilaterally. Thus, Nash equilibria provide a concept of optimality for competitive agents, where each agent is focused on maximizing her own objective and takes the

controls of the other agents as given. In the literature on *n*-agent games, there are two different kinds of Nash equilibria which need to be distinguished – *open-loop* and *closed-loop Nash equilibria*. Carmona and Delarue (2018a, p. 72) argue that the above definition of a Nash equilibrium only makes sense once it is properly defined how the strategies of the players $j \neq i$ are "frozen" from the perspective of agent *i*. To be more specific, it is important to explain how the players update their strategies. In an open-loop equilibrium, agent *i* treats the strategies of the other agents as fixed in the sense that the strategies do not change when agent *i* changes her strategy in order to find the optimal one. On the other hand, when looking for a closed-loop equilibrium, agent *i* considers the strategies of the other players in feedback form. Thus, if agent *i* changes her strategy, the strategies of the other players change as well. Carmona and Delarue (2018a, pp. 72-75) give rigorous definitions of open-loop and closed-loop equilibria.

Definition 2.15 (see Carmona and Delarue, 2018a, Definition 2.4). Let $(\varphi^{1,*}, \ldots, \varphi^{n,*})$ be a vector of admissible strategies such that (2.11) holds for all $i \in \{1, \ldots, n\}$. If, for all $i \in \{1, \ldots, n\}$, the vector $(\varphi^{1,*}, \ldots, \varphi^{i-1,*}, \varphi^{i+1,*}, \ldots, \varphi^{n,*})$ remains the same if player *i* changes her strategy from $\varphi^{i,*}$ to a different strategy φ^i , $(\varphi^{1,*}, \ldots, \varphi^{n,*})$ is called an *open-loop Nash equilibrium*.

Definition 2.16 (see Carmona and Delarue, 2018a, Definition 2.6). Let $(\varphi^{1,*}, \ldots, \varphi^{n,*})$ be a vector of admissible strategies of the form $\varphi^{i,*} = \phi^{i,*}(t, X_{[0,t]}^{1,\varphi^{1,*}}, \ldots, X_{[0,t]}^{n,\varphi^{n,*}})$. If

$$J_{i}\left(\phi^{1,*}(t,X_{[0,t]}),\ldots,\phi^{n,*}(t,X_{[0,t]})\right) \geq J_{i}\left(\phi^{1,*}(t,\widetilde{X}_{[0,t]}),\ldots,\phi^{i-1,*}(t,\widetilde{X}_{[0,t]}),\phi^{i}(t,\widetilde{X}_{[0,t]}),\phi^{i+1,*}(t,\widetilde{X}_{[0,t]}),\ldots,\phi^{n,*}(t,\widetilde{X}_{[0,t]})\right)$$

holds for all $i \in \{1, \ldots, n\}$, where $\varphi^i = \phi^i(t, \widetilde{X}_{[0,t]}), X_t = \left(X_t^{1,\varphi^{1,*}}, \ldots, X_t^{i,\varphi^{i,*}}, \ldots, X_t^{n,\varphi^{n,*}}\right)$, and $\widetilde{X}_t = \left(X_t^{1,\varphi^{1,*}}, \ldots, X_t^{i-1,\varphi^{i-1,*}}, X_t^{i,\varphi^i}, X_t^{i+1,\varphi^{i+1,*}}, \ldots, X_t^{n,\varphi^{n,*}}\right)$, then $(\varphi^{1,*}, \ldots, \varphi^{n,*})$ is called a *closed-loop Nash equilibrium*.

In general, closed-loop equilibria are preferable in terms of a more realistic interpretation, whereas open-loop equilibria are mathematically more tractable. Thus, open-loop equilibria are often preferred in the literature on dynamic optimization for *n*-agent games (Carmona and Delarue, 2018a, Remark 2.8). Throughout this thesis, we only consider open-loop Nash equilibria and thus, refer to them simply as Nash equilibria. It should be noted that for constant Nash equilibria open-loop and closed-loop Nash equilibria coincide (see, e.g., Lacker and Zariphopoulou (2019)).

As the games considered in this thesis are motivated by competitive investors, Nash equilibria are the most used notion of optimality. However, we also want to consider a different type of optimality, namely, *Pareto optimality*. A Pareto optimum is defined as follows.

Definition 2.17 (Miettinen, 1999, Definition 2.2.1). A vector $(\varphi^{1,*}, \ldots, \varphi^{n,*})$ of admissible strategies $\varphi^{i,*} \in \mathcal{A}_i, i = 1, \ldots, n$, is called *Pareto optimal* if there is no vector $(\varphi^1, \ldots, \varphi^n)$ of admissible strategies such that

$$J_i(\varphi^1, \dots, \varphi^n) \ge J_i(\varphi^{1,*}, \dots, \varphi^{n,*})$$
 for all $i = 1, \dots, n$

and

$$J_i(\varphi^1, \dots, \varphi^n) > J_i(\varphi^{1,*}, \dots, \varphi^{n,*})$$
 for at least one $i \in \{1, \dots, n\}$.

In contrast to Nash equilibria, Pareto optima are related to optimization of the public good instead of each individuals objective. A Pareto optimum is attained if no player can increase her objective without decreasing the objective of another player. One could also imagine some kind of central planner or manager that controls the state processes of n clients simultaneously. Thus, the manager wants to satisfy all clients at the same time, and therefore aims to achieve a Pareto optimum. We revisit this interpretation in Chapter 6.

Chapter 3

NASH EQUILIBRIA FOR RELATIVE INVESTORS

Dating back to the pioneering work of Von Neumann and Morgenstern (1947) and Markowitz (1952), portfolio optimization problems have been treated extensively in the mathematical finance literature. It is usually assumed that one single investor aims to maximize an objective function applied to her terminal wealth, possibly under additional constraints. Such problems include expected utility maximization, mean-variance optimization, and many others. Although these problems might look very different, they have one thing in common: all of them consider only a single investor. However, due to the widespread empirical evidence on competition between fund managers (briefly presented in the introduction of this thesis), there is a rapidly growing strand of literature on competitive optimal investment and related issues.

In general, there are two different types of competitive optimization problems – zero-sum¹ and non-zero-sum games. Our focus lies solely on non-zero-sum games, for an example of a zero-sum investment game between two players we refer to Browne (2000).

Espinosa and Touzi (2015) argue that the most natural approach to modeling the interaction between competing investors is a general equilibrium model, where the behavior of the agents is coupled via market equilibrium² constraints. However, this leads to intractable calculations, which create the need for a different method to include interactions. It turns out that a different kind of equilibrium is better suited to treat the model mathematically. Instead of searching for market equilibria, portfolio optimization problems can be transformed to multi-objective portfolio optimization problems in which the search for Nash equilibria becomes the main goal. Most of the literature on many player games of wealth optimization includes the competitive feature into the problem by changing the argument of the objective function of a classical portfolio optimization

¹Zero-sum games are competitive games in which the objective functions of all participants add up to zero (see, e.g., Carmona, 2016, p. 177).

²A financial market is in equilibrium if market-clearing is realized. That is, if the supply and demand for assets in the market coincide (see, e.g., Yan, 2018, p. 47).

problem. Instead of the (terminal) wealth of a single investor, a so-called *relative performance metric* is used inside the objective function (see, e.g., Geng and Zariphopoulou, 2017; Dos Reis and Platonov, 2021, 2022). With the relative performance metric, each agent places a certain weight on the optimization of her performance compared to competitors in the same market. Hence, maximizing the relative performance metric allows the agent to simultaneously maximize her own state variable (e.g., her terminal wealth) while also comparing her outcome to that of her competitors. Using a relative performance metric transforms the single-agent portfolio optimization problem into a multi-objective portfolio optimization problem.

Most competitive portfolio optimization problems use either the additive or the multiplicative relative performance metric. For n agents with state variables X^1, \ldots, X^n , the *additive relative performance metric*

$$X^{i} - \frac{\theta_{i}}{n} \sum_{\substack{j=1\\j\neq i}}^{n} X^{j}$$

$$(3.1)$$

of agent $i \in \{1, ..., n\}$ is given as the difference of her own state variable and a weighted arithmetic mean of the other agents' state variables. Alternatively, the *multiplicative relative performance metric*

$$X^{i} \left(\prod_{j=1, \, j \neq i}^{n} X^{j}\right)^{-\frac{\theta_{j}}{n}} \tag{3.2}$$

of agent $i \in \{1, ..., n\}$ describes the quotient of her own state variable and a weighted geometric mean of those of the other agents. In most cases, the state variables $X^1, ..., X^n$ describe the wealth of n agents at the end of some time period [0, T], where T > 0 is fixed beforehand. In both expressions, the parameters $\theta_i \in [0, 1]$, i = 1, ..., n, indicate how much agent i cares about her performance with respect to her competitors. A larger choice of θ_i implies that the agent is more concerned with her relative performance. Hence, we refer to θ_i as the *competition weight* of agent i. Sometimes θ_i is also called concern rate or jealousy factor (see Bielagk et al., 2017). Throughout this thesis, we refer to the composition of a utility function and one of the relative performance metrics as the *competitive utility function*.

To the best of our knowledge, the additive relative performance metric first appeared in Palomino (2005) in a non-zero-sum game of n fund managers competing to maximize their compensation. As the argument of an expected utility criterion, it first appeared in the PhD thesis of Espinosa (2010). The results were later published by Espinosa and Touzi (2015). The linear structure of the additive relative performance metric makes the resulting optimization particularly tractable, especially when combined with an expected utility of CARA type. The combination of the additive relative performance metric and CARA utility was, for instance, used by Frei and Dos Reis (2011) as well as Lacker and Zariphopoulou (2019) in stock markets modeled by Itô diffusions with constant parameters, and by Deng et al. (2020) in a model with unobservable drift of the risky stock. Moreover, Hu and Zariphopoulou (2022) used CARA utility functions combined with the additive relative performance metric in a market model with stock price dynamics that contain possibly non-Markovian drift and volatility processes. Fu et al. (2020) modified (3.1) by allowing for a random competition weight parameter. Tangpi and Zhou (2022) used expected CARA utility in combination with a modified version of (3.1) as well. In their model, each agent i assigns a

specific competition weight θ_{ij} to each competitor $j \neq i$. Apart from CARA utility, the additive relative performance metric was also used in combination with a mean-variance criterion by Guan and Hu (2022) as well as Yang and Chen (2022), or a convex, monetary risk measure by Bielagk et al. (2017).

The multiplicative relative performance metric was (seemingly) first applied by Basak and Makarov (2014) in a model that contains n competing agents aiming to maximize expected (kinked) CRRA utility applied to (3.2) for the terminal wealth of the *n* investors. The multiplicative structure of (3.2) works well combined with CRRA utility and hence, this combination is very popular in the literature on many player games of wealth optimization. Basak and Makarov (2015) continued to work with expected CRRA utility applied to the multiplicative relative performance metric of two agents including an asset specialization incentive. Their work was later extended to an n-agent game by Whitmeyer (2019), allowing for competition weights θ_{ij} specifically assigned by agent i to competitor j. Similar problems were also considered by Lacker and Zariphopoulou (2019) as well as Lacker and Soret (2020), where the latter were focused on finding optimal investment-consumption strategies for n competing agents. The problem of optimal investment and consumption was also considered by Bo and Li (2022) in a stochastic growth model with jumps. In a generalized Heston model, Kraft et al. (2020) solved the problem of optimal investment for two players with CRRA utility combined with the multiplicative relative performance metric. Moreover, Wang and Ye (2023) maximized the running reward of n competing agents, where the objective function consists of the accumulated, exponentially discounted CRRA utility applied to a modified version of (3.2). A similar objective function, including a running as well as a terminal reward, was also used in the previously mentioned article by Bo and Li (2022).

Although the problem of (classical) expected utility maximization involving some kind of relative performance metric takes up a large portion of the competitive portfolio optimization literature, there are different strands of research that emerged from the seminal works of Espinosa (2010) and Basak and Makarov (2014). For example, Geng and Zariphopoulou (2017) combine the multiplicative relative performance metric with forward utilities. Their work has later been taken up and extended by Dos Reis and Platonov (2021, 2022) as well as Anthropelos et al. (2022). Another application of the relative performance metric lies in the study of competing insurance companies. This problem was considered by Deng et al. (2018), Guan and Hu (2022), and Yang and Chen (2022). A third strand of literature is focused on so-called market impact games in which competing agents are interested in the optimal execution or liquidation of a fixed position. Such issues arise in financial markets where the agents are "large" in the sense that their investment has an impact on the stock price movements. We give an explanation of such models and a short literature overview in Chapter 7.

In what follows, we are able to solve an *n*-player competitive portfolio optimization problem in which the agents maximize an expected utility function applied to the *additive* relative performance metric of terminal wealth. In contrast to most of the articles mentioned above, we do not need to specify the utility function. Moreover, we are able to solve the problem in a very general semimartingale financial market that covers most settings in the existing literature. The foundation lies in our ability to reduce the *n*-player problem to a classical single-investor portfolio optimization

problem. We can therefore rely on the wide variety of literature on single-investor portfolio optimization problems. However, we are not able to relax all assumptions used in the previously mentioned articles. In the following, we discuss the restrictions used throughout this chapter. Note that these limitations are present in the other chapters as well.

For one thing, we pose the assumption of full diversification for all n agents. This means that the assets traded in the financial market are available to each of the n investors. This restriction stands in contrast to the asset specialization models which are used frequently in the literature. In these settings, each agents specializes to a specific stock (or a set of stocks) and does not consider the other assets available in the market. The assumption of asset specialization was, for instance, used by Basak and Makarov (2015), Lacker and Zariphopoulou (2019), Lacker and Soret (2020), and Dos Reis and Platonov (2021, 2022), to name just a few. The main justification is that investors tend to invest into stocks they are familiar with (see Basak and Makarov, 2015, and their references). Other reasons include the reduction of trading or learning cost (see, e.g., Dos Reis and Platonov, 2022, and references therein). However, there is also an increasing number of research papers that consider a model of full diversification. Some examples are Frei and Dos Reis (2011), Espinosa and Touzi (2015), and Geng and Zariphopoulou (2017). The article by Basak and Makarov (2015) treats both asset specialization and diversification and provides a comparison of the optimal values in the two settings. It turns out that choosing a fully diverse portfolio is beneficial for risk-averse investors. Since our focus lies solely on risk-averse agents, the assumption of full diversification instead of asset specialization is reasonable. Moreover, full diversification has the additional advantage of being mathematically more manageable than problems where agents specialize to a specific stock.

Additionally, we need to assume that the agents agree upon their beliefs on the stock price dynamics. This assumption is of purely technical nature, a model allowing for heterogeneous beliefs on the stock price dynamics would, apparently, be more realistic. There are a few articles allowing for heterogeneous beliefs, like the market impact game considered by Evangelista and Thamsten (2021). However, the literature on n-agent games under heterogeneous beliefs on the market dynamics is still very sparse.

Another rather restrictive assumption is that each agent faces the same time horizon T > 0. This assumption is solely for mathematical purposes, since the empirical evidence suggests the exact opposite. Geng and Zariphopoulou (2017) and Anthropelos et al. (2022) discuss this restriction in more detail. We merely give a short summary of their arguments. While it is typical for managers to report their performance within standardized time frames (e.g., quarterly or annually), agents do not act in a single trading horizon but consider their past performance. Hence, we cannot divide the optimization problems, which they face over a longer time period, into a sequence of decisions over standardized time intervals. Moreover, horizons of performance evaluation are company specific factors and cannot be generalized for a large number of agents. However, as explained at the beginning of this paragraph, we need to make the assumption of a common time horizon for tractability reasons. It should be noted that the assumption is very common throughout the literature. To the best of our knowledge, Anthropelos et al. (2022) were the only authors to allow for different time horizons among the competing agents. Finally, we need to make the assumption of each agent having full information about their competitors' model parameters (e.g., initial capital, risk tolerance, and competition weight). We can refer to Basak and Makarov (2015), Wang and Ye (2023), and the references therein for how these parameters can be estimated from observed data.

To conclude this introduction, we give a short overview of the current chapter. In Section 3.1, we state the *n*-player portfolio optimization problem based on the semimartingale financial market presented in Subsection 2.3.1. The statement of the problem is followed by Section 3.2, where we find the unique Nash equilibrium for the *n*-player optimization problem. More specifically, questions of existence and uniqueness of a Nash equilibrium are reduced to asking for the existence and uniqueness of an optimal solution to the corresponding single-investor problem. At the end of this chapter, we provide a brief discussion of the general equilibrium, which is taken up and extended in Chapter 4 in a variety of examples covered by the semimartingale model from Subsection 2.3.1.

3.1. Relative performance problem under general utility

We assume that there are n agents trading in the semimartingale market from Subsection 2.3.1. To summarize, there are d + 1 assets – a riskless bond with zero interest rate and d risky stocks with price process $(S(t))_{t \in [0,T]} = (S_1(t), \ldots, S_d(t))_{t \in [0,T]}$, defined on the finite time interval [0,T]. The stock prices are given as $L^2(\mathbb{P})$ -semimartingales with càdlàg paths. To exclude arbitrage strategies, we require the existence of an equivalent σ -martingale measure \mathbb{Q} .

Each of the n agents has the same investment opportunities provided by the semimartingale financial market. Thus, we work in the framework of full diversification. To include interaction into the optimization problem, we combine the additive relative performance metric with the expected utility of terminal wealth. The n agents are allowed to be heterogeneous in the sense that their initial wealth, their preference parameters, and their utility functions are allowed to be distinct.

To be more specific, we consider the following setting. Each investor i is endowed with an initial capital $x_0^i \in \mathbb{R}$. The preferences of agent i are measured with respect to a utility function $U_i : \mathcal{D}_i \to \mathbb{R}, \ \mathcal{D}_i \in \{(0, \infty), \mathbb{R}\}$, and a competition weight $\theta_i \in [0, 1], i = 1, \ldots, n$. The utility function can be rather general here. We only require U_i to be continuous and strictly increasing (see also Definition 2.11). For convenience, we extend all utility functions to functions on \mathbb{R} by setting $U_i(x) = -\infty$ if $x \notin \mathcal{D}$. A similar convention was used by Kramkov and Schachermayer (1999). Note that this does not affect the optimal value of the problem. The objective function of agent $i \in \{1, \ldots, n\}$ consists of her expected utility applied to the additive relative performance metric of her own as well as the other agents' terminal wealth. Moreover, she is limited to her initial capital. Hence, agent i aims to solve the following optimization problem

$$\begin{cases} \sup_{\varphi^i \in \mathcal{A}} \mathbb{E} \left[U_i \left(X_T^{i,\varphi^i} - \theta_i \bar{X}_T^{-i,\varphi} \right) \right], \\ \text{s.t.} \quad X_T^{i,\varphi^i} = x_0^i + (\varphi^i \cdot S)_T, \end{cases} \quad i = 1, \dots, n,$$
(3.3)

where $\bar{X}_T^{-i,\varphi} = \frac{1}{n} \sum_{j \neq i} X_T^{j,\varphi^j}$ and $\varphi^j \in \mathcal{A}, j \neq i$, are fixed admissible strategies and \mathcal{A} denotes the set of admissible strategies from (2.4). For the later analysis, it is slightly more convenient to scale the sum by n instead of n-1.

Using a weighted difference of the terminal wealth of all n agents instead of only her own terminal wealth incorporates the fact that agent i wants to maximize her own terminal wealth, while also making a good performance with respect to the other agents. We sometimes refer to the function $X \mapsto U_i \left(X - \theta_i \bar{X}_T^{-i,\varphi} \right)$ as the competitive utility function of agent i.

Remark 3.1. We need to ensure that agent *i* is able to attain a terminal wealth X_T^{i,φ^i} such that $X_T^{i,\varphi^i} - \theta_i \bar{X}_T^{-i,\varphi} \in \mathcal{D}_i$ P-almost surely. Otherwise, the problem is trivial. We will later see that this condition is satisfied when we choose the competition weight $\theta_i \in [0, 1]$ under the constraint $x_0^i - \frac{\theta_i}{n} \sum_{j \neq i} x_0^j \in \mathcal{D}_i$. Obviously, this constraint is only relevant if $\mathcal{D}_i = (0, \infty)$. In this case, we need to make sure that

$$x_0^i - \frac{\theta_i}{n} \sum_{j \neq i} x_0^j > 0$$

is satisfied. This is equivalent to

$$\left(1+\frac{\theta_i}{n}\right)x_0^i - \frac{\theta_i}{n}\sum_{j=1}^n x_0^j > 0 \Longleftrightarrow \frac{\frac{\theta_i}{n}}{1+\frac{\theta_i}{n}} < \frac{x_0^i}{\sum_{j=1}^n x_0^j} \Longleftrightarrow \frac{\theta_i}{n} < \frac{\alpha_i}{1-\alpha_i}$$

where

$$\alpha_i := \frac{x_0^i}{\sum_{j=1}^n x_0^j}$$

describes the fraction of the collective initial capital originating from investor *i*. Therefore, the constraint $x_0^i - \frac{\theta_i}{n} \sum_{j \neq i} x_0^j > 0$ implies an upper bound on the choice of θ_i , which is increasing in the fraction α_i . Hence, the upper bound is increasing in terms of the *i*-th agent's initial capital and decreasing in terms of the other n-1 agents' initial investment. We can interpret this observation as follows: The more an investor contributes in the beginning, the more she may care about the other agents' investment behavior. Moreover, the upper bound on θ_i is strictly smaller than 1 if, and only if, $\alpha_i < (n+1)^{-1}$, i.e., if the fraction of initial wealth originating from agent *i* is smaller than $(n+1)^{-1}$. If, for example, each agent invests the same amount in the beginning, there are no restrictions on the competition weight parameters.

The optimization problem (3.3) contains n objective functions, where each objective function depends on all n strategies φ^j , $j = 1, \ldots, n$. Since these objective functions need to be maximized simultaneously, (3.3) is a multi-objective optimization problem. There is no a priori given notion of optimality for such problems. We work with the concept of *Nash equilibria*, i.e., each investor tries to maximize her own objective function while the strategies of the other investors are assumed to be fixed. For the definition of a Nash equilibrium, we refer to Definition 2.13 as well as Definition 2.15. To summarize, a Nash equilibrium is a vector of admissible strategies, chosen by the players, such that no agent would benefit from deviating unilaterally from her equilibrium strategy. To be more specific, we search for open-loop Nash equilibria. For a differentiation between open-loop and closed-loop Nash equilibria, we refer to Section 2.4.

3.2. Solution method via problem reduction

Next, we explain how to find (open-loop) Nash equilibria for (3.3). Thus, each investor tries to maximize her own objective function while the strategies of the other investors are assumed to be fixed. This maximization, often called *best response problem*, results in the optimal strategy of agent *i* in terms of the strategies of her competitors. The second step of the optimization process is a fixed point problem in order to find the *n*-tuple of admissible strategies to satisfy each investors' preference determined in the first step.

Generally, one would proceed in the previously described way by fixing some investor i, fixing the other agents' strategies, maximizing the *i*-th objective function, and solving the fixed point problem afterwards. While being very intuitive, the described approach brings some disadvantages. First, the fixed point problem in the second step can be nonlinear and hence, hard or even impossible to solve. Moreover, the method depends strongly on the explicit model and often requires a very restrictive mathematical environment. For example, Lacker and Zariphopoulou (2019) were only able to consider constant Nash equilibria and a Black-Scholes type market with constant market parameters. Note that the restrictive environment is also partly caused by the asset specialization framework in their paper.

To avoid these technical difficulties, we choose a different approach to find a Nash equilibrium and discuss its uniqueness, which enables us to solve the problem in a very general financial market without any specific assumptions on the utility function. Moreover, the method can even be extended to problems beyond utility maximization (see Remark 3.4 and Section 4.5).

First, we consider the expression inside the utility function in (3.3). Since agent *i* can only control her own strategy, the random variable $\theta_i \bar{X}_T^{-i,\varphi}$ can be understood as a fixed asset claim. An arbitrary strategy $\varphi \in \mathcal{A}$ can due to linearity always be decomposed into

$$X_T^{\varphi} = X_T^{\varphi} - X_T^{\varphi'} + X_T^{\varphi'} = X_T^{\varphi-\varphi'} + X_T^{\varphi'},$$

for arbitrary $\varphi' \in \mathcal{A}$. Investor *i* can, without loss of generality, invest some fraction of her initial capital in order to hedge the claim $\theta_i \bar{X}_T^{-i,\varphi}$. The remaining part of her initial capital can then be used to maximize her own terminal wealth. This idea leads to a much simpler method of determining Nash equilibria in the given context.

The first step is to determine the price of the claim $\theta_i \bar{X}_T^{-i,\varphi}$. Each investor j, where $j \neq i$, has some initial capital x_0^j . Hence, the time zero price of X_T^{j,φ^j} equals the initial capital x_0^j (see (2.6) in Subsection 2.3.1). By linearity, the price of the claim $\theta_i \bar{X}_T^{-i,\varphi}$ is simply given by

$$\theta_i \bar{x}_0^{-i} := \frac{\theta_i}{n} \sum_{j \neq i} x_0^j.$$

This shows that the time zero price is independent of the strategies φ^j , $j \neq i$, chosen by the other investors. Hence, the maximization problem in the second step does not depend on the other n-1 agents' strategies.

In the second step, investor i needs to solve a classical portfolio optimization problem, using the

utility function U_i and the reduced initial capital $\tilde{x}_0^i := x_0^i - \theta_i \bar{x}_0^{-i}$. The portfolio optimization problem

$$\begin{cases} \sup_{\substack{\psi^i \in \mathcal{A} \\ \text{s.t.} \quad Y_T^{i,\psi^i} = \widetilde{x}_0^i + (\psi^i \cdot S)_T, \end{cases}} (3.4) \end{cases}$$

can be solved using standard methods. Existence and uniqueness of a solution does of course depend on the specific choice of the utility function and underlying financial market. We assume here that there exists a unique optimal strategy $\psi^{i,*} \in \mathcal{A}$ for (3.4).

In the last step, the Nash equilibria are determined using the linearity of the price process. By construction, the process $(Y_t^{i,\psi^{i,*}})_{t\in[0,T]}$ can be written as

$$Y_t^{i,\psi^{i,*}} = X_t^{i,\varphi^i} - \frac{\theta_i}{n} \sum_{j \neq i} X_t^{j,\varphi^j}$$

 \mathbb{P} -almost surely for all $t \in [0, T]$. Then, using the linearity of the wealth process (recall that we assumed in Subsection 2.3.1 that stock prices are not identical) and the uniqueness of $\psi^{i,*}$, the strategies $\varphi^j \in \mathcal{A}$ are obtained from

$$\psi_k^{i,*}(t) = \varphi_k^i(t) - \frac{\theta_i}{n} \sum_{j \neq i} \varphi_k^j(t), \quad k = 1, \dots, d, \ i = 1, \dots, n,$$
(3.5)

 \mathbb{P} -almost surely for all $t \in [0, T]$. Hence, the Nash equilibria can be determined as the solution to the system of linear equations (3.5), where the solutions $\psi_k^{i,*}$ to the classical problems are given. If (3.4) and (3.5) have a unique solution, the resulting Nash equilibrium is unique as well. The question of uniqueness has been posed as an open problem in Lacker and Zariphopoulou (2019). Our setting gives a partial answer to their question due to the asset specialization feature in their model.

Theorem 3.2. If (3.4) has a unique (up to modifications) optimal portfolio strategy $\psi^{i,*}$ for all $i \in \{1, ..., n\}$, then there exists a unique Nash equilibrium for (3.3) given by

$$\varphi_k^i(t) := \frac{n}{n+\theta_i} \psi_k^{i,*}(t) + \frac{\theta_i}{(n+\theta_i)(1-\hat{\theta})} \sum_{j=1}^n \frac{n}{n+\theta_j} \psi_k^{j,*}(t),$$
(3.6)

 $k = 1, \ldots, d, i = 1, \ldots, n, \mathbb{P}$ -almost surely for all $t \in [0, T]$, where $\hat{\theta} \coloneqq \sum_{i=1}^{n} \frac{\theta_i}{n+\theta_i}$.

The proof of Theorem 3.2 uses the following lemma.

Lemma 3.3. Let $\theta_i \in [0, 1], i = 1, ..., n$. Then

$$\hat{\theta} = \sum_{i=1}^{n} \frac{\theta_i}{n + \theta_i} \in [0, 1).$$

Proof. Since $\theta_i \in [0, 1], i = 1, \ldots, n$, we obtain

$$0 \le \hat{\theta} = \sum_{i=1}^{n} \frac{\theta_i}{n + \theta_i} = \sum_{i=1}^{n} \left(1 + \frac{n}{\theta_i} \right)^{-1} \le \sum_{i=1}^{n} (1 + n)^{-1} = \frac{n}{n+1} < 1.$$
Proof (Theorem 3.2). Without knowing $\psi^{i,*}$ explicitly, we can determine the solution to (3.5) in terms of $\psi^{i,*}$, $i = 1, \ldots, n$, for all $t \in [0, T]$ and \mathbb{P} -almost all $\omega \in \Omega$. Therefore, we fix some arbitrary $t \in [0, T]$ and $\omega \in \Omega$ (we omit the argument ω in the following calculations), and define

$$\widehat{\varphi}_k(t) := \sum_{j=1}^n \varphi_k^j(t), \quad k = 1, \dots, d.$$
(3.7)

Hence, (3.5) is equivalent to

$$\psi_k^{i,*}(t) = \frac{n+\theta_i}{n}\varphi_k^i(t) - \frac{\theta_i}{n}\widehat{\varphi}_k(t)$$

and therefore, $\varphi_k^i(t)$ is implicitly given by

$$\varphi_k^i(t) = \frac{n}{n+\theta_i} \left(\psi_k^{i,*}(t) + \frac{\theta_i}{n} \widehat{\varphi}_k(t) \right).$$
(3.8)

Inserting (3.8) into (3.7) yields

$$\widehat{\varphi}_k(t) = \sum_{i=1}^n \varphi_k^i(t) = \sum_{i=1}^n \frac{n}{n+\theta_i} \psi_k^{i,*}(t) + \widehat{\varphi}_k(t) \sum_{i=1}^n \frac{\theta_i}{n+\theta_i}$$
$$= \sum_{i=1}^n \frac{n}{n+\theta_i} \psi_k^{i,*}(t) + \widehat{\theta}\widehat{\varphi}_k(t),$$

which is equivalent to

$$(1-\hat{\theta})\widehat{\varphi}_k(t) = \sum_{i=1}^n \frac{n}{n+\theta_i} \psi_k^{i,*}(t),$$

where we used the abbreviation of $\hat{\theta}$. Therefore, using Lemma 3.3, we can deduce further that

$$\widehat{\varphi}_k(t) = \frac{1}{1 - \widehat{\theta}} \sum_{i=1}^n \frac{n}{n + \theta_i} \psi_k^{i,*}(t)$$
(3.9)

holds \mathbb{P} -almost surely for all $t \in [0, T]$. Finally, inserting (3.9) into (3.8) yields the stated expression for the Nash equilibrium. The line of arguments implies that there exists a unique (up to modifications) Nash equilibrium given by (3.6) if, and only if, there exists a unique (up to modifications) optimal portfolio strategy $\psi^{i,*}$ to the auxiliary problem (3.4).

Remark 3.4. The described method is not limited to classical expected utility maximization. Some examples of other types of optimization problems that can be treated with the described method are the VaR-based optimization by Basak and Shapiro (2001), the rank-dependent utility model with a VaR-based constraint by He and Zhou (2016), or general mean-variance problems that can be found for example in Korn (1997). Another example is the cumulative prospect theory (CPT) by Tversky and Kahneman (1992), that is further analyzed in Section 4.5. Moreover, the arguments presented in the current section also apply for financial markets in discrete time. In order to keep the setting as simple, but also as general as possible, we consider a setting in continuous time that covers many important market models (see Chapter 4 for some examples). However, the arguments in the current section do not depend at all on the underlying financial market, so that one could also have a general arbitrage-free financial market in discrete time in mind. Later, we take a closer look at a discrete, time financial market in Section 4.3.

Additionally, we want to point out that the use of the *additive* relative performance metric is crucial here. One could argue that a similar approach might work for the multiplicative relative performance metric if one uses the invested fraction instead of the number of shares. In that case, the wealth of an agent is a stochastic exponential and thus, the multiplicative relative performance metric applied to the terminal wealth of the agents has an exponential structure as well. However, the expression inside the exponential function is not linear since, for general semimartingales Xand Y,

$$\mathcal{E}(X) \cdot \mathcal{E}(Y) = \mathcal{E}(X + Y + \langle X, Y \rangle) \neq \mathcal{E}(X + Y)$$

holds in the (interesting) cases where $\langle X, Y \rangle \neq 0$. Recall that $\mathcal{E}(X)$ denotes the stochastic exponential of X (see Theorem 2.4). For the first equality, we refer to Theorem 38 in Protter (2005, p. 86).

Remark 3.5. Since the Nash equilibrium given in Theorem 3.2 explicitly contains the optimal solutions to the associated classical portfolio optimization problems, we can compare the optimal solution $\psi_k^{i,*}$ of the classical problem to the associated component φ_k^i of the Nash equilibrium. Indeed, if we set $\theta_i = 0$, the agent does not care about the other agents and just solves the classical portfolio problem, i.e., $\varphi_k^i = \psi_k^{i,*}$. If $\psi_k^{i,*} > 0$ (< 0) and $\sum_{j \neq i} \frac{n}{n+\theta_j} \psi_k^{j,*} > 0$ (< 0), $\varphi_k^i(t)$ is increasing (decreasing) in terms of θ_i , which can be seen directly considering the partial derivative

$$\begin{split} \frac{\partial}{\partial \theta_i} \varphi_k^i(t) = & \frac{n}{(n+\theta_i)^2} \left(1 + \frac{\theta_i}{(n+\theta_i)(1-\hat{\theta})} \right) \left(\frac{n}{(n+\theta_i)(1-\hat{\theta})} - 1 \right) \psi_k^{i,*}(t) \\ & + \frac{n}{(n+\theta_i)^2(1-\hat{\theta})} \left(1 + \frac{\theta_i}{(n+\theta_i)(1-\hat{\theta})} \right) \sum_{j \neq i} \frac{n}{n+\theta_j} \psi_k^{j,*}(t). \end{split}$$

Except from the third factor in the first summand, all factors are obviously non-negative. Moreover,

$$\frac{n}{(n+\theta_i)(1-\hat{\theta})} \ge \frac{n}{(n+\theta_i)\left(1-\frac{\theta_i}{n+\theta_i}\right)} = 1,$$
(3.10)

using the definition of $\hat{\theta}$. Hence, the remaining factor is non-negative as well. Thus, under these conditions in a competitive environment ($\theta_i > 0$), agent *i* invests more in the stocks than in a classical non-competitive environment ($\theta_i = 0$). A similar result was obtained by Espinosa and Touzi (2015) (in a less general setting).

CHAPTER 4

Application of the method from Section 3.2

In the following, we apply the method described in Section 3.2 to different examples from the literature, including a discrete-time market, a market with jumps, and a stochastic volatility model, as well as a problem that goes beyond expected utility maximization. In Section 4.1, we consider agents applying exponential utility functions in a Lévy market with jumps. A special case of the Lévy market, a Black-Scholes market, is considered in Section 4.2. There, we consider exponential utilities in Subsection 4.2.1, as well as power and logarithmic utility functions in Subsection 4.2.2. Section 4.3 covers a financial market in discrete time, the Cox-Ross-Rubinstein market, in which agents use exponential utilities. In Section 4.4, we consider a model with stochastic volatility, and the final Section 4.5 contains an example that goes beyond expected utility maximization, namely, cumulative prospect theory.

It should be noted that the definition of the set \mathcal{A} of admissible strategies changes throughout this chapter. In each example, we take \mathcal{A} to be the set of strategies satisfying the necessary assumptions for the specific example according to the paper from which the example is taken.

4.1. Lévy market

The first example is a Lévy market taken from Bäuerle and Blatter (2011). The market consists of a riskless bond with interest rate r = 0 and d stocks. Let W be a d-dimensional Brownian motion and N a Poisson random measure on $[0,T] \times (-1,\infty)^d$, i.e., $N([0,t] \times B)$ is the number of all jumps taking values in the set $B \in \mathcal{B}((-1,\infty)^d)$ up to the time $t \in [0,T]$, where $\mathcal{B}((-1,\infty)^d)$ denotes the Borel σ -algebra on $(-1,\infty)^d$. For the definition of a Poisson random measure (or Poisson point process), we refer to Definition 19.1 of Sato (1999) as well as Last and Penrose (2017, p. 19). We denote the associated Lévy measure by ν , i.e., $\nu(B) = \mathbb{E}[N([0,1] \times B)]$ gives the expected number of jumps per unit time whose size belongs to B (see, e.g., Cont and Tankov, 2004, Definition 3.4). For notational convenience, let us define $\overline{N}(dt, dz) \coloneqq N(dt, dz) - \mathbb{1}_{\{||z|| < 1\}}\nu(dz)dt$. The price processes for $k \in \{1, ..., d\}$ are given by the stochastic differential equation

$$dS_k(t) = S_k(t_-) \left(\mu_k \, dt + \sum_{\ell=1}^d \sigma_{k\ell} \, dW_\ell(t) + \int_{(-1,\infty)^d} z_k \, \bar{N}(dt, dz) \right), \tag{4.1}$$

where $S_k(0) = 1$, $\sigma_{k\ell} \ge 0$, $k, \ell = 1, \ldots, d$. By $\mu = (\mu_1, \ldots, \mu_d)$ we denote the drift vector and by $\sigma = (\sigma_{k\ell})_{1 \le k, \ell \le d}$ the volatility matrix, which is assumed to be regular. The restriction of N onto $[0, T] \times (-1, \infty)^d$, which has the interpretation that the size of any jump is strictly larger than -1, ensures that the stock price processes (4.1) stay positive (see, e.g., Proposition 8.21 in Cont and Tankov, 2004, or Theorem 2.4).

There are *n* investors trading in the Lévy market, each endowed with an initial capital of $x_0^i \in \mathbb{R}$, i = 1, ..., n. We assume that each investor uses an exponential utility function

$$U_i : \mathbb{R} \to \mathbb{R}, \ U_i(x) = -\exp\left(-\frac{1}{\delta_i}x\right),$$

for parameters $\delta_i > 0$, i = 1, ..., n. Hence, the objective function of agent *i* is given by

$$\mathbb{E}\left[-\exp\left(-\frac{1}{\delta_i}\left(X_T^{i,\varphi^i}-\theta_i\bar{X}_T^{-i,\varphi}\right)\right)\right],$$

for the competition weight parameters $\theta_i \in [0, 1]$, i = 1, ..., n. We assume that the market is free of arbitrage for an appropriate strategy class and that

$$\int_{\|z\|>1} \|z\| \exp\left(\frac{1}{\delta_i} \Lambda_i \|z\|\right) \nu(\mathrm{d}z) < \infty$$
(4.2)

for constants $0 < \Lambda_i < \infty$, i = 1, ..., n. The integrability condition (4.2) is used in the proof of Theorem 2.1 by Bäuerle and Blatter (2011), which provides the unique optimal solution to the problem of maximizing exponential utility shown below.

Now we use the method from Section 3.2 to determine the Nash equilibrium in the given situation. First, the unique (up to modifications) optimal solution to the optimization problem

$$\begin{cases} \sup_{\psi^i \in \mathcal{A}} \mathbb{E} \left[-\exp\left(-\delta_i^{-1} Y_T^{i,\psi^i} \right) \right], \\ \text{s.t.} \quad Y_T^{i,\psi^i} = \widetilde{x}_0^i + (\psi^i \cdot S)_T, \end{cases}$$

for $\widetilde{x}_0^i = x_0^i - \frac{\theta_i}{n} \sum_{j \neq i} x_0^j$, is given by

$$\psi_k^{i,*}(t) = \frac{\pi_k^{i,*}}{S_k(t)}.$$

Here, $(\pi_1^{i,*}, \ldots, \pi_d^{i,*})$ is the (deterministic and constant) solution of

$$0 = \mu_k - \frac{1}{\delta_i} \sum_{\ell=1}^d \sum_{r=1}^d \pi_r^{i,*} \sigma_{k\ell} \sigma_{r\ell} + \int_{(-1,\infty)^d} z_k \left(\exp\left(-\frac{1}{\delta_i} \sum_{r=1}^d \pi_r^{i,*} z_r\right) - \mathbb{1}_{\{\|z\| < 1\}} \right) \nu(\mathrm{d}z), \quad (4.3)$$

k = 1, ..., d, which we assume to exist and be unique (see Bäuerle and Blatter, 2011). Regarding the existence of a solution to (4.3), Blatter (2009) states that one could apply a fixed point theorem which can be found, for example, in Schäfer (2007). If the parameters of the model are chosen such that the conditions of the fixed point theorem are satisfied, we can guarantee the existence of a solution to (4.3). However, according to Blatter (2009), it is not possible to derive explicit conditions on the model parameters.

Using Theorem 3.2, we know that the unique (up to modifications) Nash equilibrium is then given by

$$\varphi_k^{i,*}(t) = \frac{n}{n+\theta_i} \psi_k^{i,*}(t) + \frac{\theta_i}{(n+\theta_i)(1-\hat{\theta})} \sum_{j=1}^n \frac{n}{n+\theta_j} \psi_k^{j,*}(t)$$

P-almost surely for all $t \in [0,T]$ and therefore, the Nash equilibrium $\pi_k^i(t) \coloneqq \varphi_k^{i,*}(t)S_k(t)$ for $k = 1, \ldots, d, i = 1, \ldots, n$, in terms of invested amounts, is deterministic and constant.

4.2. Black-Scholes Market

Our next example is a special case of the setting used in Section 4.1. We consider a Black-Scholes market consisting of a riskless bond with interest rate r = 0 and d stocks with price processes

$$dS_k(t) = S_k(t) \left(\mu_k dt + \sum_{\ell=1}^d \sigma_{k\ell} dW_\ell(t) \right), \ S_k(0) = 1, \ k = 1, \dots, d,$$

driven by a *d*-dimensional Brownian motion $W = (W_1, \ldots, W_d)$. By $\mu = (\mu_1, \ldots, \mu_d)$ we denote the drift vector and by $\sigma = (\sigma_{jk})_{1 \le j,k \le d}$ the volatility matrix, which is again assumed to be regular. This model can be obtained from the Lévy market described in Section 4.1 by omitting the jump component.

4.2.1. Optimization under exponential utility

First, we assume that agents use exponential utility functions with risk tolerance parameters $\delta_i > 0, i = 1, ..., n$. The method from Section 3.2 can be used to find all Nash equilibria for

$$\mathbb{E}\bigg[-\exp\bigg(-\frac{1}{\delta_i}\Big(X_T^{i,\varphi^i}-\theta_i\bar{X}_T^{-i,\varphi}\Big)\bigg)\bigg],\ i=1,\ldots,n$$

The unique optimal solution to the auxiliary problem

$$\begin{cases} \sup_{\psi^i \in \mathcal{A}} \mathbb{E} \Big[-\exp\left(-\delta_i^{-1} Y_T^{i,\psi^i}\right) \Big], \\ \text{s.t.} \quad Y_T^{i,\psi^i} = \widetilde{x}_0^i + (\psi^i \cdot S)_T, \end{cases}$$

is given by

$$\psi_k^i(t)S_k(t) = \delta_i(\sigma\sigma^{\top})^{-1}\mu, \ k = 1, \dots, d, \ i = 1, \dots, n$$

(see, e.g., Eberlein and Kallsen, 2019, Example 10.20, for one stock – the extension to d stocks is straightforward). Therefore, the amount invested into the d stocks is constant in time and

deterministic. Hence, Theorem 3.2 implies that the unique (up to modifications) Nash equilibrium in terms of invested amounts is given by

$$\pi^{i,*}(t) = \left(\frac{n\delta_i}{n+\theta_i} + \frac{\theta_i}{(n+\theta_i)(1-\hat{\theta})}\sum_{j=1}^n \frac{n\delta_j}{n+\theta_j}\right) \cdot (\sigma\sigma^{\top})^{-1}\mu, \ t \in [0,T], \ i = 1, \dots, n.$$
(4.4)

Thus, the Nash equilibrium in terms of invested amounts is deterministic and constant in time.

Remark 4.1. If we set d = 1, $\mu_1 = \mu$, and $\sigma_1 = \sigma > 0$, we obtain the constant Nash equilibrium found by Lacker and Zariphopoulou (2019, Corollary 2.4). They use the slightly different objective function

$$\mathbb{E}\bigg[-\exp\bigg(-\frac{1}{\delta_i}\bigg(X_T^{i,\varphi^i}-\frac{\theta_i}{n}\sum_{j=1}^n X_T^{j,\varphi^j}\bigg)\bigg)\bigg],$$

which can be rewritten as

$$\mathbb{E}\left[-\exp\left(-\frac{1}{\tilde{\delta}_{i}}\left(X_{T}^{i,\varphi^{i}}-\tilde{\theta}_{i}\bar{X}_{T}^{-i,\varphi}\right)\right)\right]$$
(4.5)

by introducing the parameters $\tilde{\delta}_i = \frac{\delta_i}{1-\frac{\theta_i}{n}}$ and $\tilde{\theta}_i = \frac{\theta_i}{1-\frac{\theta_i}{n}}$. Hence, solving (4.5) for $i = 1, \ldots, n$ yields the (one-dimensional) Nash equilibrium

$$\pi_i^* = \left(\delta_i + \theta_i \frac{\bar{\delta}_n}{1 - \bar{\theta}_n}\right) \frac{\mu}{\sigma^2}$$

where we used the auxiliary calculations

$$\frac{n\tilde{\delta}_i}{n+\tilde{\theta}_i} = \delta_i, \ \sum_{j=1}^n \frac{\tilde{\theta}_j}{n+\tilde{\theta}_j} = \bar{\theta}_n,$$

and the abbreviations $\bar{\theta}_n \coloneqq \frac{1}{n} \sum_{j=1}^n \theta_j$, $\bar{\delta}_n = \frac{1}{n} \sum_{j=1}^n \delta_j$. It is important to notice here that Lacker and Zariphopoulou (2019) had to restrict their problem to finding *constant* Nash equilibria while we are able to consider a more general Nash equilibrium which turns out to be constant. \diamond

COMPARISON TO THE SINGLE-AGENT OPTIMIZATION PROBLEM

In the single-agent problem of maximizing the expected exponential utility of the terminal wealth in the underlying Black-Scholes market, the optimally invested amount is known to be given by

$$\pi^{i,*} = \delta_i (\sigma \sigma^\top)^{-1} \mu. \tag{4.6}$$

A comparison of the optimal portfolio (4.6) in the single-agent problem and the Nash equilibrium (4.4) shows that the overall structure is the same. In both cases, the optimal amount invested into the *d* stocks is constant over time and deterministic. Moreover, the optimally invested amount is given as $(\sigma\sigma^{\top})^{-1}\mu$ multiplied by a constant. We define the constant in (4.4) by C_i , i.e.,

$$C_i \coloneqq \frac{n\delta_i}{n+\theta_i} + \frac{\theta_i}{(n+\theta_i)(1-\hat{\theta})} \sum_{j=1}^n \frac{n\delta_j}{n+\theta_j}.$$
(4.7)

In the single-agent scenario ($\theta_i = 0$), the constant is simply given by δ_i .

Let us discuss some properties of C_i . First, we rewrite C_i as

$$C_i = \frac{n\delta_i}{n+\theta_i} \left(1 + \frac{\theta_i}{(n+\theta_i)(1-\hat{\theta})} \right) + \frac{\theta_i}{(n+\theta_i)(1-\hat{\theta})} \sum_{j \neq i} \frac{n\delta_j}{n+\theta_j}.$$

Regarding the monotonicity of C_i in terms of θ_i , we take the first order partial derivative with respect to θ_i . Thus,

$$\frac{\partial}{\partial \theta_i} C_i = -\frac{n\delta_i}{(n+\theta_i)^2} \left(1 + \frac{\theta_i}{(n+\theta_i)(1-\hat{\theta})} \right) \\
+ \frac{n\delta_i}{n+\theta_i} \left(\frac{n}{(n+\theta_i)^2} \frac{1}{1-\hat{\theta}} + \frac{\theta_i}{n+\theta_i} \frac{1}{(1-\hat{\theta})^2} \frac{n}{(n+\theta_i)^2} \right) \\
+ \sum_{j \neq i} \frac{n\delta_j}{n+\theta_j} \left(\frac{n}{(n+\theta_i)^2} \frac{1}{1-\hat{\theta}} + \frac{\theta_i}{n+\theta_i} \frac{1}{(1-\hat{\theta})^2} \frac{n}{(n+\theta_i)^2} \right) \\
= \frac{n\delta_i}{(n+\theta_i)^2} \left(\frac{n}{n+\theta_i} \frac{1}{1-\hat{\theta}} - 1 \right) \left(1 + \frac{\theta_i}{(n+\theta_i)(1-\hat{\theta})} \right)$$
(4.8)

$$+\sum_{j\neq i}\frac{n\delta_j}{n+\theta_j}\left(\frac{n}{(n+\theta_i)^2}\frac{1}{1-\hat{\theta}}+\frac{\theta_i}{n+\theta_i}\frac{1}{(1-\hat{\theta})^2}\frac{n}{(n+\theta_i)^2}\right).$$
(4.9)

The second summand (4.9) is strictly positive since $\hat{\theta} < 1$ for every admissible choice of $\theta_1, \ldots, \theta_n$ (see Lemma 3.3). Moreover, all factors in the first summand (4.8) are non-negative (see (3.10) for the second factor). Hence, the constant C_i is strictly increasing in terms of θ_i . Moreover, the factors C_i and δ_i coincide if, and only if, $\theta_i = 0$.

Let us now analyze the influence of θ_i on the optimal terminal wealth of agent *i*. As the optimally invested amount for agent *i* is deterministic and constant in time, the optimal terminal wealth is given by

$$\begin{aligned} X_T^{i,\varphi^{i,*}} &= x_0^i + C_i \mu^\top (\sigma \sigma^\top)^{-1} \mu T + C_i \mu^\top (\sigma \sigma^\top)^{-1} \sigma W(T) \\ &= x_0^i + C_i \| \sigma^{-1} \mu \|^2 T + C_i (\sigma^{-1} \mu)^\top W(T). \end{aligned}$$

Therefore, the expected optimal terminal wealth is given by

$$\mathbb{E}\Big[X_T^{i,\varphi^{i,*}}\Big] = x_0^i + C_i \|\sigma^{-1}\mu\|^2 T$$
(4.10)

and thus, strictly increasing in terms of θ_i . Hence, in order to maximize the expected terminal wealth, agent *i* should choose the competition weight $\theta_i = 1$.

However, choosing a high competition weight also brings the disadvantage of increasing the probability of a loss (with respect to the initial capital x_0^i). To justify this assertion, we choose a constant $K < x_0^i$ and consider the probability that the optimal terminal wealth $X_T^{i,\varphi^{i,*}}$ is less or equal than K. Since W(T) has a d-dimensional normal distribution, we obtain

$$(\sigma^{-1}\mu)^{\top}W(T) \sim \mathcal{N}(0, \|\sigma^{-1}\mu\|^2 T),$$

and therefore, it follows

$$\mathbb{P}\left(X_{T}^{i,\varphi^{i,*}} \leq K\right) = \mathbb{P}\left(x_{0}^{i} + C_{i} \|\sigma^{-1}\mu\|^{2}T + C_{i}(\sigma^{-1}\mu)^{\top}W(T) \leq K\right) \\
= \mathbb{P}\left(\frac{(\sigma^{-1}\mu)^{\top}W(T)}{\|\sigma^{-1}\mu\|\sqrt{T}} \leq \frac{K - x_{0}^{i}}{C_{i}\|\sigma^{-1}\mu\|\sqrt{T}} - \|\sigma^{-1}\mu\|\sqrt{T}\right) \\
= \Phi\left(\frac{K - x_{0}^{i}}{C_{i}\|\sigma^{-1}\mu\|\sqrt{T}} - \|\sigma^{-1}\mu\|\sqrt{T}\right),$$
(4.11)

where Φ denotes the distribution function of the standard normal distribution. The expression inside Φ is increasing in θ_i since $K - x_0^i < 0$. Hence, the probability of the terminal wealth being significantly smaller than the initial wealth is strictly increasing in terms of θ_i . Thus, a larger choice of θ_i results in a riskier strategy.

The question remains which of the two factors is more significant - the expected optimal terminal wealth or the probability of a loss. More precisely, we are interested in how the parameter θ_i should be optimally chosen (depending on the investor's risk tolerance). We discuss this question based on numerical results obtained from the explicit representations (4.10) and (4.11). For this purpose, let d = 1 (one stock), n = 12, and consider the expected optimal terminal wealth of agent 1 while the other agents use risk tolerance parameters increasing from 0.5 to 2.7 in steps of size 0.2, and competition weights increasing from 0 to 1 in steps of size 0.1. Further, let T = 3, $x_0^1 = 10$, $\mu = 0.03$, and $\sigma = 0.2$. To simplify the representation of our numerical results, define $\varepsilon(\theta_1) \coloneqq \mathbb{E}\left[X_T^{1,\varphi^{1,*}}\right]$ and $\rho(\theta_1) = \rho(\theta_1, K) \coloneqq \mathbb{P}\left(X_T^{1,\varphi^{1,*}} \leq K\right)$, given that agent 1 uses the competition weight θ_1 . The results are displayed in Figure 4.2.1 and Table 4.1.

Figure 4.2.1 illustrates a comparison of the expected optimal terminal wealth and the probability of a loss for agent 1 in terms of $\theta_1 \in [0, 1]$. The underlying data was generated using $K = 0.99x_0^1$ and three different choices for the risk tolerance parameter $\delta_1 \in \{0.5, 1, 2\}$. Note that the expected terminal wealth $\varepsilon(\theta_1)$ was shifted appropriately (scaled by x_0^1 and moved down by the value 0.75) to simplify the comparison between the expected optimal terminal wealth and the probability of a loss. Of course, the shift does not change the overall growth behavior of ε . Figure 4.2.1 shows that, as explained earlier, both expressions are strictly increasing in terms of θ_1 . However, the probability of a loss ρ grows steeper than the expected terminal wealth ε . The difference in slope between ε and ρ decreases as the risk tolerance parameter δ_1 increases.

A similar observation can be made in Table 4.1. The table shows the change of ε and ρ between $\theta_1 = 0$ and $\theta_1 = 1$ for five different values of δ_1 . For both ε and ρ , we displayed the value at $\theta_1 = 0$, and the total and relative change between $\theta_1 = 0$ and $\theta_1 = 1$. We notice again that ε and ρ are increasing in terms of θ_1 since, for both ε and ρ , the total difference is larger than 0 and the relative change is larger than 1 for each parameter choice. Moreover, we observe an increase in both $\varepsilon(0)$ and $\rho(0)$ as the risk tolerance δ_1 increases. However, we also notice that the total and relative change in ε is increasing in δ_1 , whereas the total and relative change in ρ is decreasing in terms of δ_1 . We can interpret this observation as follows. If an agent is more careful due to her strong aversion against risk, the change between $\theta_1 = 0$ and $\theta_1 = 1$ has a relatively small effect on the expected optimal terminal wealth and a relatively large effect on the probability of a loss. It should also be noted that the loss probability $\rho(0)$ is much smaller for a highly risk-averse agent



Figure 4.2.1.: Illustration of (4.10) and (4.11) for the first of n = 12 agents for market parameters $d = 1, T = 3, \mu = 0.03$, and $\sigma = 0.2$. The agents $j \ge 2$ use θ_j and $\delta_j, j \ge 2$, increasing from 0 to 1 with step size 0.1, and from 0.5 to 2.7 by step size 0.2, respectively. The figure shows $\varepsilon(\theta_1)$ (shifted appropriately) in a solid and $\rho(\theta_1)$ in a dashed line, in terms of θ_1 with $\delta_1 \in \{0.5, 1, 2\}$ and fixed $K = 0.99x_0^1$.

	$\delta_1 = 0.5$	$\delta_1 = 1$	$\delta_1 = 2$	$\delta_1 = 5$	$\delta_1 = 10$
$\varepsilon(0)$	10.0338	10.0675	10.1350	10.3375	10.6750
$\varepsilon(1) - \varepsilon(0)$	0.1764	0.1787	0.1833	0.1971	0.2201
$\varepsilon(1)/\varepsilon(0)$	1.0176	1.0178	1.0181	1.0191	1.0206
ho(0)	0.1516	0.2596	0.3255	0.3681	0.3827
$\rho(1)-\rho(0)$	0.1991	0.0979	0.0408	0.0107	0.0036
$\rho(1)/\rho(0)$	2.3134	1.3771	1.1255	1.0292	1.0094

Table 4.1.: Comparison of (4.10) and (4.11) for the first of n = 12 agents. Agent 1 uses $\theta_1 \in \{0, 1\}$ and $\delta_1 \in \{0.5, 1, 2, 5, 10\}$ for comparison while the other agents use θ_j and δ_j , $j \ge 2$, increasing from 0 to 1 with step size 0.1, and from 0.5 to 2.7 by step size 0.2, respectively. The market parameters are d = 1, T = 3, $\mu = 0.03$, and $\sigma = 0.2$.

 $(\delta_1 = 0.5)$ than for a more risk-tolerant agent $(\delta_1 \in \{5, 10\})$. If, in contrast, the agent is more tolerant towards risk, the change in the expected optimal terminal wealth is relatively large and the change in the loss probability is a lot smaller than the change observed for a highly risk-averse agent. It should be noted that the more risk-tolerant agent faces a high loss probability even for $\theta_1 = 0$.

To summarize, in the current example a highly risk-averse agent should choose a value of θ_1 close to 0 as the benefit from an increase in expected return does not outweigh the increase in the probability of a loss. If, however, the agent is more tolerant towards risk, a choice of θ_1 close to one seems more appropriate since the increase in the loss probability is small while the increase in expected return is relatively high. We note at this point that, due to the strictly concave utility function chosen in this example, all investors are assumed to be risk-averse. However, some agents are more tolerant towards risk (larger value of δ_1) than others (smaller value of δ_1).

4.2.2. Optimization under power utility

Let the underlying financial market again be given by the Black-Scholes market introduced at the beginning of Section 4.2. Further, we assume that each agent uses a power utility function of the form

$$U_i: (0,\infty) \to \mathbb{R}, \ U_i(x) = \left(1 - \frac{1}{\delta_i}\right)^{-1} x^{1 - \frac{1}{\delta_i}},$$

for risk tolerance parameters $\delta_i > 0$, $\delta_i \neq 1$, i = 1, ..., n. The objective function of agent *i* is then given by

$$\mathbb{E}\bigg[\left(1-\frac{1}{\delta_i}\right)^{-1}\left(X_T^{i,\varphi^i}-\theta_i\bar{X}_T^{-i,\varphi}\right)^{1-\frac{1}{\delta_i}}\bigg].$$

The competition weights $\theta_i \in [0, 1]$ are chosen with respect to the condition $x_0^i - \frac{\theta_i}{n} \sum_{j \neq i} x_0^j > 0$ for the initial capitals $x_0^i > 0$, i = 1, ..., n. The unique optimal solution to the auxiliary problem

$$\begin{cases} \sup_{\psi^i \in \mathcal{A}} \mathbb{E}\left[\left(1 - \frac{1}{\delta_i}\right)^{-1} \left(Y_T^{i,\psi^i}\right)^{1 - \frac{1}{\delta_i}} \right] \\ \text{s.t.} \quad Y_T^{i,\psi^i} = \widetilde{x}_0^i + (\psi^i \cdot S)_T, \end{cases}$$

is given by

$$\psi_k^i(t)S_k(t) = \delta_i \widetilde{x}_0^i \Big((\sigma\sigma^\top)^{-1} \mu \Big)_k \exp\left(\delta_i (\sigma^{-1}\mu)^\top W(t) + \left(\delta_i - \frac{\delta_i^2}{2} \right) \mu^\top (\sigma\sigma^\top)^{-1} \mu t \right),$$

k = 1, ..., d, i = 1, ..., n (using Korn, 1997, p. 58), i.e., the optimally invested fraction of wealth is constant. Hence, using Theorem 3.2, the unique (up to modifications) Nash equilibrium is given by

$$\varphi_k^i(t) \coloneqq \frac{n}{n+\theta_i} \psi_k^{i,*}(t) + \frac{\theta_i}{(n+\theta_i)(1-\hat{\theta})} \sum_{j=1}^n \frac{n}{n+\theta_j} \psi_k^{j,*}(t),$$

for $t \in [0, T]$, k = 1, ..., d, and i = 1, ..., n.

Remark 4.2. Lacker and Zariphopoulou (2019) suggest that the multiplicative relative performance metric is better suited for CRRA utility functions. Consequently, they use objective functions of the form

$$\mathbb{E}\left[\left(1-\frac{1}{\delta_i}\right)^{-1}\left(X_T^{i,\varphi^i}\left(\prod_{j=1}^n X_T^{j,\varphi^j}\right)^{-\frac{\theta_i}{n}}\right)^{1-\frac{1}{\delta_i}}\right], i=1,\ldots,n.$$

Indeed, the multiplicative structure of the argument does simplify some calculations and is also used in this thesis (see Section 7.4). Moreover, the *constant* Nash equilibrium (in terms of invested fractions) has a shorter and overall "representation (see Lacker and Zariphopoulou, 2019, Corollary 3.2). However, the solution is less general since the method described in Section 3.2 cannot be applied here (see Remark 3.4) and thus, only constant Nash equilibria are considered. \Diamond

4.3. BINOMIAL MODEL IN DISCRETE TIME

The next example is a Cox-Ross-Rubinstein market in discrete time (see, for example, Bäuerle and Rieder, 2011, p. 60). The model brings the advantage of being very simple, but also quite popular among financial markets in discrete time. Moreover, the Cox-Ross-Rubinstein model is a discrete-time approximation of the Black-Scholes market. Hence, it does not come as a surprise that the overall structures of the Nash equilibria in the Black-Scholes market and the Cox-Ross-Rubinstein market coincide.

The market consists of one riskless bond with zero interest rate and one stock (d = 1) with price process $(S(t_k))_{k=0,\ldots,N}$, where $t_k = k \cdot T/N$, $k = 0, \ldots, N$. The stock price process is given by

$$S(t_n) = S(0) \prod_{k=1}^n R_k, \ n = 0, \dots, N.$$

The random variables R_k for $k \in \{1, ..., N\}$ are independent and identically distributed with $\mathbb{P}(R_n = u) = p = 1 - \mathbb{P}(R_n = d)$ for 0 < d < u and $p \in (0, 1)$. We assume that d < 1 < u to exclude arbitrage (see, e.g., Bäuerle and Rieder, 2011, Example 3.1.7). For a self financing strategy $\varphi = (\varphi(t_k))_{k=0,...,N-1}$ and some initial capital $x_0 \in \mathbb{R}$, the wealth process of an investor is given by (see, e.g., Bäuerle and Rieder, 2017, Lemma 2.2)

$$X_{t_k}^{\varphi} = x_0 + \sum_{\ell=1}^k \varphi(t_{\ell-1})(S(t_\ell) - S(t_{\ell-1})).$$

Thus, the wealth process is linear in φ . Moreover, we assume that investors use exponential utility functions given by

$$U_i : \mathbb{R} \to \mathbb{R}, U_i(x) = -\exp\left(-\frac{1}{\delta_i}x\right),$$

for risk tolerance parameters $\delta_i > 0$, i = 1, ..., n. The unique optimal solution to the classical optimization problem

$$\begin{cases} \sup_{\psi^i \in \mathcal{A}} \mathbb{E} \left[-\exp\left(-\frac{1}{\delta_i} Y_T^{i,\psi^i}\right) \right], \\ \text{s.t.} \quad Y_T^{i,\psi^i} = \widetilde{x}_0^i + \sum_{\ell=1}^N \psi^i(t_{\ell-1})(S(t_\ell) - S(t_{\ell-1})), \end{cases} \end{cases}$$

is given by

$$\psi^{i,*}(t_k) = \frac{\delta_i}{S(t_k)} \frac{\log\left(\frac{1-q}{1-p}\right) - \log\left(\frac{q}{p}\right)}{u-d}$$

where $q = \frac{1-d}{u-d}$ (see Bäuerle and Rieder, 2011, p. 92). Theorem 3.2 yields the unique (up to modifications) Nash equilibrium $\pi^{i,*}$, i = 1, ..., n, where

$$\pi_i^* = \left(\frac{n}{n+\theta_i}\delta_i + \frac{\theta_i}{(n+\theta_i)(1-\hat{\theta})}\sum_{j=1}^n \frac{n}{n+\theta_j}\delta_j\right)\frac{\log\left(\frac{1-q}{1-p}\right) - \log\left(\frac{q}{p}\right)}{u-d}$$

describes the invested amount. Similar to the Nash equilibrium (4.4) in Subsection 4.2.1, the invested amount is constant in time and given as the constant C_i from (4.7) multiplied by some

expression depending only on the market parameters. If the expression $\frac{\log(\frac{1-q}{1-p})-\log(\frac{q}{p})}{u-d}$ is strictly positive (this is equivalent to p > q), we can use the same argumentation regarding the monotonicity of $\pi^{i,*}$ in terms of θ_i as in Subsection 4.2.1.

4.4. Market with stochastic volatility

In this section, we consider a stochastic volatility model, the so-called *Heston model* (see Heston, 1993; Kallsen and Muhle-Karbe, 2010). The model consists of one risky asset (a stock) and a riskless bond with zero interest rate. There are two correlated Brownian motions W^S and W^Z with correlation ρ . The price process S of the risky asset is described by the system of stochastic differential equations

$$dS_t = S_t \left(\lambda Z_t dt + \sqrt{Z_t} dW_t^S \right), S_0 > 0,$$

$$dZ_t = \kappa (\vartheta - Z_t) dt + \sigma \sqrt{Z_t} dW_t^Z, Z_0 > 0,$$
(4.12)

 $t \in [0, T]$. The constants $\lambda, \kappa, \vartheta$, and σ are assumed to be positive and satisfy the Feller condition $2\kappa\vartheta \ge \sigma^2$ to ensure that Z is strictly positive (see, e.g., Jeanblanc et al., 2009, p. 357).

Remark 4.3. The volatility process $(Z_t)_{t \in [0,T]}$ describes a so-called square root diffusion. The process originates from the idea to model the square root of the volatility process as an Ornstein-Uhlenbeck process, i.e.,

$$\mathrm{d}\sqrt{Z_t} = -\beta\sqrt{Z_t}\mathrm{d}t + \delta\mathrm{d}W_t^Z$$

for constants β , $\delta > 0$ (see Heston, 1993). Using the Itô-Doeblin formula (Theorem 2.1), it can be shown that the process Z solves (4.12) if the constants are chosen appropriately. \diamond

We assume here that each agent uses a power utility function of the form

$$U_i: (0,\infty) \to \mathbb{R}, \ U_i(x) = \left(1 - \frac{1}{\delta_i}\right)^{-1} x^{1 - \frac{1}{\delta_i}}$$

for risk tolerance parameters $\delta_i > 0$, $\delta_i \neq 1$, i = 1, ..., n. The objective functions are given by

$$\mathbb{E}\left[\left(1-\frac{1}{\delta_i}\right)^{-1}\left(X_T^{i,\varphi^i}-\theta_i\bar{X}_T^{-i,\varphi}\right)^{1-\frac{1}{\delta_i}}\right],\ i=1,\ldots,n.$$

The competition weights $\theta_i \in [0, 1]$ are chosen with respect to the condition $\tilde{x}_0^i = x_0^i - \frac{\theta_i}{n} \sum_{j \neq i} x_0^j > 0$ for the initial capital $x_0^i > 0$, i = 1, ..., n. The unique (up to modifications) optimal solution of the classical portfolio optimization problem

$$\begin{cases} \sup_{\psi^i \in \mathcal{A}} \mathbb{E}\left[\left(1 - \frac{1}{\delta_i}\right)^{-1} \left(Y_T^{i,\psi^i}\right)^{1 - \frac{1}{\delta_i}} \right], \\ \text{s.t.} \quad Y_T^{i,\psi^i} = \widetilde{x}_0^i + (\psi^i \cdot S)_T, \end{cases}$$
(4.13)

is given by

$$\frac{\psi^{i,*}(t)S(t)}{Y_t^{i,*}} = \delta_i \lambda + f_i(t)$$
(4.14)

 \mathbb{P} -almost surely for all $t \in [0, T]$, where the deterministic function f_i can be given explicitly (see Kallsen and Muhle-Karbe, 2010). By $(Y_t^{i,*})_{t \in [0,T]}$, we denote the optimal wealth process for (4.13), i.e., the expression on the right-hand side of (4.14) describes the optimally invested fraction of wealth. Finally, $\varphi^{i,*}$, $i = 1, \ldots, n$, given by

$$\varphi^{i,*}(t) = \frac{n}{n+\theta_i}\psi^{i,*}(t) + \frac{\theta_i}{(n+\theta_i)(1-\hat{\theta})}\sum_{j=1}^n \frac{n}{n+\theta_j}\psi^{i,*}(t)$$

 \mathbb{P} -almost surely for all $t \in [0, T]$, is the unique (up to modifications) Nash equilibrium.

4.5. Cumulative prospect theory

In the cumulative prospect theory (CPT), introduced by Tversky and Kahneman (1992), investors evaluate their wealth relative to some reference point $\xi > 0$. Values smaller than ξ are treated as losses while values larger than ξ are seen as gains. Studies have shown that people tend to act risk-seeking when dealing with losses and risk-averse when dealing with gains (see Tversky and Kahneman, 1992, and references therein). This effect is captured by S-shaped utility functions $U: (0, \infty) \to \mathbb{R}$, for example of the form

$$U(x) = \begin{cases} -a \cdot (\xi - x)^{\delta}, & x \le \xi, \\ b \cdot (x - \xi)^{\gamma}, & x > \xi, \end{cases}$$

$$(4.15)$$

for $0 < \delta \le 1$, $0 < \gamma < 1$ and a > b > 0. Figure 4.5.1 shows the S-shaped utility function (4.15) for the parameters¹ a = 2.25, b = 1, and $\delta = \gamma = 0.5$. The reference point is chosen as $\xi = 1$ which could, for example, describe the initial capital of some investor.

Berkelaar et al. (2004) found the unique optimal solution to the associated single investor portfolio optimization problem in a Black-Scholes market with constant market parameters. The stock price processes are hence given by

$$dS_k(t) = S_k(t) \left(\mu_k dt + \sum_{\ell=1}^d \sigma_{k\ell} dW_\ell(t) \right), \ t \in [0, T], \ k = 1, \dots, d.$$

Moreover, there exists a riskless bond with zero interest rate. Since this thesis is focused on investors evaluating their wealth with respect to the wealth of their competitors, we use a reference point in terms of the weighted arithmetic mean of the other investors' wealth. Therefore, the objective function of agent i is given by

$$\mathbb{E}\left[-a_{i}\cdot\left(\frac{\theta_{i}}{n}\sum_{j\neq i}X_{T}^{j,\varphi^{j}}-X_{T}^{i,\varphi^{i}}\right)^{\delta_{i}}\mathbb{1}\left\{X_{T}^{i,\varphi^{i}}\leq\frac{\theta_{i}}{n}\sum_{j\neq i}X_{T}^{j,\varphi^{j}}\right\}\right. \\ \left.+b_{i}\cdot\left(X_{T}^{i,\varphi^{i}}-\frac{\theta_{i}}{n}\sum_{j\neq i}X_{T}^{j,\varphi^{j}}\right)^{\gamma_{i}}\mathbb{1}\left\{X_{T}^{i,\varphi^{i}}>\frac{\theta_{i}}{n}\sum_{j\neq i}X_{T}^{j,\varphi^{j}}\right\}\right]$$

¹Berkelaar et al. (2004) used the similar parameter choice a = 2.25, b = 1, $\xi = 1$, and $\delta = \gamma = 0.88$. However, we use $\delta = \gamma = 0.5$ for a better illustration of the S-shape of U.



Figure 4.5.1.: Illustration of the S-shaped utility function (4.15) for $\xi = 1$, a = 2.25, b = 1, and $\delta = \gamma = 0.5$.

for $0 < \delta_i \leq 1, \ 0 < \gamma_i < 1$, and $a_i > b_i > 0, \ i = 1, ..., n$. We further introduce the constraint $X_T^{i,\varphi^i} - \frac{\theta_i}{n} \sum_{j \neq i} X_T^{j,\varphi^j} \geq -\xi_i$ for some $\xi_i > 0$. Economically, this means that the investor only accepts a downward deviation from the weighted average wealth of the other investors by a constant ξ_i . The introduction of ξ_i ensures that the corresponding classical problem is similar to the optimization considered by Berkelaar et al. (2004). Hence, the optimization problem reads as

$$\begin{cases} \sup_{\varphi^i \in \mathcal{A}} \quad \mathbb{E}\Big[-a_i \cdot \left(\frac{\theta_i}{n} \sum_{j \neq i} X_T^{j,\varphi^j} - X_T^{i,\varphi^i}\right)^{\delta_i} \mathbbm{1} \left\{ X_T^{i,\varphi^i} \leq \frac{\theta_i}{n} \sum_{j \neq i} X_T^{j,\varphi^j} \right\} \\ + b_i \cdot \left(X_T^{i,\varphi^i} - \frac{\theta_i}{n} \sum_{j \neq i} X_T^{j,\varphi^j} \right)^{\gamma_i} \mathbbm{1} \left\{ X_T^{i,\varphi^i} > \frac{\theta_i}{n} \sum_{j \neq i} X_T^{j,\varphi^j} \right\} \Big], \\ \text{s.t.} \qquad X_T^{i,\varphi^i} = x_0^i + (\varphi^i \cdot S)_T, \ X_T^{i,\varphi^i} - \frac{\theta_i}{n} \sum_{j \neq i} X_T^{j,\varphi^j} \geq -\xi_i, \end{cases}$$

for i = 1, ..., n. The unique solution to the associated classical problem

$$\begin{cases} \sup_{\psi^i \in \mathcal{A}} \mathbb{E} \left[-a_i \left(\xi_i - Y_T^{i,\psi^i} \right)^{\delta_i} \mathbbm{1} \left\{ Y_T^{i,\psi^i} \le \xi_i \right\} + b_i \left(Y_T^{i,\psi^i} - \xi_i \right)^{\gamma_i} \mathbbm{1} \left\{ Y_T^{i,\psi^i} > \xi_i \right\} \right] \\ \text{s.t.} \quad Y_T^{i,\psi^i} = \widetilde{x}_0^i + \xi_i + (\psi^i \cdot S)_T, \ Y_T^{i,\psi^i} \ge 0, \end{cases}$$

is then given by (Berkelaar et al., 2004, Proposition 6)

$$\begin{split} \psi_k^{i,*}(t)S_k(t) = & \left(\left(\sigma \sigma^\top \right)^{-1} \mu \right)_k \cdot \left\{ \begin{array}{l} \frac{\xi_i \phi(g(t,\bar{Z}_i))}{\|\kappa\|\sqrt{T-t}} \\ &+ \left(\frac{b_i \gamma_i}{\lambda_i Z(t)} \right)^{\frac{1}{1-\gamma_i}} \mathrm{e}^{\Gamma_i(t)} \left(\begin{array}{l} \frac{\phi\left(g(t,\bar{Z}_i) + \frac{\|\kappa\|\sqrt{T-t}}{1-\gamma_i}\right)}{\|\kappa\|\sqrt{T-t}} + \frac{\Phi\left(g(t,\bar{Z}_i) + \frac{\|\kappa\|\sqrt{T-t}}{1-\gamma_i}\right)}{1-\gamma_i} \end{array} \right) \right\}, \end{split}$$

where κ and $(Z(t))_{t \in [0,T]}$ are the market price of risk and state price density process in the given Black-Scholes market, i.e., $\kappa = \sigma^{-1}\mu$ and $Z(t) = \mathcal{E}(-\kappa \cdot W)_t$. By ϕ and Φ we denote the density and cumulative distribution function of the standard normal distribution. Moreover, g and Γ_i are functions in terms of the market parameters, where $g(t, \cdot)$ also depends on Z(t). Finally, $\lambda_i > 0$ is the Lagrange multiplier to the constraint $\mathbb{E}[Z(T)Y_T^{i,\psi^i}] = \tilde{x}_0^i + \xi_i$, and \bar{Z}_i is the unique root of some additional function f_i , which depends only on the market and preference parameters of the problem. Explicit representations of the mentioned functions can be found in Proposition 6 in Berkelaar et al. (2004).

Finally, $\varphi_k^{i,*}$, $k = 1, \dots, d$, $i = 1, \dots, n$, given by

$$\varphi_{k}^{i,*}(t) = \frac{n}{n+\theta_{i}}\psi_{k}^{i,*}(t) + \frac{\theta_{i}}{(n+\theta_{i})(1-\hat{\theta})}\sum_{j=1}^{n}\frac{n}{n+\theta_{j}}\psi_{k}^{i,*}(t)$$

P-almost surely for all $t \in [0,T]$ and i = 1,...,n, is the unique (up to modifications) Nash equilibrium. Apparently, $\psi_k^{i,*}(t) > 0$ holds for all $t \in [0,T]$ if, and only if, $((\sigma\sigma^{\top})^{-1}\mu)_k > 0$, k = 1,...,d. If this is the case, it follows from Remark 3.5 that the associated component $\varphi_k^{i,*}(t)$ of the Nash equilibrium is increasing in terms of θ_i .

CHAPTER 5

MEAN FIELD EQUILIBRIA FOR RELATIVE INVESTORS

In the following, we give a short introduction to the extensive field of research on mean field games. Inspired by the mean field theory from physics for large particle systems, mean field games were introduced independently by Huang et al. (2006) and Lasry and Lions (2007) to simplify the solution to *n*-player games where the number of agents is large. We refer to Lacker (2018) for a general introduction and motivation of the topic, and to Carmona and Delarue (2018a,b) for a thorough treatment of mean field game theory and its numerous applications. Still, we give a brief motivation and highlight the key assumptions and features of mean field games considered in the literature. After the discussion of general mean field games, we take a look at the literature on mean field portfolio optimization for relative investors, which is closely related to the results presented in this chapter.

The basic motivation to introducing mean field games was to find a mathematically tractable approximation of *n*-player games with a large number of players. As Carmona and Delarue (2013) state, "large stochastic differential games are notoriously untractable". Similar to the mean field theory used in physics, where infinite particle systems are used to approximate the finite particle setting, a model containing an infinite number of players (a continuum) is used to approximate the corresponding *n*-player game. Heuristically, due to the law of large numbers, some form of averaging is expected when the number of agents tends to infinity (see Carmona and Delarue, 2013; Lacker, 2016). To provide a mathematically rigorous explanation of this heuristic idea, one has to make two fundamental assumptions on the model. The first is that the agents are "small" in the sense that each player has very little influence on the overall system and the influence becomes negligible for large n (see Lacker, 2016; Carmona and Delarue, 2013). The second basic assumption is that the agents are indistinguishable or, as Carmona and Delarue (2013) describe it, "statistically identical". Lacker (2018) phrases this assumption by asserting that players are "interchangeable". The symmetry among the players implies that the system does not depend on

the individual states of the players, but only on their collective empirical distribution. Roughly speaking, we are only interested in which states are present, but not to whom each state belongs. If these basic assumptions are satisfied, we can expect the mean field game to provide a sufficiently accurate approximation of the underlying *n*-player game.

Generally, the mean field game is more tractable than its n-player counterpart since, instead of solving a system of n coupled problems, we only need to deal with a standard control problem faced by a single representative agent who interacts with the environment (Carmona et al., 2016). By the term "environment" we describe the state of the continuum of agents, which can be assumed to be fixed since the influence of the representative agent on the system is negligible. After solving the control problem for a single agent, the next step is to find a fixed point of the so-called consistency condition. A rough intuition for this condition is that one has to make sure that the representative agent is, in fact, representative for the whole population.

There are two different approaches to mean field games in the literature. The first is to solve the problem with a continuum of agents and then construct an approximate solution for the corresponding *n*-player game (see, among others, Huang et al., 2006). The second approach is to solve the *n*-player game first and then consider the limit as *n* tends to infinity (see, e.g., Lasry and Lions, 2007). The relationship between the *n*-player and the mean field game is one of the three key questions tackled by the literature. More precisely, it contains two subquestions: Does each *n*-player solution converge to a mean field solution and, conversely, can any mean field solution be obtained as the limit of some (approximate) *n*-player solution? Under fairly general assumptions, it is possible to show that the answer to both subquestions is affirmative (see Lacker, 2016). The other main questions considered in the context of mean field games comprise the existence and uniqueness of solutions, both for the *n*-player and the mean field game. Of course, if there is a one-to-one correspondence between mean field solutions and sequences of *n*-player solutions, existence and uniqueness of solutions for the *n*-player and the mean field game are equivalent. It is possible to prove under general assumptions on the model parameters, that both the *n*-player and the mean field game admit a unique solution (see, e.g., Lacker, 2018; Lasry and Lions, 2007).

Furthermore, one key feature of a mean field game model is whether or not it contains common noise. In the early literature on the topic, the state diffusions of individual agents were assumed to be governed by independent Brownian motions. To include the more realistic scenario of random perturbations affecting each agent simultaneously, Carmona et al. (2016) introduced mean field games with common noise by adding an additional independent Brownian motion to the state diffusion of each agent. Later in this chapter, we consider a problem of competitive investment in a financial market common to all agents. Thus, in contrast to the majority of literature on mean field games, we consider common noise only.

Finally, let us mention the two different ways of solving mean field games as they appear in the literature (see Lacker, 2015). The first is an analytic approach based on two tightly coupled partial differential equations. One of the partial differential equations has a terminal condition and has to be solved backward in time while the second has an initial condition and needs to be solved forward in time. This approach was, for instance, used by Lasry and Lions (2007) as well as by Guéant et al. (2011). Carmona and Delarue (2013) chose a different, probabilistic approach to

tackle mean field games. Their method, based on the stochastic maximum principle, results in coupled forward-backward stochastic differential equations.

Later in this chapter, we consider a special mean field game which is not solved using the classical methods described above. Thus, we do not go into any more detail on the general theory on mean field games. Let us instead consider the special case of mean field portfolio problems for relative investors, which are closely related to the problem solved in this chapter. To be more specific, we focus on portfolio optimization problems where agents measure their preferences with respect to the expected utility of an additive relative performance metric applied to their own as well as their competitors' terminal wealth. For the definition of relative performance metrics in n-player games, we refer to the introduction of Chapter 3.

Lacker and Zariphopoulou (2019) were the initiators of the study of mean field games for relative investors. In a continuous-time financial market with constant market parameters, they consider agents which specialize in a single stock each, and they incorporate both an independent as well as a common noise term. After finding explicit representations of unique constant Nash equilibria for CARA and CRRA utility functions, they consider the (heuristic) limit of the *n*-player equilibria as the number of agents tends to infinity. Following this informal derivation, they define the mean field game rigorously for both the additive and the multiplicative relative performance metric. Finally, they give explicit representations of the unique mean field equilibrium for both relative performance metrics which turn out to coincide with the previously derived informal limits.

The mean field problem introduced by Lacker and Zariphopoulou (2019) was soon extended in various ways. For reasons of tractability, each of the subsequent authors used either the additive relative performance metric in combination with CARA utility, or the multiplicative relative performance metric together with CRRA utility. The setting closest to the original was considered by Lacker and Soret (2020). They added consumption to the CRRA problem with a multiplicative relative performance metric. The assumption of constant market parameters was lifted by Fu et al. (2020) and Fu and Zhou (2023), who consider a model with random coefficients for CARA and CRRA utility, respectively. Hu and Zariphopoulou (2022) allowed for the market coefficients of the stock, which is common to all agents, to depend on an additional independent Brownian motion. Bo, Wang and Yu (2023) extended the financial market by allowing for jumps in the stock price process.

Instead of classical utility functions, some authors consider forward utilities. For example, Dos Reis and Platonov (2021, 2022) used forward utilities with CARA and CRRA priors, respectively. The underlying stock price processes coincide with those used by Lacker and Zariphopoulou (2019). Park (2022) considered forward utilities with CARA priors as well, allowing the stock price coefficients to be suitable stochastic processes. In contrast to the articles mentioned up to this point, which feature only risk-averse agents, Nguyen (2022) considered a system consisting of both risk-averse and risk-seeking investors. Moreover, he allowed for competition weights taking values between -1 and 1, which previously had to be chosen between 0 and 1.

Finally, we want to mention an extension of the original problem which considers competing insurance companies instead of investors. For example, Guan and Hu (2022), Bo, Wang and Zhou

(2023), and He et al. (2023) searched for optimal investment and (re-)insurance strategies for competing insurance companies.

Apart from Hu and Zariphopoulou (2022), all of the previously mentioned authors assumed that investors specialize in a particular stock. Furthermore, each of the above articles requires specific assumptions regarding the utility function employed by the agents and the underlying financial market. In contrast to these models, we consider a mean field game for relative investors using a general utility function to evaluate their terminal wealth with respect to the continuum of competitors in the market. The relative concerns of an investor are measured with respect to the additive relative performance metric. Moreover, the underlying financial market, which is common to all investors, takes a very general form. We use the semimartingale financial market introduced in Subsection 2.3.1, but assume that the probability space now contains three additional random variables. These are assumed to be independent of the stock price processes and represent the initial capital, the risk tolerance parameter, and the competition weight of a representative agent taken from the continuum of infinitely many investors. Similar to Section 3.1, we are able to reduce the mean field game to a single-agent portfolio optimization problem and to express the mean field equilibrium in terms of the unique solution of the corresponding classical problem. Although the setting is rather general (especially compared to the existing literature), we need to assume that each agent in the continuum uses the same utility function U. Only the risk tolerance parameter is allowed to differ between agents. In the corresponding n-agent game in Section 3.1, we did not require this assumption. However, mean field games call for a lot of symmetry among the agents. At least, the specific choice of U is not relevant for the proof displayed below.

The chapter is organized as follows. In Section 5.1, we provide an informal derivation of the mean field equilibrium as the limit of the n-player Nash equilibrium from Theorem 3.2 when the number of agents tends to infinity. Afterwards, we define the mean field game properly, similar to Lacker and Zariphopoulou (2019). Finally, we reduce the mean field game to a suitable single-agent problem and display the unique mean field equilibrium in terms of the solution of the corresponding single-agent problem. It turns out that the mean field equilibrium coincides with the informal limit derived earlier.

5.1. Motivation and heuristic derivation of the mean field game

In Chapter 3, we modeled *n* agents through their type vector $\zeta_i = (x_0^i, \delta_i, \theta_i)$ which contains the initial wealth and preference parameters of agent $i \in \{1, \ldots, n\}$. In the mean field game, the type vector of some representative agent will be given as the realization of a suitable random vector $\zeta = (\xi, \delta, \theta)$, which is independent of the price processes in the underlying financial market. Before we properly define the mean field game, we provide a heuristic derivation of its solution. More specifically, we provide an informal derivation of the limit of the Nash equilibrium (3.6) as *n* tends to infinity. A similar heuristic can be found in Lacker and Zariphopoulou (2019).

Recall that the *i*-th entry of the Nash equilibrium (3.6) is given by

$$\varphi_k^{i,*}(t) \coloneqq \frac{n}{n+\theta_i} \psi_k^{i,*}(t) + \frac{\theta_i}{(n+\theta_i)(1-\hat{\theta})} \sum_{j=1}^n \frac{n}{n+\theta_j} \psi_k^{j,*}(t), \, k = 1, \dots, d, \, t \in [0,T],$$

where $\psi^{j,*}$, $j \in \{1, \ldots, n\}$, describes the unique optimal portfolio for a suitable auxiliary problem. First, we obtain

$$\lim_{n \to \infty} \frac{n}{n + \theta_i} = 1, \quad \lim_{n \to \infty} \frac{n\theta_i}{n + \theta_i} = \theta_i.$$

Moreover, if we assume that $\theta_1, \theta_2, \ldots$ are independent and identically distributed random variables, independent of the underlying filtration $(\mathcal{F}_t)_{t \in [0,T]}$, we obtain

$$\hat{\theta} = \sum_{j=1}^{n} \frac{\theta_j}{n + \theta_j} \xrightarrow{\text{a.s.}} \mathbb{E}[\theta_1], n \to \infty,$$

since, using the law of large numbers¹,

$$\hat{\theta} = \sum_{j=1}^{n} \frac{\theta_j}{n} \cdot \frac{n}{n+\theta_j} \le \sum_{j=1}^{n} \frac{\theta_j}{n} \xrightarrow{\text{a.s.}} \mathbb{E}[\theta_1], \ n \to \infty,$$
$$\hat{\theta} = \sum_{j=1}^{n} \frac{\theta_j}{n} \cdot \frac{n}{n+\theta_j} \ge \frac{n}{n+1} \sum_{j=1}^{n} \frac{\theta_j}{n} \xrightarrow{\text{a.s.}} \mathbb{E}[\theta_1], \ n \to \infty$$

Finally, if we assume that, conditional on \mathcal{F}_T , $\psi_k^{1,*}(t), \psi_k^{2,*}(t), \ldots$ are independent and identically distributed random variables (for any $t \in [0, T]$, $k \in \{1, \ldots, d\}$) we obtain (analogously, using the law of large numbers and a sandwich argument)

$$\sum_{j=1}^{n} \frac{1}{n+\theta_j} \psi_k^{j,*}(t) \xrightarrow{\text{a.s.}} \mathbb{E}[\psi_k^{1,*}(t)|\mathcal{F}_T], n \to \infty.$$

Note that we can only assume that $\psi_k^{j,*}(t)$ are independent and identically distributed given \mathcal{F}_T as they are solutions to portfolio optimization problems at time T.

Hence, we expect that the components $\varphi_k^{i,*}(t)$ of the Nash equilibrium (3.6) converge to

$$\psi_k^*(t) + \frac{\theta}{1 - \mathbb{E}[\theta]} \mathbb{E}[\psi_k^*(t)|\mathcal{F}_T], \quad k = 1, \dots, d_k$$

as $n \to \infty$, where $\theta \stackrel{\mathcal{D}}{=} \theta_1$ and $\psi_k^*(t) \stackrel{\mathcal{D}}{=} \psi_k^{1,*}(t)$ given \mathcal{F}_T . By $\stackrel{\mathcal{D}}{=}$ we denote equality in distribution.

The convergence of the Nash equilibrium as the number n of agents tends to infinity can also be observed in Figure 5.1.1. There, the first component of the Nash equilibrium (in terms of the number of stocks) for n agents using exponential utility functions is displayed in terms of the number of agents. The underlying financial market is a one-dimensional Black-Scholes market with zero interest rate, constant drift $\mu = 0.03$, and constant volatility $\sigma = 0.2$. Moreover, the competition weights θ_i and risk tolerance parameters δ_i , $i = 1, \ldots, n$, are realizations of independent and identically $\mathcal{U}(0, 1)$ - and $\mathcal{U}(0.5, 3)$ -distributed random variables, respectively.

¹By $\xrightarrow{\text{a.s.}}$ we denote \mathbb{P} -almost sure convergence.



Figure 5.1.1.: Illustration of $\varphi^{1,*}(t)$ in a Black-Scholes market with $d = 1, r = 0, \mu = 0.03$, and $\sigma = 0.2$, at t = 1 for $n \in \{1, \ldots, 200\}$ under CARA utility, where θ_i and δ_i are realizations of i.i.d. $\mathcal{U}(0, 1)$ - and $\mathcal{U}(0.5, 3)$ -distributed random variables, respectively. The dashed line marks the optimal solution to the single-agent problem ($\theta_1 = 0$) for the realization δ_1 of the $\mathcal{U}(0.5, 3)$ -distributed random variable used for agent 1.

Considering Figure 5.1.1, we notice that for small values of n, $\varphi^{1,*}$ grows very fast compared to the behavior for larger values of n. Moreover, Figure 5.1.1 indicates that, for $n \ge 2$, the first component of the Nash equilibrium is always larger than the strategy obtained for n = 1, which corresponds to the optimal strategy without competition. This observation can be explained as follows

$$\varphi_{k}^{i,*}(t) = \left(\frac{n}{n+\theta_{i}} + \frac{\theta_{i}}{(n+\theta_{i})(1-\hat{\theta})}\right) \psi_{k}^{i,*}(t) + \frac{\theta_{i}}{(n+\theta_{i})(1-\hat{\theta})} \sum_{\substack{j=1\\j\neq i}}^{n} \frac{n}{n+\theta_{j}} \psi_{k}^{j,*}(t)$$
$$\geq \psi_{k}^{i,*}(t) + \frac{\theta_{i}}{(n+\theta_{i})(1-\hat{\theta})} \sum_{\substack{j=1\\j\neq i}}^{n} \frac{n}{n+\theta_{j}} \psi_{k}^{j,*}(t) \geq \psi_{k}^{i,*}(t)$$
(5.1)

where $\varphi_k^{i,*}(t)$ is taken from (3.6). The chain of inequalities in (5.1) holds if $\psi_k^{j,*}(t) \ge 0$ for all $j = 1, \ldots, n$, which is the case in the framework used to generate Figure 5.1.1. Moreover, we also used that $\hat{\theta} \in [0, 1)$ (see Lemma 3.3).

Remark 5.1. One might argue that it is not necessary to consider the mean field game associated to problem (3.3) since there exists a unique explicit solution to the *n*-player game given in Theorem 3.2. However, as Lacker and Zariphopoulou (2019) argue, the mean field model presented here might be generalized to more involved models in which the mean field game formulation is mathematically tractable whereas the *n*-player game brings some serious difficulties. We do not elaborate this any further and refer to Lacker and Zariphopoulou (2019) for examples on models with difficult *n*-player games but promising corresponding mean field game formulations. \diamond

5.2. Formal definition of the mean field game

Let us now give a formal definition of the mean field game and a mean field equilibrium. Further, we prove that the mean field equilibrium coincides with the informally derived limit above. The structure of the mean field game is similar to the problem introduced by Lacker and Zariphopoulou (2019, Section 2.2). We use the semimartingale financial market described in Subsection 2.3.1 again. To summarize, the market consists of a riskless bond with zero interest rate and d stocks. The stock price processes $(S_k(t))_{t\in[0,T]}$, $k = 1, \ldots, d$, are assumed to be semimartingales with càdlàg paths (i.e., the paths are right continuous with existing left limits). Further, it is assumed that there exists an equivalent σ -martingale measure \mathbb{Q} . Additionally, we assume that the underlying probability space contains a $(0, \infty) \times (0, \Delta] \times [0, 1]$ -valued random vector $\zeta = (\xi, \delta, \theta)$ independent of the filtration $(\mathcal{F}_t)_{t\in[0,T]}$ with $\Delta > 0$. We assume that $\mathbb{E}[\xi^2] < \infty$. Finally, we define an additional filtration $(\mathcal{F}_t^{\mathrm{MF}})_{t\in[0,T]}$ given by

$$\mathcal{F}_t^{\mathrm{MF}} \coloneqq \sigma\left(\mathcal{F}_t, \zeta\right), \, t \in [0, T].$$

The random variables ξ , δ , and θ denote the initial capital and preference parameters of a representative investor. In this setting, the wealth process of a representative agent is given by

$$X_t^{\varphi} = \xi + (\varphi \cdot S)_t, \ t \in [0, T],$$

where φ is an admissible strategy representing the number of stocks held at time $t \in [0, T]$. We say that φ is an admissible strategy if $\varphi \in \mathcal{A}^{MF}$, where

$$\mathcal{A}^{\mathrm{MF}} \coloneqq \left\{ \varphi \in L^{\mathrm{MF}}(S) : \, (\varphi \cdot S)_T \in L^2(\mathbb{P}), \, (\varphi \cdot S)Z^{\mathbb{Q}} \text{ is a } \mathbb{P}\text{-martingale for all } S\sigma \mathrm{MM} \mathbb{Q} \right.$$
with density process $Z^{\mathbb{Q}}$ and $\frac{\mathrm{d}\mathbb{Q}}{\mathrm{d}\mathbb{P}} \in L^2(\mathbb{P}) \right\}.$

By $L^{\rm MF}(S)$ we denote the set of $(\mathcal{F}_t^{\rm MF})$ -predictable, S-integrable stochastic processes.

Further, we assume that $U: \mathcal{D} \to \mathbb{R}$ is a utility function, in the sense of Definition 2.11, defined on a domain $\mathcal{D} \in \{\mathbb{R}, (0, \infty)\}$ including some parameter $\delta > 0$. We again extend the definition of U to the whole real line by setting $U(x) = -\infty$ if $x \notin \mathcal{D}$. Now assume that δ is part of the characterization of the representative investor. Then the combination of U and δ (by inserting δ as the parameter in U) yields a (stochastic) utility function denoted by U_{δ} . Now a representative investor faces the following optimization problem

$$\begin{cases} \sup_{\varphi \in \mathcal{A}^{\mathrm{MF}}} \mathbb{E} \left[U_{\delta} \left(X_{T}^{\varphi} - \theta \bar{X} \right) \right], \\ \text{s.t.} \quad X_{T}^{\varphi} = \xi + (\varphi \cdot S)_{T}, \ \bar{X} = \mathbb{E} [X_{T}^{\varphi} | \mathcal{F}_{T}]. \end{cases}$$
(5.2)

The consistency condition $\bar{X} = \mathbb{E}[X_T^{\varphi}|\mathcal{F}_T]$ (see Lacker and Zariphopoulou, 2019) models the feature that each agent in the continuum faces the same type vector ζ and the same randomness from the financial market. Hence, if we condition on the information \mathcal{F}_T provided by the financial market at time T, each agent faces an independent and identically distributed copy of the same optimization problem. Roughly speaking, the consistency condition ensures that the representative agent is indeed representative for the whole population of agents.

Remark 5.2. We need to ensure that there is at least one strategy $\varphi \in \mathcal{A}^{\mathrm{MF}}$ such that $X_T^{\varphi} - \theta \bar{X} \in \mathcal{D}$ \mathbb{P} -almost surely. If $\mathcal{D} = (0, \infty)$, we can therefore only allow choices of ξ and θ that satisfy $\xi - \theta \bar{\xi} > 0$ \mathbb{P} -almost surely, where $\bar{\xi} \coloneqq \mathbb{E}[\xi]$.

The optimal solution to (5.2) is called *mean field equilibrium*.

Definition 5.3 (Lacker and Zariphopoulou, 2019, Definition 2.9). A strategy $\varphi^* \in \mathcal{A}^{MF}$ is called a *mean field equilibrium*, if it is an optimal solution to the optimization problem (5.2). This means in particular that φ^* needs to satisfy the consistency condition $\bar{X} = \mathbb{E}[X_T^{\varphi^*} | \mathcal{F}_T]$.

5.3. Solution method via problem reduction

The optimization problem (5.2) can be solved similarly to the *n*-agent problem. Therefore, we define the auxiliary problem

$$\begin{cases} \sup_{\psi \in \mathcal{A}^{\mathrm{MF}}} \mathbb{E}\left[U_{\delta}(Z_{T}^{\psi})\right],\\ \text{s.t.} \quad Z_{T}^{\psi} = \xi - \theta \bar{\xi} + (\psi \cdot S)_{T}. \end{cases}$$
(5.3)

Then the mean field equilibrium for (5.2) is given in the following theorem in terms of the optimal solution to the auxiliary problem (5.3). Note that we obtain exactly the representation we derived heuristically in Section 5.1.

Theorem 5.4. Let $\mathbb{E}[\theta] =: \overline{\theta} < 1$. If (5.3) has a unique (up to modifications) optimal portfolio strategy ψ^* , then there exists a unique mean field equilibrium for (5.2) given by

$$\varphi_k^*(t) = \psi_k^*(t) + \frac{\theta}{1 - \bar{\theta}} \mathbb{E}\left[\psi_k^*(t) | \mathcal{F}_T\right], \ k = 1, \dots, d,$$
(5.4)

 \mathbb{P} -almost surely for all $t \in [0, T]$.

Proof. In order to solve the optimization problem (5.2), we assume that \bar{X} is an \mathcal{F}_T -measurable random variable of the form

$$\bar{X} = \mathbb{E}\left[X_T^{\alpha} | \mathcal{F}_T\right]$$

for an admissible strategy $\alpha \in \mathcal{A}^{\mathrm{MF}}$ with $X_0^{\alpha} = \xi$. Moreover, we define the process

$$\bar{X}_t^{\alpha} \coloneqq \mathbb{E}\left[X_t^{\alpha} | \mathcal{F}_T\right], \ t \in [0, T].$$

Since X_t^{α} can be written as

$$X_t^{\alpha} = \xi + \sum_{k=1}^d \int_0^t \alpha_k(u) \mathrm{d}S_k(u),$$

we obtain

$$\bar{X}_{t}^{\alpha} = \bar{\xi} + \sum_{k=1}^{d} \int_{0}^{t} \bar{\alpha}_{k}(u) \mathrm{d}S_{k}(u), \qquad (5.5)$$

where $\bar{\alpha}_k(u) \coloneqq \mathbb{E}[\alpha_k(u)|\mathcal{F}_T]$, k = 1, ..., d, and $\bar{\xi} = \mathbb{E}[\xi]$. The representation (5.5) of \bar{X}_t^{α} requires more explanation. As a first step, we used the independence of ξ and $(\mathcal{F}_t)_{t \in [0,T]}$, and the linearity of the conditional expectation. To obtain

$$\mathbb{E}\left[\int_0^t \alpha_k(u) \mathrm{d}S_k(u) \Big| \mathcal{F}_T\right] = \int_0^t \bar{\alpha}_k(u) \mathrm{d}S_k(u), \tag{5.6}$$

we first observe that the sample space can be written as $\Omega = \Omega_1 \times \Omega_2$ with σ -algebras $\mathcal{F}^{(1)}$ and $\mathcal{F}^{(2)}$ on Ω_1 , Ω_2 , respectively, and probability measures \mathbb{P}_1 and \mathbb{P}_2 on $\mathcal{F}^{(1)}$, $\mathcal{F}^{(2)}$, satisfying $\mathbb{P} = \mathbb{P}_1 \otimes \mathbb{P}_2$, due to the independence of (\mathcal{F}_t) and ζ . Hence, Ω_1 and Ω_2 are associated to (\mathcal{F}_t) and ζ , respectively. It follows that for any $F \in \mathcal{F}_T$, there exists a unique $G \in \mathcal{F}^{(1)}$ such that $F = G \times \Omega_2$. Now we prove that the conditional expectation with respect to \mathcal{F}_T of some $\mathcal{F}_T^{\text{MF}}$ -measurable, \mathbb{P} -integrable random variable $Y = Y(\omega_1, \omega_2)$ can be written as an integral over Ω_2 with respect to \mathbb{P}_2 . First, we notice that $\mathbb{E}[Y|\mathcal{F}_T]$ is a random variable that can, due to the \mathcal{F}_T -measurability, be written in terms of ω_1 only. Now let $F \in \mathcal{F}_T$ with decomposition $F = G \times \Omega_2$. It follows (using Fubini's theorem and the definition of conditional expectation²)

$$\begin{split} \int_{G} \int_{\Omega_{2}} Y(\omega_{1}, \omega_{2}) \mathrm{d}\mathbb{P}_{2}(\omega_{2}) \mathrm{d}\mathbb{P}_{1}(\omega_{1}) &= \int_{F} Y(\omega_{1}, \omega_{2}) \mathrm{d}\mathbb{P}(\omega_{1}, \omega_{2}) \\ &= \int_{F} \mathbb{E}[Y|\mathcal{F}_{T}](\omega_{1}, \omega_{2}) \mathrm{d}\mathbb{P}(\omega_{1}, \omega_{2}) \\ &= \int_{G} \int_{\Omega_{2}} \mathbb{E}[Y|\mathcal{F}_{T}](\omega_{1}, \omega_{2}) \mathrm{d}\mathbb{P}_{2}(\omega_{2}) \mathrm{d}\mathbb{P}_{1}(\omega_{1}) \\ &= \int_{G} \int_{\Omega_{2}} \mathbb{E}[Y|\mathcal{F}_{T}](\omega_{1}) \mathrm{d}\mathbb{P}_{2}(\omega_{2}) \mathrm{d}\mathbb{P}_{1}(\omega_{1}) \\ &= \int_{G} \mathbb{E}[Y|\mathcal{F}_{T}](\omega_{1}) \mathrm{d}\mathbb{P}_{1}(\omega_{1}). \end{split}$$

Therefore, we obtain

$$\mathbb{E}[Y|\mathcal{F}_T](\omega_1) = \int_{\Omega_2} Y(\omega_1, \omega_2) d\mathbb{P}_2(\omega_2)$$

for \mathbb{P}_1 -almost all $\omega_1 \in \Omega_1$ (following the arguments of the proof of Theorem 8.12 in Klenke, 2020). Now we can use this result to prove (5.6)

$$\mathbb{E}\left[\int_{0}^{t} \alpha_{k}(u) \mathrm{d}S_{k}(u) \Big| \mathcal{F}_{T}\right] = \int_{\Omega_{2}} \int_{0}^{t} \alpha_{k}(u) \mathrm{d}S_{k}(u) \mathrm{d}\mathbb{P}_{2}(\omega_{2})$$
$$= \int_{0}^{t} \int_{\Omega_{2}} \alpha_{k}(u) \mathrm{d}\mathbb{P}_{2}(\omega_{2}) \mathrm{d}S_{k}(u)$$
$$= \int_{0}^{t} \mathbb{E}[\alpha_{k}(u)|\mathcal{F}_{T}] \mathrm{d}S_{k}(u)$$
$$= \int_{0}^{t} \bar{\alpha}_{k}(u) \mathrm{d}S_{k}(u).$$

The second equality holds due to a stochastic version of Fubini's theorem by Protter (2005) (see Theorem 2.5). Hence, the representation (5.5) is in fact correct and we can proceed with the

²If X is an integrable random variable and \mathcal{G} a sub σ -algebra of \mathcal{F} , the conditional expectation of X given \mathcal{G} is the almost surely unique, \mathcal{G} -measurable random variable Y such that $\int_G X \, d\mathbb{P} = \int_G Y \, d\mathbb{P}$ for all $G \in \mathcal{G}$ (see, for example, Klenke, 2020, Definition 8.11 and Theorem 8.12).

solution of problem (5.2). By construction, the equation $\bar{X}_T^{\alpha} = \bar{X}$ holds. Using the previously defined process (\bar{X}_t^{α}) , we define another process $(Z_t)_{t \in [0,T]}$ for $\varphi \in \mathcal{A}^{\mathrm{MF}}$ and $X_0^{\varphi} = \xi$ by

$$Z_t \coloneqq X_t^{\varphi} - \theta \bar{X}_t^{\alpha}.$$

Thus, Z_t can be written as

$$Z_t = \xi - \theta \bar{\xi} + \sum_{k=1}^d \int_0^t \underbrace{(\varphi_k(u) - \theta \bar{\alpha}_k(u))}_{=:\psi_k(u)} \mathrm{d}S_k(u) =: Z_t^{\psi}.$$

The random variable Z_T^{ψ} coincides with the argument of the objective function in (5.2). Therefore, we consider the auxiliary problem (5.3)

$$\begin{cases} \sup_{\psi \in \mathcal{A}^{\mathrm{MF}}} \mathbb{E}\left[U_{\delta}(Z_{T}^{\psi})\right], \\ \text{s.t.} \quad Z_{T}^{\psi} = \xi - \theta \bar{\xi} + (\psi \cdot S)_{T}. \end{cases}$$

If ψ^* is the unique optimal portfolio strategy to (5.3), we can determine the solution to (5.2) as follows. By definition, we have $Z_T^{\psi^*} = X_T^{\varphi} - \theta \bar{X}$ or equivalently $X_T^{\varphi} = Z_T^{\psi^*} + \theta \bar{X}$. Moreover, the random variable \bar{X} needs to satisfy the consistency condition $\bar{X} = \mathbb{E}[X_T^{\varphi}|\mathcal{F}_T]$. Hence, it follows

$$\bar{X} = \mathbb{E}[X_T^{\varphi}|\mathcal{F}_T] = \mathbb{E}\left[Z_T^{\psi^*} + \theta \bar{X}|\mathcal{F}_T\right] = \mathbb{E}\left[Z_T^{\psi^*}|\mathcal{F}_T\right] + \bar{X}\mathbb{E}[\theta|\mathcal{F}_T] = \mathbb{E}\left[Z_T^{\psi^*}|\mathcal{F}_T\right] + \bar{\theta}\bar{X},$$

where we used that \bar{X} is \mathcal{F}_T -measurable and that θ is independent of (\mathcal{F}_t) . Moreover, we introduced the notation $\bar{\theta} := \mathbb{E}[\theta]$. Under the assumption that $\bar{\theta} < 1$, we obtain

$$\bar{X} = \frac{1}{1 - \bar{\theta}} \mathbb{E} \left[Z_T^{\psi^*} | \mathcal{F}_T \right].$$

Therefore, the optimal wealth X_T^{φ} is given by

$$X_T^{\varphi} = Z_T^{\psi^*} + \theta \bar{X} = Z_T^{\psi^*} + \frac{\theta}{1 - \bar{\theta}} \mathbb{E}\left[Z_T^{\psi^*} | \mathcal{F}_T\right].$$

Since the wealth process is linear in terms of the strategy and the solution ψ^* is unique, it follows that

$$\varphi(t) = \psi^*(t) + \frac{\theta}{1-\overline{\theta}} \mathbb{E}\left[\psi^*(t)|\mathcal{F}_T\right]$$

componentwise \mathbb{P} -almost surely for all $t \in [0, T]$. The line of arguments implies that there exists a unique Nash equilibrium given by (5.4) if, and only if, the auxiliary problem (5.3) is uniquely solvable.

Remark 5.5. The mean field equilibrium (5.4) shows, similar to Remark 3.5, that a larger value of θ results in a more risky investment behavior of a representative agent. If we substitute θ by a different, [0, 1]-valued random variable $\tilde{\theta}$ with $\mathbb{E}[\tilde{\theta}] < 1$ and $\tilde{\theta} > \theta$ P-almost surely, the resulting Nash equilibrium becomes more risky in the sense that more shares of the risky asset are purchased or sold short depending on whether the realization of $\psi_k^*(t)$ is positive or negative.

Finally, we present an application of Theorem 5.4 which describes a mean field counterpart of the model in Subsection 4.2.1.

Example 5.6. We consider a one-dimensional Black-Scholes financial market with constant drift $\mu > 0$ and volatility $\sigma > 0$. Moreover, let $U_{\delta}(x) = -\exp(-\delta^{-1}x), x \in \mathbb{R}$. Then the solution to the auxiliary problem (5.3) in terms of amounts is known to be given by

$$\pi^Z = \delta \cdot \frac{\mu}{\sigma^2}.$$

Therefore, the mean field equilibrium to (5.2) in terms of amounts is given by

$$\pi = \delta \frac{\mu}{\sigma^2} + \frac{\theta}{1 - \bar{\theta}} \mathbb{E} \left[\delta \frac{\mu}{\sigma^2} \Big| \mathcal{F}_T \right] = \left(\delta + \bar{\delta} \frac{\theta}{1 - \bar{\theta}} \right) \frac{\mu}{\sigma^2}.$$

CHAPTER 6

PARETO OPTIMA FOR RELATIVE INVESTORS

In general, there are two main optimality criteria used in *n*-person stochastic games: Nash equilibria and Pareto optimality. Until now, we have only looked for Nash equilibria in the multi-objective optimization problem (3.3) and the corresponding mean field game. In this section, we search for Pareto optimal strategies. The criterion is named after Vilfredo Pareto who developed the concept further in 1896 after it had already been used in 1881 (see Miettinen, 1999, pp. 10-11, and the references therein). In contrast to a Nash equilibrium, where n agents maximize their objectives simultaneously, each assuming that the strategies of the other players are fixed, a Pareto optimal vector of strategies is found if no player can improve her objective without worsening the objective of another player. Thus, Pareto optima are related to maximizing the common good of all players as opposed to the more self-centered Nash equilibria.

The literature on Pareto optimal strategies for n-player stochastic games is rather sparse and, to the best of our knowledge, the problem of optimal investment for competing agents has not yet been solved using the concept of Pareto optimality. Of course, as the agents are assumed to be competitive, it is more intuitive to search for Nash equilibria. However, instead of interpreting Pareto optimality as optimizing the common good, we can have some central planner in mind that manages the portfolios of n competing agents. Hence, the central planner aims to maximize the common good of her clients without this behavior contradicting the competitive incentive of the n agents.

Closest to our work is the article by Branger et al. (2023). They considered n non-competitive agents aiming to maximize the expected utility of their terminal wealth given some fixed initial capital. While the solution to such problems for a single investor is well known in many interesting special cases, Branger et al. (2023) considered the modified problem of collective asset allocation and searched for Pareto optimal vectors of strategies. They defined a novel type of utility function, the so-called collective utility function, which is used to find a Pareto optimal vector of strategies for the *n* agents. An adapted version of their collective utility function is used in this chapter. The collective utility function has the advantage of containing only one state variable, as opposed to the "classical" approach of maximizing a weighted sum of the *n* players' objectives, which results in an objective function that depends on all *n* state variables. The classical approach of maximizing a weighted sum of the *n* agents' objective functions was, e.g., used by Ferrari et al. (2017) and Guo and Xu (2020) in problems other than portfolio optimization. The paper by Ferrari et al. (2017) considers the optimal allocation of the initial wealth of *n* agents between personal consumption and irreversible contributions to increase the level of some public good. The paper by Guo and Xu (2020) considers a central planner problem for *n* players aiming to minimize the running cost corresponding to some general diffusion state process. We refer to Branger et al. (2023) for a more detailed literature overview of collective optimal investment and related topics.

We conclude this introduction with a summary of this chapter. The underlying financial market and the optimization problem are explained in Section 6.1. In Section 6.2, we adapt the scalarization method introduced by Branger et al. (2023) to competitive utility functions (see Section 3.1 for the definition of competitive utility functions). In Section 6.3, we determine a Pareto optimum in terms of optimal terminal wealth. In order to achieve this, we determine the optimal collective terminal wealth first and then deduce the optimal terminal wealth for the individual agents. In the final Section 6.4, we compare the Pareto optimum to the Nash equilibrium from Chapter 3.

6.1. PROBLEM FORMULATION

We base our analysis on the semimartingale financial market from Subsection 2.3.1. In summary, there are d + 1 assets in which n agents can invest. The assets consist of one riskless bond with zero interest rate and d risky stocks. The stock price processes $(S_k(t))_{t \in [0,T]}$, $k = 1, \ldots, d$, are $L^2(\mathbb{P})$ -semimartingales with càdlàg paths. To exclude arbitrage, we require the existence of an equivalent σ -martingale measure \mathbb{Q} . In contrast to Subsection 2.3.1, we make the additional assumption that \mathbb{Q} is the *unique* equivalent σ -martingale measure. Under this assumption, the associated density process

$$Z_t^{\mathbb{Q}} = \mathbb{E}\left[\frac{\mathrm{d}\mathbb{Q}}{\mathrm{d}\mathbb{P}}\Big|\mathcal{F}_t\right], t \in [0,T],$$

is unique as well. Thus, we write $Z_t := Z_t^{\mathbb{Q}}$, $t \in [0, T]$, throughout the present chapter. The assumption that \mathbb{Q} is the unique equivalent σ -martingale measure has the important consequence that each claim $X \in L^2(\mathbb{P})$ has a *unique* fair price given by (cf. (2.6))

$$\mathbb{E}_{\mathbb{Q}}[X] = \mathbb{E}\left[Z_T X\right].$$

Moreover, the existence of a unique equivalent (σ -/local) martingale measure is usually related to a complete financial market (see Subsection 2.3.2). Although at this point we cannot deduce that the financial market is complete, we conjecture that this is the case. However, the possible lack of completeness is not a problem for the later analysis. We refer to Remark 6.10 in which we revisit the issue of completeness. Let us now state the central optimization problem of this chapter. Similar to Chapter 3, we consider n agents with objectives

$$\max_{\pi^i \in \mathcal{A}} \mathbb{E}\left[U_i\left(X_T^{i,\pi^i} - \frac{\theta_i}{n} \sum_{j \neq i} X_T^{j,\pi^j}\right)\right], \ i = 1, \dots, n.$$
(6.1)

The functions $U_i : \mathcal{D}_i \to \mathbb{R}$, defined on $\mathcal{D}_i \in \{(0, \infty), \mathbb{R}\}$, $i = 1, \ldots, n$, are assumed to be Inada utility functions (see Definition 2.11). Note that we did not have to impose the Inada conditions in Chapter 3, but it is crucial for the proofs displayed later in the current chapter. The assumption is not a severe restriction, however, since many classical examples such as the natural logarithm and the exponential or power utility satisfy the Inada conditions.

There are two main differences between the analysis in Chapter 3 and the current chapter. The first one is that in Chapter 3 we were interested in finding the optimal strategy $\pi^{i,*}$, while in this chapter we only search for the optimal terminal wealth. We adapt the solution method proposed by Branger et al. (2023). They use the martingale approach (see, for example, Korn, 1997, Section 3.4) to solve the problem of finding the optimal collective terminal wealth. Thus, they search for the optimal terminal wealth first and find the replicating strategy afterwards. If the financial market is complete, it is always possible to hedge the optimal terminal wealth. However, we do not make the assumption of a complete financial market.

Thus, the objective of agent i is to solve the static optimization problem corresponding to the dynamic problem (6.1). The objective of agent $i \in \{1, ..., n\}$ then reads as

$$\begin{cases} \max_{X_i} J_i(X_1, \dots, X_n) \coloneqq \mathbb{E} \left[U_i \left(X_i - \frac{\theta_i}{n} \sum_{j \neq i} X_j \right) \right], \\ \text{s.t.} \quad X_i \text{ is } \mathcal{F}_T - \text{measurable}, \ \mathbb{E}[Z_T X_i] \le x_0^i, \end{cases}$$
(6.2)

where x_0^i denotes the initial capital and X_i the terminal wealth of agent $i \in \{1, \ldots, n\}$.

The second major difference to Chapter 3 is that the goal in the current chapter is to find vectors of Pareto optimal strategies instead of Nash equilibria. We already gave a definition of a Pareto optimal vector of *strategies* in Section 2.4. However, as we optimize the terminal wealth X_i instead of the portfolio strategy, we repeat the definition in terms of terminal wealth.

Definition 6.1 (Miettinen, 1999, Definition 2.2.1). A vector (X_1^*, \ldots, X_n^*) of admissible states X_i^* , $i = 1, \ldots, n$, such that there is no admissible vector (X_1, \ldots, X_n) with

$$J_i(X_1, \dots, X_n) \ge J_i(X_1^*, \dots, X_n^*) \quad \text{for all } i = 1, \dots, n$$

and

 $J_i(X_1,\ldots,X_n) > J_i(X_1^*,\ldots,X_n^*) \quad \text{for at least one } i \in \{1,\ldots,n\},$

is called *Pareto optimal*. A random variable X_i is considered admissible if it satisfies the conditions of (6.2).

Let us now describe how to find a Pareto optimum for (6.2). The definition of Pareto optimality introduces an optimization problem including an *n*-dimensional objective function which consists of the *n* agents' expected competitive utilities. In general, the first step in solving such problems is to scalarize the objective function. A common and fairly intuitive method of scalarization is to maximize a weighted sum of the *n* objective functions. More specifically, let $\beta_1, \ldots, \beta_n \in (0, 1)$ with $\sum_{i=1}^n \beta_i = 1$ be weights assigned to the *n* agents. Then the scalarized objective function takes the form

$$\sum_{i=1}^n \beta_i J_i(X_1,\ldots,X_n).$$

For any choice of weights β_1, \ldots, β_n , the resulting solution is Pareto optimal (see, e.g., Aubin, 2003, p. 193). Although this method seems simple at first, using it to search for an analytical solution to the problem (6.2) results in some serious difficulties. The main disadvantage we discovered was a system of nonlinear equations emerging in the search for the maximum. Therefore, we use a different scalarization approach which is carried out in the next section.

6.2. Scalarization of the objective function

In the following, we apply a scalarization method introduced by Branger et al. (2023) to the competitive utility functions

$$U_i\left(x_i - \theta_i \bar{x}^{-i}\right)$$

for Inada utility functions $U_i : \mathcal{D}_i \to \mathbb{R}, \ \mathcal{D}_i \in \{(0,\infty), \mathbb{R}\}$, and competition weights $\theta_i \in [0,1]$, $i = 1, \ldots, n$, where $\bar{x}^{-i} = n^{-1} \sum_{j \neq i} x_j$. The scalarized utility function, denoted by \widetilde{U}_{β} , is determined as the optimal value of the optimization problem

$$\begin{cases} \widetilde{U}_{\beta}(x) = \max_{x_1,\dots,x_n} \sum_{i=1}^n \beta_i U_i \left(x_i - \theta_i \bar{x}^{-i} \right), \\ \text{s.t.} \quad \sum_{i=1}^n x_i = x, \ x_i - \theta_i \bar{x}^{-i} \in \mathcal{D}_i, \ i = 1,\dots,n, \end{cases}$$
(6.3)

where $\beta_1, \ldots, \beta_n \in (0, 1)$ describe weights, assigned to the *n* agents, with $\sum_{i=1}^n \beta_i = 1$.

Note that it is not necessary that the utility functions are of the same "type". For example, it would be possible that agent 1 uses an exponential utility function while agent 2 uses the natural logarithm. Moreover, they do not need to be defined on the same domain. We just need to assume that the domain is either the whole real numbers or the strictly positive real numbers, which is a reasonable requirement in the context of utility maximization.

Now we can state the following lemma which gives the optimal solution to (6.3) and the resulting scalarized objective function \tilde{U}_{β} . It turns out that \tilde{U}_{β} is an Inada utility function (see Definition 2.11).

Lemma 6.2. For $n \in \mathbb{N}$, let $U_i : \mathcal{D}_i \to \mathbb{R}$, i = 1, ..., n, be Inada utility functions defined on domains $\mathcal{D}_i \in \{(0, \infty), \mathbb{R}\}$. Let $I_i : (0, \infty) \to \mathcal{D}_i$ denote the inverse of the first order derivative U'_i of U_i . Further, let $\theta_i \in [0, 1]$ and $\beta_i \in (0, 1)$, i = 1, ..., n, be chosen so that $\sum_{j=1}^n \beta_j = 1$. Finally, define a function $\tilde{I} : (0, \infty) \to \bigcup_{j=1}^n \mathcal{D}_j$ by

$$\tilde{I}(y) = \frac{1}{1-\hat{\theta}} \sum_{j=1}^{n} \frac{n}{n+\theta_j} I_j\left(\frac{ny}{\beta_j(n+\theta_j)(1-\hat{\theta})}\right),$$

where $\hat{\theta} = \sum_{j=1}^{n} \frac{\theta_j}{n+\theta_j}$. Then, for any $x \in \bigcup_{j=1}^{n} \mathcal{D}_j$, the optimization problem (6.3) has a unique solution x_1^*, \ldots, x_n^* given by

$$x_i^* = f_i(x) \coloneqq \frac{\theta_i}{n + \theta_i} \cdot x + \frac{n}{n + \theta_i} I_i\left(\frac{n\tilde{I}^{-1}(x)}{\beta_i(n + \theta_i)(1 - \hat{\theta})}\right), \ i = 1, \dots, n,$$
(6.4)

in terms of the inverse function $\tilde{I}^{-1} : \bigcup_{j=1}^{n} \mathcal{D}_{j} \to (0, \infty)$ of \tilde{I} . The functions f_{i} , i = 1, ..., n, are bijections from $\bigcup_{j=1}^{n} \mathcal{D}_{j}$ to $\bigcup_{j=1}^{n} \mathcal{D}_{j}$. Finally, for $x \in \bigcup_{j=1}^{n} \mathcal{D}_{j}$, the optimal value of (6.3) is given by

$$\widetilde{U}_{\beta}(x) \coloneqq \sum_{j=1}^{n} \beta_j U_j \left(I_j \left(\frac{n \widetilde{I}^{-1}(x)}{\beta_j (n+\theta_j)(1-\widehat{\theta})} \right) \right),$$

which defines an Inada utility function $\widetilde{U}_{\beta} : \bigcup_{j=1}^{n} \mathcal{D}_{j} \to \mathbb{R}$.

Proof. The optimization problem is solved using the Lagrange dual method. The constraints $x_i - \theta_i \bar{x}^{-i} \in \mathcal{D}_i, i = 1, ..., n$, are only relevant if $\mathcal{D}_i = (0, \infty)$. We solve the optimization problem without the constraints and prove that the optimal solution satisfies them afterwards. Hence, we consider the optimization problem

$$\begin{cases} \max_{x_1,\dots,x_n} \sum_{i=1}^n \beta_i U_i \left(x_i - \theta_i \bar{x}^{-i} \right), \\ \text{s.t.} \quad \sum_{i=1}^n x_i = x. \end{cases}$$
(6.5)

The associated Lagrangian function is given by

$$\mathcal{L}(x_1,\ldots,x_n;\lambda) = \sum_{i=1}^n \beta_i U_i \left(x_i - \theta_i \bar{x}^{-i} \right) - \lambda \left(\sum_{i=1}^n x_i - x \right)$$

for the Lagrange multiplier $\lambda > 0$. Note that the objective function is concave and the constraint is linear. Thus, to find a solution to (6.5), it is sufficient to determine x_1, \ldots, x_n such that $\frac{\partial}{\partial x_j} \mathcal{L}(x_1, \ldots, x_n; \lambda) = 0$ for all $j \in \{1, \ldots, n\}$, and such that $\sum_{j=1}^n x_j = x$. The first order partial derivative of \mathcal{L} with respect to x_j for some $j \in \{1, \ldots, n\}$ is given by

$$\frac{\partial}{\partial x_j} \mathcal{L}(x_1, \dots, x_n; \lambda) = \sum_{i=1}^n \beta_i \frac{\partial}{\partial x_j} U_i \left(x_i - \theta_i \bar{x}^{-i} \right) - \lambda$$
$$= \beta_j U_j' \left(x_j - \theta_j \bar{x}^{-j} \right) - \sum_{i \neq j} \frac{\beta_i \theta_i}{n} U_i' \left(x_i - \theta_i \bar{x}^{-i} \right) - \lambda \stackrel{!}{=} 0$$

If we set

$$u_i \coloneqq U'_i \left(x_i - \theta_i \bar{x}^{-i} \right), \quad \hat{u} \coloneqq \sum_{i=1}^n \frac{\beta_i \theta_i}{n} u_i,$$

the resulting system of linear equations can be written as

$$\beta_i \left(1 + \frac{\theta_i}{n}\right) u_i - \hat{u} = \lambda, \quad i = 1, \dots, n$$

The unique (implicit) solution to this system of linear equations is given by

$$u_i = \frac{n(\lambda + \hat{u})}{\beta_i \left(n + \theta_i\right)}.\tag{6.6}$$

Inserting (6.6) into the definition of \hat{u} yields

$$\hat{u} = \sum_{i=1}^{n} \frac{\beta_i \theta_i}{n} u_i = (\lambda + \hat{u}) \sum_{i=1}^{n} \frac{\theta_i}{n + \theta_i} = \hat{\theta}(\lambda + \hat{u}).$$

Since $0 \leq \hat{\theta} < 1$ (see Lemma 3.3), it follows

$$\hat{u} = \frac{\lambda \hat{\theta}}{1 - \hat{\theta}}$$

Therefore, using (6.6), an explicit representation of u_i is given by

$$u_{i} = \frac{n}{\beta_{i} \left(n + \theta_{i}\right)} \left(\lambda + \frac{\hat{\theta}}{1 - \hat{\theta}}\lambda\right) = \frac{n\lambda}{\beta_{i} \left(n + \theta_{i}\right) \left(1 - \hat{\theta}\right)}$$

Thus, using the definition of u_i and the inverse I_i of U'_i ,

$$\left(1+\frac{\theta_i}{n}\right)x_i - \frac{\theta_i}{n}\hat{x} \coloneqq x_i - \theta_i \bar{x}^{-i} = I_i\left(\frac{n\lambda}{\beta_i\left(n+\theta_i\right)\left(1-\hat{\theta}\right)}\right),$$

which yields the following implicit representation of x_i

$$x_{i} = \frac{\theta_{i}}{n + \theta_{i}}\hat{x} + \frac{n}{n + \theta_{i}}I_{i}\left(\frac{n\lambda}{\beta_{i}\left(n + \theta_{i}\right)\left(1 - \hat{\theta}\right)}\right),\tag{6.7}$$

where we defined $\hat{x} \coloneqq \sum_{i=1}^{n} x_i$. Inserting (6.7) into the definition of \hat{x} yields

$$\hat{x} = \sum_{i=1}^{n} x_i = \hat{\theta}\hat{x} + \sum_{i=1}^{n} \frac{n}{n+\theta_i} I_i\left(\frac{n\lambda}{\beta_i \left(n+\theta_i\right) \left(1-\hat{\theta}\right)}\right)$$

Since $\hat{\theta} < 1$ (see Lemma 3.3), it follows

$$\hat{x} = \frac{1}{1-\hat{\theta}} \sum_{i=1}^{n} \frac{n}{n+\theta_i} I_i \left(\frac{n\lambda}{\beta_i \left(n+\theta_i \right) \left(1-\hat{\theta} \right)} \right),$$

and therefore,

$$x_{i} = \frac{\theta_{i}}{\left(n+\theta_{i}\right)\left(1-\hat{\theta}\right)} \sum_{j=1}^{n} \frac{n}{n+\theta_{j}} I_{j}\left(\frac{n\lambda}{\beta_{j}\left(n+\theta_{j}\right)\left(1-\hat{\theta}\right)}\right) + \frac{n}{n+\theta_{i}} I_{i}\left(\frac{n\lambda}{\beta_{i}\left(n+\theta_{i}\right)\left(1-\hat{\theta}\right)}\right).$$

Finally, the Lagrange multiplier λ needs to be chosen such that the constraint $\sum_{i=1}^{n} x_i = x$ is

satisfied. Therefore, we define the function $\tilde{I}: (0,\infty) \to \bigcup_{j=1}^n \mathcal{D}_j$ by

$$\tilde{I}(\lambda) = \frac{1}{1-\hat{\theta}} \sum_{j=1}^{n} \frac{n}{n+\theta_j} I_j\left(\frac{n\lambda}{\beta_j \left(n+\theta_j\right) \left(1-\hat{\theta}\right)}\right) (=\hat{x}).$$

Since the functions I_i take values in $\mathcal{D}_i \in \{(0, \infty), \mathbb{R}\}, i = 1, ..., n$, and all constant factors are strictly positive, the function \tilde{I} takes values in $\bigcup_{j=1}^n \mathcal{D}_j$. The functions I_i are strictly decreasing and continuously differentiable. Thus, \tilde{I} also has these properties. Moreover, U_i satisfies the Inada conditions, which implies

$$\lim_{x \searrow 0} I_i(x) = \infty, \quad \lim_{x \to \infty} I_i(x) = \inf \mathcal{D}_i \in \{0, -\infty\}$$

for all $i \in \{1, \ldots, n\}$. Hence, it follows that $\lim_{\lambda \searrow 0} \tilde{I}(\lambda) = \infty$.

When λ tends to ∞ , there are two possibilities we need to consider. If $\mathcal{D}_i = (0, \infty)$ for all $i \in \{1, \ldots, n\}$, then each summand in the definition of \tilde{I} tends to 0 and therefore, \tilde{I} converges to 0 as λ tends to ∞ . If at least one of the utility functions is defined on the whole real numbers, at least one summand in the definition of \tilde{I} tends to $-\infty$, while the other summands tend to 0 or to $-\infty$ as well. Hence, in this case \tilde{I} converges to $-\infty$ as λ tends to ∞ .

In summary, the asymptotic behavior of \tilde{I} is given as follows

$$\lim_{\lambda \searrow 0} \tilde{I}(\lambda) = \infty, \ \lim_{\lambda \to \infty} \tilde{I}(\lambda) = \inf \bigcup_{j=1}^{n} \mathcal{D}_j \in \{0, -\infty\}.$$

The previous observations imply that \tilde{I} is a bijection from $(0, \infty)$ to $\bigcup_{j=1}^{n} \mathcal{D}_{j}$ and therefore, we can consider the inverse of \tilde{I} denoted by $\tilde{I}^{-1} : \bigcup_{j=1}^{n} \mathcal{D}_{j} \to (0, \infty)$. Since \tilde{I} is strictly decreasing and continuously differentiable, \tilde{I}^{-1} inherits those properties.

Finally, since \tilde{I}^{-1} is a bijection from $\bigcup_{j=1}^{n} \mathcal{D}_{j}$ to $(0, \infty)$, we can conclude that there exists a unique $\lambda^{*} \in (0, \infty)$ such that $\tilde{I}(\lambda^{*}) = x$, which is given by $\lambda^{*} = \tilde{I}^{-1}(x)$. Hence, a candidate for the optimal solution is given by

$$x_{i} = \frac{\theta_{i}}{\left(n+\theta_{i}\right)\left(1-\hat{\theta}\right)} \sum_{j=1}^{n} \frac{n}{n+\theta_{j}} I_{j}\left(\frac{n\tilde{I}^{-1}(x)}{\beta_{j}\left(n+\theta_{j}\right)\left(1-\hat{\theta}\right)}\right) + \frac{n}{n+\theta_{i}} I_{i}\left(\frac{n\tilde{I}^{-1}(x)}{\beta_{i}\left(n+\theta_{i}\right)\left(1-\hat{\theta}\right)}\right)$$
$$= \frac{\theta_{i}}{n+\theta_{i}} x + \frac{n}{n+\theta_{i}} I_{i}\left(\frac{n\tilde{I}^{-1}(x)}{\beta_{i}\left(n+\theta_{i}\right)\left(1-\hat{\theta}\right)}\right). \tag{6.8}$$

By construction, the values x_i , i = 1, ..., n, maximize the objective function and satisfy the constraint $\sum_{i=1}^{n} x_i = x$. Therefore, the only thing left to show is that the remaining constraint $x_i - \theta_i \bar{x}^{-i} \in \mathcal{D}_i$ is also satisfied. It follows

$$x_i - \theta_i \bar{x}^{-i} = \left(1 + \frac{\theta_i}{n}\right) x_i - \frac{\theta_i}{n} \sum_{j=1}^n x_j = \left(1 + \frac{\theta_i}{n}\right) x_i - \frac{\theta_i}{n} x = I_i \left(\frac{n\tilde{I}^{-1}(x)}{\beta_i \left(n + \theta_i\right) \left(1 - \hat{\theta}\right)}\right),$$

where we used (6.8) in the last step. Since I_i takes values in \mathcal{D}_i , the constraint $x_i - \theta_i \bar{x}^{-i} \in \mathcal{D}_i$ is satisfied.

Therefore, the optimal solution to the optimization problem is indeed given by

$$x_i^* = \frac{\theta_i}{n + \theta_i} x + \frac{n}{n + \theta_i} I_i \left(\frac{n \tilde{I}^{-1}(x)}{\beta_i \left(n + \theta_i \right) \left(1 - \hat{\theta} \right)} \right)$$

The optimal value of the objective function is then given by

$$\widetilde{U}_{\beta}(x) \coloneqq \sum_{i=1}^{n} \beta_i U_i \left(I_i \left(\frac{n \widetilde{I}^{-1}(x)}{\beta_i \left(n + \theta_i \right) \left(1 - \widehat{\theta} \right)} \right) \right),$$

defining a function $\widetilde{U}_{\beta} : \bigcup_{i=1}^{n} \mathcal{D}_{i} \to \mathbb{R}$. In the following, we prove that \widetilde{U}_{β} satisfies the properties of an Inada utility function. To begin with, we determine the first and second order derivative of \widetilde{U}_{β} :

$$\begin{split} \tilde{U}_{\beta}'(x) &= \sum_{i=1}^{n} \left\{ \beta_{i} U_{i}' \left(I_{i} \left(\frac{n \tilde{I}^{-1}(x)}{\beta_{i} \left(n + \theta_{i} \right) \left(1 - \hat{\theta} \right)} \right) \right) I_{i}' \left(\frac{n \tilde{I}^{-1}(x)}{\beta_{i} \left(n + \theta_{i} \right) \left(1 - \hat{\theta} \right)} \right) \cdot \frac{n \left(\tilde{I}^{-1} \right)'(x)}{\beta_{i} \left(n + \theta_{i} \right) \left(1 - \hat{\theta} \right)} \right\} \\ &= \tilde{I}^{-1}(x) \left(\tilde{I}^{-1} \right)'(x) \sum_{i=1}^{n} \frac{n^{2}}{\beta_{i} \left(n + \theta_{i} \right)^{2} \left(1 - \hat{\theta} \right)^{2}} I_{i}' \left(\frac{n \tilde{I}^{-1}(x)}{\beta_{i} \left(n + \theta_{i} \right) \left(1 - \hat{\theta} \right)} \right) = \tilde{I}^{-1}(x). \end{split}$$

In the last step, we used the following auxiliary calculation

$$\left(\tilde{I}^{-1}\right)'(x) = \frac{1}{\tilde{I}'(\tilde{I}^{-1}(x))} = \left(\frac{\mathrm{d}}{\mathrm{d}y}\sum_{i=1}^{n}\frac{n}{(n+\theta_i)(1-\hat{\theta})}I_i\left(\frac{ny}{\beta_i(n+\theta_i)(1-\hat{\theta})}\right)\Big|_{y=\tilde{I}^{-1}(x)}\right)^{-1} \\ = \left(\sum_{i=1}^{n}\frac{n^2}{\beta_i(n+\theta_i)^2(1-\hat{\theta})^2}I_i'\left(\frac{n\tilde{I}^{-1}(x)}{\beta_i(n+\theta_i)(1-\hat{\theta})}\right)\right)^{-1}.$$

Hence, the second order derivative of \widetilde{U}_{β} is simply given by

$$\tilde{U}_{\beta}^{''}(x) = \left(\tilde{I}^{-1}\right)^{'}(x) = \left(\sum_{i=1}^{n} \frac{n^2}{\beta_i \left(n + \theta_i\right)^2 (1 - \hat{\theta})^2} I_i^{'}\left(\frac{n\tilde{I}^{-1}(x)}{\beta_i \left(n + \theta_i\right) (1 - \hat{\theta})}\right)\right)^{-1}.$$
(6.9)

Since the utility functions U_i were assumed to be twice continuously differentiable, the previous representations of the first and second order derivative of \tilde{U}_{β} imply that \tilde{U}_{β} is twice continuously differentiable as well. Additionally, as \tilde{I}^{-1} takes values in $(0, \infty)$, the first order derivative of \tilde{U}_{β} is strictly positive and \tilde{U}_{β} is strictly increasing. Moreover, the functions I_i are strictly decreasing by the strict concavity of U_i and therefore, (6.9) implies that \tilde{U}_{β} is strictly concave.

Finally, we need to examine the asymptotic behavior of $\tilde{U}'_{\beta} = \tilde{I}^{-1}$. The Inada conditions of U_i , i = 1, ..., n, imply

$$\lim_{y \searrow 0} \tilde{I}(y) = \sum_{i=1}^{n} \frac{n}{(n+\theta_i)(1-\hat{\theta})} \underbrace{\lim_{y \searrow 0} I_i\left(\frac{ny}{\beta_i(n+\theta_i)(1-\hat{\theta})}\right)}_{=\infty} = \infty,$$
$$\lim_{y \to \infty} \tilde{I}(y) = \sum_{i=1}^{n} \frac{n}{(n+\theta_i)(1-\hat{\theta})} \underbrace{\lim_{y \to \infty} I_i\left(\frac{ny}{\beta_i(n+\theta_i)(1-\hat{\theta})}\right)}_{=\inf \mathcal{D}_i} = \inf \bigcup_{i=1}^{n} \mathcal{D}_i.$$
(6.10)
The last equation (6.10) follows from the assumption $\mathcal{D}_i \in \{(0,\infty),\mathbb{R}\}$ and the fact that the constant factors are strictly positive. Hence, the inverse of \tilde{I} satisfies

$$\lim_{x \to \infty} \tilde{I}^{-1}(x) = 0, \lim_{x \to \inf \bigcup_{i=1}^n \mathcal{D}_i} \tilde{I}^{-1}(x) = \infty.$$

In summary, it follows that \widetilde{U}_{β} is an Inada utility function.

Finally, it remains to show that the functions f_i from (6.4) are bijections. The function f_i is strictly increasing in x, since I_i and \tilde{I}^{-1} are both strictly decreasing and all constant factors are strictly positive. Moreover, f_i is continuous as a composition of continuous functions. Finally, using the asymptotic behavior of I_i and \tilde{I}^{-1} ,

$$\lim_{x \to \infty} f_i(x) = \infty, \lim_{x \to \inf \bigcup_{j=1}^n \mathcal{D}_j} f_i(x) = \inf \bigcup_{j=1}^n \mathcal{D}_j.$$

Hence, f_i is a bijection from $\bigcup_{j=1}^n \mathcal{D}_j$ onto $\bigcup_{j=1}^n \mathcal{D}_j$ for each $i \in \{1, \ldots, n\}$. This concludes our proof.

The collective competitive utility function \tilde{U}_{β} can be used to find a Pareto optimum (X_1^*, \ldots, X_n^*) , where X_i^* represents the optimal terminal wealth of agent *i*. First, the optimal collective wealth X^* is determined with respect to some classical portfolio optimization problem under the utility function \tilde{U}_{β} . Afterwards, choosing $X_i^* = f_i(X^*)$, $i = 1, \ldots, n$, gives a Pareto optimum with respect to the original objective functions $\mathbb{E}[U_i(X_i - \theta_i \bar{X}^{-i})]$. To verify that this yields a Pareto optimal solution to the original problem, let X^* be the optimal solution to

$$\begin{cases} \max_X \mathbb{E}\left[\widetilde{U}_{\beta}(X)\right],\\ \text{s.t.} \quad X \text{ is } \mathcal{F}_T\text{-measurable}, \ \mathbb{E}[Z_T X] \le x_0 = \sum_{i=1}^n x_0^i. \end{cases}$$
(6.11)

Lemma 6.2 provides bijections f_1, \ldots, f_n for which the random variables $X_i^* \coloneqq f_i(X^*), i = 1, \ldots, n$, satisfy

$$\widetilde{U}_{\beta}(X^*) = \max_{\sum_{i=1}^n X_i = X^*} \sum_{i=1}^n \beta_i U_i(X_1, \dots, X_n) = \sum_{i=1}^n \beta_i U_i(X_1^*, \dots, X_n^*).$$

The expressions $U_i(X_1, \ldots, X_n)$ denote the competitive utility functions

$$U_i(X_1,\ldots,X_n) \coloneqq U_i\left(X_i - \frac{\theta_i}{n}\sum_{j\neq i}X_j\right), i = 1,\ldots,n.$$

Further, assume that there exist admissible $\widetilde{X}_1, \ldots, \widetilde{X}_n$ (i.e., \widetilde{X}_j satisfies the constraints of (6.2), $j = 1, \ldots, n$) such that

$$\mathbb{E}[U_i(X_1^*,\ldots,X_n^*)] \le \mathbb{E}[U_i(\widetilde{X}_1,\ldots,\widetilde{X}_n)] \text{ for all } i=1,\ldots,n,$$
$$\mathbb{E}[U_j(X_1^*,\ldots,X_n^*)] < \mathbb{E}[U_j(\widetilde{X}_1,\ldots,\widetilde{X}_n)] \text{ for at least one } j \in \{1,\ldots,n\}.$$

Moreover, define $\widetilde{X} \coloneqq \sum_{i=1}^{n} \widetilde{X}_i$. Then, using that $\beta_i > 0$ for all $i = 1, \ldots, n$,

$$\mathbb{E}\left[\widetilde{U}_{\beta}(X^{*})\right] = \sum_{i=1}^{n} \beta_{i} \mathbb{E}\left[U_{i}(X_{1}^{*}, \dots, X_{n}^{*})\right]$$

$$< \sum_{i=1}^{n} \beta_{i} \mathbb{E}\left[U_{i}(\widetilde{X}_{1}, \dots, \widetilde{X}_{n})\right]$$

$$= \mathbb{E}\left[\sum_{i=1}^{n} \beta_{i} U_{i}(\widetilde{X}_{1}, \dots, \widetilde{X}_{n})\right]$$

$$\leq \mathbb{E}\left[\max_{\sum_{i=1}^{n} X_{i} = \widetilde{X}} \sum_{i=1}^{n} \beta_{i} U_{i}(X_{1}, \dots, X_{n})\right] = \mathbb{E}\left[\widetilde{U}_{\beta}(\widetilde{X})\right]$$

This contradicts the assumption that X^* is an optimal solution to (6.11). Hence, (X_1^*, \ldots, X_n^*) is Pareto optimal.

Remark 6.3. If we replace the constraint $\sum_{i=1}^{n} x_i = x$ in (6.3) by $\sum_{i=1}^{n} \alpha_i x_i = x$, where $\alpha_i \in (0, 1)$ are weights with $\sum_{i=1}^{n} \alpha_i = 1$, it follows

$$\widetilde{U}_{\beta}(x) = \sum_{i=1}^{n} \beta_{i} U_{i} \left(I_{i} \left(\frac{n \widetilde{I}^{-1}(x)}{\beta_{i}(n+\theta_{i})} \left(\alpha_{i} + \frac{\widehat{\alpha \theta}}{1-\widehat{\theta}} \right) \right) \right).$$

The function \tilde{I}^{-1} denotes the inverse of $\tilde{I}: (0,\infty) \to \bigcup_{j=1}^n \mathcal{D}_j$ defined by

$$\tilde{I}(y) = \sum_{j=1}^{n} \frac{n}{n+\theta_j} \left(\alpha_j + \frac{\widehat{\alpha}\widehat{\theta}}{1-\widehat{\theta}} \right) I_j \left(\frac{ny}{\beta_j(n+\theta_j)} \left(\alpha_j + \frac{\widehat{\alpha}\widehat{\theta}}{1-\widehat{\theta}} \right) \right),$$

where $\widehat{\alpha\theta} \coloneqq \sum_{j=1}^{n} \frac{\alpha_j \theta_j}{n+\theta_j}$. The introduction of $\alpha_1, \ldots, \alpha_n$ extends the function \widetilde{U}_β by *n* additional free parameters that can be chosen with respect to some further conditions on the functions f_i or the resulting terminal wealth $X_i = f_i(X)$, where X denotes the collective terminal wealth. \Diamond Before we use the scalarized objective function \widetilde{U}_{β} to find a Pareto optimum for (6.2), let us determine U_{β} for two examples of utility functions U_1, \ldots, U_n .

Example 6.4. Let us apply Lemma 6.2 to two special cases of Inada utility functions.

a) Let $n \in \mathbb{N}$ and $U_i(x) = \log(x)$, i = 1, ..., n, with $\mathcal{D}_i = (0, \infty)$. Then the collective utility function $\widetilde{U}_{\beta}: (0,\infty) \to \mathbb{R}$ from Lemma 6.2 is given by

$$\widetilde{U}_{\beta}(x) = \log(x) + \sum_{i=1}^{n} \beta_i \log\left(\beta_i \left(1 + \frac{\theta_i}{n}\right) (1 - \hat{\theta})\right).$$

b) Let $n \in \mathbb{N}$ and $U_i(x) = -\exp(-\delta_i^{-1}x)$ on $\mathcal{D}_i = \mathbb{R}$ for parameters $\delta_i > 0, i = 1, \dots, n$. Then the collective utility function $\widetilde{U}_{\beta} : \mathbb{R} \to \mathbb{R}$ from Lemma 6.2 is given by

$$\widetilde{U}_{\beta}(x) = -\frac{\widehat{\delta\theta}}{1-\widehat{\theta}} \cdot \exp\left(-\frac{1}{\widehat{\delta\theta}} \sum_{j=1}^{n} \frac{n\delta_{j}}{n+\theta_{j}} \log\left(\frac{n\delta_{j}}{\beta_{j} \left(n+\theta_{j}\right) \left(1-\widehat{\theta}\right)}\right)\right) \cdot \exp\left(-\frac{1-\widehat{\theta}}{\widehat{\delta\theta}} \cdot x\right),$$

here $\widehat{\delta\theta} \coloneqq \sum_{j=1}^{n} \frac{n\delta_{j}}{n+\theta_{j}}.$

where $\delta \theta \coloneqq \sum_{j=1}^{n} \frac{m_j}{n+\theta_j}$.

Remark 6.5. Branger et al. (2023) introduced the collective utility function as the optimal value of the optimization problem

$$\widetilde{U}_{\beta}(X) = \max_{\sum_{i=1}^{n} X_i = X} \sum_{i=1}^{n} \beta_i U_i(X_i).$$

They obtained

$$\widetilde{U}_{\beta}(X) = \sum_{i=1}^{n} \beta_i U_i \left(I_i \left(\frac{y}{\beta_i} \right) \right),$$

where y is chosen such that

$$\sum_{i=1}^{n} I_i\left(\frac{y}{\beta_i}\right) = X.$$

Thus, up to constants depending only on the preference parameters of the agents, the optimal value coincides with our result displayed in Lemma 6.2.

6.3. PARETO OPTIMUM FOR THE RELATIVE PERFORMANCE PROBLEM

In the following, the scalarized utility function \tilde{U}_{β} is used to find a Pareto optimum for (6.2).

6.3.1. MAXIMIZATION OF THE COLLECTIVE TERMINAL WEALTH

The first step towards a Pareto optimum for (6.2) is to determine the collective optimal terminal wealth with respect to the collective competitive utility function \tilde{U}_{β} . The optimal collective wealth is determined as the optimal solution to

$$\begin{cases} \max_{X} \mathbb{E}\left[\widetilde{U}_{\beta}(X)\right], \\ \text{s.t.} \quad X \text{ is } \mathcal{F}_{T}\text{-measurable}, \mathbb{E}\left[Z_{T}X\right] \leq x_{0} \coloneqq \sum_{j=1}^{n} x_{0}^{j}. \end{cases}$$
(6.12)

The following lemma provides the unique optimal solution to (6.12).

Lemma 6.6. Let $U_i : \mathcal{D}_i \to \mathbb{R}$ be Inada utility functions on $\mathcal{D}_i \in \{(0, \infty), \mathbb{R}\}, i = 1, ..., n, and$ let \widetilde{U}_β be the scalarized objective function from Lemma 6.2. Further, let $x_0 \in \bigcup_{i=1}^n \mathcal{D}_i$. Finally, assume that

$$\mathbb{E}\left[\left|Z_{T}I_{i}\left(\frac{nyZ_{T}}{\beta_{i}\left(n+\theta_{i}\right)\left(1-\hat{\theta}\right)}\right)\right|\right] < \infty$$
(6.13)

for all $y \in (0,\infty)$ and i = 1, ..., n. Then the unique optimal solution to (6.12) is given by

$$X^* = \tilde{I}\left(\lambda^* Z_T\right) = \sum_{i=1}^n \frac{n}{\left(n + \theta_i\right)\left(1 - \hat{\theta}\right)} I_i\left(\frac{n\lambda^* Z_T}{\beta_i\left(n + \theta_i\right)\left(1 - \hat{\theta}\right)}\right),$$

where $\lambda^* > 0$ is the unique value such that $\mathbb{E}[Z_T X^*] = x_0$.

Proof. Using a standard result (see, e.g., Kramkov and Schachermayer, 1999) and Lemma 6.2, which provides that \tilde{U}_{β} is an Inada utility function, the unique optimal solution to (6.12) is given by

$$X^* = \left(\widetilde{U}_{\beta}'\right)^{-1} (\lambda Z_T).$$

The Lagrange multiplier $\lambda > 0$ needs to be chosen with respect to the constraint $\mathbb{E}[Z_T X^*] = x_0$. Using results from the proof of Lemma 6.2, the first order derivative of \tilde{U}_{β} coincides with the inverse \tilde{I}^{-1} of

$$\tilde{I}: (0,\infty) \to \bigcup_{i=1}^{n} \mathcal{D}_{i}, \ y \mapsto \sum_{i=1}^{n} \frac{n}{(n+\theta_{i})(1-\hat{\theta})} I_{i}\left(\frac{ny}{\beta_{i}(n+\theta_{i})(1-\hat{\theta})}\right).$$

Therefore, $(\widetilde{U}_{\beta})^{-1} = \widetilde{I}$ and X^* is given by

$$X^* = \tilde{I}(\lambda Z_T) = \sum_{i=1}^n \frac{n}{(n+\theta_i)(1-\hat{\theta})} I_i\left(\frac{n\lambda Z_T}{\beta_i(n+\theta_i)(1-\hat{\theta})}\right).$$

To find the Lagrange multiplier, define a function \mathcal{H} on $(0,\infty)$ by

$$\mathcal{H}(\lambda) \coloneqq \mathbb{E}\left[Z_T X^*\right] = \sum_{i=1}^n \frac{n}{(n+\theta_i)\left(1-\hat{\theta}\right)} \mathbb{E}\left[Z_T \cdot I_i\left(\frac{n\lambda Z_T}{\beta_i\left(n+\theta_i\right)\left(1-\hat{\theta}\right)}\right)\right]$$

Now $\lambda^* > 0$ needs to be chosen such that $\mathcal{H}(\lambda^*) = x_0$. Existence and uniqueness of λ^* are shown using the intermediate value theorem. To verify that \mathcal{H} is continuous, fix some arbitrary $\lambda > 0$ and a sequence $(\lambda_k)_{k \in \mathbb{N}} \subset (0, \infty)$ with $\lim_{k \to \infty} \lambda_k = \lambda$. Then it follows that

$$\lim_{k \to \infty} \mathcal{H}(\lambda_k) = \sum_{i=1}^n \frac{n}{(n+\theta_i)(1-\hat{\theta})} \lim_{k \to \infty} \mathbb{E}\left[Z_T \cdot I_i\left(\frac{n\lambda_k Z_T}{\beta_i(n+\theta_i)(1-\hat{\theta})}\right)\right]$$
$$= \sum_{i=1}^n \frac{n}{(n+\theta_i)(1-\hat{\theta})} \mathbb{E}\left[Z_T \cdot I_i\left(\frac{n\lambda Z_T}{\beta_i(n+\theta_i)(1-\hat{\theta})}\right)\right]$$
$$= \mathcal{H}(\lambda).$$
(6.14)

Equation (6.14) holds since each function I_i , i = 1, ..., n, is continuous and strictly decreasing, in combination with the dominated convergence theorem and the integrability assumption (6.13) (see Section A.1 in the appendix for the derivation of an integrable majorant). Thus, \mathcal{H} is continuous.

Now we need to examine the asymptotic behavior of \mathcal{H} . Using a monotone convergence theorem for increasing and decreasing sequences of functions which are not necessarily non-negative (Theorem 11.1 in Schilling, 2005), we obtain

$$\lim_{\lambda \searrow 0} \mathcal{H}(\lambda) = \lim_{\lambda \searrow 0} \mathbb{E} \left[Z_T \tilde{I} \left(\lambda Z_T \right) \right] = \infty,$$

since \tilde{I} is strictly decreasing with $\lim_{y \searrow 0} \tilde{I}(y) = \infty$ (which was shown in the proof of Lemma 6.2). If $\mathcal{D}_i = \mathbb{R}$ for at least one $i \in \{1, \ldots, n\}$, we need to decompose the argument of the expectation into its positive and negative part. Analogously, it follows

$$\lim_{\lambda \to \infty} \mathcal{H}(\lambda) = \lim_{\lambda \to \infty} \mathbb{E}\left[Z_T \tilde{I}(\lambda Z_T) \right] = \inf \bigcup_{i=1}^n \mathcal{D}_i,$$

since $\lim_{y\to\infty} \tilde{I}(y) = \inf \bigcup_{j=1}^n \mathcal{D}_j$. Now the intermediate value theorem in combination with the

strict monotonicity of \mathcal{H} implies the existence of a unique value $\lambda^* \in (0, \infty)$ with $\mathcal{H}(\lambda^*) = x_0$. In summary,

$$X^* = \sum_{i=1}^n \frac{n}{(n+\theta_i)(1-\hat{\theta})} I_i\left(\frac{n\lambda^* Z_T}{\beta_i(n+\theta_i)(1-\hat{\theta})}\right)$$

describes the unique optimal solution to (6.12).

Remark 6.7. The integrability assumption (6.13) is, for example, satisfied in the important special cases of a lognormal state price density in combination with logarithmic, power or exponential utility. In the logarithmic and power case, this observation follows easily using the lognormal distribution of Z_T . The integrability under exponential utility follows from Lemma A.1 in the Appendix.

6.3.2. Pareto optimal terminal wealth of the individual investors

The second step in the search for a Pareto optimum is to combine the optimal collective terminal wealth X^* from Lemma 6.6 and the bijections f_i , i = 1, ..., n, from Lemma 6.2. However, we need to make sure that the budget constraint

$$\mathbb{E}\left[Z_T f_i(X^*)\right] = x_0^i \tag{6.15}$$

is satisfied for each agent $i \in \{1, ..., n\}$. It turns out that this is only the case if the weights $\beta_1, ..., \beta_n$ are chosen correctly. The subsequent lemma provides the unique choice of weights $\beta_1, ..., \beta_n$ subject to (6.15).

Lemma 6.8. Assume that the conditions of Lemmas 6.2 and 6.6 are satisfied. Moreover, let $\lambda^* > 0$ be the Lagrange multiplier chosen in Lemma 6.6, X^* the associated optimal collective wealth at time T, and f_i , i = 1, ..., n, the bijections from Lemma 6.2. Then the unique solution β_i^* , i = 1, ..., n, to the system of equations

$$\mathbb{E}[Z_T f_i(X^*)] = x_0^i, \ i = 1, \dots, n, \ \sum_{i=1}^n \beta_i = 1,$$

is given by

$$\beta_i^* = \frac{h_i \left(x_0^i - \theta_i \bar{x}_0^{-i} \right)}{\sum_{j=1}^n h_j \left(x_0^j - \theta_j \bar{x}_0^{-j} \right)}$$

The functions $h_i : \mathcal{D}_i \to (0, \infty)$ are defined as $h_i = \frac{1}{H_i^{-1}}$, where H_i^{-1} denotes the inverse of $H_i : (0, \infty) \to \mathcal{D}_i$, where

$$H_i(y) = \mathbb{E}\left[Z_T I_i\left(\frac{nyZ_T}{(n+\theta_i)(1-\hat{\theta})}\right)\right], \ i = 1, \dots, n$$

Proof. To begin with, it is important to understand that the *n*-dimensional system of equations given by $\mathbb{E}[Z_T f_i(X^*)] = x_0^i$, i = 1, ..., n, is underdetermined. Assume that $\mathbb{E}[Z_T f_i(X^*)] = x_0^i$ is

satisfied for all $i \in \{1, \ldots, n-1\}$. Then it follows

$$\sum_{i=1}^{n} x_{0}^{i} = x_{0} = \mathbb{E}\left[Z_{T}X^{*}\right] = \mathbb{E}\left[Z_{T}\sum_{i=1}^{n} f_{i}(X^{*})\right] = \sum_{i=1}^{n-1} x_{0}^{i} + \mathbb{E}\left[Z_{T}f_{n}(X^{*})\right]$$

and, thus, $\mathbb{E}[Z_T f_n(X^*)] = x_0 - \sum_{i=1}^{n-1} x_0^i = x_0^n$. Hence, solving the system of equations given by $\mathbb{E}[Z_T f_i(X^*)] = x_0^i, i = 1, ..., n$, leaves one free variable which can then be chosen with respect to the condition $\sum_{i=1}^n \beta_i = 1$.

Now let $X^* = \tilde{I}(\lambda^* Z_T)$ be the collective optimal terminal wealth displayed in Lemma 6.6 and, for $i \in \{1, \ldots, n\}$, let f_i be the bijection given in Lemma 6.2, i.e.,

$$f_i(X^*) = \frac{\theta_i}{n+\theta_i} X^* + \frac{n}{n+\theta_i} I_i \left(\frac{n\tilde{I}^{-1}(X^*)}{\beta_i (n+\theta_i) (1-\hat{\theta})} \right)$$
$$= \frac{\theta_i}{n+\theta_i} \tilde{I}(\lambda^* Z_T) + \frac{n}{n+\theta_i} I_i \left(\frac{n\lambda^* Z_T}{\beta_i (n+\theta_i) (1-\hat{\theta})} \right).$$

The Lagrange multiplier $\lambda^* > 0$ was chosen in Lemma 6.6 subject to

$$\mathbb{E}\left[Z_T \tilde{I}\left(\lambda^* Z_T\right)\right] = x_0 = \sum_{i=1}^n x_0^i$$

Hence, it follows

$$\mathbb{E}\left[Z_T f_i(X^*)\right] = \frac{\theta_i}{n+\theta_i} x_0 + \frac{n}{n+\theta_i} \mathbb{E}\left[Z_T I_i\left(\frac{n\lambda^* Z_T}{\beta_i \left(n+\theta_i\right) \left(1-\hat{\theta}\right)}\right)\right] \stackrel{!}{=} x_0^i.$$

This can equivalently be written as

$$H_i\left(\frac{\lambda^*}{\beta_i}\right) \coloneqq \mathbb{E}\left[Z_T I_i\left(\frac{n\lambda^* Z_T}{\beta_i \left(n+\theta_i\right) \left(1-\hat{\theta}\right)}\right)\right] = x_0^i - \theta_i \bar{x}_0^{-i},\tag{6.16}$$

where $\bar{x}_0^{-i} = \frac{1}{n} \sum_{j \neq i} x_0^j$, i = 1, ..., n. The proof of Lemma 6.6 provides that the functions H_i are continuous, strictly decreasing, and satisfy

$$\lim_{y \searrow 0} H_i(y) = \infty, \ \lim_{y \to \infty} H_i(y) = \inf \mathcal{D}_i, \ i = 1, \dots, n.$$

Hence, H_i has an inverse $H_i^{-1}: \mathcal{D}_i \to (0, \infty)$ and we can equivalently rewrite (6.16) as

$$\frac{\lambda^*}{\beta_i} = H_i^{-1} \left(x_0^i - \theta_i \bar{x}_0^{-i} \right) \tag{6.17}$$

for all $i \in \{1, ..., n\}$. Note that (6.17) only gives an implicit representation of β_i since λ^* depends on $\beta_1, ..., \beta_n$. Since the system of equations $\mathbb{E}[Z_T f_i(X^*)] = x_0^i$, i = 1, ..., n, is underdetermined, we find $\beta_1, ..., \beta_{n-1}$ in terms of β_n and choose $\beta_n = 1 - \sum_{i=1}^{n-1} \beta_i$ afterwards. As (6.17) needs to hold for each $i \in \{1, ..., n\}$, we deduce

$$\beta_i H_i^{-1} (x_0^i - \theta_i \bar{x}_0^{-i}) = \lambda^* = \beta_n H_n^{-1} (x_0^n - \theta_n \bar{x}_0^{-n})$$

for all $i \in \{1, \ldots, n\}$. Equivalently, β_i is given by

$$\beta_i = \beta_n \frac{H_n^{-1}(x_0^n - \theta_n \bar{x}_0^{-n})}{H_i^{-1}(x_0^i - \theta_i \bar{x}_0^{-i})}, \ i = 1, \dots, n-1,$$
(6.18)

in terms of β_n . Now the remaining condition $\sum_{i=1}^n \beta_i = 1$ implies

$$\beta_n = 1 - \sum_{i=1}^{n-1} \beta_i = 1 - \beta_n H_n^{-1} \left(x_0^n - \theta_n \bar{x}_0^{-n} \right) \sum_{i=1}^{n-1} \frac{1}{H_i^{-1} \left(x_0^i - \theta_i \bar{x}_0^{-i} \right)}, \quad i = 1, \dots, n.$$
(6.19)

The functions H_i^{-1} take strictly positive values. Thus, we can rewrite (6.19) as

$$\beta_n = \left(1 + H_n^{-1}(x_0^n - \theta_n \bar{x}_0^{-n}) \sum_{i=1}^{n-1} \frac{1}{H_i^{-1}(x_0^i - \theta_i \bar{x}_0^{-i})}\right)^{-1} \\ = \left(H_n^{-1}(x_0^n - \theta_n \bar{x}_0^{-n}) \sum_{i=1}^n \frac{1}{H_i^{-1}(x_0^i - \theta_i \bar{x}_0^{-i})}\right)^{-1} = \frac{h_n(x_0^n - \theta_n \bar{x}_0^{-n})}{\sum_{i=1}^n h_i(x_0^i - \theta_i \bar{x}_0^{-i})},$$

where we introduced $h_i := \frac{1}{H_i^{-1}}, i = 1, ..., n$. Inserting β_n into (6.18) yields

$$\beta_{i} = \beta_{n} \frac{h_{i} \left(x_{0}^{i} - \theta_{i} \bar{x}_{0}^{-i} \right)}{h_{n} \left(x_{0}^{n} - \theta_{n} \bar{x}_{0}^{-n} \right)} = \frac{h_{i} \left(x_{0}^{i} - \theta_{i} \bar{x}_{0}^{-i} \right)}{\sum_{j=1}^{n} h_{j} \left(x_{0}^{j} - \theta_{j} \bar{x}_{0}^{-j} \right)}.$$
(6.20)

Since $\beta_i \in (0, 1)$, i = 1, ..., n, is obviously satisfied, the unique choice of β_i , i = 1, ..., n, with the desired properties is given by (6.20).

If the weights β_i , i = 1, ..., n, are chosen with respect to Lemma 6.8, the budget constraints

$$\mathbb{E}\left[Z_T f_i(X^*)\right] = x_0^i, \quad i = 1, \dots, n,$$

are satisfied by construction. Hence, the following theorem gives a Pareto optimum for the multi-objective optimization problem

$$\begin{cases} \max_{X_i} \mathbb{E}\left[U_i\left(X_i - \theta_i \bar{X}^{-i}\right)\right], \\ \text{s.t.} \quad X_i \text{ is } \mathcal{F}_T - \text{measurable}, \mathbb{E}\left[Z_T X_i\right] \leq x_0^i, \end{cases} \quad i = 1, \dots, n.$$

Theorem 6.9. Assume that the conditions of Lemmas 6.2 and 6.6 are satisfied. Moreover, let $\lambda^* > 0$ be the unique Lagrange multiplier chosen in the proof of Lemma 6.6 and β_i^* , i = 1, ..., n, the weights from Lemma 6.8. Then

$$X_{i}^{*} = \frac{n}{n+\theta_{i}} I_{i} \left(\frac{n\lambda_{i}^{*} Z_{T}}{\left(n+\theta_{i}\right) \left(1-\hat{\theta}\right)} \right) + \frac{\theta_{i}}{\left(n+\theta_{i}\right) \left(1-\hat{\theta}\right)} \sum_{j=1}^{n} \frac{n}{n+\theta_{j}} I_{j} \left(\frac{n\lambda_{j}^{*} Z_{T}}{\left(n+\theta_{j}\right) \left(1-\hat{\theta}\right)} \right), \quad (6.21)$$

i = 1, ..., n, is a Pareto optimum for (6.2). The constants λ_i^* , i = 1, ..., n, are given by $\lambda_i^* = H_i^{-1} \left(x_0^i - \theta_i \bar{x}_0^{-i} \right)$ for the function H_i^{-1} introduced in Lemma 6.8.

Proof. The assertion is a direct consequence of the statements of Lemmas 6.2, 6.6, and 6.8. For the introduction of the constants λ_i^* , we used that (6.17) holds for all $i \in \{1, \ldots, n\}$.

Remark 6.10. In Section 6.1, we explained that the search for the optimal terminal wealth instead of the optimal investment strategy originates from the martingale approach, a commonly used method to solve portfolio optimization problems (see, for example, Section 3.4 in Korn, 1997, Kramkov and Schachermayer, 1999, or Subsection 2.3.3). The martingale approach consists of two subproblems. The first one is the static optimization problem in which the optimal terminal wealth X^* is determined. The second one is the representation problem in which a replicating strategy is determined, i.e., an admissible strategy φ for which $X^* = X_T^{\varphi} = x_0 + (\varphi \cdot S)_T$ holds. By x_0 we denote the corresponding initial capital. The martingale approach is usually applied in complete financial markets because in such markets, it can be guaranteed that the representation problem is solvable. We do not explicitly assume here that the financial market is complete. Under the additional assumption of completeness, we could ensure that the representation problem has a solution. Of course, even without completeness it might be the case that the specific optimal terminal wealth obtained in Theorem 6.9 is attainable by some admissible strategy. However, we leave this problem open for future research.

Example 6.11. Let us apply Theorem 6.9 to the two utility functions used in Example 6.4.

a) First, assume that all *n* agents use the natural logarithm as their utility function. We do not have to make any further assumptions regarding the underlying financial market. The reason is that the inverse of the first order derivative of the natural logarithm is given by $I(x) = x^{-1}$, so that the expected value $\mathbb{E}[Z_T I(\lambda Z_T)] = \lambda^{-1}$ can be determined directly without specifying the distribution of Z_T . Thus, $H_i(y) = \frac{(n+\theta_i)(1-\hat{\theta})}{ny} = H_i^{-1}(y)$ and therefore,

$$\lambda_i^* = \frac{(n+\theta_i)(1-\theta)}{n(x_0^i - \theta_i \bar{x}_0^{-i})}, \quad i = 1, \dots, n$$

Inserting $I_i(x) = x^{-1}$ and λ_i^* into (6.21) yields

$$X_i^* = \frac{x_0^i}{Z_T}, i = 1, \dots, n, \text{ and } X^* = \frac{x_0}{Z_T}$$

Hence, the optimal terminal wealth for the individual agents coincides with the optimal terminal wealth for a single agent (without interaction) using the natural logarithm as her utility function and the initial capital x_0^i , i = 1, ..., n.

b) Now assume that all agents use exponential utility functions of the form $U_i(x) = -\exp(-\frac{1}{\delta_i}x)$ for risk tolerance parameters $\delta_i > 0$, i = 1, ..., n. Moreover, in order to determine the Lagrange multipliers λ^* and $\lambda_1^*, ..., \lambda_n^*$ explicitly, we need to specify the underlying financial market. We consider a one-dimensional Black-Scholes market with drift $\mu \in \mathbb{R}$, volatility $\sigma > 0$, and interest rate r = 0. Hence, the discounted state price density is given by

$$Z_T = \exp\left(-\frac{\mu}{\sigma}W_T - \frac{\mu^2}{2\sigma^2}T\right),$$

where W is a one-dimensional Brownian motion (see, e.g., Eberlein and Kallsen, 2019, Example 9.17). Theorem 6.9 implies

$$X_i^* = x_0^i + \left(\frac{n\delta_i}{n+\theta_i} + \frac{\theta_i\widehat{\delta\theta}}{(n+\theta_i)(1-\widehat{\theta})}\right) \left(\frac{\mu}{\sigma}W_T + \frac{\mu^2}{\sigma^2}T\right)$$

and

$$X^* = x_0 + \frac{\widehat{\delta\theta}}{1 - \hat{\theta}} \left(\frac{\mu}{\sigma} W_T + \frac{\mu^2}{\sigma^2} T \right).$$

where $\widehat{\delta\theta} = \sum_{j=1}^{n} \frac{n\delta_j}{n+\theta_j}$. Note that we used the following auxiliary result to determine the Lagrange multipliers: If $X \sim \mathcal{N}(\mu, \sigma^2)$, then $\mathbb{E}[X \cdot \exp(X)] = (\mu + \sigma^2) \exp(\mu + \sigma^2/2)$. The auxiliary result can be shown through a direct calculation (see Lemma A.1 in the Appendix).

 \Diamond

6.4. Comparison with the Nash equilibrium from Section 3.2

In the previous chapters, we searched for Nash equilibria. In general, Nash equilibria are not Pareto optimal (see, e.g., Carmona, 2016, p. 169). In the underlying situation however, the Nash equilibrium determined in Chapter 3 coincides with the Pareto optimum given in Theorem 6.9. In the following, we justify this observation.

Theorem 6.9 displays the optimal terminal wealth of agent $i \in \{1, ..., n\}$ in the Pareto optimum as

$$X_i^{*,P} = \frac{n}{n+\theta_i} I_i \left(\frac{n\lambda_i^* Z_T}{\left(n+\theta_i\right) \left(1-\hat{\theta}\right)} \right) + \frac{\theta_i}{\left(n+\theta_i\right) \left(1-\hat{\theta}\right)} \sum_{j=1}^n \frac{n}{n+\theta_j} I_j \left(\frac{n\lambda_j^* Z_T}{\left(n+\theta_j\right) \left(1-\hat{\theta}\right)} \right),$$

where $\lambda_i^* = H_i^{-1} \left(x_0^i - \theta_i \bar{x}_0^{-i} \right)$. The function H_i^{-1} was defined in Lemma 6.8 as the inverse of H_i , where

$$H_{i}(y) = \mathbb{E}\left[Z_{T}I_{i}\left(\frac{nyZ_{T}}{(n+\theta_{i})(1-\hat{\theta})}\right)\right], y \in (0,\infty)$$

Now we compare $X_i^{*,P}$ to the optimal terminal wealth $X_i^{*,N}$ in the Nash equilibrium. Thus, Theorem 3.2 implies

$$X_{i}^{*,N} = \frac{n}{n+\theta_{i}}Y_{i}^{*} + \frac{\theta_{i}}{(n+\theta_{i})(1-\hat{\theta})}\sum_{j=1}^{n}\frac{n}{n+\theta_{j}}Y_{j}^{*},$$

where Y_i^* is the unique optimal solution to

$$\begin{cases} \max_{Y_i} \mathbb{E}\left[U_i(Y_i)\right],\\ \text{s.t.} \quad Y_i \text{ is } \mathcal{F}_T - \text{measurable}, \ \mathbb{E}\left[Z_T Y_i\right] \leq \tilde{x}_0^i \end{cases}$$

for $\tilde{x}_0^i = x_0^i - \frac{\theta_i}{n} \sum_{j \neq i} x_0^j$. Using standard arguments (see Kramkov and Schachermayer, 1999), Y_i^* is given by

$$Y_i^* = I_i \left(G_i^{-1}(\widetilde{x}_0^i) Z_T \right),$$

where G_i^{-1} is defined as the inverse of $G_i: (0, \infty) \to \mathbb{R}$ given by

$$G_i(y) = \mathbb{E}\left[Z_T I_i(yZ_T)\right] = H_i\left(y\left(1 + \frac{\theta_i}{n}\right)(1 - \hat{\theta})\right).$$

Thus,

$$X_{i}^{*,N} = \frac{n}{n+\theta_{i}} I_{i} \left(G_{i}^{-1}(\tilde{x}_{0}^{i}) Z_{T} \right) + \frac{\theta_{i}}{\left(n+\theta_{i} \right) \left(1-\hat{\theta} \right)} \sum_{j=1}^{n} \frac{n}{n+\theta_{j}} I_{j} \left(G_{j}^{-1}(\tilde{x}_{0}^{j}) Z_{T} \right).$$

 ${\rm Since}^1$

$$G_i^{-1}(\cdot) = \frac{nH_i^{-1}(\cdot)}{(n+\theta_i)\left(1-\hat{\theta}\right)},$$

it follows that $X_i^{*,N} = X_i^{*,P}$. Thus, the optimal terminal wealth in the Nash equilibrium and the Pareto optimum are identical.

¹For functions f and g with inverse functions f^{-1} and g^{-1} , and a constant $\alpha \neq 0$ with $f(x) = g(\alpha x)$ for all x, it follows that $f^{-1} = \frac{g^{-1}}{\alpha}$

CHAPTER 7

NASH EQUILIBRIA FOR RELATIVE INVESTORS UNDER (NON)LINEAR PRICE IMPACT

In the classical mathematical finance literature, it is usually assumed that the underlying financial markets are perfectly elastic, meaning that asset prices are not affected by trades (see, among others, Merton, 1969, 1975; Black and Scholes, 1973). While this assumption is reasonable for small investors, there is plenty of empirical evidence for large traders having a significant impact on asset prices (see, for example, Bouchaud, 2009; Cronqvist and Fahlenbrach, 2009). Thus, there is a rapidly growing strand of mathematical literature regarding financial markets without perfect elasticity. Webster (2023) gives a historical overview of the literature on price impact dating back to the seminal work of Kyle (1985). According to Webster (2023, p. 5), the general economic idea behind price impact is that "trading of [a] stock cause[s] price moves for the stock that otherwise would not have happened". The literature lists several different explanations for the appearance of price impact. Jarrow (1992, 1994) explains that the reason a large trader (i.e., an investor that affects prices through her trades) is "large" might either be significant wealth or that other traders think she has private information. DeMarzo and Urošević (2006) argue that large investors are often institutions or companies, but it is also possible for a private individual to be a large trader. Bouchaud (2009) lists additional reasons for price impact and gives a short review on the different possibilities for mathematical market models to incorporate price impact. A survey by Gatheral and Schied (2013) gives a more detailed overview of the various price impact models covered by the literature. In the following, we explain some features of these models, but our introduction cannot claim to give an exhaustive overview of the diverse literature on price impact.

As stated by Schöneborn and Schied (2009), the literature on price impact follows two separate paths. The first is to derive models that describe empirical findings for price impact appropriately. The second line of research takes exogenously given models and solves classical mathematical finance problems such as absence of arbitrage, asset pricing, and optimal trading. As we aim to find optimal investment strategies for risk-averse agents in a competitive price impact model, our focus lies solely on the second line of research. For literature on the optimal model choice based on empirical data, we refer to Schöneborn and Schied (2009) and references therein. Moreover, Webster (2023) includes empirical data and compares it to different models from the mathematical literature.

It is important to understand that there are two different types of price impact – temporary and permanent price impact. Almgren and Chriss (2001) explain how temporary price impact is caused by a large order being placed, leading to a temporary imbalance in supply. Hence, the asset price deviates from equilibrium¹ for a short period of time and then returns. In contrast, permanent price impact describes a change in the equilibrium price which lasts over the whole time span covered by the model. There is a wide variety of articles displaying models which contain both temporary and permanent price impact (Almgren and Chriss, 2001; Schöneborn and Schied, 2009; Horst and Naujokat, 2010; Schied and Zhang, 2017, 2019; Schied et al., 2017; Luo and Schied, 2019, to name a few). The model used throughout this chapter includes only permanent price impact. Nevertheless, let us explain the difference between these two features and the resulting interpretations.

In general, temporary price impact is unfavorable for the large trader as she has to pay the price *after* it has been changed by her order. As Bank and Baum (2004) argue, the large trader "always has to trade on the bad side". Hence, in such market models, questions of optimal execution of an order or optimal liquidation of a fixed position play an important role. Such problems have been considered by, among others, He and Mamaysky (2005), Schied et al. (2017), Schied and Zhang (2017, 2019), and Schöneborn and Schied (2009). Moreover, we should also mention the popular linear price impact model by Obizhaeva and Wang (2013), in which the authors find the optimal order execution strategy that minimizes execution cost. Finally, in the model of Bank and Dolinsky (2023), the large trader has additional information on the course of the future stock price. Specifically, they introduce price impact as a penalty for the large trader in order to exclude arbitrage strategies arising from the additional information available to the trader.

For permanent price impact, it is generally not possible to tell whether or not the price impact is beneficial for the large trader. In this case, it depends on a particular feature of the model. If the market is constructed such that prices change immediately and the large investor trades on the bad side of the price, permanent price impact admits the same interpretation as temporary price impact and can be considered a disadvantage for the large trader. This is often the case if the stock price is affected by the large trader's order directly through some reaction function (see, for example, Bank and Baum, 2004; Jarrow, 1992, 1994). In these models, so-called market manipulation strategies are often of interest. A market manipulation strategy is an arbitrage strategy with respect to real wealth instead of paper wealth, i.e., the amount of money attained after liquidating the current position (Jarrow, 1992, 1994). Identifying under what conditions such strategies do (not) exist is a central topic in the literature on price impact. If prices do not change immediately, the large trader could use her impact to her own benefit. In such models, finding criteria for the absence of arbitrage portfolios is of significant importance. Cvitanić and Ma (1996)

¹A financial market is in equilibrium if market-clearing is realized. That is, if the supply and demand for assets in the market coincide (see, e.g., Yan, 2018, p. 47).

and Cuoco and Cvitanić (1998) defined a market model with indirect price impact, where the large trader's orders only affect the model parameters, i.e., the interest rate of the riskless asset, and the drift and volatility process of the risky stocks. A special case of their model was later used by Kraft and Kühn (2011). Eksi and Ku (2017) and Ku and Zhang (2018) described similar models where the large trader only affects the drift process of the risky asset. The model specified in Section 7.1 provides a special case of the aforementioned models. A different approach to define an indirect price impact model that reacts slowly to the large trader's orders was given by Busch et al. (2013). There, the large trader only affects the regime switching intensity of the risky asset's price process. Eksi and Ku (2017) argue that in models where price impact is beneficial for the large trader, it is counterintuitive to pose questions of optimal order execution. Thus, typical issues considered in such models are the pricing of asset claims as well as optimal investment and consumption over a fixed time interval.

Although this feature is not relevant to the analysis in this thesis, let us mention a third type of price impact, namely, transient price impact. Under transient price impact, the effect of a large order is strongest immediately after the order has been placed and vanishes over time. Common ways to model this feature are exponential decay as used by Schied et al. (2017), or a more general decay kernel as used by Luo and Schied (2019) as well as Schied and Zhang (2019).

Throughout this thesis, we are interested in problems of optimal portfolio choice for strategically interacting agents. Competition between agents can be seen as an effect of price impact if the agents are aware of their own as well as the others' impact on asset prices. Problems of strategically interacting large traders, often called *market impact games*, have been considered by Carlin et al. (2007), Horst and Naujokat (2010), Schöneborn and Schied (2009), Schied et al. (2017), and Schied and Zhang (2019). In each of these papers, a finite number of agents aims to minimize liquidation/execution cost in a competitive environment. Risk-averse investors competing to maximize expected utility of terminal wealth have, for instance, been considered by Curatola (2019, 2022) and Schied and Zhang (2017). In the aforementioned articles, the authors search for open-loop Nash equilibria for agents interacting strategic agents who cumulatively affect the price process of a risky stock. They also elaborate the difference between open-loop and closed-loop Nash equilibria (see also Section 2.4).

To the best of our knowledge, the existing literature only considers strategic interaction between competing investors caused by either relative concerns included into the objective function or by their joint impact on asset prices. In the current chapter, we combine these features in order to display both the effect of large investments on the stock price (caused by a large number of small investors) and the investors' desire to outperform their opponents. Thus, we find the unique solution to a multi-objective portfolio optimization problem, similar to the one introduced in Chapter 3, in a financial market where the n agents' investment affects the stock price dynamics linearly. Our model provides a generalization of the financial market used by Kraft and Kühn (2011) by including multiple risky assets instead of just one. First, we determine the unique constant Nash equilibrium if investors use exponential utility functions. Afterwards, we study the influence of the parameter α , which measures the sensitivity of the stock price to the agents' investment, on the Nash equilibrium and the stock price process. Furthermore, as empirical data suggests that price impact is *not* linear, we also consider the nonlinear case. We prove that, as long as the price impact grows sublinearly, the emerging best response problems are solvable. If the price impact grows superlinearly, there is no optimal solution to these best response problems. Finally, we switch from exponential to power utility and introduce a different type of objective function including the multiplicative relative performance metric (3.2). This criterion was previously used by Basak and Makarov (2015) and Lacker and Zariphopoulou (2019), among others. In that case, using the linear price impact model again, we are able to find the unique constant Nash equilibrium.

7.1. INTRODUCTION OF THE PRICE IMPACT MARKET

Let $(\Omega, \mathcal{F}, (\mathcal{F}_t)_{t \in [0,T]}, \mathbb{P})$ be a filtered probability space and T > 0 a finite time horizon. Further, let $W = (W_1, \ldots, W_d)$ be a *d*-dimensional Brownian motion. The underlying financial market contains a riskless bond, which is for simplicity assumed to be identical to 1. Moreover, there are *d* stocks. Our goal is to define their price processes as the solution to the following stochastic differential equation

$$dS_k(t) = S_k(t) \left((\mu_k(t) + \alpha_k \bar{\pi}_k(t)) dt + \sum_{\ell=1}^d \sigma_{k\ell}(t) dW_\ell(t) \right), \ S_k(0) = 1.$$
(7.1)

The expression $\bar{\pi}_k(t) = \frac{1}{n} \sum_{j=1}^n \pi_k^j(t)$ describes the arithmetic mean of the *n* agents investment into the *k*-th stock at time $t \in [0, T]$.

Before we discuss the requirements we need to impose on the components of (7.1), let us discuss the structure of the stock price and its interpretations in general. First, we notice that the price impact of the n agents is homogeneous in the sense that the weight $\frac{\alpha_k}{n}$ of the investment into the k-th stock is the same for each of the n agents. This feature of our model is based on the idea that the n agents are small if we consider them by themselves. However, as n is supposed to be large, we can treat the group of n agents like one large agent. Hence, it is reasonable to assume that the influence of each agent's investment on the stock price is assigned the same weight. Moreover, it makes the model mathematically more tractable. Second, we assume that the agents' investment only affects the drift and not the volatility of the stock price. Cuoco and Cvitanić (1998) give a detailed explanation of this assumption. To summarize, they base the assumption on two different factors. On one hand, they argue that market equilibrium conditions affect the stock price as a whole, and not the drift and volatility processes separately. Hence, one could fix the volatility matrix and only manipulate the drift in order to clear the market. On the other hand, they refer to empirical data in which the volatility is independent of the quantities invested into the asset. A third feature of our model is the affine linear structure of the drift process, which was also used by Kraft and Kühn (2011). They argue that, additionally to resulting in a more tractable model, the affine linear structure is a typical outcome of equilibrium models and hence, a reasonable assumption. Moreover, they explain that the affine linear structure can be interpreted as a first-order approximation of a more general price impact function.

Now we proceed with the conditions we need to impose on the processes contained in the stock price dynamics (7.1). The conditions are taken from Section 1.1 in Karatzas and Shreve (1998). The drift process $\mu = (\mu_1, \ldots, \mu_d)$ is an (\mathcal{F}_t) -progressively measurable, *d*-dimensional stochastic process satisfying

$$\mathbb{P}\left(\int_0^T \|\mu(t)\| \mathrm{d}t < \infty\right) = 1.$$
(7.2)

The volatility process $\sigma = (\sigma_{k\ell})_{1 \le k, \ell \le d}$ is an $\mathbb{R}^{d \times d}$ -valued stochastic process such that $\sigma_{k\ell}$ is (\mathcal{F}_t) -progressively measurable with

$$\mathbb{P}\left(\int_0^T \sigma_{k\ell}(t)^2 \mathrm{d}t < \infty\right) = 1$$

for all $k, \ell \in \{1, \ldots, d\}$. Moreover, we assume that $\sigma(t)$ is regular \mathbb{P} -almost surely for all $t \in [0, T]$. The expression $\overline{\pi}_k(t)$ denotes the arithmetic mean of the investment of n agents into the k-th stock at time $t \in [0, T]$, i.e.,

$$\bar{\pi}_k(t) = \frac{1}{n} \sum_{i=1}^n \pi_k^i(t),$$

where $\pi_k^i(t)$ describes either the amount or the fraction of wealth agent *i* invests into the *k*-th stock at time $t \in [0, T]$.

Finally, $\alpha_k \in \mathbb{R}$, $k = 1, \ldots, d$, are constants that describe the impact of the *n* agents' investment into the *k*-th stock. Hence, we refer to them as *price impact parameters*. Some authors argue that α_k should take both positive and negative values due to the fact that (large) investors may have both positive and negative impact on stock returns (see, e.g., Cronqvist and Fahlenbrach, 2009; Curatola, 2019). On the other hand, Bank and Baum (2004) prove that stock prices need to be increasing in terms of a large trader's investment. Otherwise, it would be possible to construct some "In & Out" arbitrage strategy. However, such arbitrage strategies arise due to the direct change in the share price in their model and are therefore not an issue in our case. Moreover, since the optimization problems considered in the linear price impact market have finite optimal solutions, our market is free of arbitrage. Hence, we allow for general $\alpha_k \in \mathbb{R}$, $k = 1, \ldots, d$. The constants $\alpha_1, \ldots, \alpha_n$ are collected in the diagonal matrix

$$A \coloneqq \operatorname{diag}\left(\alpha_1, \ldots, \alpha_d\right),$$

which will be used later on.

Before we can use the stock price processes (7.1), we need to pose some assumptions on the investment processes π^i , i = 1, ..., n, of the *n* investors. The assumptions are again based on the financial market in Section 1.1 of Karatzas and Shreve (1998). Since the drift in (7.1) needs to satisfy (7.2), we obtain the condition

$$\mathbb{P}\left(\int_0^T \|\mu(t) + A\bar{\pi}_t\| \mathrm{d}t < \infty\right) = 1.$$

However, using the triangle inequality, it suffices² to request that the strategies π^i satisfy

$$\mathbb{P}\left(\int_0^T \left\|\pi^i(t)\right\|^2 \mathrm{d}t < \infty\right) = 1, \ i = 1, \dots, n,$$

since μ satisfies (7.2) and A is a deterministic, constant diagonal matrix.

Regarding the investment strategies, Karatzas and Shreve (1998) request, additional to progressive measurability, the following conditions for a portfolio process π

$$\mathbb{P}\left(\int_0^T \left|\pi(t)^\top \mu(t)\right| \mathrm{d}t < \infty\right) = 1,\tag{7.3}$$

$$\mathbb{P}\left(\int_0^T \left\|\sigma(t)^{\top} \pi(t)\right\|^2 \mathrm{d}t < \infty\right) = 1.$$
(7.4)

The second assumption (7.4) is the same in our setting, whereas the first assumption (7.3) translates to

$$\mathbb{P}\left(\int_0^T \left|\pi^i(t)^\top \left(\mu(t) + A\bar{\pi}_t\right)\right| \mathrm{d}t < \infty\right) = 1$$

Using the triangle and Cauchy-Schwarz inequality, we can request instead that

$$\mathbb{P}\left(\int_0^T \left|\mu(t)^\top \pi^i(t)\right| \mathrm{d}t < \infty\right) = 1,$$
$$\mathbb{P}\left(\int_0^T \left\|\pi^i(t)\right\|^2 \mathrm{d}t < \infty\right) = 1$$

hold for all $i \in \{1, ..., n\}$. To summarize, the sufficient integrability assumptions for a strategy π^i are

$$\mathbb{P}\left(\int_0^T \left\|\pi^i(t)\right\|^2 \mathrm{d}t < \infty\right) = 1,\tag{7.5}$$

$$\mathbb{P}\left(\int_0^T \left|\mu(t)^\top \pi^i(t)\right| \mathrm{d}t < \infty\right) = 1,\tag{7.6}$$

$$\mathbb{P}\left(\int_0^T \left\|\sigma(t)^\top \pi^i(t)\right\|^2 \, \mathrm{d}t < \infty\right) = 1.$$
(7.7)

The set \mathcal{A} of admissible strategies is, thus, given by

$$\mathcal{A} \coloneqq \left\{ \pi : \pi \text{ is an } \mathbb{R}^d \text{-valued, } (\mathcal{F}_t)_{t \in [0,T]} \text{-progressively measurable process that satisfies} \right.$$
the integrability conditions (7.5) - (7.7)
$$\left. (7.8)^{-1} \right\}$$

Remark 7.1. If μ and σ are constant, the set \mathcal{A} is simply given by the set of \mathbb{R}^d -valued, progressively measurable processes π such that $\mathbb{P}\left(\int_0^T \|\pi(t)\|^2 \, dt < \infty\right) = 1.$

²Note that it would be sufficient to assume that $\mathbb{P}(\int_0^T \|\pi^i(t)\| dt < \infty) = 1$, but since we need the stronger quadratic assumption later, we already pose it at this point.

For admissible strategies π^j , j = 1, ..., n, we denote the wealth process of investor $i \in \{1, ..., n\}$ by $(X_t^{i,\pi^i})_{t \in [0,T]}$. It should be noted that X^{i,π^i} depends on all n strategies $\pi^1, ..., \pi^n$. The expression $\pi^j_k(t)$ will be interpreted as either the amount or the fraction of wealth invested by investor j into stock k at time $t \in [0,T]$. This depends on the particular example since, for CARA utilities, it is more convenient to optimize the invested amount, while the invested fraction works better in combination with CRRA utility functions.

Now let x_0^i denote the initial capital of agent $i \in \{1, ..., n\}$. If $\pi_k^i(t)$ describes the amount investor i invests into stock k at time t, the associated wealth process is given by

$$\begin{aligned} X_t^{i,\pi^i} &= x_0^i + \sum_{k=1}^d \int_0^t \pi_k^i(s) \left((\mu_k(s) + \alpha_k \bar{\pi}_k(s)) \, \mathrm{d}s + \sum_{\ell=1}^d \sigma_{k\ell}(s) \mathrm{d}W_\ell(s) \right) \\ &= x_0^i + \int_0^t \pi^i(s)^\top ((\mu(s) + A\bar{\pi}(s)) \mathrm{d}s + \sigma \mathrm{d}W(s)), \ t \in [0,T]. \end{aligned}$$

In contrast, if $\pi_k^i(t)$ describes the fraction of investor *i*'s wealth invested into stock k at time t, the wealth process is given by

$$\begin{split} X_t^{i,\pi^i} &= x_0^i \exp\left(\int_0^t \sum_{k=1}^d \pi_k^i(s) \Big(\mu_k(s) + \alpha_k \bar{\pi}_k(s) - \frac{1}{2} \sum_{\ell,p=1}^d \pi_p^i(s) \sigma_{k\ell}(s) \sigma_{p\ell}(s) \Big) \mathrm{d}s \\ &+ \int_0^t \sum_{k,\ell=1}^d \pi_k^i(s) \sigma_{k\ell}(s) \mathrm{d}W_\ell(s) \Big) \\ &= x_0^i \exp\left(\int_0^t \pi^i(s)^\top \left(\mu(s) + A\bar{\pi}(s) - \frac{1}{2} \sigma(s) \sigma(s)^\top \pi^i(s) \right) \mathrm{d}s + \int_0^t \pi^i(s)^\top \sigma(s) \mathrm{d}W(s) \right), \, t \in [0,T]. \end{split}$$

7.2. Relative performance problem under general utility

As explained at the beginning of this chapter, our goal is to solve a problem similar to (3.3) in the previously described price impact market. In the following, we describe a method to find Nash equilibria in terms of invested amounts for objective functions including the additive relative performance metric. Hence, our goal is to find all Nash equilibria to the multi-objective optimization problem

$$\begin{cases} \sup_{\pi^{i} \in \mathcal{A}} \mathbb{E} \left[U_{i} \left(X_{T}^{i,\pi^{i}} - \frac{\theta_{i}}{n} \sum_{j \neq i} X_{T}^{j,\pi^{j}} \right) \right], \\ \text{s.t.} \quad X_{T}^{i,\pi^{i}} = x_{0}^{i} + \int_{0}^{T} \pi^{i}(t)^{\top} \left((\mu(t) + A\bar{\pi}(t)) \, \mathrm{d}t + \sigma(t) \mathrm{d}W(t) \right), \end{cases}$$
(7.9)

i = 1, ..., n, where $U_i : \mathcal{D}_i \to \mathbb{R}$ describes a general utility function defined on some domain $\mathcal{D}_i \in \{(0, \infty), \mathbb{R}\}$ (see Definition 2.11). Since the difference in the argument of U_i may become negative, we extend U_i to the whole real line by setting $U_i(x) = -\infty$ if $x \notin \mathcal{D}_i$. Obviously, this does not change the optimal value in (7.9). Nevertheless, we assume that $x_0^i - \frac{\theta_i}{n} \sum_{j \neq i} x_0^j \in \mathcal{D}_i$ for all $i \in \{1, ..., n\}$ (see also Remark 3.1).

In order to solve (7.9), we choose some investor $i \in \{1, ..., n\}$ and assume that the strategies π^j , $j \neq i$, of the other agents are given. Under these conditions, we can rewrite the optimization problem (7.9) as a classical portfolio optimization problem in a different price impact market.

Afterwards, Nash equilibria can be determined using the solution to the classical problem. First, define the process $(Y_t^{i,\varphi^i})_{t\in[0,T]}$ by

$$Y_t^{i,\varphi^i} = X_t^{i,\pi^i} - \frac{\theta_i}{n} \sum_{j \neq i} X_t^{j,\pi^j}, \ t \in [0,T], \ i = 1, \dots, n,$$
(7.10)

where we further defined the strategy φ^i by

$$\varphi_k^i(t) = \pi_k^i(t) - \frac{\theta_i}{n} \sum_{j \neq i} \pi_k^j(t), \ t \in [0, T], \ k = 1, \dots, d, \ i = 1, \dots, n$$

Note that φ^i is admissible since the conditions imposed in the definition of \mathcal{A} are preserved under linear combinations of elements in \mathcal{A} . Then we can write Y_T^{i,φ^i} as

$$\begin{split} Y_T^{i,\varphi^i} &= X_T^{i,\pi^i} - \frac{\theta_i}{n} \sum_{j \neq i} X_T^{j,\pi^j} \\ &= \underbrace{x_0^i - \frac{\theta_i}{n} \sum_{j \neq i} x_0^j + \sum_{k=1}^d \int_0^T \left(\underbrace{\pi_k^i(t) - \frac{\theta_i}{n} \sum_{j \neq i} \pi_k^j(t)}_{=\varphi_k^i(t)} \right) \left(\left(\mu_k(t) + \alpha_k \bar{\pi}_k(t) \right) dt + \sum_{\ell=1}^d \sigma_{k\ell}(t) dW_\ell(t) \right) \\ &= : \widetilde{x}_0^i + \sum_{k=1}^d \int_0^T \varphi_k^i(t) \left(\left(\mu_k(t) + \frac{\alpha_k}{n} \varphi_k^i(t) + \alpha_k \frac{n+\theta_i}{n} \bar{\pi}_k^{-i}(t) \right) dt + \sum_{\ell=1}^d \sigma_{k\ell}(t) dW_\ell(t) \right) \\ &= : \widetilde{x}_0^i + \sum_{k=1}^d \int_0^T \varphi_k^i(t) \left(\left(\left(\widetilde{\mu}_k^{-i}(t) + \frac{\alpha_k}{n} \varphi_k^i(t) \right) dt + \sum_{\ell=1}^d \sigma_{k\ell}(t) dW_\ell(t) \right), \end{split}$$

where we introduced the following notation

$$\bar{\pi}_k^{-i}(t) \coloneqq \frac{1}{n} \sum_{j \neq i} \pi_k^j(t), \quad \tilde{\mu}_k^{-i}(t) \coloneqq \mu_k(t) + \alpha_k \frac{n + \theta_i}{n} \bar{\pi}_k^{-i}(t).$$

Hence, in order to solve the best response problem associated to (7.9), we can solve the single investor portfolio optimization problem

$$\begin{cases} \sup_{\varphi^i \in \mathcal{A}} \mathbb{E} \Big[U_i \Big(Y_T^{i,\varphi^i} \Big) \Big], \\ \text{s.t.} \quad Y_T^{i,\varphi^i} = \widetilde{x}_0^i + \int_0^T \varphi^i(t)^\top \Big(\big(\widetilde{\mu}^{-i}(t) + \frac{1}{n} A \varphi^i(t) \big) \mathrm{d}t + \sigma(t) \mathrm{d}W(t) \Big), \end{cases}$$
(7.11)

in a different price impact market. Now assume that $\varphi^{i,*} = \varphi^{i,*}(\tilde{\mu}^{-i})$ is the unique optimal solution to (7.11) depending on the drift process $\tilde{\mu}^{-i}$. Then the optimal solution to the best response problem with respect to (7.9) is uniquely determined by

$$\pi_k^i = \varphi_k^{i,*}(\tilde{\mu}^{-i}) + \frac{\theta_i}{n} \sum_{j \neq i} \pi_k^j, \, k = 1, \dots, d, \, i = 1, \dots, n.$$
(7.12)

Note that we can find a unique Nash equilibrium if, and only if, problem (7.11) and the system of equations (7.12) are uniquely solvable.

7.3. Optimization under exponential utility

Given the situation from Sections 7.1 and 7.2, suppose that the n investors use exponential utility functions of the form

$$U_i: \mathbb{R} \to \mathbb{R}, x \mapsto -\exp\left(-\delta_i^{-1}x\right)$$

for parameters $\delta_i > 0$, i = 1, ..., n. Moreover, let the market parameters σ and μ be constant.

In order to solve the auxiliary problem (7.11), we need to restrict ourselves to constant Nash equilibria. Hence, for the best response problem of agent i, we assume that the strategies of the other agents $j \neq i$ are constant while we still need to allow agent i to consider any admissible strategy in \mathcal{A} (see Remark 2.14). If the optimal strategy of agent i turns out to be constant as well, we found a constant Nash equilibrium.

Lacker and Zariphopoulou (2019) give a justification of the restriction to constant strategies. They argue that the assumption is natural in the case of lognormal stock prices in combination with CARA or CRRA utilities. Indeed, Merton (1969) considered lognormal stock prices and obtained constant expressions for the optimally invested amount and fraction in the cases of CARA and CRRA utility, respectively. Thus, constant strategies are somehow expected and the restriction to such strategies is not too severe.

7.3.1. UNIQUE CONSTANT NASH EQUILIBRIUM

Let us now search for constant Nash equilibria for (7.9). Each agent *i* aims to solve an optimization problem of the form

$$\begin{cases} \sup_{\pi^{i} \in \mathcal{A}} \mathbb{E} \left[-\exp\left(-\frac{1}{\delta_{i}} \left(X_{T}^{i,\pi^{i}} - \frac{\theta_{i}}{n} \sum_{j \neq i} X_{T}^{j,\pi^{j}}\right) \right) \right], \\ \text{s.t.} \quad X_{T}^{i,\pi^{i}} = x_{0}^{i} + \int_{0}^{T} \pi^{i}(t)^{\top} \left((\mu + A\bar{\pi}_{t}) \, \mathrm{d}t + \sigma \mathrm{d}W(t) \right), \end{cases}$$
(7.13)

where $\delta_i > 0$ and $\theta_i \in [0, 1]$ denote the risk tolerance parameter and competition weight of agent $i \in \{1, \ldots, n\}$, respectively. Then the following theorem displays the unique constant Nash equilibrium for (7.13).

Theorem 7.2. Assume that the following assumptions hold

a) $\sigma \sigma^{\top} - \frac{2\delta_i}{n} A$ is positive-definite for all $i \in \{1, ..., n\}$, b) $(1 - \hat{\theta})I_d - \sum_{j=1}^n \frac{\delta_j}{n+\theta_i} \left(\sigma \sigma^{\top} - \frac{\delta_j}{n} A\right)^{-1} A$ is regular,

where $\hat{\theta} = \sum_{i=1}^{n} \frac{\theta_i}{n+\theta_i} < 1$ (see Lemma 3.3). Then the unique constant Nash equilibrium to (7.13) is given by

$$\pi^{i,*} = \frac{n\delta_i}{n+\theta_i} \left(\sigma\sigma^{\top} - \frac{\delta_i}{n}A\right)^{-1} \mu + \left(\frac{\delta_i}{n+\theta_i} \left(\sigma\sigma^{\top} - \frac{\delta_i}{n}A\right)^{-1}A + \frac{\theta_i}{n+\theta_i}I_d\right)$$
(7.14)

$$\cdot \left((1-\hat{\theta})I_d - \sum_{j=1}^n \frac{\delta_j}{n+\theta_j} \left(\sigma\sigma^{\top} - \frac{\delta_j}{n}A\right)^{-1}A\right)^{-1} \sum_{j=1}^n \frac{n\delta_j}{n+\theta_j} \left(\sigma\sigma^{\top} - \frac{\delta_j}{n}A\right)^{-1} \mu$$

for $i \in \{1, ..., n\}$.

Remark 7.3. The first condition of the theorem implies that the matrix $\sigma\sigma^{\top} - \frac{\delta_i}{n}A$ is also positivedefinite since

$$\frac{n}{\delta_i}\sigma\sigma^\top - A >_L \frac{n}{2\delta_i}\sigma\sigma^\top - A >_L 0,$$

where $>_L$ describes the Loewner order for Hermitian matrices (see, for example, Horn and Johnson, 2013, Section 7.7). The first inequality is valid since σ was assumed to be regular and hence, $\sigma\sigma^{\top}$ is positive-definite. The second inequality is equivalent to condition a) from Theorem 7.2.

The following example provides a class of matrices that satisfy the conditions of Theorem 7.2. Roughly speaking, the conditions of Theorem 7.2 state that the price impact parameters α_k cannot be too large. Similar requirements were, for example, used by Kraft and Kühn (2011).

Example 7.4. In the previous theorem, assume that $A = \alpha \cdot I_d$ for some constant $\alpha \in \mathbb{R}$. Let $\lambda_1, \ldots, \lambda_d$ be the eigenvalues of $\sigma\sigma^{\top}$ which are strictly positive since $\sigma\sigma^{\top}$ is positive-definite. Then the eigenvalues of the matrix in condition a) of Theorem 7.2 are given by $\lambda_{\ell} - \frac{2\delta_i}{n}\alpha$, $\ell = 1, \ldots, d$ (Bernstein, 2009, Proposition 4.4.5). Hence, condition a) is satisfied if, and only if,

$$\alpha < \frac{n\lambda_{\min}}{2\delta_{\max}},\tag{7.15}$$

where $\lambda_{\min} \coloneqq \min{\{\lambda_1, \ldots, \lambda_d\}}$ and $\delta_{\max} \coloneqq \max{\{\delta_1, \ldots, \delta_n\}}$. If *n* is sufficiently large, (7.15) is satisfied and the matrix in a) is positive-definite.

Now we need to consider condition b) of Theorem 7.2. If α is chosen with respect to (7.15), the matrix $n = \delta_{1} + (1 - \delta_{2})^{-1}$

$$\sum_{j=1}^{n} \frac{\delta_j}{n+\theta_j} \left(\sigma \sigma^{\top} - \frac{\delta_j}{n} A\right)^{-1}$$

is positive-definite (see Remark 7.3). Hence, its eigenvalues μ_1, \ldots, μ_d are strictly positive. Moreover, using $A = \alpha \cdot I_d$ and Proposition 4.4.5 in Bernstein (2009), the eigenvalues of the matrix in b) are given by

$$-\alpha\mu_{\ell}+1-\hat{\theta},\,\ell=1,\ldots,d.$$

We can now ensure that the conditions of the theorem are satisfied if we choose α sufficiently small (with respect to (7.15)) and so that $\alpha \neq \frac{1-\hat{\theta}}{\mu_{\ell}}$ for all $\ell = 1, \ldots, d$.

Now we proceed with the proof of Theorem 7.2 using the auxiliary problem (7.11).

Proof (Theorem 7.2). We solve the auxiliary problem (7.11) for exponential utility to find a constant Nash equilibrium for the multi-objective optimization problem (7.13). Therefore, let $i \in \{1, \ldots, n\}$ be arbitrary but fixed and assume that the strategies π^j of the investors $j \neq i$ are fixed and constant. Then we determine the optimal strategy for investor i, which turns out to be constant as well. Afterwards, we can solve a system of linear equations to determine a Nash equilibrium for the original problem.

Since the strategies π^j , $j \neq i$, are constant, the drift $\tilde{\mu}^{-i}$ is also constant. The process Y^{i,φ^i} from (7.10) is therefore given by

$$Y_t^{i,\varphi^i} = \widetilde{x}_0^i + \int_0^t \varphi^i(s)^\top \left(\left(\widetilde{\mu}^{-i} + \frac{1}{n} A \varphi^i(s) \right) \mathrm{d}s + \sigma \mathrm{d}W(s) \right), \ t \in [0,T].$$
(7.16)

In order to derive an HJB equation, we introduce the value functions

$$J(t,y;\varphi^i) \coloneqq \mathbb{E}^{t,y}\left[-\exp\left(-\frac{1}{\delta_i}Y_T^{i,\varphi^i}\right)\right], \quad J(t,y) \coloneqq \sup_{\varphi^i \in \mathcal{A}} J(t,y;\varphi^i),$$

 $t \in [0, T], y \in \mathbb{R}$. By $\mathbb{E}^{t,y}$ we denote the conditional expectation given that $Y_t^{i,\varphi^i} = y$. Since the parameters of the adapted price impact market are constant, we can use the *Bellman principle* (see, for example, Equation (3.20) in Pham, 2009)

$$J(t,y) = \sup_{\varphi^i \in \mathcal{A}} \mathbb{E}^{t,y} \left[J\left(t', Y_{t'}^{i,\varphi^i}\right) \right],$$
(7.17)

that holds for each $t \in [0, T]$ and $t' \in [t, T]$.

Now let $t \leq t' \leq T$ and let J_t , J_y , J_{yy} denote the first and second order partial derivatives of J in terms of t and y. We assume here that $J \in C^{1,2}([0,T] \times \mathbb{R})$ for the heuristic derivation of the HJB equation. Using the Itô-Doeblin formula (Theorem 2.1) on the interval [t, t'] in combination with Lemma 2.2 implies

$$J(t', Y_{t'}^{i,\varphi^{i}})$$

$$= J(t, Y_{t}^{i,\varphi^{i}}) + \int_{t}^{t'} J_{t}(s, Y_{s}^{i,\varphi^{i}}) ds + \int_{t}^{t'} J_{y}(s, Y_{s}^{i,\varphi^{i}}) dY_{s}^{i,\varphi^{i}} + \frac{1}{2} \int_{t}^{t'} J_{yy}(s, Y_{s}^{i,\varphi^{i}}) d\langle Y^{i,\varphi^{i}} \rangle_{s}$$

$$= J(t, Y_{t}^{i,\varphi^{i}}) + \int_{t}^{t'} J_{t}(s, Y_{s}^{i,\varphi^{i}}) ds + \int_{t}^{t'} J_{y}(s, Y_{s}^{i,\varphi^{i}}) \varphi^{i}(s)^{\top} \left(\widetilde{\mu}^{-i} + \frac{1}{n} A \varphi^{i}(s) \right) ds$$

$$+ \int_{t}^{t'} J_{y}(s, Y_{s}^{i,\varphi^{i}}) \varphi^{i}(s)^{\top} \sigma dW(s) + \frac{1}{2} \int_{t}^{t'} J_{yy}(s, Y_{s}^{i,\varphi^{i}}) \varphi^{i}(s)^{\top} \sigma \sigma^{\top} \varphi^{i}(s) ds.$$
(7.18)

Now we can apply the Bellman principle (7.17), divide by t' - t, and take the limit $t' \to t$ to obtain the following HJB equation

$$0 = G_t + \sup_{\varphi^i \in \mathbb{R}^d} \left\{ G_y \cdot (\varphi^i)^\top \left(\widetilde{\mu}^{-i} + \frac{1}{n} A \varphi^i \right) + \frac{1}{2} G_{yy} \cdot (\varphi^i)^\top \sigma \sigma^\top \varphi^i \right\},\tag{7.19}$$

with terminal condition $G(T, y) = -\exp\left(-\frac{1}{\delta_i}y\right)$, $y \in \mathbb{R}$. We omitted the arguments of the partial derivatives of G to simplify notation. In the previous step, we assumed that the stochastic integral in (7.18) is a martingale and hence, the conditional expectation vanishes. Further, we assumed that the limit $t' \to t$ and the conditional expectation can be interchanged. These assumptions are no limitation since we only use them in the heuristic derivation of the HJB equation. Lemma B.1 in the Appendix verifies that a solution to the HJB equation does, in fact, provide a unique optimal portfolio strategy for (7.11). In the proof of Lemma B.1, it is also shown that the stochastic integral in (7.18) is a martingale.

Let $h(\varphi^i)$ describe the argument inside the supremum, i.e.,

$$h(\varphi^{i}) = G_{y} \cdot (\varphi^{i})^{\top} \widetilde{\mu}^{-i} + (\varphi^{i})^{\top} \left(\frac{1}{n} G_{y} A + \frac{1}{2} G_{yy} \sigma \sigma^{\top}\right) \varphi^{i}.$$

To find the maximizer of h, we determine the gradient. Thus,

$$\nabla h(\varphi^i) = G_y \tilde{\mu}^{-i} + 2\left(\frac{1}{n}G_y A + \frac{1}{2}G_{yy}\sigma\sigma^{\top}\right)\varphi^i \stackrel{!}{=} 0.$$

Solving the previous equation for φ^i would yield a possible maximizer of h. However, since we cannot ensure that the matrix $\frac{1}{n}G_yA + \frac{1}{2}G_{yy}\sigma\sigma^{\top}$ is regular, we first insert an ansatz for G and then solve for φ^i using the assumptions of the theorem. Let

$$G(t,y) = f(t)\tilde{U}(y) \coloneqq f(t) \cdot \left(-\exp\left(-\frac{1}{\delta_i}y\right)\right), \ t \in [0,T], \ y \in \mathbb{R},$$

for some continuously differentiable function $f : [0,T] \to (0,\infty)$ with f(T) = 1. Then G is sufficiently differentiable, i.e., $G \in \mathcal{C}^{1,2}([0,T],\mathbb{R})$, and the partial derivatives of G with respect to t and y are given by

$$G_t(t,y) = f'(t)\widetilde{U}(y), \quad G_y(t,y) = -\frac{1}{\delta_i} \cdot f(t)\widetilde{U}(y), \quad G_{yy}(t,y) = \frac{1}{\delta_i^2} \cdot f(t)\widetilde{U}(y).$$

Thus, ∇h simplifies to

$$\nabla h(\varphi^{i}) = \frac{f(t)\widetilde{U}(y)}{\delta_{i}^{2}} \left(-\delta_{i}\widetilde{\mu}^{-i} + \left(\sigma\sigma^{\top} - \frac{2\delta_{i}}{n}A\right)\varphi^{i}\right) \stackrel{!}{=} 0,$$

which we can equivalently rewrite as

$$\left(\sigma\sigma^{\top} - \frac{2\delta_i}{n}A\right)\varphi^{i,*} \coloneqq \left(\sigma\sigma^{\top} - \frac{2\delta_i}{n}A\right)\varphi^i = \delta_i\tilde{\mu}^{-i},\tag{7.20}$$

since $f(t)\widetilde{U}(y) \neq 0$ for all $(t,y) \in [0,T] \times \mathbb{R}$. Moreover, the Hessian matrix of h is given by

$$H_h(\varphi^i) = \frac{f(t)\tilde{U}(y)}{\delta_i^2} \left(\sigma\sigma^{\top} - \frac{2\delta_i}{n}A\right),\,$$

which is negative-definite since f > 0, $\tilde{U} < 0$, and $\sigma \sigma^{\top} - \frac{2\delta_i}{n}A$ is positive-definite by assumption. Hence, $\varphi^{i,*}$ from (7.20) is the unique maximizer of h. Further, we can solve (7.20) for φ^i since $\sigma \sigma^{\top} - \frac{2\delta_i}{n}A$ is regular by assumption.

It remains to show that the ansatz for G solves (7.19). Inserting the ansatz into the definition of h implies (where we omitted the arguments of f and \tilde{U} to simplify notation)

$$h(\varphi) = \frac{f\widetilde{U}}{\delta_i^2} \left(\frac{1}{2} \varphi^\top \left(\sigma \sigma^\top - \frac{2\delta_i}{n} A \right) \varphi - \delta_i \varphi^\top \widetilde{\mu}^{-i} \right).$$

Therefore, $h(\varphi^{i,*})$ is given by

$$h(\varphi^{i,*}) = -\frac{1}{2} f \widetilde{U} \cdot \left(\widetilde{\mu}^{-i}\right)^{\top} \left(\sigma \sigma^{\top} - \frac{2\delta_i}{n} A\right)^{-1} \widetilde{\mu}^{-i},$$

where we used that $\sigma\sigma^{\top} - \frac{2\delta_i}{n}A$ is symmetric. Now we can finally insert the ansatz for G into the

HJB equation (7.19), which yields

$$0 = f'(t)\widetilde{U}(y) - \frac{1}{2}f(t)\widetilde{U}(y)(\widetilde{\mu}^{-i})^{\top} \left(\sigma\sigma^{\top} - \frac{2\delta_i}{n}A\right)^{-1}\widetilde{\mu}^{-i}$$

=: $\widetilde{U}(y) \left(f'(t) - \rho f(t)\right)$ (7.21)

with terminal condition f(T) = 1, where $\rho = \frac{1}{2} \left(\tilde{\mu}^{-i} \right)^{\top} \left(\sigma \sigma^{\top} - \frac{2\delta_i}{n} A \right)^{-1} \tilde{\mu}^{-i}$. Since $\tilde{U}(y) \neq 0$ for all $y \in \mathbb{R}$, f needs to solve

$$f'(t) = \rho f(t), t \in [0, T], \quad f(T) = 1.$$

The unique solution to this ordinary differential equation is given by

$$f(t) = e^{-\rho(T-t)}, t \in [0, T]$$

Hence, the function $G: [0,T] \times \mathbb{R} \to \mathbb{R}$ with $G(t,y) = -\exp\left(-\frac{1}{\delta_i}y - \rho(T-t)\right)$ solves (7.19) and

$$\varphi^{i,*} = \delta_i \left(\sigma \sigma^\top - \frac{2\delta_i}{n} A \right)^{-1} \widetilde{\mu}^{-i}$$

is a candidate for the optimal solution to the auxiliary problem. Lemma B.1 in the Appendix verifies that $\varphi^{i,*}$ is, in fact, an optimal solution to the auxiliary problem. Moreover, Lemma B.1 also states that the above solution G and the value function J are equal, which also implies that the solution $\varphi^{i,*}$ is unique (up to modifications).

Now we can reinsert the definition of $\tilde{\mu}^{-i}$ and $\varphi^{i,*}$ to determine the Nash equilibrium. Recall that

$$\widetilde{\mu}^{-i} = \mu + \frac{n+\theta_i}{n^2} A \sum_{j \neq i} \pi^j, \ \varphi^i = \pi^i - \frac{\theta_i}{n} \sum_{j \neq i} \pi^j.$$

Therefore, we obtain the system of linear equations

$$\left(\sigma\sigma^{\top} - \frac{2\delta_i}{n}A\right) \cdot \left(\pi^i - \frac{\theta_i}{n}\sum_{j\neq i}\pi^j\right) = \delta_i\widetilde{\mu}^{-i} = \delta_i\mu + \frac{n+\theta_i}{n} \cdot \frac{\delta_i}{n}A\sum_{j\neq i}\pi^j,$$

which can be equivalently rewritten as

$$\left(\sigma\sigma^{\top} - \frac{2\delta_i}{n}A\right)\pi^i = \delta_i\mu + \left(\frac{\delta_i}{n}A + \frac{\theta_i}{n}\left(\sigma\sigma^{\top} - \frac{\delta_i}{n}A\right)\right)\sum_{j\neq i}\pi^j.$$
(7.22)

In order to solve (7.22), we add $\left(\frac{\delta_i}{n}A + \frac{\theta_i}{n}\left(\sigma\sigma^{\top} - \frac{\delta_i}{n}A\right)\right)\pi^i$ on both sides to obtain

$$\frac{n+\theta_i}{n} \left(\sigma \sigma^\top - \frac{\delta_i}{n} A \right) \pi^i = \delta_i \mu + \left(\frac{\delta_i}{n} A + \frac{\theta_i}{n} \left(\sigma \sigma^\top - \frac{\delta_i}{n} A \right) \right) \sum_{j=1}^n \pi^j.$$

Since the matrix $\sigma \sigma^{\top} - \frac{\delta_i}{n} A$ is regular by assumption and the arguments in Remark 7.3, it follows

that

$$\pi^{i} = \frac{n\delta_{i}}{n+\theta_{i}} \left(\sigma\sigma^{\top} - \frac{\delta_{i}}{n}A\right)^{-1} \mu + \frac{n}{n+\theta_{i}} \left(\sigma\sigma^{\top} - \frac{\delta_{i}}{n}A\right)^{-1} \left(\frac{\delta_{i}}{n}A + \frac{\theta_{i}}{n} \left(\sigma\sigma^{\top} - \frac{\delta_{i}}{n}A\right)\right) \sum_{j=1}^{n} \pi^{j}$$
$$= \frac{n\delta_{i}}{n+\theta_{i}} \left(\sigma\sigma^{\top} - \frac{\delta_{i}}{n}A\right)^{-1} \mu + \left(\frac{\delta_{i}}{n+\theta_{i}} \left(\sigma\sigma^{\top} - \frac{\delta_{i}}{n}A\right)^{-1}A + \frac{\theta_{i}}{n+\theta_{i}}I_{d}\right) \sum_{j=1}^{n} \pi^{j}.$$
(7.23)

Taking the sum over all $i \in \{1, ..., n\}$ on both sides yields

$$\sum_{j=1}^{n} \pi^{j} = \sum_{j=1}^{n} \frac{n\delta_{j}}{n+\theta_{j}} \left(\sigma\sigma^{\top} - \frac{\delta_{j}}{n}A\right)^{-1} \mu + \left(\sum_{j=1}^{n} \frac{\delta_{j}}{n+\theta_{j}} \left(\sigma\sigma^{\top} - \frac{\delta_{j}}{n}A\right)^{-1}A + \hat{\theta} \cdot I_{d}\right) \sum_{j=1}^{n} \pi^{j}.$$

This can equivalently be rewritten as

$$\left((1-\hat{\theta})I_d - \sum_{j=1}^n \frac{\delta_j}{n+\theta_j} \left(\sigma\sigma^\top - \frac{\delta_j}{n}A\right)^{-1}A\right) \sum_{j=1}^n \pi^j = \sum_{j=1}^n \frac{n\delta_j}{n+\theta_j} \left(\sigma\sigma^\top - \frac{\delta_j}{n}A\right)^{-1}\mu.$$

The matrix on the left-hand side is regular by assumption, so we can further deduce

$$\sum_{j=1}^{n} \pi^{j} = \left((1-\hat{\theta})I_{d} - \sum_{j=1}^{n} \frac{\delta_{j}}{n+\theta_{j}} \left(\sigma \sigma^{\top} - \frac{\delta_{j}}{n} A \right)^{-1} A \right)^{-1} \sum_{j=1}^{n} \frac{n\delta_{j}}{n+\theta_{j}} \left(\sigma \sigma^{\top} - \frac{\delta_{j}}{n} A \right)^{-1} \mu.$$

Finally, we insert $\sum_{j=1}^{n} \pi^{j}$ into (7.23) to obtain the asserted constant Nash equilibrium given by

$$\pi^{i,*} = \frac{n\delta_i}{n+\theta_i} \left(\sigma\sigma^{\top} - \frac{\delta_i}{n}A\right)^{-1} \mu + \left(\frac{\delta_i}{n+\theta_i} \left(\sigma\sigma^{\top} - \frac{\delta_i}{n}A\right)^{-1}A + \frac{\theta_i}{n+\theta_i}I_d\right) \\ \cdot \left((1-\hat{\theta})I_d - \sum_{j=1}^n \frac{\delta_j}{n+\theta_j} \left(\sigma\sigma^{\top} - \frac{\delta_j}{n}A\right)^{-1}A\right)^{-1} \sum_{j=1}^n \frac{n\delta_j}{n+\theta_j} \left(\sigma\sigma^{\top} - \frac{\delta_j}{n}A\right)^{-1}\mu. \quad \Box$$

- Remark 7.5. a) In the previous proof, we used condition a) of Theorem 7.2 to ensure that the local extremum of the function h is, in fact, a maximum. Roughly speaking, condition a) prevents the price impact of the agents from becoming too large. Otherwise, they might be able to exploit their price impact by investing an infinite amount into the stock and gaining an infinite amount of utility. A similar assumption has also been used by Kraft and Kühn (2011). They explain that without the assumption that the price impact parameter is sufficiently small, "the demand for stocks is infinite".
- b) If assumption b) in Theorem 7.2 is not satisfied, there is no constant Nash equilibrium (in most cases) since, in general, the system of linear equations (7.22) is not solvable without condition b).

Remark 7.6. Inserting A = 0 into the Nash equilibrium (7.14) yields

$$\pi^{i,*} = \left(\frac{n\delta_i}{n+\theta_i} + \frac{\theta_i}{(n+\theta_i)(1-\hat{\theta})}\sum_{j=1}^n \frac{n\delta_j}{n+\theta_j}\right) \left(\sigma\sigma^{\top}\right)^{-1}\mu,$$

which is exactly the solution to the problem without price impact (see Subsection 4.2.1). This does not come as a surprise since the matrix A contains the coefficients α_k , which model the impact of the arithmetic mean of the n agents' investment on the price process of the k-th stock, $k = 1, \ldots, d$. Moreover, if we let A be arbitrary and set $\theta_i = 0$, we can deduce the optimal solution to the optimization problem without relative concerns. In this case, interaction is caused solely by the cumulative price impact. Then

$$\pi^{i,*} = \delta_i \left(\sigma \sigma^\top - \frac{\delta_i}{n} A \right)^{-1} \mu + \frac{\delta_i}{n} \left(\sigma \sigma^\top - \frac{\delta_i}{n} A \right)^{-1} A$$
$$\cdot \left(I_d - \sum_{j=1}^n \frac{\delta_j}{n} \left(\sigma \sigma^\top - \frac{\delta_j}{n} A \right)^{-1} A \right)^{-1} \sum_{j=1}^n \delta_j \left(\sigma \sigma^\top - \frac{\delta_j}{n} A \right)^{-1} \mu$$

describes the unique Nash equilibrium.

We conclude this subsection with a version of Theorem 7.2 for a market with only one stock.

Corollary 7.7. Let d = 1, $A = \alpha$, and assume that

$$n\sigma^2 - 2\alpha\delta_j > 0, \ j = 1, \dots, n, \quad \sum_{j=1}^n \frac{n\alpha\delta_j}{(n+\theta_j)(n\sigma^2 - \delta_j\alpha)} \neq 1 - \hat{\theta}.$$

Then the unique constant Nash equilibrium for (7.13) is given by

$$\pi^{i,*} = \frac{n^2 \delta_i \mu}{\left(n + \theta_i\right) \left(n\sigma^2 - \delta_i \alpha\right)} + \left(\frac{\theta_i}{n + \theta_i} + \frac{n\alpha\delta_i}{\left(n + \theta_i\right) \left(n\sigma^2 - \delta_i \alpha\right)}\right) \cdot \frac{\sum_{j=1}^n \frac{n^2 \delta_j}{\left(n + \theta_j\right) \left(n\sigma^2 - \delta_j \alpha\right)} \cdot \mu}{1 - \hat{\theta} - \sum_{j=1}^n \frac{n\alpha\delta_j}{\left(n + \theta_j\right) \left(n\sigma^2 - \delta_j \alpha\right)}}$$

for $i \in \{1, ..., n\}$.

7.3.2. INFLUENCE OF THE PRICE IMPACT PARAMETER

The goal of this subsection is to study the influence of the price impact parameter α on the components of the constant Nash equilibrium and on the stock price. To do this, assume that there is only one stock (d = 1) with constant and deterministic parameters $\mu_1 \equiv \mu > 0$, $\sigma_{11} \equiv \sigma > 0$, and $\alpha_1 = \alpha$, driven by a one-dimensional Brownian motion.

The first step is to take a closer look at the assumptions of Corollary 7.7. The first assumption can be written as

$$\alpha < \alpha_{\max} \coloneqq \frac{n\sigma^2}{2\delta_{\max}},$$

where $\delta_{\max} \coloneqq \max{\{\delta_1, \ldots, \delta_n\}}$. For the second assumption, we define $s : (-\infty, \alpha_{\max}] \to \mathbb{R}$ by

$$s(\alpha) \coloneqq \sum_{j=1}^{n} \frac{n\alpha\delta_j}{(n+\theta_j)(n\sigma^2 - \alpha\delta_j)}.$$
(7.24)

Then we can deduce that there exists a unique value α_0 such that $s(\alpha_0) = 1 - \hat{\theta}$. Moreover, we have $\alpha_0 \in (0, \alpha_{\max})$. To verify this observation, we introduce functions $f_j : (-\infty, \alpha_{\max}] \to \mathbb{R}$,

 \Diamond

 $j = 1, \ldots, n$, given by

$$f_j(\alpha) \coloneqq \frac{n\delta_j}{(n+\theta_j)(n\sigma^2 - \alpha\delta_j)}$$

Note that, for any $j \in \{1, ..., n\}$, f_j is strictly positive and continuously differentiable in α for $\alpha \in (-\infty, \alpha_{\max}]$. Now we can display the components of the Nash equilibrium from Corollary 7.7 in terms of the functions $f_1, ..., f_n$ as

$$\pi^{i,*} = \left(nf_i(\alpha) + \left(\frac{\theta_i}{n + \theta_i} + \alpha f_i(\alpha) \right) \cdot \frac{\sum_{j=1}^n nf_j(\alpha)}{1 - \hat{\theta} - s(\alpha)} \right) \cdot \mu,$$
(7.25)

and the function s from (7.24) can be written as

$$s(\alpha) = \alpha \sum_{j=1}^{n} f_j(\alpha), \quad \alpha \in (-\infty, \alpha_{\max}].$$

The functions f_j are strictly increasing in terms of α since

$$\frac{\partial}{\partial \alpha} f_j(\alpha) = f'_j(\alpha) = \frac{n \delta_j^2}{(n+\theta_j)(n\sigma^2 - \alpha\delta_j)^2} > 0.$$

Moreover, the functions $\alpha \mapsto \alpha f_j(\alpha), j = 1, \dots, n$, are strictly increasing in α since

$$\frac{\partial}{\partial \alpha} (\alpha f_j(\alpha)) = \frac{n\delta_j}{(n+\theta_j)(n\sigma^2 - \alpha\delta_j)} + \alpha \cdot \frac{n\delta_j^2}{(n+\theta_j)(n\sigma^2 - \alpha\delta_j)^2} \\ = \frac{n^2\sigma^2\delta_j}{(n+\theta_j)(n\sigma^2 - \alpha\delta_j)^2} > 0$$

holds for any $\alpha \leq \alpha_{\max}$. Hence, the function s is strictly increasing in α as well. Moreover, s(0) = 0 and

$$s(\alpha_{\max}) + \hat{\theta} = \sum_{j=1}^{n} \left(\frac{n\delta_j}{(n+\theta_j)(2\delta_{\max} - \delta_j)} + \frac{\theta_j}{n+\theta_j} \right)$$
$$= \sum_{j=1}^{n} \frac{n\delta_j + \theta_j(2\delta_{\max} - \delta_j)}{(n+\theta_j)(2\delta_{\max} - \delta_j)}$$
$$> \frac{n\delta_{\max} + \theta_k(2\delta_{\max} - \delta_{\max})}{(n+\theta_k)(2\delta_{\max} - \delta_{\max})} = 1,$$
(7.26)

where $k \in \{1, \ldots, n\}$ is chosen so that $\delta_k = \delta_{\max}$. Hence, $s(\alpha_{\max}) > 1 - \hat{\theta}$. Finally, since s is continuous on $(-\infty, \alpha_{\max}]$, the intermediate value theorem implies that there exists a unique point $\alpha_0 \in (0, \alpha_{\max})$ such that $s(\alpha_0) = 1 - \hat{\theta}$.

To summarize, $\alpha \in \mathbb{R}$ is an admissible price impact parameter if, and only if, $\alpha < \alpha_{\max}$ and $\alpha \neq \alpha_0$, where admissible means that α satisfies the conditions of Corollary 7.7. This can now be used to analyze the influence of the price impact parameter.

INFLUENCE ON THE NASH EQUILIBRIUM

First, we consider the influence of the price impact parameter α on the components of the Nash equilibrium from Corollary 7.7. The representation of $\pi^{i,*}$ in (7.25) implies

$$(1 - \hat{\theta} - s(\alpha))\frac{\pi^{i,*}}{\mu} = nf_i(\alpha)(1 - \hat{\theta} - s(\alpha)) + \left(\frac{\theta_i}{n + \theta_i} + \alpha f_i(\alpha)\right)\sum_{j=1}^n nf_j(\alpha)$$
$$= nf_i(\alpha)(1 - \hat{\theta} - s(\alpha)) + \frac{n\theta_i}{n + \theta_i}\sum_{j=1}^n f_j(\alpha) + nf_i(\alpha)s(\alpha)$$
$$= n(1 - \hat{\theta})f_i(\alpha) + \frac{n\theta_i}{n + \theta_i}\sum_{j=1}^n f_j(\alpha) > 0,$$

since $\hat{\theta} \in [0, 1)$ (see Lemma 3.3) and $f_j(\alpha) > 0$ for all $\alpha < \alpha_{\max}$ and all $j \in \{1, \ldots, n\}$. Hence, since $\mu > 0$, the sign of $\pi^{i,*}$ equals the sign of $1 - \hat{\theta} - s(\alpha)$. By definition of $\alpha_0, 1 - \hat{\theta} - s(\alpha) > 0$ holds if, and only if, $\alpha < \alpha_0$. Thus, $\pi^{i,*} > 0$ if, and only if, $\alpha < \alpha_0$. The behavior of $\pi^{i,*}$ at the discontinuity α_0 and the left boundary $(\alpha \to -\infty)$ can be further specified. If α becomes increasingly small, $\pi^{i,*}$ tends to zero. To verify this, we consider the asymptotic behavior of the functions $f_j, j = 1, \ldots, n$, and s. It follows

$$\lim_{\alpha \to -\infty} f_j(\alpha) = \lim_{\alpha \to -\infty} \frac{n\delta_j}{(n+\theta_j)(n\sigma^2 - \alpha\delta_j)} = 0,$$

$$\lim_{\alpha \to -\infty} \alpha f_j(\alpha) = \lim_{\alpha \to -\infty} \frac{n\delta_j\alpha}{(n+\theta_j)(n\sigma^2 - \alpha\delta_j)} = \frac{n\delta_j}{n+\theta_j} \lim_{\alpha \to -\infty} \left(\frac{n\sigma^2}{\alpha} - \delta_j\right)^{-1} = -\frac{n}{n+\theta_j},$$

$$\lim_{\alpha \to -\infty} s(\alpha) = -\sum_{j=1}^n \frac{n}{n+\theta_j}.$$

(7.27)

Hence,

$$\lim_{\alpha \to -\infty} \pi^{i,*} = \lim_{\alpha \to -\infty} \left(nf_i(\alpha) + \left(\frac{\theta_i}{n + \theta_i} + \alpha f_i(\alpha) \right) \cdot \frac{\sum_{j=1}^n nf_j(\alpha)}{1 - \hat{\theta} - s(\alpha)} \right) \cdot \mu = 0$$

Moreover, since $\lim_{\alpha \searrow \alpha_0} 1 - \hat{\theta} - s(\alpha) = 0 = \lim_{\alpha \nearrow \alpha_0} 1 - \hat{\theta} - s(\alpha)$ and $\pi^{i,*} > 0$ if, and only if, $\alpha < \alpha_0$, it follows

$$\lim_{\alpha \nearrow \alpha_0} \pi^{i,*} = \infty, \quad \lim_{\alpha \searrow \alpha_0} \pi^{i,*} = -\infty.$$

Furthermore, we can consider the monotonicity of $\pi^{i,*}$ in terms of α . The first order derivative of $\pi^{i,*}$ from (7.25) is given by

$$\frac{\partial}{\partial \alpha} \frac{\pi^{i,*}}{\mu} = nf'_i(\alpha) + \left(f_i(\alpha) + \alpha f'_i(\alpha)\right) \cdot \frac{n\sum_{j=1}^n f_j(\alpha)}{1 - \hat{\theta} - \sum_{j=1}^n \alpha f_j(\alpha)} + \left(\frac{\theta_i}{n + \theta_i} + \alpha f_i(\alpha)\right) \\ \cdot \frac{n\sum_{j=1}^n f'_j(\alpha) \left(1 - \hat{\theta} - \sum_{j=1}^n \alpha f_j(\alpha)\right) + n\sum_{j=1}^n f_j(\alpha) \left(\sum_{j=1}^n f_j(\alpha) + \alpha f'_j(\alpha)\right)}{\left(1 - \hat{\theta} - \sum_{j=1}^n \alpha f_j(\alpha)\right)^2},$$

which can be rewritten as

$$\frac{\partial}{\partial \alpha} \frac{\pi^{i,*}}{\mu} = \left(1 - \hat{\theta} - \alpha \sum_{j=1}^{n} f_j(\alpha)\right)^{-2} \left\{ n(1 - \hat{\theta})^2 f'_i(\alpha) - n\alpha(1 - \hat{\theta}) f'_i(\alpha) \sum_{j=1}^{n} f_j(\alpha) \\
+ n(1 - \hat{\theta}) f_i(\alpha) \sum_{j=1}^{n} f_j(\alpha) + n\alpha(1 - \hat{\theta}) f_i(\alpha) \sum_{j=1}^{n} f'_j(\alpha) \\
+ \frac{\theta_i}{n + \theta_i} \left(n(1 - \hat{\theta}) \sum_{j=1}^{n} f'_j(\alpha) + n\left(\sum_{j=1}^{n} f_j(\alpha)\right)^2 \right) \right\}$$

$$= \left(1 - \hat{\theta} - \alpha \sum_{j=1}^{n} f_j(\alpha)\right)^{-2} \left\{ n(1 - \hat{\theta}) f'_i(\alpha) \left(1 - \hat{\theta} - \alpha \sum_{j=1}^{n} f_j(\alpha)\right) \\
+ n(1 - \hat{\theta}) f_i(\alpha) \left(\sum_{j=1}^{n} f_j(\alpha) + \alpha \sum_{j=1}^{n} f'_j(\alpha)\right) \\
+ \frac{\theta_i}{n + \theta_i} \left(n(1 - \hat{\theta}) \sum_{j=1}^{n} f'_j(\alpha) + n\left(\sum_{j=1}^{n} f_j(\alpha)\right)^2 \right) \right\}.$$
(7.29)

If $\alpha < \alpha_0$, all summands in (7.29) are strictly positive. If $\alpha_0 < \alpha < \alpha_{\text{max}}$, it follows

$$f_i(\alpha) - \alpha f'_i(\alpha) = \frac{n\delta_i}{(n+\theta_i)(n\sigma^2 - \alpha\delta_i)} - \alpha \frac{n\delta_i^2}{(n+\theta_i)(n\sigma^2 - \alpha\delta_i)^2}$$
$$= \frac{n\delta_i}{(n+\theta_i)(n\sigma^2 - \alpha\delta_i)} \cdot \frac{n\sigma^2 - 2\alpha\delta_i}{n\sigma^2 - \alpha\delta_i} > 0$$

since $\alpha < \alpha_{\text{max}}$. Hence, we obtain the following for the second and third summand in (7.28)

$$-n\alpha(1-\hat{\theta})f_{i}'(\alpha)\sum_{j=1}^{n}f_{j}(\alpha)+n(1-\hat{\theta})f_{i}(\alpha)\sum_{j=1}^{n}f_{j}(\alpha)=n(1-\hat{\theta})\sum_{j=1}^{n}f_{j}(\alpha)(f_{i}(\alpha)-\alpha f_{i}'(\alpha))>0.$$

Since the other summands in (7.28) are strictly positive as well, the derivative of $\pi^{i,*}$ is strictly positive for $\alpha \in (\alpha_0, \alpha_{\max})$ and thus, for all admissible α . Note that, due to the jump located at $\alpha_0, \pi^{i,*}$ is only piecewise strictly increasing in terms of α on $(-\infty, \alpha_0)$ and $(\alpha_0, \alpha_{\max})$.

The previously derived properties regarding the behavior of $\pi^{i,*}$ in terms of α can also be observed in Figure 7.3.1. The vertical lines (dotted) show the discontinuity α_0 for the different parameter choices. The gray horizontal line (dashed) marks the value 0 while the orange and blue horizontal lines (dashed) display the optimal solution to the classical problem of maximizing expected terminal wealth under exponential utility, without price impact and relative concerns, given by $\delta_1 \mu \sigma^{-2}$ (Merton ratio). There are two ways the agents may try to influence the stock price to their advantage. By buying the stock, they may jointly increase the stock value and thus, raise their utility, or by jointly short-selling the stock and thus, decrease its value. Our analysis shows that, in case of a small price impact, the agents go for the first option and, in case of a larger price impact, they go for the latter option. Of course, this is only true under the exponential utility where short-selling is no problem. Under an increasingly negative price impact, the investors engage less in the financial market.



Figure 7.3.1.: Illustration of $\pi^{1,*}$ from Corollary 7.7 in terms of $\alpha \in (-0.04, \alpha_{\max})$ for $n = 12, \mu = 0.03, \sigma = 0.2, \alpha_{\max} = n\sigma^2/8$. The parameters for agent 1 are $\theta_1 = 0.3, \delta_1 \in \{1, 4\}$, and the parameters θ_j and $\delta_j, j \ge 2$ are increasing from 0 to 1 with step size 0.1, and from 0.5 to 2.7 by step size 0.2, respectively.

INFLUENCE ON THE STOCK PRICE

In the second part of this subsection, we consider the influence of the price impact parameter α on the stock price. Recall that the dynamics of the stock price are given by

$$\mathrm{d}S_t = S_t((\mu + \alpha \bar{\pi})\mathrm{d}t + \sigma \mathrm{d}W_t), \ t \in [0, T], \quad S_0 = 1.$$

If we insert the constant Nash equilibrium from (7.25) and use that μ and σ are constants, Theorem 2.4 implies

$$S_t^*(\alpha) \coloneqq S_t = \exp\left(\left(\mu + \alpha \cdot \frac{\sum_{j=1}^n f_j(\alpha)}{1 - \hat{\theta} - s(\alpha)} \cdot \mu - \frac{\sigma^2}{2}\right) t + \sigma W_t\right), t \in [0, T].$$
(7.30)

Since the functions $\alpha \mapsto \alpha f_j(\alpha)$, $j = 1, \ldots, n$, and s are strictly increasing in terms of α , the stock price S_t^* is piecewise increasing on $(-\infty, \alpha_0)$ and $(\alpha_0, \alpha_{\max})$ in terms of α with a jump located at α_0 . Moreover, the summand $\frac{\alpha \sum_{j=1}^n f_j(\alpha)}{1-\hat{\theta}-s(\alpha)} \cdot \mu$ is strictly positive if, and only if, $\alpha \in (0, \alpha_0)$. Thus, $S_t^*(\alpha) > S_t^*(0)$ if, and only if, $\alpha \in (0, \alpha_0)$. More specifically, we obtain the following ordering for any choice of $\alpha_1 < 0$, $\alpha_2 \in (0, \alpha_0)$, and $\alpha_3 \in (\alpha_0, \alpha_{\max})$

$$S_t^*(\alpha_3) < S_t^*(\alpha_1) < S_t^*(0) < S_t^*(\alpha_2).$$
(7.31)

The second and third inequality follow directly from the previous observations. For the first inequality, we consider the limit of S_t^* as α tends to $-\infty$ and α_{\max} . Obviously, it suffices to consider the expression $\frac{s(\alpha)}{1-\hat{\theta}-s(\alpha)} = \frac{\alpha \sum_{j=1}^{n} f_j(\alpha)}{1-\hat{\theta}-s(\alpha)}$. If α tends to α_{\max} , it follows

$$\lim_{\alpha \to \alpha_{\max}} \frac{s(\alpha)}{1 - \hat{\theta} - s(\alpha)} = \frac{\sum_{j=1}^{n} \frac{n\delta_j}{(n+\theta_j)(2\delta_{\max} - \delta_j)}}{1 - \hat{\theta} - \sum_{j=1}^{n} \frac{n\delta_j}{(n+\theta_j)(2\delta_{\max} - \delta_j)}}.$$
(7.32)

Further, recall that $s(\alpha) \to -\sum_{j=1}^{n} \frac{n}{n+\theta_j}$ as $\alpha \to -\infty$ (see (7.27)). Now consider the function $r: \mathbb{R} \setminus \{1 - \hat{\theta}\} \to \mathbb{R}, x \mapsto \frac{x}{1 - \hat{\theta} - x}$. Then

$$\lim_{x \to \pm \infty} r(x) = -1$$

and r is piecewise increasing on $(-\infty, 1-\hat{\theta})$ and $(1-\hat{\theta}, \infty)$ with a jump from positive to negative values located at $1-\hat{\theta}$. Hence, r(x) > -1 on $(-\infty, 1-\hat{\theta})$ and r(x) < -1 on $(1-\hat{\theta}, \infty)$ and, in particular,

$$r(x) > -1 > r(y) \quad \text{for all} \quad x < 1 - \hat{\theta} < y.$$

$$(7.33)$$

Since $-\sum_{j=1}^{n} \frac{n}{n+\theta_j} < 0 < 1 - \hat{\theta}$ and $\sum_{j=1}^{n} \frac{n\delta_j}{(n+\theta_j)(2\delta_{\max}-\delta_j)} > 1 - \hat{\theta}$ (see (7.26)), (7.27), (7.32) and (7.33) imply

$$\lim_{\alpha \to -\infty} \frac{s(\alpha)}{1 - \hat{\theta} - s(\alpha)} = r \left(-\sum_{j=1}^{n} \frac{n}{n + \theta_j} \right)$$
$$> r \left(\sum_{j=1}^{n} \frac{n \delta_j}{(n + \theta_j)(2\delta_{\max} - \delta_j)} \right) = \lim_{\alpha \to \alpha_{\max}} \frac{s(\alpha)}{1 - \hat{\theta} - s(\alpha)}.$$

Since $\frac{s(\alpha)}{1-\hat{\theta}-s(\alpha)}$ is piecewise increasing in α on $(-\infty, \alpha_0)$ and $(\alpha_0, \alpha_{\max})$, the first inequality in (7.31) follows as well.

The behavior of S_T^* in terms of α can also be observed in Figure 7.3.2. The horizontal dashed lines represent the stock price $S_T^*(0)$ without price impact for three possible realizations of W_T . The vertical dotted lines mark the values $\alpha = 0$ and $\alpha = \alpha_0$. The figure shows the monotonicity of S_T^* in terms of α as well as the ordering discussed in (7.31).

7.3.3. DISCUSSION OF NONLINEAR PRICE IMPACT

At the beginning of Section 7.1, we assumed that the price impact of the n investors is incorporated into the drift of the stock as a linear function in terms of the arithmetic mean of the investors' strategies. While the use of the arithmetic mean seems intuitive and reasonable since we assumed that the single investors are "small", one could ask whether using a different function than a linear one would lead to a different optimization problem and hence, also a different optimal investment. Moreover, Muhle-Karbe et al. (2022) (and references therein) discuss empirical data which suggests that price impact is not linear but concave in order size. Further, Bouchaud (2009) argues that the volume dependence of price impact is sublinear and behaves like a power function with an exponent approximately between 0.1 and 0.3.



Figure 7.3.2.: Illustration of $S_T^*(\alpha)$ (solid) from (7.30) in terms of $\alpha \in (-0.02, \alpha_{\max})$ for three different realizations of W_T , where T = 10, n = 11, $\mu = 0.03$, $\sigma = 0.2$. The parameters θ_j and δ_j , $j \in \{1, \ldots, 11\}$, are increasing from 0 to 1 with step size 0.1, and from 0.5 to 2.7 by step size 0.2, respectively. The horizontal dashed lines mark the stock price for $\alpha = 0$.

In Theorem 7.2 and the subsequent Corollary 7.7, we were able to find an explicit constant Nash equilibrium to the associated multi-objective portfolio optimization problem using exponential utility (if the parameters are chosen accordingly). The proof depends strongly on the linearity of the price impact. Thus, we are not be able to give an explicit solution to the optimization problem if the price impact is included into the drift via a nonlinear function g. However, we can show that using a function g that grows superlinearly results in a problem that does not have a finite optimal solution, whereas a function g that grows sublinearly yields a finite optimal solution. If g is a linear function, it depends on the parameter choices whether or not there exists a finite optimal solution (see Corollary 7.7).

In the case of linear price impact, we allowed the price impact parameter α to take both positive and negative values. However, the considerations in Subsection 7.3.2 showed that the *n* agents' investment into the stock becomes negligible for increasingly negative price impact. Hence, in the nonlinear case, we only consider price impact which is increasing in order size. Therefore, the price impact is now modeled by a continuous and strictly increasing function $g : \mathbb{R} \to \mathbb{R}$ with g(0) = 0. Thus, the stock price process is given as the solution to the stochastic differential equation

$$\mathrm{d}S_t = S_t\big(\left(\mu + g\left(\bar{\pi}_t\right)\right)\mathrm{d}t + \sigma\mathrm{d}W_t\big), \, S_0 = 1.$$

We have to restrict the problem to constant Nash equilibria again. Therefore, from the perspective of investor $i \in \{1, ..., n\}$, we can rewrite the expression $g(\bar{\pi}_t)$ in the previous stochastic differential equation as

$$g(\bar{\pi}_t) = g\left(\frac{1}{n}\sum_{j=1}^n \pi_t^j\right) = g\left(\frac{1}{n}\pi_t^i + \frac{1}{n}\sum_{j\neq i}\pi^j\right) \eqqcolon \widetilde{g}(\pi_t^i),$$

where $\tilde{g}(p) \coloneqq g\left(\frac{p}{n} + \frac{1}{n}\sum_{j\neq i}\pi^{j}\right), p \in \mathbb{R}$. We assume that the strategies $\pi^{j}, j \neq i$, of the other investors are fixed, constant, and deterministic. Moreover, we require $\pi^{i} \in \mathcal{A}$ for the set \mathcal{A} of admissible strategies from (7.8). From the definition of \tilde{g} , it follows directly that \tilde{g} is still continuous and strictly increasing and satisfies $\tilde{g}\left(-\sum_{j\neq i}\pi^{j}\right) = 0$. The wealth process of investor i is given by

$$X_t^{i,\pi^i} = x_0^i + \int_0^t \pi_s^i \left(\left(\mu + \widetilde{g}(\pi_s^i) \right) \mathrm{d}s + \sigma \mathrm{d}W_s \right), \, t \in [0,T].$$

In the following, we prove that the best response problem for

$$\begin{cases} \sup_{\pi^i \in \mathcal{A}} \mathbb{E}\left[-\exp\left(-\frac{1}{\delta_i}\left(X_T^{i,\pi^i} - \frac{\theta_i}{n}\sum_{j \neq i}X_T^{j,\pi^j}\right)\right)\right], \\ \text{s.t.} \quad X_T^{i,\pi^i} = x_0^i + \int_0^T \pi_t^i\left((\mu + g(\bar{\pi}_t))\,\mathrm{d}t + \sigma\mathrm{d}W_t\right), \end{cases} \quad i = 1,\ldots,n, \tag{7.34}$$

has a finite optimal solution if g grows sublinearly, and is not solvable if g grows superlinearly. The following proposition summarizes the first assertion of this subsection which treats the case of sublinear growth.

Proposition 7.8. Let $i \in \{1, ..., n\}$ arbitrary but fixed, and π^j , $j \neq i$, deterministic and constant. If $\lim_{x\to\pm\infty} \frac{g(x)}{x} = 0$, the best response problem for constant Nash equilibria associated to (7.34) has a finite, constant optimal solution. Moreover, each optimal solution is a global maximum point of the function $h : \mathbb{R} \to \mathbb{R}$ given by

$$h(a) = \left(a - \frac{\theta_i}{n} \sum_{j \neq i} \pi^j\right) \left(\mu + \widetilde{g}(a)\right) - \frac{\sigma^2}{2\delta_i} \left(a - \frac{\theta_i}{n} \sum_{j \neq i} \pi^j\right)^2.$$

Remark 7.9. A similar observation was made by Eksi and Ku (2017). They showed that their price impact market does not contain arbitrage strategies if the price impact function is sublinear. However, they had to restrict their set of admissible strategies to bounded strategies. \diamond

Proof. For the moment, we restrict the set of admissible strategies to bounded strategies $(\pi_t^i)_{t \in [0,T]}$, i.e. there exists a constant K > 0 such that $|\pi_t^i| \leq K \mathbb{P}$ -almost surely for all $t \in [0,T]$. For constants π^j , $j \neq i$, we obtain

$$\begin{split} &-\frac{1}{\delta_i} \left(X_T^{i,\pi^i} - \frac{\theta_i}{n} \sum_{j \neq i} X_T^{j,\pi^j} \right) \\ &= -\frac{1}{\delta_i} \left(x_0^i - \frac{\theta_i}{n} \sum_{j \neq i} x_0^j \right) - \frac{1}{\delta_i} \left(\int_0^T \left(\pi_t^i - \frac{\theta_i}{n} \sum_{j \neq i} \pi^j \right) (\mu + \widetilde{g}(\pi_t^i)) \mathrm{d}t + \sigma \int_0^T \left(\pi_t^i - \frac{\theta_i}{n} \sum_{j \neq i} \pi^j \right) \mathrm{d}W_t \right) \\ &- \frac{\sigma^2}{2\delta_i^2} \int_0^T \left(\pi_t^i - \frac{\theta_i}{n} \sum_{j \neq i} \pi^j \right)^2 \mathrm{d}t + \frac{\sigma^2}{2\delta_i^2} \int_0^T \left(\pi_t^i - \frac{\theta_i}{n} \sum_{j \neq i} \pi^j \right)^2 \mathrm{d}t. \end{split}$$

Now define a new probability measure \mathbb{Q} by

$$\frac{\mathrm{d}\mathbb{Q}}{\mathrm{d}\mathbb{P}} = \exp\left(-\frac{\sigma^2}{2\delta_i^2}\int_0^T \left(\pi_t^i - \frac{\theta_i}{n}\sum_{j\neq i}\pi^j\right)^2 \mathrm{d}t - \frac{\sigma}{\delta_i}\int_0^T \left(\pi_t^i - \frac{\theta_i}{n}\sum_{j\neq i}\pi^j\right) \mathrm{d}W_t\right).$$

Note that the expression on the right-hand side is a density since $(\pi_t^i)_{t \in [0,T]}$ is bounded. Thus, we can write the objective function of (7.34) as

$$\begin{split} & \mathbb{E}\bigg[-\exp\bigg(-\frac{1}{\delta_i}\bigg(X_T^{i,\pi^i} - \frac{\theta_i}{n}\sum_{j\neq i}X_T^{j,\pi^j}\bigg)\bigg)\bigg] \\ &= -\exp\bigg(-\frac{\widetilde{x}_0^i}{\delta_i}\bigg) \cdot \mathbb{E}_{\mathbb{Q}}\bigg[\exp\bigg(-\frac{1}{\delta_i}\int_0^T \Big(\pi_t^i - \frac{\theta_i}{n}\sum_{j\neq i}\pi^j\Big)(\mu + \widetilde{g}(\pi_t^i)) - \frac{\sigma^2}{2\delta_i}\Big(\pi_t^i - \frac{\theta_i}{n}\sum_{j\neq i}\pi^j\Big)^2 \mathrm{d}t\bigg)\bigg] \\ &= -\exp\bigg(-\frac{\widetilde{x}_0^i}{\delta_i}\bigg) \cdot \mathbb{E}_{\mathbb{Q}}\bigg[\exp\bigg(-\frac{1}{\delta_i}\int_0^T h(\pi_t^i)\,\mathrm{d}t\bigg)\bigg], \end{split}$$

where we used the abbreviation of $\tilde{x}_0^i = x_0^i - \frac{\theta_i}{n} \sum_{j \neq i} x_0^j$ and the function h from the proposition. But now in order to maximize the expectation, we can do this pointwise under the integral which leads to maximizing the function h. If a tends to $\pm \infty$, the function h converges to $-\infty$ since g, and thus \tilde{g} , grows sublinearly. Hence, the maximizing point of h is not at the boundary and the assumption of bounded strategies is no restriction. Moreover, h is continuous since \tilde{g} is continuous and hence, h takes its maximum. Thus, an optimal strategy to (7.34) exists and is finite. This concludes our proof.

Remark 7.10. The structure of the function h in Proposition 7.8 suggests that there are at least one and at most two global maxima. To verify this, we consider the components of h. The first factor $a - \frac{\theta_i}{n} \sum_{j \neq i} \pi^j$ is strictly increasing in a for all $a \in \mathbb{R}$ and strictly positive if, and only if, $a > \frac{\theta_i}{n} \sum_{j \neq i} \pi^j$. The second factor $\mu + \tilde{g}(a)$ is also strictly increasing for all $a \in \mathbb{R}$ and strictly positive if, and only if, $\tilde{g}(a) > -\mu$. Given the continuity and monotonicity of \tilde{g} , there exists a unique point $a' \in [-\infty, -\sum_{j \neq i} \pi^j)$ such that $\tilde{g}(a) > -\mu$ if, and only if, a > a'. Note that $a' = -\infty$ is possible since \tilde{g} might be bounded from below by a constant strictly larger than $-\mu$. The last component of h, given by

$$-\frac{\sigma^2}{2\delta_i} \Big(a - \frac{\theta_i}{n} \sum_{j \neq i} \pi^j \Big)^2,$$

is strictly negative for all $a \in \mathbb{R}$, strictly increasing for $a < \frac{\theta_i}{n} \sum_{j \neq i} \pi^j$, and strictly decreasing, otherwise. A combination of these observations, together with the asymptotic behavior of h, yields that h can have at most three local extrema, as there are only three points at which the monotonicity behavior of h might change. More specifically, there is either only one local maximum, which is also the global maximum of h, or there are two local maxima and one local minimum. \diamond

A combination of Corollary 7.7 and Proposition 7.8 yields that, as long as the asymptotic growth of g is at most linear, there exists an optimal solution to the best response problem. The next proposition shows that the opposite holds if g grows superlinearly.

Proposition 7.11. Let $i \in \{1, ..., n\}$ arbitrary but fixed, and π^j , $j \neq i$, deterministic and constant. If $\lim_{x\to\pm\infty} \frac{g(x)}{x} = \infty$, the best response problem for constant Nash equilibria associated to (7.34) does not have an optimal solution.

Proof. To prove that the best response problem for (7.34) does not have a finite optimal solution, we show that even if we consider only constant strategies for agent i, the optimal value is 0 and the associated strategy is infinite. If π^j is constant for all $j \in \{1, \ldots, n\}$, it follows

$$X_T^{i,\pi^i} - \frac{\theta_i}{n} \sum_{j \neq i} X_T^{j,\pi^j}$$

$$= x_0^i - \frac{\theta_i}{n} \sum_{j \neq i} x_0^j + \left(\pi^i - \frac{\theta_i}{n} \sum_{j \neq i} \pi^j\right) (\mu + \tilde{g}(\pi^i)) T + \left(\pi^i - \frac{\theta_i}{n} \sum_{j \neq i} \pi^j\right) \sigma W_T$$

$$=: \tilde{x}_0^i + \mu(\pi^i) T + \sigma(\pi^i) W_T.$$
(7.35)

The random variable in (7.35) has a normal distribution with mean $\mu(\pi^i)$ and variance $\sigma(\pi^i)^2 T$. Therefore, if we apply the exponential utility function to (7.35), we obtain a lognormally distributed random variable and the expectation can be determined explicitly. Hence, for constant π^j , $j = 1, \ldots, n$, the value of the objective function in (7.34) is given by (see Johnson et al., 1994, p. 212)

$$\mathbb{E}\left[-\exp\left(-\frac{1}{\delta_i}\left(\tilde{x}_0^i+\mu(\pi^i)T+\sigma(\pi^i)W_T\right)\right)\right] = -\exp\left(-\frac{1}{\delta_i}\tilde{x}_0^i\right)\cdot\exp\left(-\frac{1}{\delta_i}\left(\mu(\pi^i)-\frac{\sigma(\pi^i)^2}{2\delta_i}\right)T\right).$$

Thus, maximizing the objective function of (7.34) with respect to constant strategies π^i is equivalent to maximizing $\mu(\pi^i) - \frac{\sigma(\pi^i)^2}{2\delta_i}$. Reinserting the definition of $\mu(\pi^i)$ and $\sigma(\pi^i)$ yields

$$\begin{split} &\mu(\pi^i) - \frac{\sigma(\pi^i)^2}{2\delta_i} \\ &= \pi^i \widetilde{g}(\pi^i) - \frac{\sigma^2}{2\delta_i} (\pi^i)^2 + \pi^i \left(\mu + \frac{\sigma^2 \theta_i}{n\delta_i} \sum_{j \neq i} \pi^j \right) - \frac{\theta_i}{n} \widetilde{g}(\pi^i) \sum_{j \neq i} \pi^j - \frac{\theta_i}{n} \sum_{j \neq i} \pi^j \left(\mu + \frac{\sigma^2 \theta_i}{2n\delta_i} \sum_{j \neq i} \pi^j \right), \end{split}$$

which converges to ∞ if π^i converges to $\pm \infty$ using the assumption that g (and thus, \tilde{g}) grows superlinearly. Therefore,

$$0 \ge \sup_{\pi^{i} \in \mathcal{A}} \mathbb{E} \left[-\exp\left(-\frac{1}{\delta_{i}} \left(X_{T}^{i,\pi^{i}} - \frac{\theta_{i}}{n} \sum_{j \neq i} X_{T}^{j,\pi^{j}}\right)\right) \right]$$
$$\ge \sup_{\substack{\pi^{i} \in \mathcal{A} \\ \pi^{i} \text{ constant}}} \mathbb{E} \left[-\exp\left(-\frac{1}{\delta_{i}} \left(X_{T}^{i,\pi^{i}} - \frac{\theta_{i}}{n} \sum_{j \neq i} X_{T}^{j,\pi^{j}}\right)\right) \right] = 0.$$

Hence, the optimal value of (7.34) is 0 which implies that the argument inside the exponential function needs to be infinite. Thus, the problem does not have a finite optimal solution.

7.4. Optimization under power utility

In the previous two sections, we used the additive relative performance metric to measure the relative concerns of n agents. For the definition of relative performance metrics, we refer to the introduction of Chapter 3. Afterwards, we found the unique constant Nash equilibrium in a more specific setting, where the market parameters were assumed to be constant and the investors used exponential utility functions. In combination with exponential utility, it was convenient to use the arithmetic mean in the objective function and to optimize the amount invested into each of the d stocks. However, when using power utility functions, it is more convenient to optimize the invested fraction of wealth and use the geometric mean in the objective function (see, among others, Lacker and Zariphopoulou, 2019). Indeed, it is possible to determine a constant Nash equilibrium to the resulting portfolio optimization problem explicitly. We refer to the beginning of Section 7.3 for a justification of the restriction to constant strategies. Another benefit of the geometric mean (compared to the arithmetic mean) is that we do not need any restrictions on the competition weight θ_i , since the argument inside the utility function is strictly positive if, and only if, the initial capital x_0^i of each agent $i \in \{1, \ldots, n\}$ is strictly positive.

Let us assume that the agents use power utility functions of the form

$$U_i: (0,\infty) \to \mathbb{R}, \ x \mapsto \left(1 - \frac{1}{\delta_i}\right)^{-1} x^{1 - \frac{1}{\delta_i}} = \frac{\delta_i}{\delta_i - 1} x^{\frac{\delta_i - 1}{\delta_i}},$$

for risk tolerance parameters $\delta_i > 0$, $\delta_i \neq 1$, i = 1, ..., n. The underlying financial market again consists of one riskless bond, without loss of generality assumed to be identical to 1, and d stocks with price processes determined by

$$dS_k(t) = S_k(t) \left((\mu_k + \alpha_k \bar{\pi}_k(t)) dt + \sum_{\ell=1}^d \sigma_{k\ell} dW_\ell(t) \right), \ t \in [0, T], \ S_k(0) = 1, \ k = 1, \dots, d.$$

For tractability reasons, we assume that the market parameters μ and σ are constant and deterministic. The volatility matrix σ is again assumed to be regular and the strategies π^j , $j = 1, \ldots, n$, are taken from the set \mathcal{A} of admissible strategies defined in (7.8).

In contrast to the model presented in Section 7.1, the process $(\pi^j(t))_{t\in[0,T]} = (\pi_1^j(t), \ldots, \pi_d^j(t))_{t\in[0,T]}$ now represents the fractions of agent *j*'s wealth invested into the *d* stocks. Nevertheless, $\bar{\pi}$ still describes the arithmetic mean of the *n* investors' strategies. The wealth process of agent $i \in \{1, \ldots, n\}$ solves the stochastic differential equation

$$dX_t^{i,\pi^i} = X_t^{i,\pi^i} \sum_{k=1}^d \pi_k^i(t) \left((\mu_k + \alpha_k \bar{\pi}_k(t)) dt + \sum_{\ell=1}^d \sigma_{k\ell} dW_\ell(t) \right), \ X_0^{i,\pi^i} = x_0^i$$

An explicit representation of X^{i,π^i} is thus given by (see Theorem 2.4 and Lemma 2.2)

$$X_{t}^{i,\pi^{i}} = x_{0}^{i} \exp\left(\int_{0}^{t} \pi^{i}(s)^{\top} \left(\mu + A\bar{\pi}(s) - \frac{1}{2}\sigma\sigma^{\top}\pi^{i}(s)\right) \mathrm{d}s + \int_{0}^{t} \pi^{i}(s)^{\top}\sigma \mathrm{d}W(s)\right),$$
(7.36)

 $t \in [0, T]$. Hence, the portfolio optimization problem of agent i is given by

$$\begin{cases} \max_{\pi^{i} \in \mathcal{A}} \mathbb{E} \left[\frac{\delta_{i}}{\delta_{i}-1} \left(X_{T}^{i,\pi^{i}} \left(\prod_{j \neq i} X_{T}^{j,\pi^{j}} \right)^{-\frac{\theta_{i}}{n}} \right)^{\frac{\delta_{i}-1}{\delta_{i}}} \right], \\ \text{s.t.} \quad \mathrm{d} X_{t}^{i,\pi^{i}} = X_{t}^{i,\pi^{i}} \pi^{i}(t)^{\top} \left(\left(\mu + A\bar{\pi}(t) \right) \mathrm{d} t + \sigma \mathrm{d} W(t) \right), X_{0}^{i,\pi^{i}} = x_{0}^{i}. \end{cases}$$
(7.37)

In order to determine a Nash equilibrium explicitly, we need to restrict the problem to constant Nash equilibria again. Then the unique constant Nash equilibrium is given in the following theorem.

Theorem 7.12. Assume that the following assumptions hold

a) $\frac{n+\theta_i}{n} \left(\sigma \sigma^\top - \frac{\delta_i}{n} A \right) - \frac{\theta_i \delta_i}{n} \sigma \sigma^\top$ is regular for all $i \in \{1, \dots, n\}$, b) $\sigma \sigma^\top - \frac{2\delta_i}{n} A$ is positive-definite for all $i \in \{1, \dots, n\}$,

c)
$$I_d - \sum_{j=1}^n \left(\frac{n+\theta_j}{n} \left(\sigma \sigma^\top - \frac{\delta_j}{n} A \right) - \frac{\theta_j \delta_j}{n} \sigma \sigma^\top \right)^{-1} \left(\frac{\delta_j}{n} \left(1 - \frac{\theta_i}{n} \right) A - \frac{\theta_j}{n} (\delta_j - 1) \sigma \sigma^\top \right)$$
 is regular.

Then the unique constant Nash equilibrium for (7.37) in terms of invested fractions is given by

$$\pi^{i,*} = \delta_i \left(\frac{n+\theta_i}{n} \left(\sigma \sigma^\top - \frac{\delta_i}{n} A \right) - \frac{\theta_i \delta_i}{n} \sigma \sigma^\top \right)^{-1} \mu + \left(\frac{n+\theta_i}{n} \left(\sigma \sigma^\top - \frac{\delta_i}{n} A \right) - \frac{\theta_i \delta_i}{n} \sigma \sigma^\top \right)^{-1} \left(\frac{\delta_j}{n} \left(1 - \frac{\theta_i}{n} \right) A - \frac{\theta_j}{n} (\delta_j - 1) \sigma \sigma^\top \right) \cdot \left(I_d - \sum_{j=1}^n \left(\frac{n+\theta_j}{n} \left(\sigma \sigma^\top - \frac{\delta_j}{n} A \right) - \frac{\theta_j \delta_j}{n} \sigma \sigma^\top \right)^{-1} \left(\frac{\delta_j}{n} \left(1 - \frac{\theta_i}{n} \right) A - \frac{\theta_j}{n} (\delta_j - 1) \sigma \sigma^\top \right) \right)^{-1} \cdot \sum_{j=1}^n \delta_j \left(\frac{n+\theta_j}{n} \left(\sigma \sigma^\top - \frac{\delta_j}{n} A \right) - \frac{\theta_j \delta_j}{n} \sigma \sigma^\top \right)^{-1} \mu, \quad i = 1, \dots, n.$$

Before we display the proof of Theorem 7.12, we provide an example of a class of matrices A that satisfy the conditions of the theorem. Roughly speaking, the conditions a) – c) of Theorem 7.12 are satisfied if the price impact parameters α_k , $k = 1, \ldots, d$, are not too large.

Example 7.13. Let $A = \alpha I_d$ for some constant $\alpha \in \mathbb{R}$. In order to ensure that A satisfies the conditions of Theorem 7.12, we can find sufficient conditions on α .

Since the volatility matrix σ is regular by assumption, $\sigma\sigma^{\top}$ is positive-definite. Let $\lambda_1, \ldots, \lambda_d$ denote the (strictly positive) eigenvalues of $\sigma\sigma^{\top}$. It turns out that the condition

$$\alpha < \frac{n\theta_j(\delta_j - 1)}{\delta_j(n - \theta_j)}\lambda_\ell \quad \text{for all } \ell \in \{1, \dots, d\}, \ j \in \{1, \dots, n\},$$
(7.38)

is sufficient to ensure that $A = \alpha I_d$ satisfies the assumptions of Theorem 7.12, given that n is sufficiently large. The following shows why this is the case.

For $j \in \{1, \ldots, n\}$, the eigenvalues of the matrix

$$\frac{\theta_j(\delta_j-1)}{n}\sigma\sigma^{\top} - \frac{\delta_j(n-\theta_j)}{n^2}A$$
(that appears in condition c) of the previous theorem) are given by

$$\frac{\theta_j(\delta_j - 1)}{n}\lambda_\ell - \frac{\delta_j(n - \theta_j)}{n^2}\alpha,\tag{7.39}$$

 $\ell = 1, \ldots, d$ (see Proposition 4.4.5 in Bernstein, 2009). The expression in (7.39) is strictly positive if, and only if,

$$\alpha < \frac{n\theta_j(\delta_j - 1)}{\delta_j(n - \theta_j)}\lambda_\ell$$

for all $\ell \in \{1, \ldots, d\}$ and $j \in \{1, \ldots, n\}$, i.e., if the asserted sufficient condition (7.38) is satisfied.

Using Proposition 4.4.5 by Bernstein (2009) again, the eigenvalues of the matrix in assumption b) are given by

$$\lambda_{\ell} - \frac{2\delta_j}{n}\alpha, \ \ell = 1, \dots, d$$

for $j \in \{1, \ldots, n\}$. Thus, the eigenvalues are strictly positive if, and only if,

$$\alpha < \frac{n\lambda_{\ell}}{2\delta_j}$$
 for all $\ell = 1, \dots, d, j = 1, \dots, n$.

If n is chosen large enough, i.e., if $n > \theta_j(2\delta_j - 1)$ for all $j \in \{1, \ldots, n\}$, the following holds

$$\frac{n\theta_j(\delta_j - 1)}{\delta_j(n - \theta_j)} < \frac{n}{2\delta_j}.$$

Therefore, condition (7.38) implies that A satisfies condition b) of Theorem 7.12 if n is sufficiently large. Moreover, the eigenvalues of the matrix in a) are given by

$$\frac{n+\theta_j(1-\delta_j)}{n}\lambda_\ell - \frac{\delta_j(n+\theta_j)}{n^2}\alpha, \quad \ell = 1,\dots,d,$$
(7.40)

for $j \in \{1, \ldots, n\}$. The expression in (7.40) is strictly positive if, and only if,

$$\alpha < \frac{n(n+\theta_j(1-\delta_j))}{\delta_j(n+\theta_j)}\lambda_\ell$$

for all $\ell = 1, \ldots, d, j = 1, \ldots, n$. Note that

$$\frac{\theta_j(\delta_j - 1)}{n - \theta_j} < \frac{n + \theta_j(1 - \delta_j)}{n + \theta_j}$$

holds if, and only if, $n > \theta_j(2\delta_j - 1)$, which we already assumed earlier. Hence, the matrix in a) is positive-definite and thus, regular, if (7.38) holds. Finally, this implies that the matrix in condition c) of Theorem 7.12, which can be rewritten as

$$I_d + \sum_{j=1}^n \left(\frac{n+\theta_j}{n} \left(\sigma \sigma^\top - \frac{\delta_j}{n} A \right) - \frac{\theta_j \delta_j}{n} \sigma \sigma^\top \right)^{-1} \left(\frac{\theta_j (\delta_j - 1)}{n} \sigma \sigma^\top - \frac{\delta_j (n-\theta_j)}{n^2} A \right),$$

is a sum of positive-definite matrices. Thus, the matrix is positive-definite as well and hence, regular. Altogether, assumption (7.38) ensures that $A = \alpha I_d$ satisfies the requirements of Theorem 7.12 as long as $n > \theta_j(2\delta_j - 1)$ for all $j \in \{1, \ldots, n\}$. This is not a restriction in our case considering that *n* represents a (large) number of agents. It should be noted that condition (7.38) is sufficient, but not necessary, to ensure that the requirements of Theorem 7.12 are satisfied. Indeed, we chose (7.38) to ensure that the matrices apparent in the sum in condition c) are positive-definite while the theorem only requires that the sum of matrices in c) is regular. However, it simplifies the arguments to consider positive-definite summands instead of regularity for the whole matrix. \Diamond Now we proceed with the proof of Theorem 7.12.

Proof (Theorem 7.12). Choose some arbitrary $i \in \{1, ..., n\}$ and assume that the agents $j \neq i$ use constant strategies π^j , $j \neq i$, which are assumed to be arbitrary but fixed. Now define a stochastic process $(Y_t^{-i})_{t \in [0,T]}$ by $Y_t^{-i} = \prod_{j \neq i} X_t^{j,\pi^j}$, $t \in [0,T]$. Using (7.36), it follows

$$Y_t^{-i} = y_0^i \exp\left(\int_0^t \left(\sum_{j \neq i} \pi^j\right)^\top \left((\mu + A\bar{\pi}(s))\mathrm{d}s + \sigma \mathrm{d}W(s)\right) - \frac{1}{2}\int_0^t \sum_{j \neq i} (\pi^j)^\top \sigma \sigma^\top \pi^j \mathrm{d}s\right),$$

where $y_0^i := \prod_{i \neq i} x_0^j$. Thus, the argument of the expected utility in (7.37) can be written as follows

$$\begin{split} X_t^{i,\pi^i} \left(Y_t^{-i}\right)^{-\frac{\theta_i}{n}} & \exp\left(\int_0^t \pi^i(s)^\top \left(\mu + A\bar{\pi}(s) - \frac{1}{2}\sigma\sigma^\top \pi^i(s)\right) \mathrm{d}s + \int_0^t \pi^i(s)^\top \sigma \mathrm{d}W(s) \right. \\ & - \frac{\theta_i}{n} \left\{\int_0^t \left(\sum_{j\neq i} \pi^j\right)^\top (\mu + A\bar{\pi}(s)) \mathrm{d}s + \int_0^t \left(\sum_{j\neq i} \pi^j\right)^\top \sigma \mathrm{d}W(s) - \frac{1}{2}\int_0^t \sum_{j\neq i} (\pi^j)^\top \sigma \sigma^\top \pi^j \mathrm{d}s \right\} \right) \\ & = x_0^i \exp\left(\int_0^t \pi^i(s)^\top \left(\mu + \left(\frac{1}{n}A - \frac{1}{2}\sigma\sigma^\top\right)\pi^i(s) + \frac{1}{n}\left(1 - \frac{\theta_i}{n}\right)A\sum_{j\neq i}\pi^j\right) \mathrm{d}s + \int_0^t \pi^i(s)^\top \sigma \mathrm{d}W(s) \right) \\ & \cdot (y_0^i)^{-\frac{\theta_i}{n}} \exp\left(-\frac{\theta_i}{n} \left\{\int_0^t \left(\sum_{j\neq i}\pi^j\right)^\top \left(\mu + \frac{1}{n}A\sum_{j\neq i}\pi^j\right) \mathrm{d}s + \int_0^t \left(\sum_{j\neq i}\pi^j\right)^\top \sigma \mathrm{d}W(s) - \frac{1}{2}\int_0^t \sum_{j\neq i} (\pi^j)^\top \sigma \sigma^\top \pi^j \mathrm{d}s \right\} \right) \\ & =: \tilde{X}_t^{i,\pi^i} \left(\tilde{Y}_t^{-i}\right)^{-\frac{\theta_i}{n}}, \end{split}$$

where the process \tilde{Y}^{-i} does not depend on π^i . More specifically, the processes \tilde{X}^{i,π^i} and \tilde{Y}^{-i} are the unique solutions to the following stochastic differential equations (using Theorem 2.4 and Lemma 2.2)

$$\begin{split} \mathrm{d}\widetilde{X}_{t}^{i,\pi^{i}} &= \widetilde{X}_{t}^{i,\pi^{i}}\pi^{i}(t)^{\top} \bigg(\bigg(\mu + \frac{1}{n}A\pi^{i}(t) + \frac{1}{n}\Big(1 - \frac{\theta_{i}}{n}\Big)A\sum_{j\neq i}\pi^{j}\Big)\mathrm{d}t + \sigma\mathrm{d}W(t) \bigg), \\ \mathrm{d}\widetilde{Y}_{t}^{-i} &= \widetilde{Y}_{t}^{-i} \bigg(\Big(\sum_{j\neq i}\pi^{j}\Big)^{\top} \bigg(\bigg(\mu + \frac{1}{n}A\sum_{j\neq i}\pi^{j}\bigg)\mathrm{d}t + \sigma\mathrm{d}W(t) \bigg) + \frac{1}{2}\sum_{\substack{1\leq h\neq j\leq n\\h,j\neq i}}(\pi^{h})^{\top}\sigma\sigma^{\top}\pi^{j}\mathrm{d}t \bigg) \end{split}$$

with initial values $\tilde{X}_0^{i,\pi^i} = x_0^i$ and $\tilde{Y}_0^{-i} = y_0^i$. The introduction of \tilde{X}^{i,π^i} and \tilde{Y}^{-i} simplifies the derivation of the HJB equation below. First, we notice that the quadratic (co)variations of \tilde{X}^{i,π^i}

and \tilde{Y}^{-i} are given as follows (see Lemma 2.2)

$$\begin{split} \mathbf{d} \langle \tilde{X}^{i,\pi^{i}} \rangle_{t} &= \left(\tilde{X}_{t}^{i,\pi^{i}} \right)^{2} \pi^{i}(t)^{\top} \sigma \sigma^{\top} \pi^{i}(t) \mathbf{d}t, \quad \mathbf{d} \langle \tilde{Y}^{-i} \rangle_{t} = \left(\tilde{Y}_{t}^{-i} \right)^{2} \left(\sum_{j \neq i} \pi^{j} \right)^{\top} \sigma \sigma^{\top} \sum_{j \neq i} \pi^{j} \mathbf{d}t, \\ \mathbf{d} \langle \tilde{X}^{i,\pi^{i}}, \tilde{Y}^{-i} \rangle_{t} &= \tilde{X}_{t}^{i,\pi^{i}} \tilde{Y}_{t}^{-i} \pi^{i}(t)^{\top} \sigma \sigma^{\top} \sum_{j \neq i} \pi^{j} \mathbf{d}t, \end{split}$$

for $t \in [0,T]$. Let us now define the following value functions for $t \in [0,T]$, $x, y \in (0,\infty)$

$$\begin{split} J(t,x,y;\pi^{i}) &\coloneqq \mathbb{E}\bigg[\frac{\delta_{i}}{\delta_{i}-1} \Big(\widetilde{X}_{T}^{i,\pi^{i}}\left(\widetilde{Y}_{T}^{-i}\right)^{-\frac{\theta_{i}}{n}}\Big)^{\frac{\delta_{i}-1}{\delta_{i}}} \Big|\widetilde{X}_{t}^{i,\pi^{i}} = x, \, \widetilde{Y}_{t}^{-i} = y\bigg] \\ &=: \mathbb{E}^{t,x,y}\bigg[\frac{\delta_{i}}{\delta_{i}-1} \Big(\widetilde{X}_{T}^{i,\pi^{i}}\Big(\widetilde{Y}_{T}^{-i}\Big)^{-\frac{\theta_{i}}{n}}\Big)^{\frac{\delta_{i}-1}{\delta_{i}}}\bigg], \\ J(t,x,y) &\coloneqq \sup_{\pi^{i} \in \mathcal{A}} J(t,x,y;\pi^{i}). \end{split}$$

Then, for any $t \leq t' \leq T$, the Bellman principle

$$J(t,x,y) = \sup_{\pi^{i} \in \mathcal{A}} \mathbb{E}^{t,x,y} \left[J\left(t', \widetilde{X}_{t'}^{i,\pi^{i}}, \widetilde{Y}_{t'}^{-i}\right) \right]$$
(7.41)

holds (see, for example, Equation (3.20) by Pham, 2009). For the following heuristic derivation of the HJB equation, we assume that $J \in C^{1,2,2}([0,T] \times (0,\infty)^2)$. Now we can apply the Itô-Doeblin formula (Theorem 2.1) for J on the interval [t,t'], in combination with Lemma 2.2, to obtain

$$J\left(t', \widetilde{X}_{t'}^{i,\pi^{i}}, \widetilde{Y}_{t'}^{-i}\right) = J\left(t, \widetilde{X}_{t}^{i,\pi^{i}}, \widetilde{Y}_{t}^{-i}\right) + \int_{t}^{t'} J_{t}\left(s, \widetilde{X}_{s}^{i,\pi^{i}}, \widetilde{Y}_{s}^{-i}\right) \mathrm{d}s + \int_{t}^{t'} J_{x}\left(s, \widetilde{X}_{s}^{i,\pi^{i}}, \widetilde{Y}_{s}^{-i}\right) \mathrm{d}\widetilde{X}_{s}^{i,\pi^{i}}$$
$$+ \int_{t}^{t'} J_{y}\left(s, \widetilde{X}_{s}^{i,\pi^{i}}, \widetilde{Y}_{s}^{-i}\right) \mathrm{d}\widetilde{Y}_{s}^{-i} + \frac{1}{2} \int_{t}^{t'} J_{xx}\left(s, \widetilde{X}_{s}^{i,\pi^{i}}, \widetilde{Y}_{s}^{-i}\right) \mathrm{d}\langle\widetilde{X}^{i,\pi^{i}}\rangle_{s}$$
$$+ \int_{t}^{t'} J_{xy}\left(s, \widetilde{X}_{s}^{i,\pi^{i}}, \widetilde{Y}_{s}^{-i}\right) \mathrm{d}\langle\widetilde{X}^{i,\pi^{i}}, \widetilde{Y}^{-i}\rangle_{s} + \frac{1}{2} \int_{t}^{t'} J_{yy}\left(s, \widetilde{X}_{s}^{i,\pi^{i}}, \widetilde{Y}_{s}^{-i}\right) \mathrm{d}\langle\widetilde{Y}^{-i}\rangle_{s}.$$
(7.42)

By J_x we denote the first order partial derivative of J with respect to x. The other partial derivatives used above are denoted analogously. To simplify notation, the arguments of J and its derivatives are omitted from now on. Continuing (7.42) yields

$$\begin{split} \dots &= J\left(t, \tilde{X}_{t}^{i,\pi^{i}}, \tilde{Y}_{t}^{-i}\right) + \int_{t}^{t'} J_{t} \,\mathrm{d}s + \int_{t}^{t'} J_{x} \tilde{X}_{s}^{i,\pi^{i}} \pi^{i}(s)^{\top} \left(\mu + \frac{1}{n} A \pi^{i}(s) + \frac{1}{n} \left(1 - \frac{\theta_{i}}{n}\right) A \sum_{j \neq i} \pi^{j}\right) \mathrm{d}s \\ &+ \int_{t}^{t'} J_{x} \tilde{X}_{s}^{i,\pi^{i}} \pi^{i}(s)^{\top} \sigma \mathrm{d}W(s) + \int_{t}^{t'} J_{y} \tilde{Y}_{s}^{-i} \left(\sum_{j \neq i} \pi^{j}\right)^{\top} \sigma \mathrm{d}W(s) \tag{7.43} \\ &+ \int_{t}^{t'} J_{y} \tilde{Y}_{s}^{-i} \left(\left(\sum_{j \neq i} \pi^{j}\right)^{\top} \left(\mu + \frac{1}{n} A \sum_{j \neq i} \pi^{j}\right) + \frac{1}{2} \sum_{\substack{h \neq j \\ h, j \neq i}} (\pi^{h})^{\top} \sigma \sigma^{\top} \pi^{j}\right) \mathrm{d}s \\ &+ \frac{1}{2} \int_{t}^{t'} J_{xx} \left(\tilde{X}_{s}^{i,\pi^{i}}\right)^{2} \pi^{i}(s)^{\top} \sigma \sigma^{\top} \pi^{i}(s) \mathrm{d}s + \int_{t}^{t'} J_{xy} \tilde{X}_{s}^{i,\pi^{i}} \tilde{Y}_{s}^{-i} \pi^{i}(s)^{\top} \sigma \sigma^{\top} \sum_{j \neq i} \pi^{j} \mathrm{d}s \\ &+ \frac{1}{2} \int_{t}^{t'} J_{yy} \left(\tilde{Y}_{s}^{-i}\right)^{2} \sum_{h, j \neq i} (\pi^{h})^{\top} \sigma \sigma^{\top} \pi^{j} \mathrm{d}s. \end{aligned}$$

Now we insert the previous representation of $J(t', \tilde{X}_{t'}^{i,\pi^i}, \tilde{Y}_{t'}^{-i})$, into the Bellman equation (7.41), divide by t' - t and take the limit $t' \to t$, to obtain

$$0 = G_t + yG_y \left\{ \left(\sum_{j \neq i} \pi^j \right)^\top \left(\mu + \frac{1}{n} A \sum_{j \neq i} \pi^j \right) + \frac{1}{2} \sum_{\substack{h \neq j \\ h, j \neq i}} (\pi^h)^\top \sigma \sigma^\top \pi^j \right\} + \frac{1}{2} y^2 G_{yy} \sum_{\substack{h, j \neq i}} (\pi^h)^\top \sigma \sigma^\top \pi^j$$
$$+ \sup_{\pi^i \in \mathbb{R}^d} \left\{ xG_x \cdot (\pi^i)^\top \mu + \frac{1}{n} xG_x \cdot (\pi^i)^\top A \pi^i + xG_x \frac{1}{n} \left(1 - \frac{\theta_i}{n} \right) (\pi^i)^\top A \sum_{j \neq i} \pi^j$$
$$+ \frac{1}{2} x^2 G_{xx} \cdot (\pi^i)^\top \sigma \sigma^\top \pi^i + xy G_{xy} \cdot (\pi^i)^\top \sigma \sigma^\top \sum_{j \neq i} \pi^j \right\}$$
(7.44)

with terminal condition $G(T, x, y) = \left(1 - \frac{1}{\delta_i}\right)^{-1} \left(xy^{-\frac{\theta_i}{n}}\right)^{1-\frac{1}{\delta_i}}$, $x, y \in (0, \infty)$. Note that we assumed that the limit $t' \to t$ and the conditional expectation $\mathbb{E}^{t,x,y}$ can be interchanged. Further, we assumed that the stochastic integrals in (7.43) are martingales so that the conditional expectation vanishes. This is not a restriction as the verification theorem (Lemma B.2 in the Appendix) ensures that a solution to the HJB equation coincides with the value function and provides a unique best response $\pi^{i,*}$.

To determine the supremum in the previous equation, we define the expression inside the supremum as $h(\pi^i)$, i.e.,

$$h(\pi^{i})$$

$$= (\pi^{i})^{\top} \left(x G_{x} \mu + \left(x G_{x} \frac{1}{n} \left(1 - \frac{\theta_{i}}{n} \right) A + x y G_{xy} \sigma \sigma^{\top} \right) \sum_{j \neq i} \pi^{j} \right) + (\pi^{i})^{\top} \left(\frac{1}{n} x G_{x} A + \frac{1}{2} x^{2} G_{xx} \sigma \sigma^{\top} \right) \pi^{i}$$

$$=: (\pi^{i})^{\top} \left(x G_{x} \mu + C_{2} \sum_{j \neq i} \pi^{j} \right) + (\pi^{i})^{\top} C_{1} \pi^{i},$$

where

$$C_1 = \frac{1}{n}xG_xA + \frac{1}{2}x^2G_{xx}\sigma\sigma^{\top}, \quad C_2 = xG_x\frac{1}{n}\left(1 - \frac{\theta_i}{n}\right)A + xyG_{xy}\sigma\sigma^{\top}.$$

The gradient of h is then given by

$$\nabla h(\pi^i) = x G_x \mu + C_2 \sum_{j \neq i} \pi^j + 2C_1 \pi^i \stackrel{!}{=} 0.$$

Moreover, the Hessian matrix of h is given by

$$H_h(\pi^i) = 2C_1.$$

Hence, a candidate for the maximizer of h is determined by

$$C_1 \pi^i = -\frac{1}{2} x G_x \mu - \frac{1}{2} C_2 \sum_{j \neq i} \pi^j.$$
(7.45)

Since the matrix C_1 depends on the unknown function G, we cannot guarantee that C_1 is regular.

Therefore, we insert an ansatz for G first. We choose

$$G(t,x,y) = f(t) \cdot \frac{\delta_i}{\delta_i - 1} \left(xy^{-\frac{\theta_i}{n}} \right)^{\frac{\delta_i - 1}{\delta_i}} \eqqcolon f(t) \cdot \widetilde{U}(x,y), \, x, y \in (0,\infty),$$

where $f: [0,T] \to (0,\infty)$ is some continuously differentiable function with f(T) = 1. Then the terminal condition is obviously satisfied and $G \in C^{1,2,2}([0,T], (0,\infty)^2)$. Moreover, we can directly deduce that $G(t,x,y) \neq 0$ for all $t \in [0,T]$ and $x, y \in (0,\infty)$. Let us now find a suitable choice for f to ensure that G solves the HJB equation (7.44). First, we determine the partial derivatives of G with respect to t, x, and y:

$$\begin{aligned} G_{t}(t,x,y) &= f'(t)\widetilde{U}(x,y), \\ G_{xx}(t,x,y) &= -\frac{\delta_{i}-1}{\delta_{i}^{2}}x^{-2}G(t,x,y), \\ G_{xx}(t,x,y) &= -\frac{\delta_{i}-1}{\delta_{i}}x^{-1}G(t,x,y), \\ G_{y}(t,x,y) &= -\frac{\theta_{i}}{n}\frac{\delta_{i}-1}{\delta_{i}}y^{-1}G(t,x,y), \\ G_{yy}(t,x,y) &= -\frac{\theta_{i}}{n}\frac{\delta_{i}-1}{\delta_{i}}y^{-1}G(t,x,y), \\ G_{yy}(t,x,y) &= -\frac{\theta_{i}}{n}\frac{\delta_{i}-1}{\delta_{i}}\left(1 + \frac{\theta_{i}}{n}\frac{\delta_{i}-1}{\delta_{i}}\right)y^{-2}G(t,x,y). \end{aligned}$$

The matrices C_1 and C_2 can then be simplified to

$$C_{1} = \frac{1}{n}xG_{x}A + \frac{1}{2}x^{2}G_{xx}\sigma\sigma^{\top} = -\frac{1}{2}\frac{\delta_{i}-1}{\delta_{i}^{2}}G\cdot\left(\sigma\sigma^{\top}-\frac{2\delta_{i}}{n}A\right),$$

$$C_{2} = xG_{x}\frac{1}{n}\left(1-\frac{\theta_{i}}{n}\right)A + xyG_{xy}\sigma\sigma^{\top} = \frac{\delta_{i}-1}{\delta_{i}}G\cdot\left(\left(1-\frac{\theta_{i}}{n}\right)\frac{1}{n}A - \frac{\theta_{i}}{n}\frac{\delta_{i}-1}{\delta_{i}}\sigma\sigma^{\top}\right).$$

Hence, C_1 is regular using assumption b) as well as $G \neq 0$ and $\delta_i \neq 1$. Thus, we can insert C_1 and C_2 into (7.45) to obtain

$$\begin{aligned} \pi^{i,*} &= -\frac{1}{2} C_1^{-1} \left(x G_x \mu + C_2 \sum_{j \neq i} \pi^j \right) \\ &= \left(\sigma \sigma^\top - \frac{2\delta_i}{n} A \right)^{-1} \left(\delta_i \mu + \left(\frac{\delta_i}{n} \left(1 - \frac{\theta_i}{n} \right) A - \frac{\theta_i}{n} (\delta_i - 1) \sigma \sigma^\top \right) \sum_{j \neq i} \pi^j \right). \end{aligned}$$

Therefore, the candidate $\pi^{i,*}$ for the optimal strategy of investor *i* appears to be constant as well. However, we need to verify that *G* is in fact a solution to the HJB equation (7.44). The first step is to insert $\pi^{i,*}$ into the function *h*. This yields (since C_1 is symmetric by construction)

$$\begin{split} h(\pi^{i,*}) &= (\pi^{i,*})^\top \left(x G_x \mu + C_2 \sum_{j \neq i} \pi^j \right) + (\pi^{i,*})^\top C_1 \pi^{i,*} \\ &= -\frac{1}{4} \left(x G_x \mu + C_2 \sum_{j \neq i} \pi^j \right)^\top C_1^{-1} \left(x G_x \mu + C_2 \sum_{j \neq i} \pi^j \right) \\ &= \frac{1}{2} (\delta_i - 1) G \cdot \left(\mu + \left(\frac{1}{n} \left(1 - \frac{\theta_i}{n} \right) A - \frac{\theta_i}{n} \frac{\delta_i - 1}{\delta_i} \sigma \sigma^\top \right) \sum_{j \neq i} \pi^j \right)^\top \left(\sigma \sigma^\top - \frac{2\delta_i}{n} A \right)^{-1} \\ &\cdot \left(\mu + \left(\frac{1}{n} \left(1 - \frac{\theta_i}{n} \right) A - \frac{\theta_i}{n} \frac{\delta_i - 1}{\delta_i} \sigma \sigma^\top \right) \sum_{j \neq i} \pi^j \right) \\ &=: \rho_1 \cdot G \end{split}$$

for a constant $\rho_1 \in \mathbb{R}$, since π^j , $j \neq i$, are constant by assumption. It also follows that $\pi^{i,*}$ is indeed a maximizer of h since the Hessian matrix $H_h(\pi^{i,*}) = 2C_1$ is negative-definite due to assumption b) of the theorem and the fact that

$$-\frac{\delta_i-1}{\delta_i^2}G=-\frac{1}{\delta_i}f(t)\cdot\left(xy^{-\frac{\theta_i}{n}}\right)^{\frac{\delta_i-1}{\delta_i}}<0$$

for all $x, y \in (0, \infty), t \in [0, T]$. Moreover, we can simplify the first three summands of (7.44) as follows

$$G_{t} = f' \cdot \tilde{U},$$

$$yG_{y} \left\{ \left(\sum_{j \neq i} \pi^{j} \right)^{\top} \left(\mu + \frac{1}{n} A \sum_{j \neq i} \pi^{j} \right) + \frac{1}{2} \sum_{\substack{h \neq j \\ h, j \neq i}} (\pi^{h})^{\top} \sigma \sigma^{\top} \pi^{j} \right\} = \rho_{2} \cdot G,$$

$$\frac{1}{2} y^{2} G_{yy} \cdot \sum_{h, j \neq i} (\pi^{h})^{\top} \sigma \sigma^{\top} \pi^{j} = \rho_{3} \cdot G,$$

for constants ρ_2 , $\rho_3 \in \mathbb{R}$ that can be given explicitly but are not of interest for the rest of the proof. Hence, after inserting the ansatz for G, the HJB equation (7.44) simplifies to (using $G = f \cdot \tilde{U}$)

$$0 = f'\widetilde{U} + \rho_2 f\widetilde{U} + \rho_3 f\widetilde{U} + \rho_1 f\widetilde{U} \eqqcolon \widetilde{U} \left(f' + \rho f\right)$$
(7.46)

for $\rho = \rho_1 + \rho_2 + \rho_3$. Since $\widetilde{U}(x, y) \neq 0$ for all $x, y \in (0, \infty)$, it follows that f needs to solve

$$f'(t) = -\rho f(t), \ t \in [0, T], \tag{7.47}$$

with terminal condition f(T) = 1. The unique solution to this ordinary differential equation is given by

$$f(t) = e^{\rho(T-t)} t \in [0, T].$$

Hence, the ansatz for G does in fact yield a solution to the HJB equation if we choose f as in (7.47). Therefore, the function $G: [0,T] \times (0,\infty)^2 \to \mathbb{R}$ given by

$$G(t, x, y) = e^{\rho(T-t)} \frac{\delta_i}{\delta_i - 1} \left(x y^{-\frac{\theta_i}{n}} \right)^{\frac{\delta_i - 1}{\delta_i}}$$

is a solution to the HJB equation. Using Lemma B.2 in the Appendix and (7.45), the unique (up to modifications) optimal solution of the best response problem is given by $\pi^{i,*}$, where $\pi^{i,*}$ solves

$$\left(\sigma\sigma^{\top} - \frac{2\delta_i}{n}A\right)\pi^{i,*} = \delta_i\mu + \left(\frac{\delta_i}{n}\left(1 - \frac{\theta_i}{n}\right)A - \frac{\theta_i}{n}(\delta_i - 1)\sigma\sigma^{\top}\right)\sum_{j\neq i}\pi^j.$$
(7.48)

Moreover, Lemma B.2 implies that the solution G and the value function J are equal which also implies that G is the unique solution to the HJB equation (7.44).

The only task left in order to find the Nash equilibrium is to solve the system of linear equations

given by (7.48). First, we add $\left(\frac{\delta_i}{n}\left(1-\frac{\theta_i}{n}\right)A-\frac{\theta_i}{n}(\delta_i-1)\sigma\sigma^{\top}\right)\pi^i$ on both sides to obtain

$$\left(\frac{n+\theta_i}{n}\left(\sigma\sigma^{\top}-\frac{\delta_i}{n}A\right)-\frac{\theta_i\delta_i}{n}\sigma\sigma^{\top}\right)\pi^i=\delta_i\mu+\left(\frac{\delta_i}{n}\left(1-\frac{\theta_i}{n}\right)A-\frac{\theta_i}{n}(\delta_i-1)\sigma\sigma^{\top}\right)\sum_{j=1}^n\pi^j$$

By assumption a), the matrix on the left-hand side of the previous equation is regular, so it follows that

$$\pi^{i} = \delta_{i} \left(\frac{n + \theta_{i}}{n} \left(\sigma \sigma^{\top} - \frac{\delta_{i}}{n} A \right) - \frac{\theta_{i} \delta_{i}}{n} \sigma \sigma^{\top} \right)^{-1} \mu \\ + \left(\frac{n + \theta_{i}}{n} \left(\sigma \sigma^{\top} - \frac{\delta_{i}}{n} A \right) - \frac{\theta_{i} \delta_{i}}{n} \sigma \sigma^{\top} \right)^{-1} \left(\frac{\delta_{i}}{n} \frac{n - \theta_{i}}{n} A - \frac{\theta_{i}}{n} (\delta_{i} - 1) \sigma \sigma^{\top} \right) \sum_{j=1}^{n} \pi^{j}.$$

Summing over all $i \in \{1, ..., n\}$ on both sides of the previous equation implies

$$\sum_{j=1}^{n} \pi^{j} = \sum_{j=1}^{n} \delta_{j} \left(\frac{n+\theta_{j}}{n} \left(\sigma \sigma^{\top} - \frac{\delta_{j}}{n} A \right) - \frac{\theta_{j} \delta_{j}}{n} \sigma \sigma^{\top} \right)^{-1} \cdot \mu$$

+
$$\sum_{j=1}^{n} \left(\frac{n+\theta_{j}}{n} \left(\sigma \sigma^{\top} - \frac{\delta_{j}}{n} A \right) - \frac{\theta_{j} \delta_{j}}{n} \sigma \sigma^{\top} \right)^{-1} \left(\frac{\delta_{j}}{n} \frac{n-\theta_{j}}{n} A - \frac{\theta_{j}}{n} (\delta_{j} - 1) \sigma \sigma^{\top} \right) \sum_{j=1}^{n} \pi^{j}.$$

By assumption c), the matrix

$$I_d - \sum_{j=1}^n \left(\frac{n+\theta_j}{n} \left(\sigma \sigma^\top - \frac{\delta_j}{n} A \right) - \frac{\theta_j \delta_j}{n} \sigma \sigma^\top \right)^{-1} \left(\frac{\delta_j}{n} \frac{n-\theta_j}{n} A - \frac{\theta_j}{n} (\delta_j - 1) \sigma \sigma^\top \right)$$

is regular. Hence, the sum of all π^j , j = 1, ..., n, is given by

$$\sum_{j=1}^{n} \pi^{j} = \left(I_{d} - \sum_{j=1}^{n} \left(\frac{n+\theta_{j}}{n} \left(\sigma \sigma^{\top} - \frac{\delta_{j}}{n} A \right) - \frac{\theta_{j} \delta_{j}}{n} \sigma \sigma^{\top} \right)^{-1} \left(\frac{\delta_{j}}{n} \frac{n-\theta_{j}}{n} A - \frac{\theta_{j}}{n} (\delta_{j} - 1) \sigma \sigma^{\top} \right) \right)^{-1} \cdot \sum_{j=1}^{n} \delta_{j} \left(\frac{n+\theta_{j}}{n} \left(\sigma \sigma^{\top} - \frac{\delta_{j}}{n} A \right) - \frac{\theta_{j} \delta_{j}}{n} \sigma \sigma^{\top} \right)^{-1} \cdot \mu.$$

Finally, the unique constant Nash equilibrium is given by

$$\pi^{i,*} = \delta_i \left(\frac{n+\theta_i}{n} \left(\sigma \sigma^\top - \frac{\delta_i}{n} A \right) - \frac{\theta_i \delta_i}{n} \sigma \sigma^\top \right)^{-1} \mu \\ + \left(\frac{n+\theta_i}{n} \left(\sigma \sigma^\top - \frac{\delta_i}{n} A \right) - \frac{\theta_i \delta_i}{n} \sigma \sigma^\top \right)^{-1} \left(\frac{\delta_i}{n} \frac{n-\theta_i}{n} A - \frac{\theta_i}{n} (\delta_i - 1) \sigma \sigma^\top \right) \\ \cdot \left(I_d - \sum_{j=1}^n \left(\frac{n+\theta_j}{n} \left(\sigma \sigma^\top - \frac{\delta_j}{n} A \right) - \frac{\theta_j \delta_j}{n} \sigma \sigma^\top \right)^{-1} \left(\frac{\delta_j}{n} \frac{n-\theta_j}{n} A - \frac{\theta_j}{n} (\delta_j - 1) \sigma \sigma^\top \right) \right)^{-1} \\ \cdot \sum_{j=1}^n \delta_j \left(\frac{n+\theta_j}{n} \left(\sigma \sigma^\top - \frac{\delta_j}{n} A \right) - \frac{\theta_j \delta_j}{n} \sigma \sigma^\top \right)^{-1} \cdot \mu.$$

Similarly to Corollary 7.7, we can deduce the unique constant Nash equilibrium for the case of only one stock (d = 1).

Corollary 7.14. Let d = 1, $A = \alpha$, and assume that the following assumptions hold

a)
$$(n + \theta_i) (n\sigma^2 - \delta_i \alpha) - n\theta_i \delta_i \sigma^2 \neq 0$$
 for all $i \in \{1, ..., n\}$,
b) $n\sigma^2 - 2\delta_i \alpha > 0$ for all $i \in \{1, ..., n\}$,

c)
$$\sum_{j=1}^{n} \frac{(n-\theta_j)\alpha \delta_j - n\theta_j(\delta_j - 1)\sigma^2}{(n+\theta_j)(n\sigma^2 - \alpha\delta_j) - n\theta_j\delta_j\sigma^2} \neq 1$$

Then the unique constant Nash equilibrium to (7.37) in terms of invested fractions is given by

$$\pi^{i,*} = \frac{n^2 \delta_i \mu}{(n+\theta_i)(n\sigma^2 - \delta_i \alpha) - n\theta_i \delta_i \sigma^2} + \frac{(n-\theta_i)\alpha \delta_i - n\theta_i (\delta_i - 1)\sigma^2}{(n+\theta_i)(n\sigma^2 - \delta_i \alpha) - n\theta_i \delta_i \sigma^2} \\ \cdot \left(1 - \sum_{j=1}^n \frac{(n-\theta_j)\alpha \delta_j - n\theta_j (\delta_j - 1)\sigma^2}{(n+\theta_j)(n\sigma^2 - \alpha \delta_j) - n\theta_j \delta_j \sigma^2}\right)^{-1} \sum_{j=1}^n \frac{n^2 \delta_j \mu}{(n+\theta_j)(n\sigma^2 - \delta_j \alpha) - n\theta_j \delta_j \sigma^2}$$

for $i \in \{1, ..., n\}$.

Subsequent to Corollary 7.7, we analyzed the influence of the price impact parameter α on the entries of the constant Nash equilibrium in terms of invested amounts under exponential utility. Due to the more complicated structure of the constant Nash equilibrium in Corollary 7.14, we do not discuss the influence of α as detailed as in Subsection 7.3.2. However, Corollary 7.14 enables us to compute the value of a component of the Nash equilibrium for specific parameter choices. The results are illustrated in Figure 7.4.1.



Figure 7.4.1.: Illustration of $\pi^{1,*}$ from Corollary 7.14 in terms of $\alpha \in (-0.02, \alpha_{\max})$ for n = 12, $\mu = 0.03$, $\sigma = 0.2$, and $\alpha_{\max} = n\sigma^2/8$. Further, $\theta_1 = 0.3$, $\delta_1 \in \{1,4\}$ and the parameters θ_j and δ_j , $j \ge 2$ are increasing from 0 to 1 with step size 0.1, and from 0.5 to 2.7 by step size 0.2, respectively.

Figure 7.4.1 displays the first component $\pi^{1,*}$ of the constant Nash equilibrium given in Corollary 7.14 in terms of α varying between -0.02 and α_{\max} for the two different risk tolerance parameters $\delta_1 = 1$ and $\delta_1 = 4$. The expression α_{\max} is defined analogously to Subsection 7.3.2 as

$$\alpha_{\rm max} = \frac{n\sigma^2}{2\delta_{\rm max}},$$

where $\delta_{\max} = \max{\{\delta_1, \ldots, \delta_n\}}$. In the example displayed in Figure 7.4.1, we used $\delta_{\max} = 4$. The market parameters are chosen as $\mu = 0.03$ and $\sigma = 0.2$. Note that all considered parameter combinations satisfy the conditions of Corollary 7.14. Similar to Figure 7.3.1, we observe a discontinuity of $\pi^{1,*}$. In Subsection 7.3.2, we provided a detailed discussion of the existence of a unique point of discontinuity. Here, we only give a short explanation regarding the discontinuity. For the specific parameter choices used in the example, conditions a) and b) of Corollary 7.14 are always satisfied. The discontinuity is due to condition c), i.e., for both parameter choices $\delta_1 \in \{1, 4\}$, there exists a unique value $\alpha_0 \in (-\infty, \alpha_{\max})$ such that the expression in condition c) is zero. In the figure, the value α_0 is highlighted by a vertical dotted line for each of the two parameter choices $\delta_1 \in \{1, 4\}$. Moreover, the blue and orange horizontal dashed lines mark the Merton ratio $\delta_1 \mu \sigma^{-2}$, i.e., the unique optimally invested fraction in the associated classical problem ($\alpha = 0, \theta_1 = 0$), for the two different values used for δ_1 . Finally, we highlighted the value zero on both axes by a grey line.

Considering the behavior of $\pi^{1,*}$ in terms of α , we notice that $\pi^{1,*}$ is strictly positive for $\alpha < \alpha_0$ and strictly negative for $\alpha > \alpha_0$. Moreover, we observe that for larger price impact (i.e., if the absolute value of α increases), the agents engage less in the financial market which is represented by a decrease in the absolute value of $\pi^{1,*}$. Overall, we notice the same behavior of $\pi^{1,*}$ in terms of α as in the case of exponential utility which we considered in Subsection 7.3.2.

CHAPTER 8

Relative performance via a VAR-type constraint

In Chapters 3 – 7, we modeled the relative concerns of investors by including a relative performance metric into their utility function. In this chapter, we take a different approach by adapting a method introduced by Basak and Shapiro (2001). They incorporated risk constraints into the classical utility maximization problem. As a risk measure, they chose the value at risk (VaR) which is a standard criterion to assess risk in the financial industry (see, e.g., Berkelaar et al., 2002). The value at risk at level $1 - \alpha \in (0, 1)$ of some payoff X describes the "smallest amount of capital which, if added to X and invested in the risk-free asset, keeps the probability of a negative outcome below the level $1 - \alpha$ ". Mathematically speaking, the value at risk at level $1 - \alpha$ of a payoff X is defined as

$$\operatorname{VaR}_{1-\alpha}(X) = \inf \Big\{ m \in \mathbb{R} \, | \, \mathbb{P}(X + m < 0) \le 1 - \alpha \Big\}.$$

Thus, $\operatorname{VaR}_{1-\alpha}(X)$ describes the lower α -quantile¹ of -X (see Föllmer and Schied, 2016, p. 231). Basak and Shapiro (2001) define the value at risk slightly different by considering the loss of wealth suffered by some portfolio over a fixed time period. They describe it as the worst loss over a given time interval under "normal market conditions". The term "normal market conditions" refers to the fact that the level $1 - \alpha$ is usually chosen as a small percentage, for example 5% or 1%. Hence, the loss is bounded from above in a fraction of α of all possible market scenarios and, for small values of $1 - \alpha$, this fraction is close to 1. They specify their definition by explaining the value at risk at level $1 - \alpha$ for some wealth process $(X(t))_{t \in [0,T]}$ over the time period [0,T] as the real number $\operatorname{VaR}_{1-\alpha}(X(T) - X(0))$ for which

$$\mathbb{P}\Big(X(0) - X(T) \le \operatorname{VaR}_{1-\alpha}(X(T) - X(0))\Big) = \alpha.$$

¹Note that α is usually chosen as the level of the value at risk instead of $1 - \alpha$. However, it simplifies the notation for the later analysis to choose $1 - \alpha$ instead of α as the level of the value at risk.

If the cumulative distribution function of X(T) - X(0) is strictly increasing and continuous on its support, this is simply the α -quantile of -(X(T) - X(0)). Note that if the cumulative distribution function is strictly increasing, the upper and lower quantiles² are equal and we simply refer to them as quantiles. Thus, the definition of Basak and Shapiro (2001) coincides with the standard definition for the payoff X(T) - X(0) if the corresponding cumulative distribution function is strictly increasing and continuous on its support.

Basak and Shapiro (2001) chose the following approach to incorporate risk management concerns into the standard utility maximization problem. To keep the risk below a certain level, they bounded the value at risk from above by some endogenously given constant K > 0. Thus, they considered the constraint

$$\operatorname{VaR}_{1-\alpha}(X(T) - X(0)) \le X(0) - K_{2}$$

which entails

$$\alpha = \mathbb{P}\Big(X(0) - X(T) \le \operatorname{VaR}_{1-\alpha}(X(T) - X(0))\Big) \le \mathbb{P}(X(T) \ge K).$$

Therefore, by including the additional constraint $\mathbb{P}(X(T) \ge K) \ge \alpha$ into the utility maximization problem, they bounded the risk arising from the portfolio while maximizing the expected utility of the terminal wealth. In support of the risk management feature of the problem, they explain that the constraint acts as a partial portfolio insurance which lies somewhere between the benchmark cases of the classical problem without the constraint ($\alpha = 0$) and the case of a portfolio insurer ($\alpha = 1$).

In order to adapt the model of Basak and Shapiro (2001) to include relative concerns, we replace the constant K in the optimization problem of agent i by a weighted arithmetic mean of the other n-1 agents' terminal wealth. Thus, we incorporate the constraint

$$\mathbb{P}\left(X_i(T) \ge \sum_{j \ne i} \beta_j X_j(T)\right) \ge \alpha_i \tag{8.1}$$

into the optimization problem of agent i, where $\alpha_i \in [0, 1]$ and $\beta_j \in (0, 1)$ for all $i, j \in \{1, \ldots, n\}$. By $X_j(T)$ we denote the terminal wealth of agent $j \in \{1, \ldots, n\}$. Under normal market conditions, i.e., in a fraction of α_i of the possible market scenarios, agent i attains a terminal wealth which is at least as large as the weighted average of her competitors' terminal wealth. As opposed to the optimization problems considered in Chapters 3 – 7, the objective function of agent i is given by the standard objective, i.e., the expected utility of her terminal wealth only.

The parameters β_j in (8.1) are custom to each agent j, but not to agent i, meaning that the weight assigned to agent j is the same in the optimization of agent i for any $i \neq j$. A possible choice is $\beta_j = \frac{1}{n-1}$, but we allow for more generality at this point. It is, for example, possible to consider weights in terms of the initial capital invested by the agents so that a larger initial investment goes along with a larger weight assigned to the corresponding agent. In contrast to the parameters

²For a random variable X and $\lambda \in (0, 1)$, the lower and upper λ -quantiles are $q_X^-(\lambda) = \inf\{x \in \mathbb{R} \mid \mathbb{P}(X \le x) \ge \lambda\}$ and $q_X^+(\lambda) = \inf\{x \in \mathbb{R} \mid \mathbb{P}(X \le x) > \lambda\}$, respectively (see, e.g., Föllmer and Schied, 2016).

 $\beta_j, j = 1, ..., n$, the level $\alpha_i \in [0, 1]$ is chosen by agent *i* herself and has a similar interpretation as the competition weight θ_i used in earlier chapters. If α_i is chosen close to 1, agent *i* wants to insure her terminal wealth against the other agents' wealth in almost all possible scenarios, while a value α_i close to 0 implies that she does not care as much about her performance with respect to the others. However, a large choice of α_i indicates risk aversion (since the agent bounds her risk in a larger percentage of cases) whereas a large choice of θ_i turns out to be more risk-seeking.

To conclude, let us give an overview of the chapter. In Section 8.1, we introduce the underlying financial market and state the optimization problem for n agents using the VaR-based constraint (8.1). We present the optimal solution to the best response problem in terms of terminal wealth in Section 8.2. Due to the complicated structure of the optimal solution, we restrict the set of strategies to be able to find Nash equilibria. This gives rise to an n-dimensional fixed point problem. In Section 8.3, we solve the fixed point problem for two agents. Afterwards, the unique fixed point for a general number of agents is found and discussed in Section 8.4.

8.1. PROBLEM FORMULATION

In order to formalize the optimization problem described above, let us specify the underlying financial market first. The financial market is the same as the one used in Chapter 6. Nevertheless, we repeat the basics at this point. We base our analysis on the semimartingale financial market explained in Subsection 2.3.1. In summary, there are d + 1 assets in which n agents can invest. The assets consist of one riskless bond with zero interest rate and d risky stocks. The stock price processes $(S_k(t))_{t\in[0,T]}$, $k = 1, \ldots, d$, are $L^2(\mathbb{P})$ -semimartingales with càdlàg paths (i.e., the paths are right continuous with left limits). To exclude arbitrage, we require the existence of an equivalent σ -martingale measure \mathbb{Q} . In contrast to Subsection 2.3.1, we make the additional assumption that \mathbb{Q} is the unique equivalent σ -martingale measure. Under this assumption, the associated density process

$$Z_t^{\mathbb{Q}} = \mathbb{E}\left[\frac{\mathrm{d}\mathbb{Q}}{\mathrm{d}\mathbb{P}}\Big|\mathcal{F}_t\right], t \in [0,T],$$

is unique as well. Thus, we write $Z_t \coloneqq Z_t^{\mathbb{Q}}$, $t \in [0, T]$, throughout this chapter. The assumption that \mathbb{Q} is the unique equivalent σ -martingale measure has the important consequence that each claim $X \in L^2(\mathbb{P})$ has a *unique* fair price given by

$$\mathbb{E}_{\mathbb{Q}}[X] = \mathbb{E}\left[Z_T X\right].$$

Let us now formalize the optimization problem motivated in the introduction to this chapter. Let $\beta_i \in (0, 1)$ and $\alpha_i \in [0, 1]$ for all $i \in \{1, \ldots, n\}$. The constant β_j describes a weight assigned to the *j*-th agent's wealth in the stochastic constraint and α_i describes the probability assigned to the stochastic constraint in the optimization problem of agent *i*. Each agent *i* uses an Inada utility function $U_i : (0, \infty) \to \mathbb{R}$ to measure her preferences (see Definition 2.11). Note that the requirements on the utility functions are a lot stricter than the ones considered in Chapter 3. First, U_i is required to satisfy the Inada conditions and second, we only allow for utility functions defined on the strictly positive real numbers. However, the proofs presented in the current chapter rely heavily on both assumptions imposed on U_1, \ldots, U_n . Fortunately, this requirement is not too restricting as there are commonly used utility functions such as the natural logarithm or power utility functions that satisfy the conditions. Finally, we assume that each agent *i* is equipped with an initial capital $x_0^i > 0$, $i = 1, \ldots, n$.

Following the motivation in the introduction to this chapter, agent $i \in \{1, ..., n\}$ aims to solve the optimization problem

$$\begin{cases} \max_{\varphi^i \in \mathcal{A}} \mathbb{E}\left[U_i(X_T^{i,\varphi^i})\right],\\ \text{s.t.} \quad X_T^{i,\varphi^i} = x_0^i + (\varphi^i \cdot S)_T, \ \mathbb{P}\left(X_T^{i,\varphi^i} \ge \sum_{j \neq i} \beta_j X_T^{j,\varphi^j}\right) \ge \alpha_i. \end{cases}$$

Similar to Chapter 6, we only consider the terminal wealth, not the replicating strategy. If the financial market is additionally assumed to be complete, replicating strategies do exist for every terminal wealth determined in this chapter. However, we do not consider the replicating strategies and leave this question open for future research. Thus, we state the optimization problem of agent $i \in \{1, \ldots, n\}$ is given as

$$\begin{cases} \max_{X_i} \mathbb{E}\left[U_i(X_i)\right], \\ \text{s.t.} \quad X_i \text{ is } \mathcal{F}_T - \text{measurable}, \ \mathbb{E}\left[Z_T X_i\right] \le x_0^i, \ \mathbb{P}\left(X_i \ge \sum_{j \ne i} \beta_j X_j\right) \ge \alpha_i, \end{cases}$$
(8.2)

where X_j denotes the terminal wealth of investor $j \in \{1, \ldots, n\}$.

Remark 8.1. In Chapter 3, we displayed a method to solve portfolio optimization problems for relative investors who include their relative concerns into their expected utility function via the additive relative performance metric. Because of the arithmetic mean in the constraint (8.1), it would be possible to adjust (8.2) to make the method from Chapter 3 applicable. However, in order to be able to reduce the problem to an auxiliary single-agent problem, we would also have to adjust the argument in the objective function of (8.2) to match the constraint. Since this does not fit the motivation presented in the introduction to this chapter, we did not choose this approach. However, the resulting optimization problem seems more tractable when compared to (8.2). \diamond

8.2. Discussion of terminal wealth

Let us consider the optimization problem (8.2) faced by agent $i \in \{1, ..., n\}$. In the following, we state and prove the optimal solution to (8.2) for agent i while the terminal wealth X_j , $j \neq i$, of the other agents is assumed to be fixed. To simplify notation, we introduce the abbreviation

$$X_{\beta}^{-i} \coloneqq \sum_{j \neq i} \beta_j X_j, \, i \in \{1, \dots, n\}.$$

Now the following proposition contains the optimal terminal wealth for agent i in the optimization problem (8.2).

Proposition 8.2. Let $\alpha_j \in [0,1]$, $\beta_j \in (0,1)$, and $x_0^j > 0$, j = 1, ..., n. Further, let $i \in \{1, ..., n\}$ and assume that X_j , $j \neq i$, are fixed and \mathbb{P} -almost surely positive. Moreover, assume that U_i : $(0,\infty) \to \mathbb{R}$ is an Inada utility function and let $I_i := (U_i')^{-1}$. Define the function $g_i : (0,\infty) \to \mathbb{R}$ by

$$g_i(Z) \coloneqq U_i(I_i(Z)) - ZI_i(Z) + ZX_{\beta}^{-i}$$

$$(8.3)$$

and define

$$X_{i}^{*} = \begin{cases} I_{i} \left(\lambda_{i}^{(1)} Z_{T} \right), & \lambda_{i}^{(1)} Z_{T} \leq U_{i}^{\prime} (X_{\beta}^{-i}), \\ X_{\beta}^{-i}, & U_{i}^{\prime} (X_{\beta}^{-i}) < \lambda_{i}^{(1)} Z_{T} \leq \bar{\xi}_{i}, \\ I_{i} \left(\lambda_{i}^{(1)} Z_{T} \right), & \lambda_{i}^{(1)} Z_{T} > \bar{\xi}_{i}, \end{cases}$$

$$(8.4)$$

where $\bar{\xi}_i$ is chosen such that $g_i(\bar{\xi}_i) = \lambda_i^{(2)} + U_i(X_{\beta}^{-i})$ and $\bar{\xi}_i > U_i'(X_{\beta}^{-i})$ P-almost surely, and $\lambda_i^{(1)}, \lambda_i^{(2)} > 0$ are chosen with respect to $\mathbb{E}[Z_T X_i^*] = x_0^i$ and $\mathbb{P}(X_i^* \ge X_{\beta}^{-i}) = \alpha_i$. We assume that $\bar{\xi}_i, \lambda_i^{(1)}$, and $\lambda_i^{(2)}$ exist. Then X_i^* is the optimal solution to (8.2).

Proof. Let $i \in \{1, ..., n\}$ be arbitrary but fixed and define

$$L(X, z_1, z_2) \coloneqq U_i(X) - z_1 X + z_2 \mathbb{1}\left\{X \ge X_\beta^{-i}\right\}.$$

For fixed $z_1, z_2 > 0$, the function $X \mapsto L(X, z_1, z_2)$ is increasing if, and only if, $X \leq I_i(z_1)$. Moreover, $L(\cdot, z_1, z_2)$ is piecewise concave with a jump located at X_{β}^{-i} . Therefore, the maximizer is located either at $I_i(z_1)$ or at X_{β}^{-i} . It turns out that

$$X_{i} = I_{i}(z_{1})\mathbb{1}\left\{z_{1} \leq U_{i}'(X_{\beta}^{-i})\right\} + X_{\beta}^{-i}\mathbb{1}\left\{U_{i}'(X_{\beta}^{-i}) < z_{1} \leq \xi_{z}\right\} + I_{i}(z_{1})\mathbb{1}\left\{z_{1} > \xi_{z}\right\},$$

is the unique maximizer of $X \mapsto L(X, z_1, z_2)$, where ξ_z is chosen such that $g_i(\xi_z) = U_i(X_{\beta}^{-i}) + z_2$ and $\xi_z > U'_i(X_{\beta}^{-i})$ P-almost surely. At this point, we assume that such a value ξ_z exists. To prove this assertion, we use a case distinction with respect to z_1 . First, let $z_1 \leq U'_i(X_{\beta}^{-i})$ and thus, $I_i(z_1) \geq X_{\beta}^{-i}$. Since the function g_i from (8.3) is increasing if, and only if, $Z \geq U'_i(X_{\beta}^{-i})$, we obtain

$$\begin{split} L(I_i(z_1), z_1, z_2) &= U_i(I_i(z_1)) - z_1 I(z_1) + z_2 \\ &= g_i(z_1) - z_1 X_{\beta}^{-i} + z_2 \\ &\geq g_i(U_i'(X_{\beta}^{-i})) - z_1 X_{\beta}^{-i} + z_2 \\ &= U_i(X_{\beta}^{-i}) - z_1 X_{\beta}^{-i} + z_2 = L(X_{\beta}^{-i}, z_1, z_2). \end{split}$$

For $U'_i(X_{\beta}^{-i}) < z_1 \leq \xi_z$, it follows

$$L(X_{\beta}^{-i}, z_{1}, z_{2}) = U_{i}(X_{\beta}^{-i}) - z_{1}X_{\beta}^{-i} + z_{2}$$

$$= g_{i}(\xi_{z}) - z_{1}X_{\beta}^{-i}$$

$$\geq g_{i}(z_{1}) - z_{1}X_{\beta}^{-i}$$

$$= U_{i}(I_{i}(z_{1})) - z_{1}I(z_{1}) = L(I(z_{1}), z_{1}, z_{2}),$$

(8.5)

using the monotonicity of g_i and the definition of ξ_z . Finally, if $z_1 > \xi_z$, the reverse of (8.5) holds. Thus, X_i is the unique maximizer of $L(\cdot, z_1, z_2)$.

To prove that X_i^* from (8.4) is optimal for (8.2), assume that \tilde{X} is another random variable satisfying the constraints of (8.2). Then it follows

$$\mathbb{E}\left[U_{i}(X_{i}^{*})\right] - \mathbb{E}\left[U_{i}(\widetilde{X})\right]$$

$$= \mathbb{E}\left[U_{i}(X_{i}^{*})\right] - \mathbb{E}\left[U_{i}(\widetilde{X})\right] - \lambda_{i}^{(1)}x_{0}^{i} + \lambda_{i}^{(1)}x_{0}^{i} + \lambda_{i}^{(2)}\alpha_{i} - \lambda_{i}^{(2)}\alpha_{i}$$

$$\geq \mathbb{E}\left[U_{i}(X_{i}^{*}) - \lambda_{i}^{(1)}Z_{T}X_{i}^{*} + \lambda_{i}^{(2)}\mathbb{1}\left\{X_{i}^{*} \geq X_{\beta}^{-i}\right\}\right] - \mathbb{E}\left[U_{i}(\widetilde{X}) - \lambda_{i}^{(1)}Z_{T}\widetilde{X} + \lambda_{i}^{(2)}\mathbb{1}\left\{\widetilde{X} \geq X_{\beta}^{-i}\right\}\right]$$

$$= \mathbb{E}\left[L\left(X_{i}^{*}, \lambda_{i}^{(1)}Z_{T}, \lambda_{i}^{(2)}\right) - L\left(\widetilde{X}, \lambda_{i}^{(1)}Z_{T}, \lambda_{i}^{(2)}\right)\right]$$

$$\geq 0,$$

since, by construction and the previous observation, X_i^* satisfies the constraints from (8.2) with equality and maximizes the function $L\left(\cdot, \lambda_i^{(1)} Z_T, \lambda_i^{(2)}\right)$ (pointwise for every $\omega \in \Omega$). This concludes our proof since $\bar{\xi}_i$, $\lambda_i^{(1)}$, and $\lambda_i^{(2)}$ exist by assumption.

The representation (8.4) of the best response for agent *i* brings some serious difficulties. In general, existence and uniqueness of the expressions $\bar{\xi}_i$, $\lambda_i^{(1)}$, $\lambda_i^{(2)}$ are not clear. Moreover, we do not expect to be able to find analytical representations for these expressions, even in special cases like logarithmic utility. Thus, in order to find Nash equilibria for (8.2), we restrict the set of possible terminal wealth, although we might lose optimality compared to the more general set of admissible wealth profiles. Nevertheless, we only consider wealth profiles for agent $i \in \{1, \ldots, n\}$ which are of the form

$$X_{i} = \begin{cases} I_{i}(\lambda_{i}Z_{T}), & I_{i}(\lambda_{i}Z_{T}) \geq X_{\beta}^{-i}, Z_{T} \leq \chi_{\alpha_{i}}, \\ X_{\beta}^{-i}, & I_{i}(\lambda_{i}Z_{T}) < X_{\beta}^{-i}, Z_{T} \leq \chi_{\alpha_{i}}, \\ I_{i}(\lambda_{i}Z_{T}), & Z_{T} > \chi_{\alpha_{i}}, \end{cases}$$
(8.6)

where χ_{α_i} is the α_i -quantile of Z_T and $X_{\beta}^{-i} = \sum_{j \neq i} \beta_j X_j$, $i = 1, \ldots, n$. The random variables X_j , $j \neq i$, describe the terminal wealth of the (other) agents $j \neq i$. The constant $\lambda_i > 0$ is chosen so that $\mathbb{E}[Z_T X_i^*] = x_0^i$. Thus, λ_i is often called Lagrange multiplier. For now, we assume that the parameters of the model are chosen such that λ_i exists and is unique. In general, this is the case if the initial capital is sufficiently large. In Section 8.3 as well as Section C.3 in the appendix, we give necessary and sufficient conditions for the existence and an explicit representation of the unique λ_i in the special case of logarithmic utility and $n \in \{2, 3\}$. Moreover, it should be noted that, by construction, X_i satisfies the constraints of (8.2).

Remark 8.3. a) The structure of (8.6) is motivated by the unique optimal terminal wealth found by Basak and Shapiro (2001). They consider the optimization problem

$$\begin{cases} \max_X \mathbb{E}[U(X)], \\ \text{s.t.} \quad \mathbb{E}[Z_T X] \le x_0, \ \mathbb{P}(X \ge K) \ge \alpha, \end{cases}$$
(8.7)

for constants K > 0, $x_0 > 0$, and $\alpha \in [0, 1]$, and an Inada utility function $U : (0, \infty) \to \mathbb{R}$. They prove that the optimal solution to (8.7) takes the form

$$X^* = \begin{cases} I(\lambda Z_T), & Z_T \leq \frac{U'(K)}{\lambda}, \\ K, & \frac{U'(K)}{\lambda} < Z_T \leq \chi_{\alpha}, \\ I(\lambda Z_T), & Z_T > \chi_{\alpha}, \end{cases}$$
(8.8)

where χ_{α} denotes the α -quantile of Z_T and I denotes the inverse of the first order derivative U' of U. Moreover, $\lambda > 0$ is chosen with respect to the budget equation $\mathbb{E}[Z_T X^*] = x_0$.

b) Due to the structure of the optimal terminal wealth (8.8), which divides the solution into three cases depending on the value of the state price density Z_T , Basak and Shapiro (2001) call these three cases good, intermediate, and bad states. The good states are associated to smaller values of Z_T while the bad states correspond to larger values of Z_T . However, this interpretation is not based on the optimal solution to the VaR-based problem (8.7) alone. It can also be justified in more generality. If, for example, we take a standard Black-Scholes market with one stock, one riskless bond with zero interest rate, and constant market parameters (see, e.g., Eberlein and Kallsen, 2019, Examples 9.1 and 9.17), the state price density at time T is given by

$$Z_T = \exp\left(-\frac{\mu}{\sigma}W_T - \frac{1}{2}\frac{\mu^2}{\sigma^2}T\right).$$

The stock value at time T is given by

$$S_T = \exp\left(\left(\mu - \frac{\sigma^2}{2}\right)T + \sigma W_T\right),$$

which can be rewritten as

$$S_T = \exp\left(-\frac{\sigma^2}{\mu}\log(Z_T) + \frac{1}{2}\left(\mu - \sigma^2\right)T\right).$$

Hence, large values of Z_T result in small stock values which justifies the association of large values of Z_T with bad market states.

 \Diamond

Since we aim to find Nash equilibria in the class of terminal wealth profiles of the form (8.6), we need to solve the emerging *n*-dimensional fixed point problem to find explicit representations of the components X_1, \ldots, X_n in the Nash equilibrium. The search for and discussion of the *n*-dimensional fixed point is the focus of the subsequent Sections 8.3 and 8.4.

8.3. Solution of the fixed point problem for two agents

In this section, we set n = 2 and determine the unique two-dimensional Nash equilibrium in the class of wealth profiles of the form (8.6). Solving the fixed point problem for general n is quite complicated, so for illustration we solve it for only two agents at first. Afterwards, Section 8.4 gives the general solution for n agents.

The following theorem displays the unique solution to the fixed point problem (8.6) for two agents.

Theorem 8.4. Let $U_i : (0, \infty) \to \mathbb{R}$ be Inada utility functions, i = 1, 2. Moreover, for $\alpha_i \in [0, 1]$, let χ_{α_i} describe the α_i -quantile of Z_T and let $\beta_i \in (0, 1)$, i = 1, 2. Define

$$X_{1}^{*}(\lambda_{1},\lambda_{2}) = \begin{cases} I_{1}(\lambda_{1}Z_{T}), & I_{1}(\lambda_{1}Z_{T}) \geq \beta_{2}I_{2}(\lambda_{2}Z_{T}), Z_{T} \leq \chi_{\alpha_{1}}, \\ \beta_{2}I_{2}(\lambda_{2}Z_{T}), & I_{1}(\lambda_{1}Z_{T}) < \beta_{2}I_{2}(\lambda_{2}Z_{T}), Z_{T} \leq \chi_{\alpha_{1}}, \\ I_{1}(\lambda_{1}Z_{T}), & Z_{T} > \chi_{\alpha_{1}}, \end{cases}$$
$$X_{2}^{*}(\lambda_{1},\lambda_{2}) = \begin{cases} I_{2}(\lambda_{2}Z_{T}), & I_{2}(\lambda_{2}Z_{T}) \geq \beta_{1}I_{1}(\lambda_{1}Z_{T}), Z_{T} \leq \chi_{\alpha_{2}}, \\ \beta_{1}I_{1}(\lambda_{1}Z_{T}), & I_{2}(\lambda_{2}Z_{T}) < \beta_{1}I_{1}(\lambda_{1}Z_{T}), Z_{T} \leq \chi_{\alpha_{2}}, \\ I_{2}(\lambda_{2}Z_{T}), & Z_{T} > \chi_{\alpha_{2}}, \end{cases}$$

and assume that the system

$$\mathbb{E}[Z_T X_i^*(\lambda_1, \lambda_2)] = x_0^i, \ i = 1, 2, \tag{8.9}$$

has a unique solution $(\lambda_1^*, \lambda_2^*) \in (0, \infty)^2$. Then $X_i^*(\lambda_1^*, \lambda_2^*)$, i = 1, 2, is the unique Nash equilibrium for (8.2) in the class of wealth profiles of the form (8.6).

- Remark 8.5. a) Since $\beta_1, \beta_2 \in (0, 1)$ and I_1 and I_2 take only strictly positive values, it cannot happen that $I_1(\lambda_1 Z_T) < \beta_2 I_2(\lambda_2 Z_T)$ and $I_2(\lambda_2 Z_T) < \beta_1 I_1(\lambda_1 Z_T)$ hold at the same time. Thus, we notice that for any valid choice of parameters, utility functions, and for any realization of Z_T , at least one of the random variables X_j^* is given by $I_j(\lambda_j Z_T)$. Note that, although the structure is the same, $I_j(\lambda_j Z_T)$ does not necessarily describe the solution of the classical problem without the VaR-constraint, since the Lagrange multiplier might be different from the one in the classical problem. It is also possible that both agents obtain the classical solution in the Nash equilibrium. This is the case if $I_1(\lambda_1 Z_T)$ and $I_2(\lambda_2 Z_T)$ are "close" in the sense that $\beta_2 I_2(\lambda_2 Z_T) \leq I_1(\lambda_1 Z_T) \leq \beta_1^{-1} I_2(\lambda_2 Z_T)$. This observation becomes more tangible in the case where both investors use the natural logarithm as their utility function (see Remark 8.10).
- b) The search for a solution of the system (8.9) of equations consists of two steps. First, we need to determine the expected value to receive a two-dimensional system of equations. Afterwards, we need to solve the emerging system of equations (which is most likely nonlinear) and make sure that there exists an admissible solution in the sense that $\lambda_i^* > 0$, i = 1, 2. If, for example, $x_0^j \geq \beta_{3-j} x_0^{3-j}$, j = 1, 2, we can ensure that for any fixed $\lambda_j > 0$ the equation $\mathbb{E}[Z_T X_{3-j}^*(\lambda_1, \lambda_2)] = x_0^{3-j}$ has a unique solution λ_{3-j}^* , depending on λ_j , using the monotonicity and continuity in terms of λ_{3-j} and the intermediate value theorem. However, this does not ensure that the emerging two-dimensional system of equations is (uniquely) solvable. It should also be noted that this condition is sufficient, but in general not necessary. Lemma 8.8 gives necessary and sufficient conditions for the parameter choice in the case of logarithmic utility for both investors.

Before we provide a proof of Theorem 8.4, we apply the result to the case that both agents use power utility functions. The example gives us more insight into the case distinction in the Nash equilibrium (X_1^*, X_2^*) . Example 8.6. Suppose that both agents use power utility functions with parameters $\delta_i > 0$, $\delta_i \neq 1$, i.e.,

$$U_i(x) = \left(1 - \frac{1}{\delta_i}\right)^{-1} x^{1 - \frac{1}{\delta_i}}, \ I_i(x) = x^{-\delta_i}, \ x > 0, \ i = 1, 2.$$

Assume without loss of generality that $\delta_2 > \delta_1$. Then

$$I_1(\lambda_1 Z_T) \ge \beta_2 I_2(\lambda_2 Z_T) \iff Z_T \ge \left(\beta_2 \frac{\lambda_1^{\delta_1}}{\lambda_2^{\delta_2}}\right)^{\frac{1}{\delta_2 - \delta_1}} =: z_1,$$

$$I_2(\lambda_2 Z_T) \ge \beta_1 I_1(\lambda_1 Z_T) \iff Z_T \le \left(\beta_1 \frac{\lambda_2^{\delta_2}}{\lambda_1^{\delta_1}}\right)^{\frac{1}{\delta_1 - \delta_2}} =: z_2.$$

Since $\beta_1\beta_2 < 1$ and $\lambda_1, \lambda_2 > 0$, we obtain $z_1 < z_2$. Hence, we obtain the Nash equilibrium

$$\begin{aligned} X_1^* &= \beta_2 I_2(\lambda_2 Z_T) \mathbb{1} \{ Z_T < z_1 \land \chi_{\alpha_1} \} + I_1(\lambda_1 Z_T) \mathbb{1} \{ Z_T \ge z_1 \land \chi_{\alpha_1} \}, \\ X_2^* &= I_2(\lambda_2 Z_T) \mathbb{1} \{ Z_T \le z_2 \land \chi_{\alpha_2} \} + \beta_1 I_1(\lambda_1 Z_T) \mathbb{1} \{ z_2 \land \chi_{\alpha_2} < Z_T \le \chi_{\alpha_2} \} \\ &+ I_2(\lambda_2 Z_T) \mathbb{1} \{ Z_T > \chi_{\alpha_2} \}. \end{aligned}$$

Note that $\delta_2 > \delta_1$ has the interpretation of investor 2 being more risk-taking than investor 1. If $\delta_1 = \delta_2 = \delta$, then

$$I_1(\lambda_1 Z_T) \ge \beta_2 I_2(\lambda_2 Z_T) \iff \lambda_2 \ge \beta_2^{1/\delta} \lambda_1,$$

$$I_2(\lambda_2 Z_T) \ge \beta_1 I_1(\lambda_1 Z_T) \iff \lambda_1 \ge \beta_1^{1/\delta} \lambda_2,$$

which shows that the case distinction does not depend on Z_T but solely on the parameters chosen by the investors. Hence, we obtain the following case distinction for the Nash equilibrium (X_1^*, X_2^*) .

a) If $\lambda_1 \ge \beta_1^{1/\delta} \lambda_2$ and $\lambda_2 \ge \beta_2^{1/\delta} \lambda_1$, then

$$X_1^* = I_1(\lambda_1 Z_T), X_2^* = I_2(\lambda_2 Z_T).$$

b) If $\lambda_2 < \beta_2^{1/\delta} \lambda_1$, then $\lambda_1 > \beta_1^{1/\delta} \lambda_2$ and

$$X_1^* = \beta_2 I_2(\lambda_2 Z_T) \mathbb{1} \{ Z_T \le \chi_{\alpha_1} \} + I_1(\lambda_1 Z_T) \mathbb{1} \{ Z_T > \chi_{\alpha_1} \}, X_2^* = I_2(\lambda_2 Z_T).$$

c) If $\lambda_1 < \beta_1^{1/\delta} \lambda_2$, then $\lambda_2 > \beta_2^{1/\delta} \lambda_1$ and $X_1^* = I_1(\lambda_1 Z_T), X_2^* = \beta_1 I_1(\lambda_1 Z_T) \mathbb{1} \{ Z_T \le \chi_{\alpha_2} \} + I_2(\lambda_2 Z_T) \mathbb{1} \{ Z_T > \chi_{\alpha_2} \}.$

Proof (Theorem 8.4). To simplify notation throughout this proof, we abbreviate $I_i = I_i(\lambda_i Z_T)$ and $X_i^* = X_i^*(\lambda_1, \lambda_2), i = 1, 2$. We define functions f_i to rewrite $X_i, i = 1, 2$, from (8.6) as

$$X_1 = f_1(X_2) \coloneqq I_1 \mathbb{1} \{ I_1 \ge \beta_2 X_2, Z_T \le \chi_{\alpha_1} \} + \beta_2 X_2 \mathbb{1} \{ I_1 < \beta_2 X_2, Z_T \le \chi_{\alpha_1} \} + I_1 \mathbb{1} \{ Z_T > \chi_{\alpha_1} \}$$

 \diamond

for agent 1 and

$$X_2 = f_2(X_1) \coloneqq I_2 \mathbb{1} \{ I_2 \ge \beta_1 X_1, Z_T \le \chi_{\alpha_2} \} + \beta_1 X_1 \mathbb{1} \{ I_2 < \beta_1 X_1, Z_T \le \chi_{\alpha_2} \} + I_2 \mathbb{1} \{ Z_T > \chi_{\alpha_2} \}$$

for agent 2. To construct the two-dimensional fixed point for (8.6), we consider the equation

$$f_1(f_2(X_1)) = X_1 \tag{8.10}$$

and prove that the unique solution to (8.10) is given by X_1^* . The analysis of $X_2 = f_2(f_1(X_2))$ proceeds analogously.

We use a case distinction based on Z_T . If $Z_T > \chi_{\alpha_1} \lor \chi_{\alpha_2}$, (8.10) simplifies to $X_1 = I_1$ and thus, the unique solution is given by $X_1 = I_1$ in this case. For the case $\chi_{\alpha_1} \land \chi_{\alpha_2} < Z_T \leq \chi_{\alpha_1} \lor \chi_{\alpha_2}$, we consider two subcases. If $\chi_{\alpha_1} < \chi_{\alpha_2}$, then $Z_T > \chi_{\alpha_1}$ and (8.10) again simplifies to $X_1 = I_1$. If $\chi_{\alpha_1} \geq \chi_{\alpha_2}$, (8.10) reads as

$$X_1 = f_1(I_2) = I_1 \mathbb{1} \{ I_1 \ge \beta_2 I_2 \} + \beta_2 I_2 \mathbb{1} \{ I_1 < \beta_2 I_2 \}.$$

Since the right-hand side is constant in X_1 , it is the unique solution to (8.10) in that case. Finally, assume that $Z_T \leq \chi_{\alpha_1} \wedge \chi_{\alpha_2}$. First, let us simplify the components of $f_1(f_2(X_1))$. We obtain

$$\mathbb{1} \{ I_1 \ge \beta_2 f_2(X_1) \} = \mathbb{1} \{ I_1 \ge \beta_2 I_2, I_2 \ge \beta_1 X_1 \} + \mathbb{1} \{ I_1 \ge \beta_1 \beta_2 X_1, I_2 < \beta_1 X_1 \} \\
= \mathbb{1} \{ I_1 \ge \beta_2 I_2, X_1 \le (\beta_1 \beta_2)^{-1} I_1 \}$$

for the first indicator,

$$\mathbb{1}\left\{I_{1} < \beta_{2}f_{2}(X_{1})\right\} = \mathbb{1}\left\{I_{1} < \beta_{2}I_{2}, I_{2} \ge \beta_{1}X_{1}\right\} + \mathbb{1}\left\{I_{1} < \beta_{1}\beta_{2}X_{1}, I_{2} < \beta_{1}X_{1}\right\} \\
= \mathbb{1}\left\{I_{1} < \beta_{2}I_{2}, X_{1} \le \beta_{1}^{-1}I_{2}\right\} + \mathbb{1}\left\{X_{1} > \max\left\{\beta_{1}^{-1}I_{2}, (\beta_{1}\beta_{2})^{-1}I_{1}\right\}\right\}$$

for the second indicator, and thus,

$$f_2(X_1)\mathbb{1}\left\{I_1 < \beta_2 f_2(X_1)\right\}$$

= $I_2\mathbb{1}\left\{I_1 < \beta_2 I_2, X_1 \le \beta_1^{-1} I_2\right\} + \beta_1 X_1\mathbb{1}\left\{X_1 > \max\left\{\beta_1^{-1} I_2, (\beta_1 \beta_2)^{-1} I_1\right\}\right\}$

Therefore, (8.10) simplifies to

$$X_{1} = I_{1} \mathbb{1} \left\{ I_{1} \ge \beta_{2} I_{2}, X_{1} \le (\beta_{1} \beta_{2})^{-1} I_{1} \right\} + \beta_{2} I_{2} \mathbb{1} \left\{ I_{1} < \beta_{2} I_{2}, X_{1} \le \beta_{1}^{-1} I_{2} \right\} + \beta_{1} \beta_{2} X_{1} \mathbb{1} \left\{ X_{1} > \max \left\{ \beta_{1}^{-1} I_{2}, (\beta_{1} \beta_{2})^{-1} I_{1} \right\} \right\}.$$
(8.11)

If $I_1 \ge \beta_2 I_2$, the unique solution to (8.11) is given by $X_1 = I_1$, whereas for $I_1 < \beta_2 I_2$, the unique solution is given by $X_1 = \beta_2 I_2$. This is based on the fact that, in both cases, the right-hand side of (8.11) is constant for small values of X_1 and linear with slope between 0 and 1 for large values of X_1 . In both cases, the unique point of intersection with the identity function is located in the interval where the right-hand side is constant.

To summarize, the unique solution to (8.10) is given by

$$X_1 = I_1 \mathbb{1} \{ I_1 \ge \beta_2 I_2, \ Z_T \le \chi_{\alpha_1} \} + \beta_2 I_2 \mathbb{1} \{ I_1 < \beta_2 I_2, \ Z_T \le \chi_{\alpha_1} \} + I_1 \mathbb{1} \{ Z_T > \chi_{\alpha_1} \}.$$

Using similar arguments for X_2 yields that (X_1^*, X_2^*) from the theorem is the unique fixed point for (8.6).

Remark 8.7. In Section C.2 in the Appendix, the *n*-dimensional fixed point problem (8.6) is reduced to an (n-1)-dimensional fixed point problem. The reduced problem (C.7) can be used to determine the fixed point for n = 2 instead of the previously displayed direct calculation. It is also used to solve the fixed point problem explicitly for n = 3 (see Section C.3 in the appendix). Afterwards, the Lagrange multipliers λ_1 , λ_2 are calculated explicitly if the agents use logarithmic utility functions and the parameters are chosen correctly. Finally, the explicit representations are used to generate some numerical results. \diamond

EXAMPLE: LOGARITHMIC UTILITY

In the following, we consider a special case of Theorem 8.4 where both agents use the natural logarithm as their utility function. In this setting, we explicitly determine the Lagrange multipliers mentioned in Theorem 8.4, which allows us to look at some numerical results.

Theorem 8.4 implies that, if both agents use logarithmic utility, the unique fixed point for (8.6) is given by

$$X_{1}^{*} = \begin{cases} \frac{1}{\lambda_{1}Z_{T}}, & \lambda_{1}\beta_{2} \leq \lambda_{2}, Z_{T} \leq \chi_{\alpha_{1}}, \\ \frac{\beta_{2}}{\lambda_{2}Z_{T}}, & \lambda_{1}\beta_{2} > \lambda_{2}, Z_{T} \leq \chi_{\alpha_{1}}, \\ \frac{1}{\lambda_{1}Z_{T}}, & Z_{T} > \chi_{\alpha_{1}}, \end{cases} \quad X_{2}^{*} = \begin{cases} \frac{1}{\lambda_{2}Z_{T}}, & \lambda_{2}\beta_{1} \leq \lambda_{1}, Z_{T} \leq \chi_{\alpha_{2}}, \\ \frac{\beta_{1}}{\lambda_{1}Z_{T}}, & \lambda_{2}\beta_{1} > \lambda_{1}, Z_{T} \leq \chi_{\alpha_{2}}, \\ \frac{1}{\lambda_{2}Z_{T}}, & Z_{T} > \chi_{\alpha_{2}}. \end{cases}$$
(8.12)

Now we can explicitly determine the Lagrange multipliers λ_1 and λ_2 with respect to the conditions $\mathbb{E}[X_i^*Z_T] = x_0^i$, i = 1, 2, if the parameters are chosen correctly.

Lemma 8.8. Let $x_0^1 > \alpha_1 \beta_2 x_0^2$, $x_0^2 > \alpha_2 \beta_1 x_0^1$, and let $X_i^* = X_i^*(\lambda_1, \lambda_2)$, i = 1, 2, be given by (8.12). Then the unique solution $\lambda_1, \lambda_2 > 0$ to the system

$$\mathbb{E}[X_1^* Z_T] = x_0^1, \ \mathbb{E}[X_2^* Z_T] = x_0^2,$$

is given by

$$\begin{split} \lambda_1 &= \frac{1 - \alpha_1}{x_0^1 - \alpha_1 \beta_2 x_0^2} \mathbb{1} \left\{ x_0^1 < \beta_2 x_0^2 \right\} + \frac{1}{x_0^1} \mathbb{1} \left\{ x_0^1 \ge \beta_2 x_0^2 \right\},\\ \lambda_2 &= \frac{1 - \alpha_2}{x_0^2 - \alpha_2 \beta_1 x_0^1} \mathbb{1} \left\{ x_0^2 < \beta_1 x_0^1 \right\} + \frac{1}{x_0^2} \mathbb{1} \left\{ x_0^2 \ge \beta_1 x_0^1 \right\}. \end{split}$$

A proof can be found in Section C.1 in the appendix. Using Lemma 8.8, the unique solution to the fixed point problem (8.6) can now be given explicitly.

Proposition 8.9. Let $x_0^1 > \alpha_1 \beta_2 x_0^2$, $x_0^2 > \alpha_2 \beta_1 x_0^1$, and $U_i(x) = \log(x)$, $x \in (0, \infty)$, i = 1, 2. Then the unique Nash equilibrium in the class of wealth profiles of the form (8.6) is given as follows.

a) If $x_0^1 \ge \beta_2 x_0^2$ and $x_0^2 \ge \beta_1 x_0^1$, then

$$X_1^* = x_0^1 Z_T^{-1}, \ X_2^* = x_0^2 Z_T^{-1}$$

b) If $x_0^1 < \beta_2 x_0^2$, then $x_0^2 > \beta_1 x_0^1$ and

$$X_1^* = \beta_2 x_0^2 Z_T^{-1} \mathbb{1} \left\{ Z_T \le \chi_{\alpha_1} \right\} + \frac{x_0^1 - \alpha_1 \beta_2 x_0^2}{1 - \alpha_1} Z_T^{-1} \mathbb{1} \left\{ Z_T > \chi_{\alpha_1} \right\}, \quad X_2^* = x_0^2 Z_T^{-1} Z_T^{-1} \mathbb{1} \left\{ Z_T > \chi_{\alpha_1} \right\},$$

c) If $x_0^2 < \beta_1 x_0^1$, then $x_0^1 > \beta_2 x_0^2$ and

$$X_1^* = x_0^1 Z_T^{-1}, \quad X_2^* = \beta_1 x_0^1 Z_T^{-1} \mathbb{1} \left\{ Z_T \le \chi_{\alpha_2} \right\} + \frac{x_0^2 - \alpha_2 \beta_1 x_0^1}{1 - \alpha_2} Z_T^{-1} \mathbb{1} \left\{ Z_T > \chi_{\alpha_2} \right\}.$$

The assertion follows directly by inserting the Lagrange multipliers from Lemma 8.8 into (8.12). Nevertheless, we displayed the proof in Section C.1 in the appendix.

Remark 8.10. Proposition 8.9 reveals an important feature of the Nash equilibrium (X_1^*, X_2^*) . First, we notice that either X_1^* or X_2^* has a discontinuity while the other is given by the solution to the classical problem. The agent whose wealth in the Nash equilibrium is given by the classical solution $x_0^j Z_T^{-1}$ is always "the better one" in the sense that her initial capital is significantly larger than the other investors' initial capital $(x_0^j > \beta_j^{-1} x_0^i > x_0^i, i \neq j)$. If the initial investments x_0^1 and x_0^2 are close in the sense that $\beta_2 x_0^2 \leq x_0^1 \leq \frac{x_0^2}{\beta_1}$, then both investors obtain the terminal wealth $x_0^j Z_T^{-1}$ in the fixed point.

NUMERICAL RESULTS

Proposition 8.9 enables us to consider some numerical results. We use the special case of a classical Black-Scholes model (one stock, constant market parameters) with zero interest rate. Thus, the discounted state price density Z_T is given by (see, e.g., Eberlein and Kallsen, 2019, Examples 9.1 and 9.17)

$$Z_T = \exp\left(-\frac{\mu}{\sigma}W_T - \frac{1}{2}\frac{\mu^2}{\sigma^2}T\right).$$

Hence, Z_T follows a lognormal distribution with parameters ν and τ^2 , where

$$\nu = -\frac{1}{2}\frac{\mu^2}{\sigma^2}T, \ \tau^2 = \frac{\mu^2}{\sigma^2}T.$$

The quantiles χ_{α} , $\alpha \in [0, 1]$, of Z_T are given by (see, e.g., Johnson et al., 1994, p. 213)

$$\chi_{\alpha} = \exp\left(\nu + \tau \cdot \Phi^{-1}(\alpha)\right),$$

where Φ^{-1} describes the quantile function of the standard normal distribution. Throughout our computations we set $\mu = 0.03$, $\sigma = 0.2$, and T = 4. The numerical results are shown below.



Figure 8.3.1.: X_1^* and X_2^* from Proposition 8.9 (solid) in terms of Z_T for $\beta_1 = 0.5$, $\beta_2 = 0.9$, $\alpha_1 = 0.2$, $\alpha_2 = 0.4$, $x_0^1 = 2$, $x_0^2 = 3$, $\mu = 0.03$, $\sigma = 0.2$, T = 4. The dashed and dotted lines mark the classical solution $x_0^1 Z_T^{-1}$ and $(\lambda_1 Z_T)^{-1}$ for λ_1 from Lemma 8.8, respectively. The parameters are chosen such that case b) from Proposition 8.9 is present.



(a) $x_0^1 = 2, x_0^2 = 3, \beta_1 = 0.9, \beta_2 = 0.5$ (case a). (b) $x_0^1 = 2, x_0^2 = 3, \beta_1 = 0.5, \beta_2 = 0.9$ (case b).

Figure 8.3.2.: X_1^* and X_2^* from Proposition 8.9 in terms of Z_T for $\mu = 0.03$, $\sigma = 0.2$, T = 4, $\alpha_1 = 0.2$ and $\alpha_2 = 0.4$.

Figure 8.3.1 gives a first glance at the behavior of the Nash equilibrium (X_1^*, X_2^*) from Proposition 8.9. It shows X_1^* and X_2^* (solid) for the parameter choices $\beta_1 = 0.5$, $\beta_2 = 0.9$, $\alpha_1 = 0.2$, $\alpha_2 = 0.4$, $x_0^1 = 2$, and $x_0^2 = 3$. Note that, due to the choice of parameters, X_1^* and X_2^* are given in part b) of Proposition 8.9. For reference, the figure also displays $(\lambda_1 Z_T)^{-1}$ (dotted) in terms of Z_T . It is important to notice that $(\lambda_1 Z_T)^{-1}$ is not the solution to the classical problem, since the Lagrange multiplier $\lambda_1 = \frac{1-\alpha_1}{x_0^1 - \alpha_1 \beta_2 x_0^2} \approx 0.5479$ is different from the Lagrange multiplier $\lambda = (x_0^1)^{-1} = 0.5$ in the solution to the classical problem. For comparison, we also included the classical solution $x_0^1 Z_T^{-1}$ (dashed) for agent 1. While X_2^* is simply given by the solution to the classical problem, X_1^* and χ_1^* is given by $\beta_2(\lambda_2 Z_T)^{-1}$ and hence, strictly larger than $(\lambda_1 Z_T)^{-1}$. If $Z_T > \chi_{\alpha_1}$ ("bad states"), X_1^* is given by

 $(\lambda_1 Z_T)^{-1}$. We observe that X_1^* is slightly smaller than the classical solution $x_0^1 Z_T^{-1}$ if $Z_T > \chi_{\alpha_1}$, but significantly larger if $Z_T \leq \chi_{\alpha_1}$.

Figure 8.3.2 illustrates cases a) and b) from Proposition 8.9. Due to the symmetry between cases b) and c), only cases a) and b) are illustrated. For the first set of parameters, the conditions of case a) in Proposition 8.9 are satisfied so that X_i^* is given by $(\lambda_i Z_T)^{-1}$, i = 1, 2, for any realization of Z_T . Hence, both X_1^* and X_2^* are continuous in Z_T . For the second set of parameters, the conditions of case b) are satisfied. In that case, X_2^* is again given by $(\lambda_2 Z_T)^{-1}$ whereas X_1^* is given by $\beta_2(\lambda_2 Z_T)^{-1}$ if $Z_T \leq \chi_{\alpha_1}$ and $(\lambda_1 Z_T)^{-1}$ if $Z_T > \chi_{\alpha_1}$, with a discontinuity located at χ_{α_1} .



Figure 8.3.3.: X_1^* from Proposition 8.9 in terms of Z_T for $\beta_1 = 0.1$, $\beta_2 = 0.9$, $\alpha_2 = 0.4$ and different values of α_1 , $x_0^1 = 2$, $x_0^2 = 3$, $\mu = 0.03$, $\sigma = 0.2$, and T = 4. The parameters are chosen such that case b) from Proposition 8.9 is present.

Figures 8.3.3 and 8.3.4 illustrate how the choice of the parameters α_i and β_i , i = 1, 2, affects the component X_1^* of the unique fixed point for two agents. All parameter choices used in the two figures satisfy the conditions of case b) in Proposition 8.9.

Figure 8.3.3 shows two different features of the influence of α_1 . We notice that the location of the discontinuity changes with α_1 , since the discontinuity is located at the quantile χ_{α_1} . We also notice that the value of X_1^* is decreasing in terms of α_1 if $Z_T > \chi_{\alpha_1}$. Hence, if a larger value of α_1 is chosen, more states are insured with respect to agent 2, which outperforms agent 1 in this case. However, in the states in which there is no insurance with respect to agent 2, the terminal wealth is smaller for a larger choice of α_1 . The reason for this observation is the Lagrange multiplier λ_1 which is increasing in α_1 in case b) of Proposition 8.9. Thus, we observe that a larger choice of α_1 is related to agent 1 being more risk-averse. She wants to insure her wealth against the other agent's wealth in a larger fraction of the possible market scenarios. However, she has to "pay the price" for this choice with an even smaller terminal wealth in the "bad" states. This reveals a general criticism of the value at risk – it only limits the probability of losses above a certain threshold; the magnitude of such losses is not taken into account (see, e.g., Föllmer and Schied, 2016, p. 231).



Figure 8.3.4.: X_1^* (solid) and X_2^* (dashed) from Proposition 8.9 in terms of Z_T for different values of β_2 , $\beta_1 = 1 - \beta_2$, $\alpha_1 = 0.2$, $\alpha_2 = 0.4$, $x_0^1 = 2$, $x_0^2 = 3$, $\mu = 0.03$, $\sigma = 0.2$, and T = 4. The parameters are chosen such that case b) from Proposition 8.9 is present.

Figure 8.3.4 illustrates the influence of the parameter β_2 on X_1^* . Here we chose $x_0^1 = 2$ and $x_0^2 = 3$, so that, as long as $\beta_2 > 2/3$, case b) of Proposition 8.9 is present. Note that the parameter β_1 has no effect in this case, since X_1^* does not depend on β_1 , nor does the condition which ensures that case b) is present. Figure 8.3.4 shows that a larger value of β_2 results in a larger value of X_1^* in the "good states" ($Z_T \leq \chi_{\alpha_1}$) and in a smaller value of X_1^* in the "bad states".

8.4. Solution of the fixed point problem for n agents

In this section, we show that, if $\beta_j \leq \frac{1}{n-1}$ for all $j \in \{1, \ldots, n\}$, a fixed point for (8.6) is either unique or all components are equal. Moreover, we prove an important property of the fixed point, which is similar to the case distinction for three investors given in Table C.1 in the appendix. It shows that any possible value of the triple (X_1^*, X_2^*, X_3^*) contains some (at least one, at most three) entries X_j^* that are given by $I_j(\lambda_j Z_T)$ while the other entries are linear combinations of these I_j 's. We provide a similar result for a general number of investors. Finally, if we consider the case $\beta_j = \frac{1}{n-1}$ and $\alpha_j = \alpha \in [0, 1]$ for all $j \in \{1, \ldots, n\}$, we can give a closed-form representation of the unique Nash equilibrium in the class of strategies of the form (8.6) (under an additional assumption which is conjectured to hold for general n and is proven to hold for $n \in \{2, 3\}$). The solution for investor i is given as a maximum of 2^{n-1} terms if $Z_T \leq \chi_{\alpha}$, and by $I_i(\lambda_i Z_T)$ if $Z_T > \chi_{\alpha}$. We also state and justify a conjecture on a statement allowing for more general choices of β_j , $j = 1, \ldots, n$.

Using (8.6), X_i equals $I_i(\lambda_i Z_T)$ if $Z_T > \chi_{\alpha_i}$, or if $Z_T \leq \chi_{\alpha_i}$ as well as $I_i \geq \sum_{j \neq i} \beta_j X_j$, and $\sum_{j \neq i} \beta_j X_j$ for other values of Z_T . Hence, to solve this *n*-dimensional fixed point problem, we can distinguish how many of the *n* investors use the "classical" solution $I_i(\lambda_i Z_T)$. Note that we need to be careful with the expression "classical solution" since the Lagrange multiplier λ_i is not necessarily given as the solution to $\mathbb{E}[Z_T I_i(\lambda_i Z_T)] = x_0^i$ since there might be values of Z_T for which X_i^* is not equal to $I_i(\lambda_i Z_T)$. Hence, although the structure is the same as in the classical solution for some values of Z_T , the value might not be equal to the value of the classical solution.

We can use this idea in the following to solve the *n*-dimensional fixed point problem emerging from (8.6). But first, we need to introduce some notation. For a finite set $S \subset \mathbb{R}$, we denote the set of *k*-combinations of pairwise distinct elements from S by $\binom{S}{k}$, i.e.,

$$\binom{S}{k} \coloneqq \{ (i_1, \dots, i_k) \, | \, i_j \in S, \, j = 1, \dots, k, \, i_1 < i_2 < \dots < i_k \}, \, k \in \{0, 1, \dots, |S|\},$$
$$\binom{S}{k} \coloneqq \emptyset, \, k \notin \{0, 1, \dots, |S|\}.$$

If k = 0, the set $\binom{S}{0}$ contains only the empty tuple (). A standard combinatorial result then states that the number of elements in $\binom{S}{k}$ equals $\binom{|S|}{k}$ if $k \in \{0, \ldots, |S|\}$, where

$$\binom{|S|}{k} = \frac{|S|!}{k!(|S|-k)!}$$

denotes the binomial coefficient (see, e.g., Harris et al., 2008, p. 133). We also use the abbreviation

$$[L] \coloneqq \{1, 2, \dots, L\}, L \in \mathbb{N}.$$

We can then define

$$d_L(j_1, \dots, j_L) \coloneqq 1 - \sum_{k=2}^{L} (k-1) \sum_{(i_1, \dots, i_k) \in \binom{[L]}{k}} \prod_{\ell=1}^{k} \beta_{j_{i_\ell}}$$
(8.13)

and

$$A_{L}(j_{1},...,j_{L}) \coloneqq \begin{pmatrix} 1 & -\beta_{j_{2}} & -\beta_{j_{3}} & \dots & -\beta_{j_{L}} \\ -\beta_{j_{1}} & 1 & -\beta_{j_{3}} & \ddots & -\beta_{j_{L}} \\ \vdots & \ddots & \ddots & \dots & \vdots \\ -\beta_{j_{1}} & \dots & -\beta_{j_{L-2}} & 1 & -\beta_{j_{L}} \\ -\beta_{j_{1}} & -\beta_{j_{2}} & \dots & -\beta_{j_{L-1}} & 1 \end{pmatrix}$$

for some $L \in \{1, \ldots, n\}$, where $A_1(j_1) \coloneqq (1)$. Hence, the entries of $A_L(j_1, \ldots, j_L)$ are given by

$$A_L(j_1,\ldots,j_L)[i,k] = \begin{cases} 1, & i=k, \\ -\beta_{j_k}, & i\neq k. \end{cases}$$

The first goal of this section is to prove the following theorem which presents an n-dimensional analogue of Table C.1.

Theorem 8.11. Let $\beta_j \leq \frac{1}{n-1}$ for all $j \in \{1, \ldots, n\}$ and let the value of Z_T be arbitrary but fixed. If there exists a solution (X_1, \ldots, X_n) to the fixed point problem (8.6), one of the following two cases is present. Either the fixed point satisfies $X_i = X_j$ for all $i, j \in \{1, \ldots, n\}$ or it is unique with the following property: If, for some $L \in \{0, \ldots, n-1\}$ and an L-combination $(j_1, \ldots, j_L) \in {[n] \choose L}$,

$$X_{j_{\ell}} \neq I_{j_{\ell}}(\lambda_{j_{\ell}}Z_T), \ \ell \in \{1, \dots, L\}, \quad X_k = I_k(\lambda_k Z_T), \ k \notin \{j_1, \dots, j_L\},$$

then

$$X_{j_{\ell}} = \frac{\prod_{i=1, i \neq \ell}^{L} (1 + \beta_{j_i})}{d_L(j_1, \dots, j_L)} \sum_{\substack{k=1\\k \notin \{j_1, \dots, j_L\}}}^{n} \beta_k I_k(\lambda_k Z_T)$$

holds for all $\ell \in \{1, \ldots, L\}$.

The proof uses Lemmas 8.12 - 8.14 stated below. Their proofs can be found in Section C.4 in the appendix.

Lemma 8.12. Let $L \in \{1, ..., n\}$ and $(j_1, ..., j_L) \in {\binom{[n]}{L}}$. If $\beta_j \leq \frac{1}{n-1}$ for all $j \in \{1, ..., n\}$, then

$$d_L(j_1,\ldots,j_L) \ge \left(\frac{n}{n-1}\right)^{L-1} \frac{n-L}{n-1}$$

Further, $d_L(j_1, \ldots, j_L) > 0$ holds if, and only if, either $L \in \{1, \ldots, n-1\}$ or L = n and $\prod_{j=1}^n \beta_j < \left(\frac{1}{n-1}\right)^n$.

Lemma 8.13. Assume that $\beta_j \leq \frac{1}{n-1}$ for all $j \in \{1, \ldots, n\}$. Then, if either $L \in \{1, \ldots, n-1\}$, or L = n and $\prod_{j=1}^n \beta_j < \left(\frac{1}{n-1}\right)^n$, the matrix $A_L(j_1, \ldots, j_L)$ is regular with inverse $A_L(j_1, \ldots, j_L)^{-1}$ given by

$$A_{L}(j_{1},\ldots,j_{L})^{-1}[i,k] = \begin{cases} 1 - \sum_{m=2}^{L-1} (m-1) \sum_{\substack{(i_{1},\ldots,i_{m}) \in \binom{\{j_{1},\ldots,j_{L}\} \setminus \{j_{i}\}}{m}} \prod_{\ell=1}^{m} \beta_{i_{\ell}}}{d_{L}(j_{1},\ldots,j_{L})}, & i = k, \\ \frac{\beta_{j_{k}} \prod_{\ell=1,\ell \neq i,k}^{L} (1+\beta_{j_{\ell}})}{d_{L}(j_{1},\ldots,j_{L})}, & i \neq k, \end{cases}$$

for any $(j_1, ..., j_L) \in {[n] \choose L}$. If L = n and $\beta_j = \frac{1}{n-1}$ for all $j \in \{1, ..., n\}$, then $A_n(1, ..., n)$ is not regular.

Lemma 8.14. Let $\beta_j \leq \frac{1}{n-1}$ for all $j \in \{1, \ldots, n\}$. Further, let either $L \in \{1, \ldots, n-1\}$ or L = n and $\prod_{j=1}^n \beta_j < \left(\frac{1}{n-1}\right)^n$. Finally, let $(j_1, \ldots, j_L) \in \binom{[n]}{L}$. Then the sum of the *i*-th row of $A_L(j_1, \ldots, j_L)^{-1}$, $i \in \{1, \ldots, L\}$, is given by

$$\sum_{k=1}^{L} A_L(j_1, \dots, j_L)^{-1}[i, k] = \frac{1}{d_L(j_1, \dots, j_L)} \prod_{\ell=1, \ell \neq i}^{L} (1 + \beta_{j_\ell})$$

Using these three lemmas, we can now prove Theorem 8.11.

Proof (Theorem 8.11). Considering the *n*-dimensional fixed point problem (8.6), any fixed point (X_1, \ldots, X_n) has the property that, for any value of Z_T , X_i is given by either $I_i(\lambda_i Z_T)$ or $\sum_{j \neq i} \beta_j X_j$. To prove the assertion in the theorem, we use a case distinction based on the number M of indices $j \in \{1, \ldots, n\}$ for which $X_j = I_j(\lambda_j Z_T)$ holds.

First, we need to consider the case that $X_i = \sum_{j \neq i} \beta_j X_j$ for all $i \in \{1, \ldots, n\}$, i.e., M = 0. This yields a system of linear equations which can be written as

$$A_n(1,\ldots,n)(X_1,\ldots,X_n)^{\top} = \mathbf{0}_n, \qquad (8.14)$$

where $\mathbf{0}_n$ denotes the *n*-dimensional vector of zeros. If $\prod_{j=1}^n \beta_j < \left(\frac{1}{n-1}\right)^n$, Lemma 8.13 implies that $A_n(1,\ldots,n)$ is regular and thus, the unique solution to this system is given by $\mathbf{0}_n$. However, $X_i = \sum_{j \neq i} \beta_j X_j$ implies that $0 = \sum_{j \neq i} \beta_j X_j > I_i(\lambda_i Z_T)$, which is a contradiction since I_i takes only strictly positive values. Now let $\beta_j = \frac{1}{n-1}$ for all $j \in \{1,\ldots,n\}$. Then (8.14) reads as

$$X_i = \frac{1}{n-1} \sum_{j \neq i} X_j$$
 or, equivalently, $X_i = \frac{1}{n} \sum_{j=1}^n X_j$,

for all $i \in \{1, ..., n\}$. Thus, all components of the fixed point are equal.

Now let $M \in \{1, ..., n\}$ describe the number of indices $j \in \{1, ..., n\}$ with $X_j = I_j(\lambda_j Z_T)$. Further, let $L \coloneqq n - M \in \{0, ..., n - 1\}$ and denote the *L*-tuple of indices ℓ for which³ $X_{\ell} \neq I_{\ell}(\lambda_{\ell} Z_T)$ by $(j_1, ..., j_L)$. For L = 0, $X_j = I_j(\lambda_j Z_T)$ holds for all $j \in \{1, ..., n\}$. Thus, we consider $L \in \{1, ..., n - 1\}$. Hence, for all $\ell \in \{1, ..., L\}$,

$$X_{j_{\ell}} = \sum_{\substack{i=1\\i \neq j_{\ell}}}^{n} \beta_{i} X_{i} = \sum_{\substack{p=1\\p \neq \ell}}^{L} \beta_{j_{p}} X_{j_{p}} + \sum_{\substack{k=1\\k \notin \{j_{1}, \dots, j_{L}\}}}^{n} \beta_{k} I_{k}(\lambda_{k} Z_{T})$$

and we obtain the system of linear equations

$$X_{j_{\ell}} - \sum_{\substack{p=1\\p \neq \ell}}^{L} \beta_{j_{p}} X_{j_{p}} = \sum_{\substack{k=1\\k \notin \{j_{1}, \dots, j_{L}\}}}^{n} \beta_{k} I_{k}(\lambda_{k} Z_{T}), \quad \ell = 1, \dots, L,$$

which we can also write as

$$A_{L}(j_{1},\ldots,j_{L})(X_{j_{1}},\ldots,X_{j_{L}})^{\top} = \sum_{\substack{k=1\\k\notin\{j_{1},\ldots,j_{L}\}}}^{n} \beta_{k}I_{k}(\lambda_{k}Z_{T})\cdot\mathbf{1}_{L},$$
(8.15)

where $\mathbf{1}_L$ denotes the *L*-dimensional vector of ones. Since, for $L \in \{1, \ldots, n-1\}$, $A_L(j_1, \ldots, j_L)$ is regular with known inverse $A_L(j_1, \ldots, j_L)^{-1}$ (see Lemma 8.13), the unique solution to (8.15) is given by

$$(X_{j_1},\ldots,X_{j_L})^{\top} = \sum_{\substack{k=1\\k\notin\{j_1,\ldots,j_L\}}}^n \beta_k I_k(\lambda_k Z_T) \cdot A_L(j_1,\ldots,j_L)^{-1} \mathbf{1}_L.$$

³Note that, although not relevant for our proof, this does imply that $Z_T \leq \chi_{\alpha_{j_\ell}}$ for all $\ell \in \{1, \ldots, L\}$.

Therefore, the ℓ -th entry of the solution is given by

$$X_{j_{\ell}} = \left(\sum_{\substack{k=1\\k \notin \{j_1, \dots, j_L\}}}^{n} \beta_k I_k(\lambda_k Z_T)\right) \cdot \sum_{p=1}^{L} A_L(j_1, \dots, j_L)^{-1}[\ell, p].$$

Combining Lemma 8.13 and Lemma 8.14 yields

$$X_{j_{\ell}} = \frac{\prod_{p=1, p\neq \ell}^{L} (1+\beta_{j_{p}})}{d_{L}(j_{1}, \dots, j_{L})} \cdot \sum_{\substack{k=1\\k\notin\{j_{1},\dots,j_{L}\}}}^{n} \beta_{k}I_{k}(\lambda_{k}Z_{T}).$$

Since we can use this argumentation for any solution of the fixed point problem, it follows that, if there exists a fixed point for (8.6), it is unique or all components are equal. Moreover, it satisfies the asserted property from the theorem.

Remark 8.15. Although the fixed point problem (8.6) contains a case distinction on whether $Z_T \leq \chi_{\alpha}$ or $Z_T > \chi_{\alpha}$, we did not need to separate these two cases in the proof of Theorem 8.11. We just needed to consider how many and which of the indices $j \in \{1, \ldots, n\}$ satisfy $X_j = I_j(\lambda_j Z_T)$, which could be due to $Z_T > \chi_{\alpha_j}$ or to $I_j(\lambda_j Z_T) > \sum_{k \neq j} \beta_k X_k$. This enables us to state the theorem without the assumption $\alpha_i = \alpha$ for all $i \in \{1, \ldots, n\}$.

Theorem 8.11 shows that if there exists a fixed point for (8.6), it is either unique or all components of the fixed point are equal. Moreover, in the case where not all components are equal, it can be expressed implicitly via a case distinction. However, it does not prove the existence of a fixed point or give an explicit representation. In the following, we provide a closed-form representation of the unique fixed point (in a slightly more restrictive setting) and thus, the unique Nash equilibrium in the case that $\alpha_i = \alpha \in [0, 1]$ and $\beta_i = \frac{1}{n-1}$ for all $i \in \{1, \ldots, n\}$. We also state and justify two conjectures about a more general theorem (see Remarks 8.17 and 8.19). The following theorem displays the mentioned closed-form representation of the unique fixed point.

Theorem 8.16. Let $\alpha_j = \alpha \in [0, 1]$ and $\beta_j = \frac{1}{n-1}$ for all $j \in \{1, \ldots, n\}$. Then the \mathbb{P} -almost surely unique Nash equilibrium for (8.2) in the class of wealth profiles X_j , $j = 1, \ldots, n$, of the form (8.6) for which

$$\mathbb{P}\left(X_i = X_j \text{ for all } i, j \in \{1, \dots, n\}\right) = 0$$
(8.16)

holds, is given by (X_1^*, \ldots, X_n^*) for

$$X_{i}^{*} = \max\left\{\{I_{i}(\lambda_{i}Z_{T})\} \cup \bigcup_{L=0}^{n-2} \bigcup_{(j_{1},\dots,j_{L})\in\binom{[n]\setminus\{i\}}{L}} \left\{\frac{\prod_{\ell=1}^{L}(1+\beta_{j_{\ell}})}{d_{L+1}(i,j_{1},\dots,j_{L})} \sum_{\substack{k=1\\k\notin\{i,j_{1},\dots,j_{L}\}}}^{n} \beta_{k}I_{k}(\lambda_{k}Z_{T})\right\}\right\}$$

$$\cdot \mathbb{1}\left\{Z_{T} \leq \chi_{\alpha}\right\} + I_{i}(\lambda_{i}Z_{T})\mathbb{1}\left\{Z_{T} > \chi_{\alpha}\right\},$$
(8.17)

i = 1, ..., n, where we assume that the n-dimensional system $\mathbb{E}[Z_T X_i^*] = x_0^i$, i = 1, ..., n, of equations has a unique solution $(\lambda_1, ..., \lambda_n) \in (0, \infty)^n$.

Remark 8.17. We conjecture that (8.16) holds for any Nash equilibrium for (8.2) in the class of wealth profiles of the form (8.6). Thus, we believe that Theorem 8.16 holds without the restriction

(8.16) to tuples (X_1, \ldots, X_n) with components that are almost surely not identical. For $n \in \{2, 3\}$, we can prove that (8.17) is the unique Nash equilibrium in the class of wealth profiles of the form (8.6) without the additional restriction (8.16) (see Theorem 8.4 and Theorem C.1 in the appendix).

The proof of Theorem 8.16 uses the following lemma.

Lemma 8.18. Let $L \in \{0, ..., n-1\}$ and $(j_1, ..., j_{L+1}) \in {\binom{[n]}{L+1}}$. Moreover, let $\beta_j \leq \frac{1}{n-1}$, j = 1, ..., n. Then

$$d_{L+1}(j_1,\ldots,j_L,j_{L+1}) = (1+\beta_{j_{L+1}})d_L(j_1,\ldots,j_L) - \beta_{j_{L+1}}\prod_{\ell=1}^L (1+\beta_{j_\ell}).$$

The proof can be found in Section C.4 in the appendix.

Proof (Theorem 8.16). The conditions of Theorem 8.16 allow us to use the statement of Theorem 8.11. We only need to consider the case $Z_T \leq \chi_{\alpha}$, because otherwise we already know that X_i^* is given by $I_i(\lambda_i Z_T)$ for all $i \in \{1, \ldots, n\}$. Thus, we assume that $Z_T \leq \chi_{\alpha}$ throughout this proof.

To prove the claimed representation of X_i^* , i = 1, ..., n, we have to show that the *n*-tuple $(X_1^*, ..., X_n^*)$ solves the fixed point problem (8.6). To keep notation simple, we consider i = 1 and assume that X_j^* , j = 2, ..., n, are given by (8.17). Then we need to show that X_1^* can also be written in the form of (8.17). For the remainder of this proof, we abbreviate $I_j = I_j(\lambda_j Z_T)$.

Analogously to the proof of Theorem 8.11, we use a case distinction based on the number $N \in \{0, \ldots, n-1\}$ of indices $j \in \{2, \ldots, n\}$ for which $X_j^* = I_j$ holds. If N = 0, i.e., if $X_j^* \neq I_j(\lambda_j Z_T)$ for all $j \in \{2, \ldots, n\}$, it follows immediately from assumption (8.16) that $X_1^* = I_1(\lambda_1 Z_T)$ P-almost surely. In fact, the proof of Theorem 8.11 implies that, if $X_j^* \neq I_j(\lambda_j Z_T)$ for all $j \in \{1, \ldots, n\}$, it follows that $X_i^* = X_j^*$ for all $i, j \in \{1, \ldots, n\}$, which is P-almost surely not the case using (8.16). Thus, we can assume that there are $N \in \{1, \ldots, n-1\}$ indices $j \in \{2, \ldots, n\}$ for which $X_j^* = I_j$ holds. Further, let $L = n - 1 - N \in \{0, \ldots, n-2\}$ and $j_1, \ldots, j_L \in \{2, \ldots, n\}$ describe the pairwise distinct indices for which $X_{j\ell}^* \neq I_{j\ell}, \ell = 1, \ldots, L$. Then it follows analogously to the proof of Theorem 8.11 that X_1^* is given by either I_1 or

$$\frac{\prod_{\ell=1}^{L} (1+\beta_{j_{\ell}})}{d_{L+1}(1,j_{1},\ldots,j_{L})} \sum_{\substack{k=2\\k\notin\{j_{1},\ldots,j_{L}\}}}^{n} \beta_{k} I_{k}$$

Hence, it remains to show that

$$\max\left\{\{I_{1}\} \cup \bigcup_{M=0}^{n-2} \bigcup_{(m_{1},\dots,m_{M})\in\binom{[n]\setminus\{1\}}{M}} \left\{\frac{\prod_{p=1}^{M}(1+\beta_{m_{p}})}{d_{M+1}(1,m_{1},\dots,m_{M})} \sum_{\substack{r=2\\r\notin\{m_{1},\dots,m_{M}\}}}^{n} \beta_{r}I_{r}\right\}\right\}$$
$$=\max\left\{I_{1}, \frac{\prod_{\ell=1}^{L}(1+\beta_{j_{\ell}})}{d_{L+1}(1,j_{1},\dots,j_{L})} \sum_{\substack{k=2\\k\notin\{j_{1},\dots,j_{L}\}}}^{n} \beta_{k}I_{k}\right\}.$$

In order to do this, we prove the inequality

$$\frac{\prod_{\ell=1}^{L} (1+\beta_{j_{\ell}})}{d_{L+1}(1,j_{1},\ldots,j_{L})} \sum_{\substack{k=2\\k\notin\{j_{1},\ldots,j_{L}\}}}^{n} \beta_{k}I_{k} \ge \frac{\prod_{p=1}^{M} (1+\beta_{m_{p}})}{d_{M+1}(1,m_{1},\ldots,m_{M})} \sum_{\substack{r=2\\r\notin\{m_{1},\ldots,m_{M}\}}}^{n} \beta_{r}I_{r}$$
(8.18)

for all $M \in \{0, \ldots, n-2\}$ and $(m_1, \ldots, m_M) \in \binom{\{2, \ldots, n\}}{M}$ with $(m_1, \ldots, m_M) \neq (j_1, \ldots, j_L)$. Hence, let $M \in \{0, \ldots, n-2\}$ and $(m_1, \ldots, m_M) \in \binom{\{2, \ldots, n\}}{M}$ with $(m_1, \ldots, m_M) \neq (j_1, \ldots, j_L)$ be arbitrary but fixed.

Using that X_j^* , j = 2, ..., n, are given by (8.17) and that $X_k^* = I_k$ for $k \in \{2, ..., n\} \setminus \{j_1, ..., j_L\}$ yields

$$I_k \ge \frac{\prod_{q=1}^{K} (1+\beta_{i_q})}{d_{K+1}(k, i_1, \dots, i_K)} \sum_{\substack{s=1\\s \notin \{k, i_1, \dots, i_K\}}}^n \beta_s I_s$$
(8.19)

for all $K \in \{0, \ldots, n-2\}$, $(i_1, \ldots, i_K) \in {[n] \setminus \{k\}}$, and $k \in \{2, \ldots, n\} \setminus \{j_1, \ldots, j_L\}$. We can use (8.19) to show (8.18) by choosing K and (i_1, \ldots, i_K) accordingly. The choice of K and (i_1, \ldots, i_K) is divided into two cases. The reason is that the lower bound (8.19) only holds if the sum on the right-hand side does not contain the summand $\beta_k I_k$. Hence, we need to distinguish the cases $k \in \{m_1, \ldots, m_M\}$ and $k \notin \{m_1, \ldots, m_M\}$.

First, we consider the indices k contained in $\{m_1, \ldots, m_M\} \setminus \{j_1, \ldots, j_L\}$. Note that the index set $\{m_1, \ldots, m_M\} \setminus \{j_1, \ldots, j_L\}$ is non-empty by assumption. For these indices, we can choose K = M and $\{i_1, \ldots, i_K\} = \{1, m_1, \ldots, m_M\} \setminus \{k\}$, and obtain

$$I_k \ge \frac{\prod_{p \in \{1, m_1, \dots, m_M\} \setminus \{k\}} (1 + \beta_p)}{d_{M+1}(1, m_1, \dots, m_M)} \sum_{\substack{r \in \{1, m_1, \dots, m_M\}}}^n \beta_r I_r.$$
(8.20)

Now we consider the case $k \in \{2, \ldots, n\} \setminus \{j_1, \ldots, j_L, m_1, \ldots, m_M\}$. Then we choose K = M + 1and $\{i_1, \ldots, i_K\} = \{1, m_1, \ldots, m_M\}$ and obtain

$$I_k \ge \frac{\prod_{p \in \{1, m_1, \dots, m_M\}} (1 + \beta_p)}{d_{M+2}(1, m_1, \dots, m_M, k)} \sum_{\substack{r=1\\r \notin \{1, m_1, \dots, m_M, k\}}}^n \beta_r I_r$$

which we can equivalently rewrite as

$$\left(1+\beta_k \frac{\prod_{p\in\{1,m_1,\dots,m_M\}}(1+\beta_p)}{d_{M+2}(1,m_1,\dots,m_M,k)}\right) I_k \ge \frac{\prod_{p\in\{1,m_1,\dots,m_M\}}(1+\beta_p)}{d_{M+2}(1,m_1,\dots,m_K,k)} \sum_{\substack{r=1\\r\notin\{1,m_1,\dots,m_M\}}}^n \beta_r I_r.$$

We can rewrite the constant factor on the left-hand side using Lemma 8.18. It follows

$$1 + \beta_k \frac{\prod_{p \in \{1, m_1, \dots, m_M\}} (1 + \beta_p)}{d_{M+2}(1, m_1, \dots, m_M, k)} = \frac{d_{M+2}(1, m_1, \dots, m_M, k) + \beta_k \prod_{p \in \{1, m_1, \dots, m_M\}} (1 + \beta_p)}{d_{M+2}(1, m_1, \dots, m_M, k)}$$
$$= \frac{(1 + \beta_k) d_{M+1}(1, m_1, \dots, m_M, k)}{d_{M+2}(1, m_1, \dots, m_M, k)},$$

which is strictly positive using Lemma 8.12. Hence, we obtain

$$I_k \ge \frac{\prod_{p \in \{1, m_1, \dots, m_M\}} (1 + \beta_p)}{(1 + \beta_k) d_{M+1}(1, m_1, \dots, m_M)} \sum_{\substack{r=1\\r \notin \{1, m_1, \dots, m_M\}}}^n \beta_r I_r.$$
(8.21)

Now we combine (8.20) and (8.21) to obtain

$$\begin{split} \sum_{\substack{k=2\\k\notin\{j_1,\dots,j_L\}}}^n \beta_k I_k &= \sum_{\substack{k\in\{m_1,\dots,m_M\}\\k\notin\{j_1,\dots,j_L\}}} \beta_k I_k + \sum_{\substack{k=2\\k\notin\{m_1,\dots,m_M\}\\k\notin\{j_1,\dots,j_L\}}}^n \beta_k I_k + \sum_{\substack{k\neq\{m_1,\dots,m_M\}\\k\notin\{j_1,\dots,j_L\}}}^n \beta_k I_k \frac{1}{d_{M+1}(1,m_1,\dots,m_M)} \sum_{\substack{r=1\\r\notin\{1,m_1,\dots,m_M\}}}^n \beta_r I_r \\ &+ \sum_{\substack{k\neq\{m_1,\dots,m_M\}\\k\notin\{j_1,\dots,j_L\}}}^n \beta_k \frac{1}{d_{M+1}(1,m_1,\dots,m_M)} \frac{1+\beta_p}{(1+\beta_k)d_{M+1}(1,m_1,\dots,m_M)} \sum_{\substack{r=1\\r\notin\{1,m_1,\dots,m_M\}}}^n \beta_r I_r \\ &= \frac{1}{d_{M+1}(1,m_1,\dots,m_M)} \frac{1+\beta_p}{d_{M+1}(1,m_1,\dots,m_M)} \sum_{\substack{k=2\\k\notin\{j_1,\dots,j_L\}}}^n \frac{\beta_k}{1+\beta_k} \sum_{\substack{r=1\\r\notin\{1,m_1,\dots,m_M\}}}^n \beta_r I_r. \end{split}$$

Therefore, we obtain

$$\frac{\prod_{\ell=1}^{L}(1+\beta_{j_{\ell}})}{d_{L+1}(1,j_{1},\ldots,j_{L})} \sum_{\substack{k=2\\k\notin\{j_{1},\ldots,j_{L}\}}}^{n} \beta_{k}I_{k} \\
\geq \frac{\prod_{\ell=1}^{L}(1+\beta_{j_{\ell}})}{d_{L+1}(1,j_{1},\ldots,j_{L})} \frac{\prod_{p\in\{1,m_{1},\ldots,m_{M}\}}(1+\beta_{p})}{d_{M+1}(1,m_{1},\ldots,m_{M})} \sum_{\substack{k\notin\{j_{1},\ldots,j_{L}\}}}^{n} \frac{\beta_{k}}{1+\beta_{k}} \sum_{\substack{r=1\\r\notin\{1,m_{1},\ldots,m_{M}\}}}^{n} \beta_{r}I_{r} \\
= \frac{\prod_{\ell=1}^{L}(1+\beta_{j_{\ell}})}{d_{L+1}(1,j_{1},\ldots,j_{L})} (1+\beta_{1}) \sum_{\substack{k=2\\k\notin\{j_{1},\ldots,j_{L}\}}}^{n} \frac{\beta_{k}}{1+\beta_{k}} \cdot \frac{\prod_{p\in\{m_{1},\ldots,m_{M}\}}(1+\beta_{p})}{d_{M+1}(1,m_{1},\ldots,m_{M})} \sum_{\substack{r\notin\{1,m_{1},\ldots,m_{M}\}}}^{n} \beta_{r}I_{r}. (8.22)$$

Finally, we need to check if $\gamma_{L,1} \ge 1$ which then concludes our proof. If we use the assumption $\beta_j \le \frac{1}{n-1}, j = 1, \dots, n$, used throughout the current section, we obtain

$$\gamma_{L,1} = \frac{\prod_{\ell=1}^{L} (1+\beta_{j_{\ell}})}{d_{L+1}(1,j_{1},\dots,j_{L})} (1+\beta_{1}) \sum_{\substack{k=2\\k\notin\{j_{1},\dots,j_{L}\}}}^{n} \frac{\beta_{k}}{1+\beta_{k}}$$
$$\leq \frac{\left(1+\frac{1}{n-1}\right)^{L}}{\frac{n-1-L}{n-1} \left(1+\frac{1}{n-1}\right)^{L}} \cdot \left(1+\frac{1}{n-1}\right) (n-1-L) \cdot \frac{\frac{1}{n-1}}{1+\frac{1}{n-1}} = 1.$$

The lower bound on $d_{L+1}(1, j_1, \dots, j_L)$ is given in Lemma 8.12. Hence, we can only achieve $\gamma_{L,1} \ge 1$ (or, to be more precise, $\gamma_{L,1} = 1$) if we choose $\beta_j = \frac{1}{n-1}$. Then we can make the final

step in (8.22)

$$\frac{\prod_{\ell=1}^{L} (1+\beta_{j_{\ell}})}{d_{L+1}(1,j_{1},\ldots,j_{L})} \sum_{\substack{k=2\\k\notin\{j_{1},\ldots,j_{L}\}}}^{n} \beta_{k}I_{k} \geq \gamma_{L,1} \cdot \frac{\prod_{p\in\{m_{1},\ldots,m_{M}\}} (1+\beta_{p})}{d_{M+1}(1,m_{1},\ldots,m_{M})} \sum_{\substack{r=1\\r\notin\{1,m_{1},\ldots,m_{M}\}}}^{n} \beta_{r}I_{r}$$

$$= \frac{\prod_{p\in\{m_{1},\ldots,m_{M}\}} (1+\beta_{p})}{d_{M+1}(1,m_{1},\ldots,m_{M})} \sum_{\substack{r=1\\r\notin\{1,m_{1},\ldots,m_{M}\}}}^{n} \beta_{r}I_{r},$$

which concludes our proof.

Remark 8.19. Note that, although we assumed that $\beta_j = \frac{1}{n-1}$ for all $j \in \{1, \ldots, n\}$, we displayed X_i^* in terms of β_1, \ldots, β_n . The reason is that we conjecture that (8.17) holds for general $\beta_j \leq \frac{1}{n-1}$, not just for $\beta_j = \frac{1}{n-1}$. We expect that the lower bounds (8.20) and (8.21) in the proof of Theorem 8.16 are too small. The lower bound (8.19) holds for any $K \in \{0, \ldots, n-2\}$ and any $(i_1, \ldots, i_K) \in {[n] \setminus \{k\}}$. Although the specific choice in the proof enables us to prove (8.17) for general n, different choices for $K \in \{0, \ldots, n-2\}$ and $(i_1, \ldots, i_K) \in {[n] \setminus \{k\}}$ might result in better (larger) lower bounds for $I_k, k \in \{2, \ldots, n\} \setminus \{j_1, \ldots, j_L\}$, and hence, also a better lower bound for the sum $\sum_k \beta_k I_k$. This might yield the desired lower bound (8.18) under the weaker assumption $\beta_j \leq \frac{1}{n-1}$ for all $j \in \{1, \ldots, n\}$.

In the case n = 3, we are able to prove the assertion for general $\beta_j \leq \frac{1}{2}$. In the proof of Table C.1, we needed to use all $3 = 2^{n-1} - 1$ lower bounds resulting from the representation of X_i^* as a maximum of four arguments. For general $n \in \mathbb{N}$, it is simply not possible to consider $2^{n-1} - 1$ different lower bounds in order to find the best one. However, we strongly believe that it would give us the desired result for general $\beta_j \leq \frac{1}{n-1}$.

BIBLIOGRAPHY

- Agarwal, V., Daniel, N. D. and Naik, N. Y. (2004). Flows, performance, and managerial incentives in hedge funds, SSRN:424369.
- Allais, M. (1953). Le comportement de l'homme rationnel devant le risque: critique des postulats et axiomes de l'école américaine, *Econometrica* **21**(4): 503–546.
- Almgren, R. and Chriss, N. (2001). Optimal execution of portfolio transactions, *Journal of Risk* **3**(2): 5–39.
- Anthropelos, M., Geng, T. and Zariphopoulou, T. (2022). Competition in fund management and forward relative performance criteria, SIAM Journal on Financial Mathematics 13(4): 1271– 1301.
- Arrow, K. J. (1974). Essays in the Theory of Risk-bearing, Markham economics series, 2. edn, North-Holland, New York.
- Aubin, J.-P. (2003). Optima and Equilibria: An Introduction to Nonlinear Analysis, Vol. 140 of Graduate texts in mathematics, 2. edn, Springer Science & Business Media, Heidelberg.
- Bank, P. and Baum, D. (2004). Hedging and portfolio optimization in financial markets with a large trader, Mathematical Finance: An International Journal of Mathematics, Statistics and Financial Economics 14(1): 1–18.
- Bank, P. and Dolinsky, Y. (2023). Optimal investment with a noisy signal of future stock prices, arXiv:2302.10485.
- Basak, S. and Makarov, D. (2014). Strategic asset allocation in money management, *The Journal* of finance **69**(1): 179–217.
- Basak, S. and Makarov, D. (2015). Competition among portfolio managers and asset specialization, SSRN:1563567.
- Basak, S. and Shapiro, A. (2001). Value-at-risk-based risk management: Optimal policies and asset prices, *The review of financial studies* 14(2): 371–405.

- Bäuerle, N. and Blatter, A. (2011). Optimal control and dependence modeling of insurance portfolios with Lévy dynamics, *Insurance: Mathematics and Economics* 48(3): 398–405.
- Bäuerle, N. and Göll, T. (2023a). Nash equilibria for relative investors via no-arbitrage arguments, Mathematical Methods of Operations Research 97: 1–23.
- Bäuerle, N. and Göll, T. (2023b). Nash equilibria for relative investors with (non)linear price impact, arXiv:2303.18161.
- Bäuerle, N. and Rieder, U. (2011). Markov Decision Processes with Applications to Finance, Universitext, 1. edn, Springer, Berlin Heidelberg.
- Bäuerle, N. and Rieder, U. (2017). Finanzmathematik in diskreter Zeit, Springer.
- Berkelaar, A. B., Kouwenberg, R. and Post, T. (2004). Optimal portfolio choice under loss aversion, *Review of Economics and Statistics* 86(4): 973–987.
- Berkelaar, A., Cumperayot, P. and Kouwenberg, R. (2002). The effect of VaR based risk management on asset prices and the volatility smile, *European Financial Management* 8(2): 139–164.
- Bernstein, D. S. (2009). *Matrix Mathematics: Theory, Facts, and Formulas*, 2. edn, Princeton university press, Princeton.
- Bielagk, J., Lionnet, A. and Reis, G. D. (2017). Equilibrium pricing under relative performance concerns, SIAM Journal on Financial Mathematics 8(1): 435–482.
- Bingham, N. H. and Kiesel, R. (2004). Risk-Neutral Valuation: Pricing and Hedging of Financial Derivatives, Springer Finance, 2. edn, Springer, London.
- Björk, T. (2004). Arbitrage Theory in Continuous Time, 2. edn, Oxford university press.
- Black, F. and Scholes, M. (1973). The pricing of options and corporate liabilities, Journal of political economy 81(3): 637–654.
- Blatter, A. (2009). Optimal control and dependence modeling of portfolios with Lévy dynamics, PhD thesis, Universität Karlsruhe (TH).
- Bo, L. J. and Li, T. Q. (2022). Approximating Nash equilibrium for optimal consumption in stochastic growth model with jumps, *Acta Mathematica Sinica, English Series* **38**: 1621–1642.
- Bo, L., Wang, S. and Yu, X. (2023). Mean field game of optimal relative investment with jump risk, *arXiv:2108.00799*.
- Bo, L., Wang, S. and Zhou, C. (2023). A mean field game approach to optimal investment and risk control for competitive insurers, *SSRN:4471924*.
- Bouchaud, J.-P. (2009). Price impact, arXiv:0903.2428.
- Branger, N., Chen, A., Mahayni, A. and Nguyen, T. (2023). Optimal collective investment: an analysis of individual welfare, *Mathematics and Financial Economics* 17: 101–125.
- Brown, S. J., Goetzmann, W. N. and Park, J. (2001). Careers and survival: Competition and risk in the hedge fund and CTA industry, *The Journal of Finance* **56**(5): 1869–1886.
- Browne, S. (2000). Stochastic differential portfolio games, *Journal of Applied Probability* **37**(1): 126–147.
- Busch, M., Korn, R. and Seifried, F. T. (2013). Optimal consumption and investment for a large investor: An intensity-based control framework, *Mathematical Finance: An International* Journal of Mathematics, Statistics and Financial Economics 23(4): 687–717.
- Carlin, B. I., Lobo, M. S. and Viswanathan, S. (2007). Episodic liquidity crises: Cooperative and predatory trading, *The Journal of Finance* 62(5): 2235–2274.
- Carmona, R. (2016). Lectures on BSDEs, Stochastic Control, and Stochastic Differential Games with Financial Applications, 1 edn, SIAM, Philadelphia.
- Carmona, R. and Delarue, F. (2013). Probabilistic analysis of mean-field games, SIAM Journal on Control and Optimization 51(4): 2705–2734.
- Carmona, R. and Delarue, F. (2018a). Probabilistic Theory of Mean Field Games with Applications I, Vol. 83 of Probability Theory and Stochastic Modelling, Springer, Cham.
- Carmona, R. and Delarue, F. (2018b). Probabilistic Theory of Mean Field Games with Applications II, Vol. 84 of Probability Theory and Stochastic Modelling, Springer, Cham.
- Carmona, R., Delarue, F. and Lacker, D. (2016). Mean field games with common noise, *The* Annals of Probability 44(6): 3740–3803.
- Cernỳ, A. and Kallsen, J. (2007). On the structure of general mean-variance hedging strategies, The Annals of Probability **35**(4): 1479–1531.
- Chevalier, J. and Ellison, G. (1997). Risk taking by mutual funds as a response to incentives, Journal of political economy **105**(6): 1167–1200.
- Cont, R. and Tankov, P. (2004). *Financial Modelling with Jump Processes*, Chapman & Hall/CRC financial mathematics series, Chapman & Hall/CRC, Boca Raton.
- Cronqvist, H. and Fahlenbrach, R. (2009). Large shareholders and corporate policies, *The Review* of Financial Studies **22**(10): 3941–3976.
- Cuoco, D. and Cvitanić, J. (1998). Optimal consumption choices for a 'large' investor, *Journal of Economic Dynamics and Control* **22**(3): 401–436.
- Curatola, G. (2019). Portfolio choice of large investors who interact strategically. Formerly available at SSRN:3404491, Accessed: June 24, 2020.
- Curatola, G. (2022). Price impact, strategic interaction and portfolio choice, *The North American Journal of Economics and Finance* **59**: 101594.

- Cvitanić, J. and Ma, J. (1996). Hedging options for a large investor and forward-backward SDE's, The Annals of Applied Probability 6(2): 370–398.
- Delbaen, F. and Schachermayer, W. (1995). The no-arbitrage property under a change of numéraire, Stochastics and Stochastic Reports 53(3-4): 213–226.
- Delbaen, F. and Schachermayer, W. (1996). Attainable claims with p'th moments, Annales de l'Institut Henri Poincaré **32**(6): 743–763.
- Delbaen, F. and Schachermayer, W. (1998). The fundamental theorem of asset pricing for unbounded stochastic processes, *Mathematische Annalen* **312**(2): 215–250.
- Delbaen, F. and Schachermayer, W. (2006). *The Mathematics of Arbitrage*, Springer Finance, 1. edn, Springer, Berlin Heidelberg.
- DeMarzo, P. M. and Urošević, B. (2006). Ownership dynamics and asset pricing with a large shareholder, *Journal of political Economy* 114(4): 774–815.
- Deng, C., Su, X. and Zhou, C. (2020). Relative wealth concerns with partial information and heterogeneous priors, arXiv:2007.11781.
- Deng, C., Zeng, X. and Zhu, H. (2018). Non-zero-sum stochastic differential reinsurance and investment games with default risk, *European Journal of Operational Research* 264(3): 1144– 1158.
- Dos Reis, G. and Platonov, V. (2021). Forward utilities and mean-field games under relative performance concerns, in C. Bernardin, F. Golse, P. Gonçalves, V. Ricci and A. J. Soares (eds), From Particle Systems to Partial Differential Equations: International Conference, Particle Systems and PDEs VI, VII and VIII, 2017-2019, Vol. 352, Springer, Cham, pp. 227–251.
- Dos Reis, G. and Platonov, V. (2022). Forward utility and market adjustments in relative investment-consumption games of many players, SIAM Journal on Financial Mathematics 13(3): 844–876.
- Eberlein, E. and Kallsen, J. (2019). *Mathematical Finance*, Springer Finance, 1. edn, Springer, Cham.
- Eksi, Z. and Ku, H. (2017). Portfolio optimization for a large investor under partial information and price impact, *Mathematical Methods of Operations Research* 86: 601–623.
- Espinosa, G.-E. (2010). Stochastic Control Methods for Optimal Portfolio Investment, PhD thesis, Ecole Polytechnique Paris.
- Espinosa, G.-E. and Touzi, N. (2015). Optimal investment under relative performance concerns, Mathematical Finance 25(2): 221–257.
- Evangelista, D. and Thamsten, Y. (2021). On finite population games of optimal trading, arXiv:2004.00790.

- Ferrari, G., Riedel, F. and Steg, J.-H. (2017). Continuous-time public good contribution under uncertainty: A stochastic control approach, Applied Mathematics & Optimization 75: 429–470.
- Föllmer, H. and Schied, A. (2016). Stochastic Finance: An Introduction in Discrete Time, 4. edn, Walter de Gruyter, Berlin/Boston.
- Frei, C. and Dos Reis, G. (2011). A financial market with interacting investors: does an equilibrium exist?, *Mathematics and financial economics* 4: 161–182.
- Fu, G., Su, X. and Zhou, C. (2020). Mean field exponential utility game: A probabilistic approach, arXiv:2006.07684.
- Fu, G. and Zhou, C. (2023). Mean field portfolio games, Finance and Stochastics 27: 189–231.
- Gatheral, J. and Schied, A. (2013). Dynamical models of market impact and algorithms for order execution, *in J.-P. Fouque and J. A. Langsam (eds)*, *Handbook on Systemic Risk*, pp. 579–599.
- Geng, T. and Zariphopoulou, T. (2017). Passive and competitive investment strategies under relative forward performance criteria, *SSRN:2870040*.
- Guan, G. and Hu, X. (2022). Time-consistent investment and reinsurance strategies for meanvariance insurers in *n*-agent and mean-field games, *North American Actuarial Journal* **26**(4): 537– 569.
- Guo, X. and Xu, R. (2020). Pareto optimality for a class of stochastic games, arXiv:2005.05766v1.
- Guéant, O., Lasry, J.-M. and Lions, P.-L. (2011). Mean field games and applications, Paris-Princeton Lectures on Mathematical Finance 2010, Vol. 2003 of Lecture Notes in Mathematics, Springer, Berlin, Heidelberg.
- Harris, J. M., Hirst, J. L. and Mossinghoff, M. J. (2008). Combinatorics and graph theory.
- He, H. and Mamaysky, H. (2005). Dynamic trading policies with price impact, Journal of Economic Dynamics and Control 29(5): 891–930.
- He, X. D. and Zhou, X. Y. (2016). Hope, fear, and aspirations, Mathematical Finance 26(1): 3–50.
- He, Y., He, L., Chen, D. and Liu, Z. (2023). Mean field and n-insurers games for robust optimal reinsurance-investment in correlated markets, *Journal of Industrial and Management Optimization* 19(9): 6806–6825.
- Heston, S. L. (1993). A closed-form solution for options with stochastic volatility with applications to bond and currency options, *The review of financial studies* **6**(2): 327–343.
- Horn, R. A. and Johnson, C. R. (2013). *Matrix Analysis*, 2. edn, Cambridge university press, New York.
- Horst, U. and Naujokat, F. (2010). Illiquidity and derivative valuation. SFB 649 Discussion Paper, http://sfb649.wiwi.hu-berlin.de/papers/pdf/SFB649DP2010-011.pdf. Accessed: August 30, 2023.

- Hu, R. and Zariphopoulou, T. (2022). n-player and mean-field games in Itô-diffusion markets with competitive or homophilous interaction, Stochastic Analysis, Filtering, and Stochastic Optimization, Springer, Cham.
- Huang, M., Malhamé, R. P. and Caines, P. E. (2006). Large population stochastic dynamic games: Closed-loop McKean-Vlasov systems and the Nash certainty equivalence principle, *Communications in information and systems* 6(3): 221–252.
- Jacod, J. and Shiryaev, A. (2003). Limit Theorems for Stochastic Processes, Vol. 288 of Grundlehren der mathematischen Wissenschaften, 2. edn, Springer, Berlin Heidelberg.
- Jarrow, R. A. (1992). Market manipulation, bubbles, corners, and short squeezes, *Journal of financial and Quantitative Analysis* **27**(3): 311–336.
- Jarrow, R. A. (1994). Derivative security markets, market manipulation, and option pricing theory, Journal of financial and quantitative analysis **29**(2): 241–261.
- Jeanblanc, M., Yor, M. and Chesney, M. (2009). Mathematical Methods for Financial Markets, Springer Finance, 1. edn, Springer, London.
- Johnson, N. L., Kotz, S. and Balakrishnan, N. (1994). Continuous Univariate Distributions -Volume 1, 2. edn, John wiley & Sons, New York.
- Kahneman, D. and Tversky, A. (1979). Prospect theory: An analysis of decision under risk, *Econometrica* 47(2): 263–292.
- Kallsen, J. and Muhle-Karbe, J. (2010). Utility maximization in affine stochastic volatility models, International Journal of Theoretical and Applied Finance 13(03): 459–477.
- Karatzas, I., Lehoczky, J. P. and Shreve, S. E. (1987). Optimal portfolio and consumption decisions for a "small investor" on a finite horizon, SIAM journal on control and optimization 25(6): 1557–1586.
- Karatzas, I. and Shreve, S. E. (1998). Methods of Mathematical Finance, Vol. 39 of Probability Theory and Stochastic Modelling, Springer, New York.
- Karmarkar, U. S. (1979). Subjectively weighted utility and the allais paradox, Organizational Behavior and Human Performance 24(1): 67–72.
- Kempf, A. and Ruenzi, S. (2008). Tournaments in mutual-fund families, The Review of Financial Studies 21(2): 1013–1036.
- Klenke, A. (2020). *Probability Theory: A Comprehensive Course*, Universitext, 3. edn, Springer, Cham.
- Korn, R. (1997). Optimal portfolios: Stochastic models for optimal investment and risk management in continuous time, 1. edn, World scientific, Singapore.
- Korn, R. and Desmettre, S. (2014). Moderne Finanzmathematik-Theorie und praktische Anwendung, Springer.

- Kraft, H. and Kühn, C. (2011). Large traders and illiquid options: Hedging vs. manipulation, Journal of Economic Dynamics and Control 35(11): 1898–1915.
- Kraft, H., Meyer-Wehmann, A. and Seifried, F. T. (2020). Dynamic asset allocation with relative wealth concerns in incomplete markets, *Journal of Economic Dynamics and Control* 113: 103857.
- Kramkov, D. and Schachermayer, W. (1999). The asymptotic elasticity of utility functions and optimal investment in incomplete markets, *The Annals of Applied Probability* **9**(3): 904–950.
- Ku, H. and Zhang, H. (2018). Option pricing for a large trader with price impact and liquidity costs, *Journal of Mathematical Analysis and Applications* **459**(1): 32–52.
- Kyle, A. S. (1985). Continuous auctions and insider trading, *Econometrica* 53(6): 1315–1335.
- Lacker, D. (2015). Mean field games via controlled martingale problems: Existence of Markovian equilibria, *Stochastic Processes and their Applications* **125**(7): 2856–2894.
- Lacker, D. (2016). A general characterization of the mean field limit for stochastic differential games, *Probability Theory and Related Fields* 165: 581–648.
- Lacker, D. (2018). Mean field games and interacting particle systems. Lecture notes, http: //www.columbia.edu/~dl3133/MFGSpring2018.pdf. Accessed: August 30, 2023.
- Lacker, D. and Soret, A. (2020). Many-player games of optimal consumption and investment under relative performance criteria, *Mathematics and Financial Economics* 14: 263–281.
- Lacker, D. and Zariphopoulou, T. (2019). Mean field and *n*-agent games for optimal investment under relative performance criteria, *Mathematical Finance* **29**(4): 1003–1038.
- Lasry, J.-M. and Lions, P.-L. (2007). Mean field games, Japanese journal of mathematics 2: 229–260.
- Last, G. and Penrose, M. (2017). *Lectures on the Poisson Process*, Institute of Mathematical Statistics Textbooks, Cambridge University Press, Cambridge.
- Luo, X. and Schied, A. (2019). Nash equilibrium for risk-averse investors in a market impact game with transient price impact, *Market Microstructure and Liquidity* 5(01n04).
- Markowitz, H. (1952). Portfolio selection, The Journal of Finance 7(1): 77–91.
- Merton, R. C. (1969). Lifetime portfolio selection under uncertainty: The continuous-time case, The review of Economics and Statistics 51(3): 247–257.
- Merton, R. C. (1975). Optimum consumption and portfolio rules in a continuous-time model, in
 W. Ziemba and R. Vickson (eds), Stochastic optimization models in finance, Elsevier, pp. 621–661.
- Micheli, A., Muhle-Karbe, J. and Neuman, E. (2021). Closed-loop Nash competition for liquidity, arXiv:2112.02961.
- Miettinen, K. (1999). Nonlinear Multiobjective Optimization, Vol. 12, Kluwer Academic Publishers, Boston.

- Muhle-Karbe, J., Wang, Z. and Webster, K. (2022). Stochastic liquidity as a proxy for nonlinear price impact, SSRN:4286108.
- Musiela, M. and Zariphopoulou, T. (2006). Investments and forward utilities. Preprint, https: //web.ma.utexas.edu/users/zariphop/pdfs/TZ-TechnicalReport-4.pdf. Accessed: August 30, 2023.
- Nash, J. (1951). Non-cooperative games, Annals of Mathematics 54(2): 286–295.
- Nguyen, M. (2022). n-agent and mean field games for optimal investment with HARA utility function and the presence of risk-seeking agents. https://gtcenter.org/wp-content/uploads/ 2022/04/N-agent-and-MFG-Minh-Nguyen.pdf. Accessed: August 30, 2023.
- Obizhaeva, A. A. and Wang, J. (2013). Optimal trading strategy and supply/demand dynamics, Journal of Financial Markets 16(1): 1–32.
- Palomino, F. (2005). Relative performance objectives in financial markets, Journal of Financial Intermediation 14(3): 351–375.
- Park, J. Y. (2022). Optimal portfolio selection of many players under relative performance criteria in the market model with random coefficients, *arXiv:2209.07411*.
- Pham, H. (2009). Continuous-time Stochastic Control and Optimization with Financial Applications, Vol. 61 of Stochastic Modelling and Applied Probability, Springer, Berlin Heidelberg.
- Pratt, J. W. (1964). Risk aversion in the small and in the large, 32(1/2): 122–136.
- Protter, P. E. (2005). Stochastic Integration and Differential Equations, Vol. 21 of Stochastic Modelling and Applied Probability, 2. edn, Springer, Berlin Heidelberg.
- Sato, K.-I. (1999). Lévy Processes and Infinitely Divisible Distributions, Vol. 68 of Cambridge Studies in Advanced Mathematics, Cambridge university press, Cambridge.
- Schäfer, U. (2007). A fixed point theorem based on Miranda, Fixed Point Theory and Applications 2007: 1–5.
- Schied, A., Strehle, E. and Zhang, T. (2017). High-frequency limit of Nash equilibria in a market impact game with transient price impact, SIAM Journal on Financial Mathematics 8(1): 589–634.
- Schied, A. and Zhang, T. (2017). A state-constrained differential game arising in optimal portfolio liquidation, *Mathematical Finance* 27(3): 779–802.
- Schied, A. and Zhang, T. (2019). A market impact game under transient price impact, *Mathematics* of Operations Research 44(1): 102–121.
- Schilling, R. L. (2005). Measures, Integrals and Martingales, 1. edn, Cambridge University Press, New York.

- Schöneborn, T. and Schied, A. (2009). Liquidation in the face of adversity: Stealth vs. sunshine trading, EFA 2008 Athens Meetings Paper.
- Shreve, S. E. (2004). *Stochastic Calculus for Finance II: Continuous-Time Models*, Springer Finance, 1. edn, Springer, New York.
- Sirri, E. R. and Tufano, P. (1998). Costly search and mutual fund flows, *The journal of finance* 53(5): 1589–1622.
- Tangpi, L. and Zhou, X. (2022). Optimal investment in a large population of competitive and heterogeneous agents, arXiv:2202.11314.
- Tversky, A. and Kahneman, D. (1992). Advances in prospect theory: Cumulative representation of uncertainty, *Journal of Risk and uncertainty* 5(4): 297–323.
- Von Neumann, J. and Morgenstern, O. (1947). *Theory of Games and Economic Behavior*, 2. edn, Princeton university press, Princeton.
- Wang, G. and Ye, J. (2023). Fund managers' competition for investment flows based on relative performance, *Journal of Optimization Theory and Applications*.
- Webster, K. T. (2023). Handbook of Price Impact Modeling, 1. edn, CRC Press, Boca Raton.
- Whitmeyer, M. (2019). Relative performance concerns among investment managers, Annals of Finance 15: 205–231.
- Yan, J.-A. (2018). Introduction to Stochastic Finance, Universitext, 1. edn, Springer, Singapore.
- Yang, P. and Chen, Z. (2022). Optimal reinsurance pricing, risk sharing and investment strategies in a joint reinsurer-insurer framework, *IMA Journal of Management Mathematics* 00: 1–34.

APPENDIX A

Additional material for Chapter 6

The following chapter contains supplementary material for Chapter 6. First, we explain the application of the dominated convergence theorem in the proof of Lemma 6.6 in more depth. Afterwards, we prove two results regarding the expected value of the product of a normal and a corresponding lognormal random variable (appearing in Remark 6.7 and Example 6.11).

A.1. Application of the dominated convergence theorem in Lemma 6.6

In Lemma 6.6, we determined the optimal collective competitive terminal wealth using the collective competitive utility function \tilde{U}_{β} from Lemma 6.2. In order to argue that there exists a unique Lagrange multiplier, we needed to apply the dominated convergence theorem. In what follows, we explain the application of the dominated convergence theorem in more depth. To be more specific, we justify the following interchange of convergence and expectation

$$\lim_{k \to \infty} \mathbb{E}\left[Z_T \cdot I_i\left(\frac{n\lambda_k Z_T}{\beta_i \left(n + \theta_i\right) \left(1 - \hat{\theta}\right)}\right)\right] = \mathbb{E}\left[Z_T \cdot I_i\left(\frac{n\lambda Z_T}{\beta_i \left(n + \theta_i\right) \left(1 - \hat{\theta}\right)}\right)\right],$$

where $(\lambda_k)_{k \in \mathbb{N}}$ is a sequence of real numbers with $\lim_{k \to \infty} \lambda_k = \lambda$.

For simplicity, let $\tilde{I}_i(x) \coloneqq I_i\left(\frac{nx}{\beta_i(n+\theta_i)(1-\hat{\theta})}\right)$, $x \in (0,\infty)$. We consider two different cases: First, assume that $\mathcal{D}_i = (0,\infty)$. Let $0 < \varepsilon < \lambda$. Since $\lambda_k \to \lambda$, $k \to \infty$, there exists some $k_0 = k_0(\varepsilon)$ such that $\lambda_k \in (\lambda - \varepsilon, \lambda + \varepsilon)$ for all $k \ge k_0$. Hence, since \tilde{I}_i is strictly decreasing, it follows

$$Z_T \tilde{I}_i \left(\lambda_k Z_T \right) \le Z_T \tilde{I}_i \left((\lambda - \varepsilon) Z_T \right) \quad \forall k \ge k_0.$$

Moreover, let $\lambda_{\min} \coloneqq \min_{1 \le k < k_0} \lambda_k$. Then

$$Z_T \tilde{I}_i \left(\lambda_k Z_T \right) \le Z_T \tilde{I}_i \left(\lambda_{\min} Z_T \right) \quad \forall \, k < k_0.$$

In summary, we obtain

$$Z_T \tilde{I}_i (\lambda_k Z_T) \le \max \left\{ Z_T \tilde{I}_i ((\lambda - \varepsilon) Z_T), Z_T \tilde{I}_i (\lambda_{\min} Z_T) \right\}$$
$$\le Z_T \tilde{I}_i ((\lambda - \varepsilon) Z_T) + Z_T \tilde{I}_i (\lambda_{\min} Z_T)$$

for all $k \in \mathbb{N}$. By assumption, the right-hand side is integrable and hence, we can apply the dominated convergence theorem.

Now assume that $\mathcal{D}_i = \mathbb{R}$. Again, let $0 < \varepsilon < \lambda$. Then there exists some $k_0 = k_0(\varepsilon) \in \mathbb{N}$ such that $\lambda_k \in (\lambda - \varepsilon, \lambda + \varepsilon)$. Hence, using the monotonicity of \tilde{I}_i , we have

$$\left| Z_T \tilde{I}_i(\lambda_k Z_T) \right| \le \max\left\{ \left| Z_T \tilde{I}_i((\lambda - \varepsilon) Z_T) \right|, \left| Z_T \tilde{I}_i((\lambda + \varepsilon) Z_T) \right| \right\}$$

for all $k \geq k_0$. Moreover, let

$$\lambda_{\max} = \arg\max_{\lambda \in \{\lambda_1, \dots, \lambda_{k_0-1}\}} \left| Z_T \tilde{I}_i(\lambda_k Z_T) \right|.$$

Then it follows

$$\left| Z_T \tilde{I}_i(\lambda_k Z_T) \right| \le \left| Z_T \tilde{I}_i(\lambda_{\max} Z_T) \right| \quad \forall k < k_0.$$

In summary, we obtain

$$\begin{aligned} &\left| Z_T \tilde{I}_i(\lambda_k Z_T) \right| \\ \leq & \max \left\{ \left| Z_T \tilde{I}_i((\lambda - \varepsilon) Z_T) \right|, \left| Z_T \tilde{I}_i((\lambda + \varepsilon) Z_T) \right|, \left| Z_T \tilde{I}_i(\lambda_{\max} Z_T) \right| \right\} \\ \leq & \left| Z_T \tilde{I}_i((\lambda - \varepsilon) Z_T) \right| + \left| Z_T \tilde{I}_i((\lambda + \varepsilon) Z_T) \right| + \left| Z_T \tilde{I}_i(\lambda_{\max} Z_T) \right| \end{aligned}$$

for all $k \in \mathbb{N}$. By assumption, the right-hand side is integrable and hence, we can apply the dominated convergence theorem.

A.2. EXPECTATION OF A NORMAL AND A LOGNORMAL RANDOM VARIABLE

The following lemma treats the expectation of the product of a normally distributed random variable and the corresponding lognormally distributed random variable. The results are used in Remark 6.7 and Example 6.11.

Lemma A.1. Let $X \sim \mathcal{N}(\mu, \sigma^2)$ for $\mu \in \mathbb{R}$, $\sigma > 0$. Then

$$\mathbb{E}[|X|\exp(X)] < \infty, \quad \mathbb{E}[X\exp(X)] = \left(\mu + \sigma^2\right)\exp\left(\frac{\sigma^2}{2} + \mu\right).$$

Proof. A straightforward calculation yields

$$\begin{split} \mathbb{E}[|X|\exp(X)] &= \frac{1}{\sqrt{2\pi\sigma}} \int_{-\infty}^{\infty} |x|\exp(x)\exp\left(-\frac{(x-\mu)^2}{2\sigma^2}\right) \mathrm{d}x \\ &= \exp\left(\mu + \frac{\sigma^2}{2}\right) \frac{1}{\sqrt{2\pi\sigma}} \int_{-\infty}^{\infty} |x|\exp\left(-\frac{(x-(\mu+\sigma^2))^2}{2\sigma^2}\right) \mathrm{d}x \\ &= \exp\left(\mu + \frac{\sigma^2}{2}\right) \mathbb{E}[|\tilde{X}|] < \infty, \end{split}$$

where $\tilde{X} \sim \mathcal{N}(\mu + \sigma^2, \sigma^2)$. The expectation in the last line is finite since $|x| \leq \frac{1}{2}(x^2 + 1)$ holds for all $x \in \mathbb{R}$ and the second moment of a normally distributed random variable is finite.

We can reuse the previous calculation for the second assertion of the lemma. Thus,

$$\mathbb{E}[X \exp(X)] = \exp\left(\mu + \frac{\sigma^2}{2}\right) \mathbb{E}[\widetilde{X}]$$
$$= \left(\mu + \sigma^2\right) \exp\left(\mu + \frac{\sigma^2}{2}\right).$$

APPENDIX B

Additional material for Chapter 7

In the following, we prove two verification results used in the proofs of Theorems 7.2 and 7.12.

B.1. Supplements for the proof of Theorem 7.2

Throughout this section, we work in the setting of Section 7.3 and use the notation defined there. Lemma B.1 verifies that the optimal strategy determined in the proof of Theorem 7.2 is, in fact, the unique optimal solution to the best response problem.

Lemma B.1. Suppose that the assumptions of Theorem 7.2 are satisfied. For some arbitrary but fixed $i \in \{1, ..., n\}$, let

$$G(t,y) = -\exp\left(-\rho(T-t) - \frac{1}{\delta_i}y\right), \ t \in [0,T], \ y \in \mathbb{R},$$
(B.1)

where ρ is taken from (7.21), and let $u^* = u^*(t, y)$ be the unique maximizer of

$$h: \mathbb{R}^d \to \mathbb{R}, \, \varphi \mapsto \frac{G(t, y)}{\delta_i^2} \left(\frac{1}{2} \varphi^\top \left(\sigma \sigma^\top - \frac{2\delta_i}{n} A \right) \varphi - \delta_i \varphi^\top \widetilde{\mu}^{-i} \right)$$

for $(t, y) \in [0, T] \times \mathbb{R}$.

Then J(t, y) = G(t, y) for all $(t, y) \in [0, T] \times \mathbb{R}$ and $\varphi_t^* = u^*(t, Y_t^*)$ is the unique optimal solution to (7.11), where Y^* solves the stochastic differential equation

$$dY_t = u^*(t, Y_t)^{\top} \left(\left(\tilde{\mu}^{-i} + \frac{1}{n} A u^*(t, Y_t) \right) dt + \sigma dW(t) \right), \ t \in [0, T], \ Y_0 = \tilde{x}_0^i.$$
(B.2)

Proof. The proof follows arguments of standard verification theorems (see, e.g., Björk, 2004, pp. 280–282; Pham, 2009, pp. 47–49). First, we define an operator \mathcal{T} to rewrite the HJB equation

(7.19) as follows

$$0 = \sup_{\varphi \in \mathbb{R}^d} \left\{ F_t + F_y \varphi^\top \left(\widetilde{\mu}^{-i} + \frac{1}{n} A \varphi \right) + \frac{1}{2} F_{yy} \varphi^\top \sigma \sigma^\top \varphi \right\} \eqqcolon \sup_{\varphi \in \mathbb{R}^d} \mathcal{T}F(t, y, \varphi),$$
(B.3)

where $F \in \mathcal{C}^{1,2}([0,T] \times \mathbb{R}), t \in [0,T]$, and $y \in \mathbb{R}$. Now we apply the Itô-Doeblin formula (Theorem 2.1) to the function G from (B.1) and the process $Y^{i,\varphi}$ from (7.16), where $\varphi \in \mathcal{A}$, over the time interval [t,T] for some $t \in [0,T]$. It follows

$$\begin{split} &G(T, Y_T^{i,\varphi}) \\ &= G(t, Y_t^{i,\varphi}) + \int_t^T G_t(s, Y_s^{i,\varphi}) \mathrm{d}s + \int_t^T G_y(s, Y_s^{i,\varphi}) \mathrm{d}Y_s^{i,\varphi} + \frac{1}{2} \int_t^T G_{yy}(s, Y_s^{i,\varphi}) \mathrm{d}\langle Y^{i,\varphi} \rangle_s \\ &= G(t, Y_t^{i,\varphi}) + \int_t^T G_t(s, Y_s^{i,\varphi}) \mathrm{d}s + \int_t^T G_y(s, Y_s^{i,\varphi}) \varphi(s)^\top \left(\widetilde{\mu}^{-i} + \frac{1}{n} A \varphi(s) \right) \mathrm{d}s \\ &+ \int_t^T G_y(s, Y_s^{i,\varphi}) \varphi(s)^\top \sigma \mathrm{d}W(s) + \frac{1}{2} \int_t^T G_{yy}(s, Y_s^{i,\varphi}) \varphi(s)^\top \sigma \sigma^\top \varphi(s) \mathrm{d}s \\ &= G(t, Y_t^{i,\varphi}) + \int_t^T \mathcal{T}G(s, Y_s^{i,\varphi}, \varphi(s)) \mathrm{d}s + \int_t^T G_y(s, Y_s^{i,\varphi}) \varphi(s)^\top \sigma \mathrm{d}W(s). \end{split}$$
(B.4)

By G_t , G_y , and G_{yy} , we denote the respective partial derivatives of G with respect to t and y. Further, we applied Lemma 2.2 to determine the quadratic variation of $Y^{i,\varphi}$. Let us now assume that the process

$$\int_0^t G_y(s, Y_s^{i,\varphi})\varphi(s)^\top \sigma \mathrm{d}W(s), \, t \in [0,T],$$
(B.5)

is a martingale for any admissible φ . Then we can apply the conditional expectation $\mathbb{E}^{t,y}$ for some $y \in \mathbb{R}$ to both sides of (B.4), using that $G(T, y) = -e^{-\frac{1}{\delta_i}y}$, to obtain

$$\mathbb{E}^{t,y} \left[-\exp\left(-\frac{1}{\delta_i} Y_T^{i,\varphi}\right) \right]$$

= $G(t,y) + \mathbb{E}^{t,y} \left[\int_t^T \mathcal{T}G(s, Y_s^{i,\varphi}, \varphi(s)) ds + \int_t^T G_y(s, Y_s^{i,\varphi}) \varphi(s)^\top \sigma dW(s) \right]$
 $\leq G(t,y) + \mathbb{E}^{t,y} \left[\int_t^T G_y(s, Y_s^{i,\varphi}) \varphi(s)^\top \sigma dW(s) \right]$ (B.6)
= $C(t,y)$

$$=G(t,y).$$
(B.7)

The last step (B.7) holds due to the assumption that (B.5) is a martingale. Moreover, (B.6) follows since G is a solution to the HJB equation and hence, $\mathcal{T}G(s, Y_s^{i,\varphi}, \varphi(s)) \leq 0$ for all $s \in [0, T]$ and any admissible φ . Then it follows by the definition of the value function J

$$J(t,y) = \sup_{\varphi \in \mathcal{A}} \mathbb{E}^{t,y} \left[-\exp\left(-\frac{1}{\delta_i} Y_T^{i,\varphi}\right) \right] \le G(t,y)$$
(B.8)

for all $(t, y) \in [0, T] \times \mathbb{R}$. Moreover, since $u^* = u^*(t, y)$ maximizes h and thus, $\mathcal{T}G(t, y, \cdot)$, we obtain equality in (B.6) for $\varphi^*(s) = u^*(s, Y_s^*)$, where Y^* solves the stochastic differential equation (B.2). Thus, we also obtain equality in (B.8). Hence, $\varphi^*(s) = u^*(s, Y_s^*)$ is an optimal solution to (7.11). Moreover, the solution G to the HJB equation is equal to the almost surely uniquely defined value function J and thus, G is the unique solution to the HJB equation. To see that φ^*

is the unique (up to modifications) optimal solution to (7.13), we note that any optimal strategy satisfies the Bellman optimality principle. Since we already found the (unique) value function, this implies that optimal strategies are precisely given by the extremal points in (B.3). Thus, the optimal strategy is unique due to the strict concavity of the expression inside the supremum.

Hence, it only remains to show that the process (B.5) is a martingale. For now, we assume that φ is bounded, i.e., there exists a constant K > 0 such that $\|\varphi(t)\| \leq K$ P-almost surely for all $t \in [0, T]$. In order to show that (B.5) is a martingale, it suffices to prove that the integrand is square integrable, i.e., that

$$\mathbb{E}\left[\int_0^T \left\|G_y(t, Y_t^{i,\varphi})\varphi(t)^{\top}\sigma\right\|^2 \mathrm{d}t\right] < \infty.$$

This is a general result arising in the construction of the Itô integral with respect to a Brownian motion. It can, for example, be found in Theorem 4.3.1 in Shreve (2004). Moreover, since we assumed that φ is bounded and σ is deterministic and constant, it is enough to show that

$$\mathbb{E}\left[\int_0^T G_y(t, Y_t^{i,\varphi})^2 \mathrm{d}t\right] = \int_0^T \mathbb{E}\left[G_y(t, Y_t^{i,\varphi})^2\right] \mathrm{d}t < \infty,\tag{B.9}$$

where we used Fubini's theorem in the first equality. Using the representation

$$Y_t^{i,\varphi} = \widetilde{x}_0^i + \int_0^t \varphi(s)^\top \left(\widetilde{\mu}^{-i} + \frac{1}{n} A \varphi(s) \right) \mathrm{d}s + \int_0^t \varphi(s)^\top \sigma \mathrm{d}W(s),$$

we can calculate the expectation in (B.9)

$$\mathbb{E}\left[G_{y}(t,Y_{t}^{i,\varphi})^{2}\right] \\
= \mathbb{E}\left[\frac{1}{\delta_{i}^{2}}\exp\left(-2\rho(T-t)-\frac{2}{\delta_{i}}Y_{t}^{i,\varphi}\right)\right] \\
= \frac{1}{\delta_{i}^{2}}e^{-2\rho(T-t)-\frac{2}{\delta_{i}}\widetilde{x}_{0}^{i}}\mathbb{E}\left[\exp\left(-\frac{2}{\delta_{i}}\int_{0}^{t}\varphi(s)^{\top}\left(\widetilde{\mu}^{-i}+\frac{1}{n}A\varphi(s)\right)ds-\frac{2}{\delta_{i}}\int_{0}^{t}\varphi(s)^{\top}\sigma dW(s)\right)\right] \\
=:\frac{1}{\delta_{i}^{2}}e^{-2\rho(T-t)-\frac{2}{\delta_{i}}\widetilde{x}_{0}^{i}}\mathbb{E}_{\widetilde{\mathbb{Q}}}\left[\exp\left(-\frac{2}{\delta_{i}}\int_{0}^{t}\varphi(s)^{\top}\left(\widetilde{\mu}^{-i}+\frac{1}{n}A\varphi(s)-\frac{1}{\delta_{i}}\sigma\sigma^{\top}\varphi(s)\right)ds\right)\right] \quad (B.10) \\
\leq \frac{1}{\delta^{2}}e^{-2\rho(T-t)-\frac{2}{\delta_{i}}\widetilde{x}_{0}^{i}}\cdot\exp(C\cdot t). \quad (B.11)$$

In (B.10), we introduced an equivalent probability measure $\widetilde{\mathbb{Q}} \sim \mathbb{P}$ with density

$$\frac{\mathrm{d}\widetilde{\mathbb{Q}}}{\mathrm{d}\mathbb{P}} = \exp\left(-\frac{2}{\delta_i}\int_0^t \varphi(s)^\top \sigma \mathrm{d}W(s) - \frac{2}{\delta_i^2}\int_0^t \varphi(s)^\top \sigma \sigma^\top \varphi(s) \mathrm{d}s\right).$$

Since φ was assumed to be bounded, this is in fact a density. In the last step (B.11), we used the assumption that φ is bounded again and thus, the integrand is bounded by some constant C > 0. Now we can deduce further that the integral in (B.9) is finite and thus, (B.5) is in fact a martingale.

Since the strategy we derived in the proof of Theorem 7.2 is constant and not at the boundary of

the interval the strategy φ was restricted to in the beginning, the dominated convergence theorem for stochastic processes (Theorem 32 in Protter, 2005, p.176) yields that the assumption that φ is bounded is not a restriction (see also Korn and Desmettre, 2014, p.294; Pham, 2009, pp.47–48). This concludes our proof.

B.2. Supplements for the proof of Theorem 7.12

We use notation taken from Section 7.4 throughout this section. The lemma below verifies that the optimal strategy determined in the proof of Theorem 7.12 is, in fact, the unique optimal solution to the best response problem (7.37) for constant Nash equilibria.

Lemma B.2. Assume that the assumptions of Theorem 7.12 are satisfied. For some arbitrary but fixed $i \in \{1, ..., n\}$, let π^j , $j \neq i$, be deterministic and constant, and

$$G(t,x,y) = \frac{\delta_i}{\delta_i - 1} e^{\rho(T-t)} \left(xy^{-\frac{\theta_i}{n}} \right)^{\frac{\delta_i - 1}{\delta_i}}, t \in [0,T], x, y \in (0,\infty),$$
(B.12)

where $\rho \in \mathbb{R}$ is taken from (7.46), and let $u^* = u^*(t, x, y)$ be the unique maximizer of $h : \mathbb{R}^d \to \mathbb{R}$,

$$h(\pi) = \frac{\delta_i - 1}{\delta_i^2} G(t, x, y) \cdot \pi^\top \left\{ \delta_i \mu + \left(\frac{\delta_i}{n} \left(1 - \frac{\theta_i}{n} \right) A - \frac{\theta_i}{n} (\delta_i - 1) \sigma \sigma^\top \right) \sum_{j \neq i} \pi^j - \frac{1}{2} \left(\sigma \sigma^\top - \frac{2\delta_i}{n} A \right) \pi \right\}$$

for $(t, x, y) \in [0, T] \times (0, \infty) \times (0, \infty)$.

Then J(t, x, y) = G(t, x, y) for all $(t, x, y) \in [0, T] \times (0, \infty) \times (0, \infty)$ and $\pi_t^* = u^* \left(t, \tilde{X}_t^*, \tilde{Y}_t^{-i} \right)$ is the unique optimal solution to (7.37), where X^* solves the stochastic differential equation

$$dX_t = X_t u^*(t, X_t, \tilde{Y}_t^{-i})^\top \left(\left(\mu + \frac{1}{n} A u^*(t, X_t, \tilde{Y}_t^{-i}) + \frac{1}{n} \left(1 - \frac{\theta_i}{n} \right) A \sum_{j \neq i} \pi^j \right) dt + \sigma dW(t) \right)$$
(B.13)

for $t \in [0, T]$ and $X_0 = x_0^i$.

Proof. The proof follows arguments of standard verification theorems (see, e.g., Björk, 2004, pp. 280–282; Pham, 2009, pp. 47–49). First, we define an operator \mathcal{T} by

$$\begin{aligned} \mathcal{T}F(t,x,y,\pi) \\ &= F_t + yF_y \bigg(\sum_{j \neq i} (\pi^j)^\top \bigg(\mu + \frac{1}{n} A \sum_{j \neq i} \pi^j \bigg) + \frac{1}{2} \sum_{\substack{h \neq j \\ h, j \neq i}} (\pi^h)^\top \sigma \sigma^\top \pi^j \bigg) + \frac{1}{2} y^2 F_{yy} \sum_{\substack{h, j \neq i}} (\pi^h)^\top \sigma \sigma^\top \pi^j \\ &+ xF_x \pi^\top \mu + \pi^\top \bigg(\frac{1}{n} xF_x A + \frac{1}{2} x^2 F_{xx} \sigma \sigma^\top \bigg) \pi + \pi^\top \bigg(\frac{1}{n} \bigg(1 - \frac{\theta_i}{n} \bigg) xF_x A + xyF_{xy} \sigma \sigma^\top \bigg) \sum_{j \neq i} \pi^j, \end{aligned}$$

where $\pi \in \mathbb{R}^d$, $t \in [0, T]$, $x, y \in (0, \infty)$, and $F \in \mathcal{C}^{1,2,2}([0, T] \times (0, \infty)^2)$. Note that the arguments of F were omitted to simplify notation. Then the HJB equation (7.44) can be written as

$$0 = \sup_{\pi \in \mathbb{R}^d} \mathcal{T}F(t, x, y, \pi)$$
(B.14)

with terminal condition $F(T, x, y) = \widetilde{U}(x, y) \coloneqq \frac{\delta_i}{\delta_i - 1} \left(xy^{-\frac{\theta_i}{n}} \right)^{\frac{\delta_i - 1}{\delta_i}}$.

Now we consider the function G given in (B.12) and apply the Itô-Doeblin formula (Theorem 2.1) to G, $\widetilde{X}^{i,\pi}$, and \widetilde{Y}^{-i} , where $\pi \in \mathcal{A}$, over the interval [t,T] for some $t \in [0,T]$. It follows

$$\begin{split} & G(T, \widetilde{X}_{T}^{i,\pi}, \widetilde{Y}_{T}^{-i}) \\ &= G(t, \widetilde{X}_{t}^{i,\pi}, \widetilde{Y}_{t}^{-i}) + \int_{t}^{T} G_{t} \mathrm{d}s + \int_{t}^{T} G_{x} \mathrm{d}\widetilde{X}_{s}^{i,\pi} + \int_{t}^{T} G_{y} \mathrm{d}\widetilde{Y}_{s}^{-i} \\ &\quad + \frac{1}{2} \int_{t}^{T} G_{xx} \mathrm{d}\langle \widetilde{X}^{i,\pi} \rangle_{s} + \int_{t}^{T} G_{xy} \mathrm{d}\langle \widetilde{X}^{i,\pi}, \widetilde{Y}^{-i} \rangle_{s} + \frac{1}{2} \int_{t}^{T} G_{yy} \mathrm{d}\langle \widetilde{Y}^{-i} \rangle_{s} \\ &= G(t, \widetilde{X}_{t}^{i,\pi}, \widetilde{Y}_{t}^{-i}) + \int_{t}^{T} \mathcal{T}G(s, \widetilde{X}_{s}^{i,\pi}, \widetilde{Y}_{s}^{-i}, \pi(s)) \mathrm{d}s + \int_{t}^{T} \left(G_{x} \widetilde{X}_{s}^{i,\pi} \pi(s)^{\top} \sigma + G_{y} \widetilde{Y}_{s}^{-i} \sum_{j \neq i} (\pi^{j})^{\top} \sigma \right) \mathrm{d}W(s) . \end{split}$$

By G_t , we denote the first order partial derivative of G with respect to t. The other partial derivatives with respect to t, x, and y are denoted similarly. Let us now assume that the process

$$\left(\int_0^t \left(G_x(s, \widetilde{X}_s^{i,\pi}, \widetilde{Y}_s^{-i})\widetilde{X}_s^{i,\pi}\pi(s)^\top \sigma + G_y(s, \widetilde{X}_s^{i,\pi}, \widetilde{Y}_s^{-i})\widetilde{Y}_s^{-i}\sum_{j\neq i} (\pi^j)^\top \sigma\right) \mathrm{d}W(s)\right)_{t\in[0,T]}$$
(B.15)

is a martingale for any admissible π . Then we can apply the conditional expectation $\mathbb{E}^{t,x,y}$ on both sides of the above representation of $G(T, \widetilde{X}_T^{i,\pi}, \widetilde{Y}_T^{-i})$ to obtain

$$\mathbb{E}^{t,x,y} \left[\frac{\delta_i}{\delta_i - 1} \left(\widetilde{X}_T^{i,\pi} (\widetilde{Y}_T^{-i})^{-\frac{\theta_i}{n}} \right)^{\frac{\delta_i - 1}{\delta_i}} \right] = \mathbb{E}^{t,x,y} \left[G(t, \widetilde{X}_t^{i,\pi}, \widetilde{Y}_t^{-i}) + \int_t^T \mathcal{T}G(s, \widetilde{X}_s^{i,\pi}, \widetilde{Y}_s^{-i}, \pi(s)) \mathrm{d}s \right] \\
\leq G(t, x, y). \tag{B.16}$$

On the left-hand side, we used that $G(T, x, y) = \frac{\delta_i}{\delta_i - 1} \left(xy^{-\frac{\theta_i}{n}}\right)^{\frac{\delta_i - 1}{\delta_i}}$ for all $x, y \in (0, \infty)$. In the first equality, we used the assumption that (B.15) is a martingale which implies that the conditional expectation of the stochastic integral vanishes. Further, the inequality in the second step follows since G solves the HJB equation and thus, the supremum of $\mathcal{T}G(t, x, y, \pi)$ over all $\pi \in \mathbb{R}^d$ is equal to 0. Therefore, it follows by the definition of the value function J that

$$J(t,x,y) = \sup_{\pi \in \mathcal{A}} \mathbb{E}^{t,x,y} \left[\frac{\delta_i}{\delta_i - 1} \left(\widetilde{X}_T^{i,\pi} (\widetilde{Y}_T^{-i})^{-\frac{\theta_i}{n}} \right)^{\frac{\delta_i - 1}{\delta_i}} \right] \le G(t,x,y),$$
(B.17)

i.e., the first assertion of the lemma holds.

Now let $u^* = u^*(t, x, y)$ be the maximizer of the function h given in the lemma. Noticing that the operator \mathcal{T} , applied to G and evaluated at t, x, y, and π , can be written as a sum of $h(\pi)$ and some expression independent of π (that takes the form $C \cdot G(t, x, y)$ for some constant $C \in \mathbb{R}$), we can deduce that using $\pi = u^*$ gives equality in (B.16) and thus, equality in (B.17). Hence, $\pi_t^* = u^*(t, X_t^*, \tilde{Y}_t^{-i})$ is an optimal control for the best response problem for (7.37) and G is equal to the value function J. By X^* , we denote the solution to the stochastic differential equation (B.13) given in the lemma. To verify the uniqueness of π^* , note that any optimal control satisfies the Bellman optimality principle. Thus, the optimal strategies are precisely the extremal points of the supremum in (B.14). Since the function inside the supremum has a unique maximizer (which we have shown in the proof of Theorem 7.12), π^* is the unique (up to modifications) optimal solution to the best response problem associated to (7.37).

The last step remaining in this proof is to show that the process (B.15) is in fact a martingale. For the moment, assume that π is bounded, i.e., that there exists a constant K > 0 such that $\|\pi(t)\| \leq K \mathbb{P}$ -almost surely for all $t \in [0, T]$. First, we notice that it suffices to prove that both summands in (B.15) are martingales to prove that the sum itself is a martingale. According to Theorem 4.3.1 in Shreve (2004), it suffices to show that the integrand is square integrable, i.e., that

$$\mathbb{E}\left[\int_0^T \left\|G_x(t, \widetilde{X}_t^{i, \pi}, \widetilde{Y}_t^{-i}) \widetilde{X}_t^{i, \pi} \pi(t)^\top \sigma\right\|^2 \mathrm{d}t\right] < \infty.$$

to prove that the first summand defines a martingale. Further, since σ is deterministic and constant and π is assumed to be bounded, it only remains to show that

$$\mathbb{E}\left[\int_0^T \left(G_x(t, \tilde{X}_t^{i,\pi}, \tilde{Y}_t^{-i}) \tilde{X}_t^{i,\pi}\right)^2 \mathrm{d}t\right] = \int_0^T \mathbb{E}\left[\left(G_x(t, \tilde{X}_t^{i,\pi}, \tilde{Y}_t^{-i}) \tilde{X}_t^{i,\pi}\right)^2\right] \mathrm{d}t < \infty.$$
(B.18)

We used Fubini's theorem to interchange integral and expectation in the previous line. Using $x \cdot G_x(t, x, y) = e^{\rho(T-t)} \left(xy^{-\frac{\theta_i}{n}}\right)^{\frac{\delta_i-1}{\delta_i}}$ for any $t \in [0, T], x, y \in (0, \infty)$, and recalling that

$$\begin{split} \widetilde{X}_t^{i,\pi} = & x_0^i \exp\left(\int_0^t \pi(s)^\top \Big(\mu + \Big(\frac{1}{n}A - \frac{1}{2}\sigma\sigma^\top\Big)\pi(s) + \frac{1}{n}\Big(1 - \frac{\theta_i}{n}\Big)A\sum_{j\neq i}\pi^j\Big)\mathrm{d}s + \int_0^t \pi(s)^\top\sigma\mathrm{d}W(s)\Big), \\ \widetilde{Y}_t^{-i} = & y_0^i \exp\left(\int_0^t \Big(\sum_{j\neq i}\pi^j\Big)^\top \Big(\mu + \frac{1}{n}A\sum_{j\neq i}\pi^j\Big)\mathrm{d}s + \int_0^t \Big(\sum_{j\neq i}\pi^j\Big)^\top\sigma\mathrm{d}W(s) - \frac{1}{2}\int_0^t \sum_{j\neq i}(\pi^j)^\top\sigma\sigma^\top\pi^j\mathrm{d}s\Big), \end{split}$$

we obtain the following bound for the expectation in (B.18)

$$\begin{split} & e^{-2\rho(T-t)} \Big(x_0^i(y_0^i)^{-\frac{\theta_i}{n}} \Big)^{-\frac{2(\delta_i-1)}{\delta_i}} \mathbb{E} \Big[\Big(G_x(t, \tilde{X}_t^{i,\pi}, \tilde{Y}_t^{-i}) \tilde{X}_t^{i,\pi} \Big)^2 \Big] \\ &= \mathbb{E} \Big[\exp \Big(\frac{2(\delta_i - 1)}{\delta_i} \Big\{ \int_0^t \pi(s)^\top \Big(\mu + \Big(\frac{1}{n} A - \frac{1}{2} \sigma \sigma^\top \Big) \pi(s) + \frac{1}{n} \Big(1 - \frac{\theta_i}{n} \Big) A \sum_{j \neq i} \pi^j \Big) \mathrm{d}s + \int_0^t \pi(s)^\top \sigma \mathrm{d}W(s) \Big\} \\ &- \frac{2\theta_i(\delta_i - 1)}{n\delta_i} \Big\{ \int_0^t \Big(\Big(\sum_{j \neq i} \pi^j \Big)^\top \Big(\mu + \frac{1}{n} A \sum_{j \neq i} \pi^j \Big) - \frac{1}{2} \sum_{j \neq i} (\pi^j)^\top \sigma \sigma^\top \pi^j \Big) \mathrm{d}s + \int_0^t \Big(\sum_{j \neq i} \pi^j \Big)^\top \sigma \mathrm{d}W(s) \Big\} \Big) \Big] \\ &= \exp \Big(- \frac{2\theta_i(\delta_i - 1)}{n\delta_i} \Big(\Big(\sum_{j \neq i} \pi^j \Big)^\top \Big(\mu + \frac{1}{n} A \sum_{j \neq i} \pi^j \Big) - \frac{1}{2} \sum_{j \neq i} (\pi^j)^\top \sigma \sigma^\top \pi^j \Big) \mathrm{d}s + \int_0^t \pi(s)^\top \sigma \mathrm{d}W(s) \Big\} \Big) \Big] \\ &\cdot \mathbb{E} \Big[\exp \Big(\frac{2(\delta_i - 1)}{\delta_i} \Big\{ \int_0^t \pi(s)^\top \Big(\mu + \Big(\frac{1}{n} A - \frac{1}{2} \sigma \sigma^\top \Big) \pi(s) + \frac{1}{n} \Big(1 - \frac{\theta_i}{n} \Big) A \sum_{j \neq i} \pi^j \Big) \mathrm{d}s + \int_0^t \pi(s)^\top \sigma \mathrm{d}W(s) \Big\} \\ &- \frac{2\theta_i(\delta_i - 1)}{n\delta_i} \int_0^t \Big(\sum_{j \neq i} \pi^j \Big)^\top \sigma \mathrm{d}W(s) \Big) \Big] \\ &=: \exp(C \cdot t) \mathbb{E}_{\widetilde{\mathbb{Q}}} \Big[\exp \Big(\frac{2(\delta_i - 1)}{\delta_i} \int_0^t \pi(s)^\top \Big(\mu + \Big(\frac{1}{n} A - \frac{1}{2} \sigma \sigma^\top \Big) \pi(s) + \frac{1}{n} \Big(1 - \frac{\theta_i}{n} \Big) A \sum_{j \neq i} \pi^j \Big) \mathrm{d}s \Big\}$$

$$\cdot \exp\left(\frac{2(\delta_i - 1)^2}{\delta_i^2} \int_0^t \left(\pi(s) - \frac{\theta_i}{n} \sum_{j \neq i} \pi^j\right)^\top \sigma \sigma^\top \left(\pi(s) - \frac{\theta_i}{n} \sum_{j \neq i} \pi^j\right) \mathrm{d}s\right)\right]$$
(B.19)

$$\leq \exp(C \cdot t).$$

In (B.19), we abbreviated the constant in the first exponential factor by C and introduced an equivalent probability measure $\widetilde{\mathbb{Q}} \sim \mathbb{P}$ with density

$$\frac{\mathrm{d}\widetilde{\mathbb{Q}}}{\mathrm{d}\mathbb{P}} = \exp\left(\frac{2(\delta_i - 1)}{\delta_i} \int_0^t \left(\pi(s) - \frac{\theta_i}{n} \sum_{j \neq i} \pi^j\right)^\top \sigma \mathrm{d}W(s) - \frac{2(\delta_i - 1)^2}{\delta_i^2} \int_0^t \left(\pi(s) - \frac{\theta_i}{n} \sum_{j \neq i} \pi^j\right)^\top \sigma \sigma^\top \left(\pi(s) - \frac{\theta_i}{n} \sum_{j \neq i} \pi^j\right) \mathrm{d}s\right). \tag{B.21}$$

Note that, since π was assumed to be bounded, (B.21) is indeed a density. Moreover, in (B.20) we used that π is bounded and that the other expressions appearing in the integral are constant. Thus, the integrand can be bounded by a constant. In combination with the constant C introduced in (B.19), we obtain the bound (B.20). Thus, the integral in (B.18) is finite and the first summand of (B.15) is a martingale. Proving that the second summand in (B.15) is also a martingale proceeds analogously since

$$y \cdot G_y(t, x, y) = -\frac{\theta_i}{n} e^{\rho(T-t)} \left(x y^{-\frac{\theta_i}{n}} \right)^{\frac{\delta_i - 1}{\delta_i}}$$

for all $t \in [0,T]$ and $x, y \in (0,\infty)$. Thus, up to a constant, the expectation

$$\mathbb{E}\left[\int_0^T \left(G_y(t, \widetilde{X}_t^{i, \pi}, \widetilde{Y}_t^{-i}) \widetilde{Y}_t^{-i}\right)^2 \mathrm{d}t\right]$$

coincides with (B.18) and is, thus, finite. Finally, let us recall that the candidate optimal strategy determined in the proof of Theorem 7.12 is constant and not at the boundary of the restriction interval. Hence, the dominated convergence theorem for stochastic integrals (Theorem 32 in Protter, 2005, p.176) yields that the assumption that π is bounded is not a restriction (see also Korn and Desmettre (2014), p.294; Pham, 2009, pp.47–48). This concludes the proof.

(B.20)

APPENDIX C

Additional material for Chapter 8

The following chapter contains additional material for Chapter 8. Section C.1 comprises proofs of Lemma 8.8 and Proposition 8.9, which provide an explicit representation of the Nash equilibrium in terms of terminal wealth of two agents using logarithmic utility. In Section C.2, we explain how the *n*-dimensional fixed point problem (8.6) can be reduced to an (n - 1)-dimensional fixed point problem. This technique is used in Section C.3 to solve the fixed point problem for three agents. Afterwards, we apply the result for three agents to the special case of logarithmic utility. At the end of Section C.3, we illustrate the Nash equilibrium for three agents under logarithmic utility for varying values of the state price density Z_T and for different parameter choices. The fourth and last section of this chapter contains the proofs of four lemmas used to prove Theorem 8.11 and Theorem 8.16.

C.1. PROOFS OF LEMMA 8.8 AND PROPOSITION 8.9

First, we display the proof of Lemma 8.8, in which the Lagrange multipliers from (8.12) are determined explicitly if both agents use the natural logarithm as their utility function.

Proof (Lemma 8.8). The proof consists of two steps. In the first step, we compute the expected values $\mathbb{E}[X_i^*Z_T]$, i = 1, 2. Afterwards, we solve a two-dimensional system of nonlinear equations to determine the Lagrange multipliers λ_1 and λ_2 .

Let us determine the expected value $\mathbb{E}[Z_T X_1^*]$. It follows

$$X_1^* Z_T = \left(\frac{1}{\lambda_1} \mathbb{1}\left\{\lambda_1 \beta_2 \le \lambda_2\right\} + \frac{\beta_2}{\lambda_2} \mathbb{1}\left\{\lambda_1 \beta_2 > \lambda_2\right\}\right) \mathbb{1}\left\{Z_T \le \chi_{\alpha_1}\right\} + \frac{1}{\lambda_1} \mathbb{1}\left\{Z_T > \chi_{\alpha_1}\right\}.$$

Hence, since χ_{α_1} denotes the α_1 -quantile of Z_T ,

$$\mathbb{E}[X_1^* Z_T] = \alpha_1 \left(\frac{1}{\lambda_1} \mathbb{1} \left\{ \lambda_1 \beta_2 \le \lambda_2 \right\} + \frac{\beta_2}{\lambda_2} \mathbb{1} \left\{ \lambda_1 \beta_2 > \lambda_2 \right\} \right) + \frac{1}{\lambda_1} (1 - \alpha_1)$$

=: $\alpha_1 \left(\gamma_1 \mathbb{1} \left\{ \gamma_1 \ge \beta_2 \gamma_2 \right\} + \beta_2 \gamma_2 \mathbb{1} \left\{ \gamma_1 < \beta_2 \gamma_2 \right\} \right) + (1 - \alpha_1) \gamma_1$
=: $f_1(\gamma_1, \gamma_2)$,

where we defined $\gamma_i \coloneqq \lambda_i^{-1}$, i = 1, 2, to simplify notation. Analogously, we obtain

$$\mathbb{E}[X_2^*Z_T] = \alpha_2 \left(\gamma_2 \mathbb{1}\left\{\gamma_2 \ge \beta_1 \gamma_1\right\} + \beta_1 \gamma_1 \mathbb{1}\left\{\gamma_2 < \beta_1 \gamma_1\right\}\right) + (1 - \alpha_2) \gamma_2$$
$$=: f_2(\gamma_1, \gamma_2).$$

Now it remains to solve the system of equations

$$f_1(\gamma_1, \gamma_2) = x_0^1, f_2(\gamma_1, \gamma_2) = x_0^2$$

Clearly, $f_1(\gamma_1, \gamma_2)$ can be rewritten as

$$f_1(\gamma_1, \gamma_2) = \gamma_1 \mathbb{1} \{ \gamma_1 \ge \beta_2 \gamma_2 \} + (\alpha_1 \beta_2 \gamma_2 + (1 - \alpha_1) \gamma_1) \mathbb{1} \{ \gamma_1 < \beta_2 \gamma_2 \} \stackrel{!}{=} x_0^1.$$

For fixed γ_2 , the unique solution γ_1 to this equation is given by (using a case distinction based on whether or not $x_0^1 \ge \beta_2 \gamma_2$ and the sketch of f_1 below)

$$\gamma_1 = x_0^1 \mathbb{1} \left\{ x_0^1 \ge \beta_2 \gamma_2 \right\} + \frac{x_0^1 - \alpha_1 \beta_2 \gamma_2}{1 - \alpha_1} \mathbb{1} \left\{ x_0^1 < \beta_2 \gamma_2 \right\}.$$
(C.1)



Since the representation (C.1) of γ_1 depends on γ_2 , we insert γ_1 into the equation $f_2(\gamma_1, \gamma_2) = x_0^2$ to find an explicit solution for γ_2 , which can then be inserted into (C.1) to determine an explicit solution for γ_1 as well. First, we rewrite $f_2(\gamma_1, \gamma_2)$ as

$$f_2(\gamma_1, \gamma_2) = \gamma_2 \mathbb{1} \{ \gamma_2 \ge \beta_1 \gamma_1 \} + (\alpha_2 \beta_1 \gamma_1 + (1 - \alpha_2) \gamma_2) \mathbb{1} \{ \gamma_2 < \beta_1 \gamma_1 \}.$$

Now it remains to find an explicit representation for γ_1 . Using (C.1), it follows

$$\{\gamma_2 \ge \beta_1 \gamma_1\} = \left(\left\{\gamma_2 \ge \beta_1 x_0^1\right\} \cap \left\{x_0^1 \ge \beta_2 \gamma_2\right\}\right) \cup \left(\left\{\gamma_2 \ge \beta_1 \frac{x_0^1 - \alpha_1 \beta_2 \gamma_2}{1 - \alpha_1}\right\} \cap \left\{x_0^1 < \beta_2 \gamma_2\right\}\right)$$

$$= \left\{ \beta_1 \beta_2 x_0^1 \le \beta_2 \gamma_2 \le x_0^1 \right\} \cup \left\{ \beta_2 \gamma_2 > x_0^1 \right\}$$

= $\left\{ \gamma_2 \ge \beta_1 x_0^1 \right\}.$ (C.2)

(C.2) holds since

$$\gamma_2 \ge \beta_1 \frac{x_0^1 - \alpha_1 \beta_2 \gamma_2}{1 - \alpha_1} \iff \gamma_2 \ge \frac{\beta_1}{1 - \alpha_1 (1 - \beta_1 \beta_2)} x_0^1$$

and

$$\beta_1 \beta_2 < 1 \iff (1 - \alpha_1) \beta_1 \beta_2 < 1 - \alpha_1$$
$$\iff \beta_1 \beta_2 < 1 - \alpha_1 (1 - \beta_1 \beta_2)$$
$$\iff \frac{\beta_1}{1 - \alpha_1 (1 - \beta_1 \beta_2)} < \frac{1}{\beta_2}.$$

Since $\frac{x_0^1}{\beta_2} > x_0^1 > \beta_1 x_0^1$, γ_1 given in (C.1) takes the value x_0^1 if $\gamma_2 < \beta_1 x_0^1$. Therefore,

$$f_2(\gamma_1, \gamma_2) = \gamma_2 \mathbb{1}\left\{\gamma_2 \ge \beta_1 x_0^1\right\} + (\alpha_2 \beta_1 x_0^1 + (1 - \alpha_2) \gamma_2) \mathbb{1}\left\{\gamma_2 < \beta_1 x_0^1\right\} \stackrel{!}{=} x_0^2.$$

The previous equation has the following unique solution (using a case distinction on whether or not $x_0^2 \ge \beta_1 x_0^1$ and the sketch of f_2 below)

$$\gamma_2 = x_0^2 \mathbb{1}\left\{x_0^2 \ge \beta_1 x_0^1\right\} + \frac{x_0^2 - \alpha_2 \beta_1 x_0^1}{1 - \alpha_2} \mathbb{1}\left\{x_0^2 < \beta_1 x_0^1\right\},\tag{C.3}$$

which is positive due to the assumption $x_0^2 > \alpha_2 \beta_1 x_0^1$ of the lemma.



Using (C.3), it follows

$$\{x_0^1 \ge \beta_2 \gamma_2\} = \left(\left\{ x_0^1 \ge \beta_2 x_0^2 \right\} \cap \left\{ x_0^2 \ge \beta_1 x_0^1 \right\} \right) \cup \left(\left\{ x_0^1 \ge \beta_2 \frac{x_0^2 - \alpha_2 \beta_1 x_0^1}{1 - \alpha_2} \right\} \cap \left\{ x_0^2 < \beta_1 x_0^1 \right\} \right)$$
$$= \left\{ \beta_1 \beta_2 x_0^2 \le \beta_1 x_0^1 \le x_0^2 \right\} \cup \left\{ x_0^2 < \beta_1 x_0^1 \right\}$$
(C.4)

$$= \left\{ x_0^1 \ge \beta_2 x_0^2 \right\}. \tag{C.5}$$

(C.4) follows analogously to (C.2). Moreover, since $\frac{x_0^2}{\beta_1} > x_0^2 > \beta_2 x_0^2$, γ_2 takes the value x_0^2 if $x_0^1 < \beta_2 x_0^2$. Now we can insert (C.3) into (C.1) and apply (C.5) to find an explicit representation

of γ_1 . It follows

$$\gamma_1 = x_0^1 \mathbb{1}\left\{x_0^1 \ge \beta_2 x_0^2\right\} + \frac{x_0^1 - \alpha_1 \beta_2 x_0^2}{1 - \alpha_1} \mathbb{1}\left\{x_0^1 < \beta_2 x_0^2\right\}.$$

Note that the assumption $x_0^1 > \alpha_1 \beta_2 x_0^2$ of the lemma implies $\gamma_1 > 0$. Using the definition of $\gamma_i = \lambda_i^{-1}$, i = 1, 2, concludes our proof.

With the Lagrange multipliers from the previous proof, we can now give an explicit representation of the fixed point for two agents using the natural logarithm as their utility functions. Thus, the proof of Proposition 8.9 proceeds as follows.

Proof (Proposition 8.9). a) If $x_0^1 \ge \beta_2 x_0^2$ and $x_0^2 \ge \beta_1 x_0^1$, using Lemma 8.8, the Lagrange multipliers are given by

$$\lambda_1 = \frac{1}{x_0^1}, \ \lambda_2 = \frac{1}{x_0^2}.$$

Hence, it follows that

$$\lambda_1\beta_2 \leq \lambda_2, \, \lambda_2\beta_1 \leq \lambda_1.$$

Therefore, (8.12) implies

$$X_1^* = \frac{1}{\lambda_1 Z_T} = x_0^1 Z_T^{-1}, \ X_2^* = \frac{1}{\lambda_2 Z_T} = x_0^2 Z_T^{-1}.$$

b) If $x_0^1 < \beta_2 x_0^2$, it follows

$$x_0^2 > \frac{x_0^1}{\beta_2} > x_0^1 > \beta_1 x_0^1$$

and thus,

$$\lambda_1 = \frac{1 - \alpha_1}{x_0^1 - \alpha_1 \beta_2 x_0^2}, \, \lambda_2 = \frac{1}{x_0^2}.$$

This implies

$$\beta_2 \lambda_1 > \lambda_2, \ \beta_1 \lambda_2 \le \lambda_1$$

and therefore, using (8.12),

$$\begin{aligned} X_1^* &= \frac{\beta_2}{\lambda_2 Z_T} \mathbb{1} \left\{ Z_T \le \chi_{\alpha_1} \right\} + \frac{1}{\lambda_1 Z_T} \mathbb{1} \left\{ Z_T > \chi_{\alpha_1} \right\} \\ &= \beta_2 x_0^2 Z_T^{-1} \mathbb{1} \left\{ Z_T \le \chi_{\alpha_1} \right\} + \frac{x_0^1 - \alpha_1 \beta_2 x_0^2}{1 - \alpha_1} Z_T^{-1} \mathbb{1} \left\{ Z_T > \chi_{\alpha_1} \right\} \\ X_2^* &= \frac{1}{\lambda_2 Z_T} = x_0^2 Z_T^{-1}. \end{aligned}$$

c) The proof in case c) follows, due to symmetry, analogously to case b).

C.2. DIMENSIONAL REDUCTION OF THE FIXED POINT PROBLEM

In the following, we demonstrate how to reduce the *n*-dimensional fixed point problem (8.6) to an (n-1)-dimensional fixed point problem with a similar structure. We use the abbreviation $I_i = I_i(\lambda_i Z_T), i = 1, ..., n$, throughout this section. We can write the fixed point problem (8.6) as

$$X_{i} = \left(I_{i}\mathbb{1}\left\{I_{i} \geq X_{\beta}^{-i}\right\} + X_{\beta}^{-i}\mathbb{1}\left\{I_{i} < X_{\beta}^{-i}\right\}\right)\mathbb{1}\left\{Z_{T} \leq \chi_{\alpha_{i}}\right\} + I_{i}\mathbb{1}\left\{Z_{T} > \chi_{\alpha_{i}}\right\}$$
(C.6)
$$= \max\left\{I_{i}, X_{\beta}^{-i}\right\}\mathbb{1}\left\{Z_{T} \leq \chi_{\alpha_{i}}\right\} + I_{i}\mathbb{1}\left\{Z_{T} > \chi_{\alpha_{i}}\right\},$$

 $i = 1, \ldots, n$, where we defined

$$X_{\beta}^{-j_1,\dots,j_k} \coloneqq \sum_{\substack{j=1\\ j \notin \{j_1,\dots,j_k\}}}^n \beta_j X_j, \ 0 \le k \le n, \ j_\ell \in \{1,\dots,n\}, \ 1 \le \ell \le k.$$

In the following, we insert X_n into X_i for all $1 \le i \le n-1$ to reduce the *n*-dimensional fixed point problem given by (C.6) to an (n-1)-dimensional fixed point problem. If we are able to solve the reduced problem, we can easily determine X_n by inserting the solution X_1, \ldots, X_{n-1} to the reduced problem.

Without loss of generality, we assume that $\alpha_1 \leq \alpha_2 \leq \ldots \leq \alpha_n$. Moreover, let $1 \leq i \leq n-1$. First, we insert X_n into the first indicator function in (C.6) to obtain

$$\mathbb{1}\left\{I_{i} \geq X_{\beta}^{-i}\right\} = \mathbb{1}\left\{I_{i} \geq X_{\beta}^{-i,n} + \beta_{n}\left(I_{n}\mathbb{1}\left\{I_{n} \geq X_{\beta}^{-n}\right\} + X_{\beta}^{-n}\mathbb{1}\left\{I_{n} < X_{\beta}^{-n}\right\}\right)\mathbb{1}\left\{Z_{T} \leq \chi_{\alpha_{n}}\right\} + \beta_{n}I_{n}\mathbb{1}\left\{Z_{T} > \chi_{\alpha_{n}}\right\}\right\} = \mathbb{1}\left\{Z_{T} \leq \chi_{\alpha_{n}}\right\}\left(\mathbb{1}\left\{I_{i} \geq X_{\beta}^{-i,n} + \beta_{n}I_{n}\right\}\mathbb{1}\left\{I_{n} \geq X_{\beta}^{-n}\right\} + \mathbb{1}\left\{I_{i} \geq X_{\beta}^{-i,n} + \beta_{n}X_{\beta}^{-n}\right\}\mathbb{1}\left\{I_{n} < X_{\beta}^{-n}\right\}\right) + \mathbb{1}\left\{Z_{T} > \chi_{\alpha_{n}}\right\}\mathbb{1}\left\{I_{i} \geq X_{\beta}^{-i,n} + \beta_{n}I_{n}\right\}.$$

The second indicator function can be simplified analogously. Since we assumed that $\alpha_1, \ldots, \alpha_n$ are in ascending order, it follows

$$\mathbb{1}\left\{I_i \ge X_{\beta}^{-i}\right\} \cdot \mathbb{1}\left\{Z_T \le \chi_{\alpha_i}\right\} \\
= \mathbb{1}\left\{Z_T \le \chi_{\alpha_i}\right\} \left(\mathbb{1}\left\{I_i \ge X_{\beta}^{-i,n} + \beta_n I_n\right\} \mathbb{1}\left\{I_n \ge X_{\beta}^{-n}\right\} + \mathbb{1}\left\{I_i \ge X_{\beta}^{-i,n} + \beta_n X_{\beta}^{-n}\right\} \mathbb{1}\left\{I_n < X_{\beta}^{-n}\right\}\right).$$

Now we insert X_n into the summand $X_{\beta}^{-i} \mathbb{1}\left\{I_i < X_{\beta}^{-i}\right\} \mathbb{1}\left\{Z_T \le \chi_{\alpha_i}\right\}$ of (C.6) to obtain

$$\begin{split} X_{\beta}^{-i} \mathbb{1} \left\{ I_{i} < X_{\beta}^{-i} \right\} \mathbb{1} \left\{ Z_{T} \leq \chi_{\alpha_{i}} \right\} \\ = \mathbb{1} \left\{ Z_{T} \leq \chi_{\alpha_{i}} \right\} \left(X_{\beta}^{-i,n} + \left(\beta_{n} I_{n} \mathbb{1} \left\{ I_{n} \geq X_{\beta}^{-n} \right\} + \beta_{n} X_{\beta}^{-n} \mathbb{1} \left\{ I_{n} < X_{\beta}^{-n} \right\} \right) \right) \\ \cdot \left(\mathbb{1} \left\{ I_{i} < X_{\beta}^{-i,n} + \beta_{n} I_{n} \right\} \mathbb{1} \left\{ I_{n} \geq X_{\beta}^{-n} \right\} + \mathbb{1} \left\{ I_{i} < X_{\beta}^{-i,n} + \beta_{n} X_{\beta}^{-n} \right\} \mathbb{1} \left\{ I_{n} < X_{\beta}^{-n} \right\} \right) \\ = \mathbb{1} \left\{ Z_{T} \leq \chi_{\alpha_{i}} \right\} \left(\left(X_{\beta}^{-i,n} + \beta_{n} I_{n} \right) \mathbb{1} \left\{ I_{i} < X_{\beta}^{-i,n} + \beta_{n} I_{n} \right\} \mathbb{1} \left\{ I_{n} \geq X_{\beta}^{-n} \right\} \\ + \left(X_{\beta}^{-i,n} + \beta_{n} X_{\beta}^{-n} \right) \mathbb{1} \left\{ I_{i} < X_{\beta}^{-i,n} + \beta_{n} X_{\beta}^{-n} \right\} \mathbb{1} \left\{ I_{n} < X_{\beta}^{-n} \right\} \right). \end{split}$$

Finally, we obtain the following implicit representation of X_i in terms of only X_ℓ , $1 \leq \ell \leq n-1$:

$$\begin{aligned} X_{i} &= \left(\mathbbm{1}\left\{I_{n} \geq X_{\beta}^{-n}\right\} \left(I_{i} \mathbbm{1}\left\{I_{i} \geq X_{\beta}^{-i,n} + \beta_{n}I_{n}\right\} + \left(X_{\beta}^{-i,n} + \beta_{n}I_{n}\right)\mathbbm{1}\left\{I_{i} < X_{\beta}^{-i,n} + \beta_{n}I_{n}\right\}\right) \\ &+ \mathbbm{1}\left\{I_{n} < X_{\beta}^{-n}\right\} \left(I_{i} \mathbbm{1}\left\{I_{i} \geq X_{\beta}^{-i,n} + \beta_{n}X_{\beta}^{-n}\right\} + \left(X_{\beta}^{-i,n} + \beta_{n}X_{\beta}^{-n}\right)\mathbbm{1}\left\{I_{i} < X_{\beta}^{-i,n} + \beta_{n}X_{\beta}^{-n}\right\}\right)\right) \\ &\cdot \mathbbm{1}\left\{Z_{T} \leq \chi_{\alpha_{i}}\right\} + I_{i} \cdot \mathbbm{1}\left\{Z_{T} > \chi_{\alpha_{i}}\right\}.\end{aligned}$$

In order to find an explicit representation of X_i , we define the right-hand side of the previous representation as $f_i(X_i)$ and solve the one-dimensional fixed point problem $f_i(X_i) = X_i$. First, we rewrite the indicator functions in the definition of f_i in terms of X_i to receive

$$\begin{split} f_i(X_i) = & \mathbb{I}\left\{Z_T \le \chi_{\alpha_i}\right\} \cdot \left(\mathbb{I}\left\{X_i \le \frac{1}{\beta_i}\left(I_n - X_{\beta}^{-i,n}\right)\right\} \left(I_i \mathbb{I}\left\{I_i \ge X_{\beta}^{-i,n} + \beta_n I_n\right\}\right) \\ & + \left(X_{\beta}^{-i,n} + \beta_n I_n\right) \mathbb{I}\left\{I_i < X_{\beta}^{-i,n} + \beta_n I_n\right\}\right) \\ & + \mathbb{I}\left\{X_i > \frac{1}{\beta_i}\left(I_n - X_{\beta}^{-i,n}\right)\right\} \left(I_i \mathbb{I}\left\{X_i \le \frac{1}{\beta_i \beta_n}\left(I_i - (1 + \beta_n) X_{\beta}^{-i,n}\right)\right\} \\ & + \left(\beta_i \beta_n X_i + (1 + \beta_n) X_{\beta}^{-i,n}\right) \mathbb{I}\left\{X_i > \frac{1}{\beta_i \beta_n}\left(I_i - (1 + \beta_n) X_{\beta}^{-i,n}\right)\right\}\right) \\ & + I_i \cdot \mathbb{I}\left\{Z_T > \chi_{\alpha_i}\right\}. \end{split}$$

In the following, the fixed point, which turns out to be unique, is determined via case distinction. If $Z_T > \chi_{\alpha_i}$, the unique fixed point is apparently given by $X_i = I_i$. Hence, we assume $Z_T \leq \chi_{\alpha_i}$ for the remainder of the fixed point search.

Case 1: $I_i \ge X_{\beta}^{-i,n} + \beta_n I_n$

In this case we have

$$\frac{1}{\beta_i \beta_n} \left(I_i - (1 + \beta_n) X_{\beta}^{-i,n} \right) \ge \frac{1}{\beta_i} \left(I_n - X_{\beta}^{-i,n} \right)$$

and therefore, we need to determine X_i such that

$$X_{i} = I_{i} \mathbb{1} \left\{ X_{i} \leq \frac{1}{\beta_{i}\beta_{n}} \left(I_{i} - (1+\beta_{n})X_{\beta}^{-i,n} \right) \right\}$$

+ $\left(\beta_{i}\beta_{n}X_{i} + (1+\beta_{n})X_{\beta}^{-i,n} \right) \mathbb{1} \left\{ X_{i} > \frac{1}{\beta_{i}\beta_{n}} \left(I_{i} - (1+\beta_{n})X_{\beta}^{-i,n} \right) \right\}$
= $\max \left\{ I_{i}, \beta_{i}\beta_{n}X_{i} + (1+\beta_{n})X_{\beta}^{-i,n} \right\}.$

Since the right-hand side is continuous in X_i , constant for $X_i \leq \frac{1}{\beta_i \beta_n} \left(I_i - (1 + \beta_n) X_{\beta}^{-i,n} \right)$ and linear with positive slope smaller than 1 otherwise, the unique fixed point is given by

$$X_i = \begin{cases} I_i, & I_i \leq \frac{1}{\beta_i \beta_n} \left(I_i - (1 + \beta_n) X_{\beta}^{-i,n} \right), \\ \frac{1 + \beta_n}{1 - \beta_i \beta_n} X_{\beta}^{-i,n}, & I_i > \frac{1}{\beta_i \beta_n} \left(I_i - (1 + \beta_n) X_{\beta}^{-i,n} \right). \end{cases}$$

Case 2: $I_i < X_{\beta}^{-i,n} + \beta_n I_n$

Then it follows that

$$\frac{1}{\beta_i \beta_n} \left(I_i - (1 + \beta_n) X_{\beta}^{-i,n} \right) < \frac{1}{\beta_i} \left(I_n - X_{\beta}^{-i,n} \right)$$

and the fixed point is given as the solution to the following equation

$$X_{i} = \left(X_{\beta}^{-i,n} + \beta_{n}I_{n}\right) \mathbb{1}\left\{X_{i} \leq \frac{1}{\beta_{i}}\left(I_{n} - X_{\beta}^{-i,n}\right)\right\} + \left(\beta_{i}\beta_{n}X_{i} + (1+\beta_{n})X_{\beta}^{-i,n}\right) \mathbb{1}\left\{X_{i} > \frac{1}{\beta_{i}}\left(I_{n} - X_{\beta}^{-i,n}\right)\right\}.$$

The right-hand side is continuous in X_i , constant for $X_i \leq \frac{1}{\beta_i} \left(I_n - X_{\beta}^{-i,n} \right)$ and linear with positive slope smaller than 1 otherwise. Hence, the unique fixed point is given by

$$X_{i} = \begin{cases} X_{\beta}^{-i,n} + \beta_{n}I_{n}, & X_{\beta}^{-i,n} + \beta_{n}I_{n} \leq \frac{1}{\beta_{i}} \left(I_{n} - X_{\beta}^{-i,n} \right), \\ \frac{1+\beta_{n}}{1-\beta_{i}\beta_{n}} X_{\beta}^{-i,n}, & X_{\beta}^{-i,n} + \beta_{n}I_{n} > \frac{1}{\beta_{i}} \left(I_{n} - X_{\beta}^{-i,n} \right). \end{cases}$$

In summary, the unique fixed point is given by

$$\begin{split} X_{i} &= \mathbb{1} \left\{ Z_{T} \leq \chi_{\alpha_{i}} \right\} \left(\mathbb{1} \left\{ I_{i} \geq X_{\beta}^{-i,n} + \beta_{n} I_{n} \right\} \left(I_{i} \mathbb{1} \left\{ I_{i} \geq \frac{1 + \beta_{n}}{1 - \beta_{i} \beta_{n}} X_{\beta}^{-i,n} \right\} \right) \\ &+ \frac{1 + \beta_{n}}{1 - \beta_{i} \beta_{n}} X_{\beta}^{-i,n} \mathbb{1} \left\{ I_{i} < \frac{1 + \beta_{n}}{1 - \beta_{i} \beta_{n}} X_{\beta}^{-i,n} \right\} \right) \\ &+ \mathbb{1} \left\{ I_{i} < X_{\beta}^{-i,n} + \beta_{n} I_{n} \right\} \left(\left(X_{\beta}^{-i,n} + \beta_{n} I_{n} \right) \mathbb{1} \left\{ I_{n} \geq \frac{1 + \beta_{i}}{1 - \beta_{i} \beta_{n}} X_{\beta}^{-i,n} \right\} \right) \\ &+ \frac{1 + \beta_{n}}{1 - \beta_{i} \beta_{n}} X_{\beta}^{-i,n} \mathbb{1} \left\{ I_{n} < \frac{1 + \beta_{i}}{1 - \beta_{i} \beta_{n}} X_{\beta}^{-i,n} \right\} \right) \right) \\ &+ I_{i} \mathbb{1} \left\{ Z_{T} > \chi_{\alpha_{i}} \right\} \\ &= \left(\max \left\{ I_{i}, \frac{1 + \beta_{n}}{1 - \beta_{i} \beta_{n}} X_{\beta}^{-i,n} \right\} \mathbb{1} \left\{ I_{i} \geq X_{\beta}^{-i,n} + \beta_{n} I_{n} \right\} \\ &+ \max \left\{ X_{\beta}^{-i,n} + \beta_{n} I_{n}, \frac{1 + \beta_{n}}{1 - \beta_{i} \beta_{n}} X_{\beta}^{-i,n} \right\} \mathbb{1} \left\{ I_{i} < X_{\beta}^{-i,n} + \beta_{n} I_{n} \right\} \right) \mathbb{1} \left\{ Z_{T} \leq \chi_{\alpha_{i}} \right\} \\ &+ I_{i} \mathbb{1} \left\{ Z_{T} > \chi_{\alpha_{i}} \right\} \\ &= \max \left\{ I_{i}, \frac{1 + \beta_{n}}{1 - \beta_{i} \beta_{n}} X_{\beta}^{-i,n}, X_{\beta}^{-i,n} + \beta_{n} I_{n} \right\} \mathbb{1} \left\{ Z_{T} \leq \chi_{\alpha_{i}} \right\} + I_{i} \mathbb{1} \left\{ Z_{T} > \chi_{\alpha_{i}} \right\}. \end{split}$$

Finally, the original n-dimensional fixed point problem can now be solved using the (n - 1)-dimensional fixed point problem

$$X_{i} = \max\left\{I_{i}, \frac{1+\beta_{n}}{1-\beta_{i}\beta_{n}}X_{\beta}^{-i,n}, X_{\beta}^{-i,n}+\beta_{n}I_{n}\right\}\mathbb{1}\left\{Z_{T} \leq \chi_{\alpha_{i}}\right\} + I_{i}\mathbb{1}\left\{Z_{T} > \chi_{\alpha_{i}}\right\}, \quad (C.7)$$

$$i = 1, \dots, n-1.$$

C.3. Solution of the fixed point problem for three agents

The following theorem gives the unique solution to the fixed point problem (8.6) for three agents.

Theorem C.1. Let n = 3, $\alpha_i = \alpha \in [0, 1]$, and $\beta_i \in (0, \frac{1}{2}]$ for all $i \in \{1, 2, 3\}$. Moreover, let $U_i : (0, \infty) \to \mathbb{R}$ be Inada utility functions and define $I_i \coloneqq I_i(\lambda_i Z_T)$ for notational convenience. Finally, define

$$X_i^*(\lambda_1, \lambda_2, \lambda_3) = \max\left\{I_i, \frac{1+\beta_k}{1-\beta_i\beta_k}\beta_j I_j, \frac{1+\beta_j}{1-\beta_i\beta_j}\beta_k I_k, \beta_j I_j + \beta_k I_k\right\} \mathbb{1}\left\{Z_T \le \chi_\alpha\right\} + I_i \mathbb{1}\left\{Z_T > \chi_\alpha\right\}$$

for pairwise distinct $i, j, k \in \{1, 2, 3\}$, and assume that the system

$$\mathbb{E}[Z_T X_i^*(\lambda_1, \lambda_2, \lambda_3)] = x_0^i, \ i = 1, 2, 3,$$

has a unique solution $(\lambda_1^*, \lambda_2^*, \lambda_3^*) \in (0, \infty)^3$. Then the unique Nash equilibrium in the class of wealth profiles of the form (8.6) is given by $X_i^*(\lambda_1^*, \lambda_2^*, \lambda_3^*)$, i = 1, 2, 3.

Remark C.2. Similar to Remark 8.5, we can ensure that, for fixed $\lambda_j > 0$, $j \neq i$, the equation $\mathbb{E}[Z_T X_i^*] = x_0^i$ has a unique solution $\lambda_i^* > 0$ in terms of λ_j , $j \neq i$, if we assume that $x_0^i \geq \sum_{j \neq i} \beta_j x_0^j$ for all i = 1, 2, 3. Again, this assumption is sufficient, but not necessary, and does also not ensure that the emerging three-dimensional system of equations is (uniquely) solvable.

Proof. First, we rewrite the wealth profile from (8.6) of agent *i* as

$$X_{i} = \max\left\{I_{i}, X_{\beta}^{-i}\right\} \mathbb{1}\left\{Z_{T} \le \chi_{\alpha}\right\} + I_{i} \mathbb{1}\left\{Z_{T} > \chi_{\alpha}\right\}, \ i = 1, 2, 3.$$

The three-dimensional fixed point problem from (8.6) can be reduced to a two-dimensional problem using the reduced problem (C.7). Hence, we need to solve the following two-dimensional fixed point problem

$$X_{1} = \max\left\{I_{1}, \frac{1+\beta_{3}}{1-\beta_{1}\beta_{3}}\beta_{2}X_{2}, \beta_{2}X_{2}+\beta_{3}I_{3}\right\} \mathbb{1}\left\{Z_{T} \leq \chi_{\alpha}\right\} + I_{1}\mathbb{1}\left\{Z_{T} > \chi_{\alpha}\right\}, X_{2} = \max\left\{I_{2}, \frac{1+\beta_{3}}{1-\beta_{2}\beta_{3}}\beta_{1}X_{1}, \beta_{1}X_{1}+\beta_{3}I_{3}\right\} \mathbb{1}\left\{Z_{T} \leq \chi_{\alpha}\right\} + I_{2}\mathbb{1}\left\{Z_{T} > \chi_{\alpha}\right\}.$$

Inserting X_2 into X_1 yields for $Z_T \leq \chi_{\alpha}$

$$X_{1} = \max\left\{I_{1}, \frac{1+\beta_{3}}{1-\beta_{1}\beta_{3}}\beta_{2}I_{2}, \frac{(1+\beta_{3})^{2}}{(1-\beta_{1}\beta_{3})(1-\beta_{2}\beta_{3})}\beta_{1}\beta_{2}X_{1}, \frac{1+\beta_{3}}{1-\beta_{1}\beta_{3}}\beta_{1}\beta_{2}X_{1} + \frac{1+\beta_{3}}{1-\beta_{1}\beta_{3}}\beta_{2}\beta_{3}I_{3}, \beta_{2}I_{2} + \beta_{3}I_{3}, \frac{1+\beta_{3}}{1-\beta_{2}\beta_{3}}\beta_{1}\beta_{2}X_{1} + \beta_{3}I_{3}, \beta_{1}\beta_{2}X_{1} + \beta_{3}I_{3}X_{1} + \beta_{3}I_{3}X_{1} + \beta_{3}I_{3}X_{1} + \beta_{3}I_{3}X_{$$

where we defined

$$m_{1} \coloneqq \max\left\{I_{1}, \frac{1+\beta_{3}}{1-\beta_{1}\beta_{3}}\beta_{2}I_{2}, \beta_{2}I_{2}+\beta_{3}I_{3}\right\},\$$

$$g_{1}(X_{1}) \coloneqq \frac{(1+\beta_{3})^{2}}{(1-\beta_{1}\beta_{3})(1-\beta_{2}\beta_{3})}\beta_{1}\beta_{2}X_{1},\qquad g_{3}(X_{1}) \coloneqq \frac{1+\beta_{3}}{1-\beta_{2}\beta_{3}}\beta_{1}\beta_{2}X_{1}+\beta_{3}I_{3},\$$

$$g_{2}(X_{1}) \coloneqq \frac{1+\beta_{3}}{1-\beta_{1}\beta_{3}}\beta_{1}\beta_{2}X_{1}+\frac{1+\beta_{3}}{1-\beta_{1}\beta_{3}}\beta_{2}\beta_{3}I_{3},\quad g_{4}(X_{1}) \coloneqq \beta_{1}\beta_{2}X_{1}+\beta_{3}(1+\beta_{2})I_{3}.$$

Now we compare $g_2(X_1)$ and $g_3(X_1)$. The constant term of $g_2(X_1)$ is smaller than the constant term of $g_3(X_1)$, since

$$\frac{1+\beta_3}{1-\beta_1\beta_3}\beta_2 \leq 1$$

using the assumption $\beta_i \in (0, 1/2]$ for all $i \in \{1, 2, 3\}$. Moreover, if we assume that $\beta_1 \leq \beta_2$, $g_2(X_1)$ is smaller than $g_3(X_1)$ (pointwise for any $X_1 > 0$) and can therefore be neglected inside the maximum. Note that we could make this conclusion because X_1 is strictly positive due to the above representation as the maximum of I_1 (which is strictly positive by assumption) and some other expressions. Hence, we now need to solve the following one-dimensional fixed point problem

$$X_{1} = \max\left\{m_{1}, \frac{(1+\beta_{3})^{2}}{(1-\beta_{1}\beta_{3})(1-\beta_{2}\beta_{3})}\beta_{1}\beta_{2}X_{1}, \frac{1+\beta_{3}}{1-\beta_{2}\beta_{3}}\beta_{1}\beta_{2}X_{1} + \beta_{3}I_{3}, \beta_{1}\beta_{2}X_{1} + \beta_{3}(1+\beta_{2})I_{3}\right\}$$
$$= \max\left\{m_{1}, g_{1}(X_{1}), g_{3}(X_{1}), g_{4}(X_{1})\right\}.$$
(C.9)

The second argument $g_1(X_1)$ inside the maximum cannot yield the fixed point because the intersection with the identity is located at 0. Thus, we consider $g_3(X_1)$ and $g_4(X_1)$. The value of g_4 at zero is strictly larger than the value of g_3 , while the slope of g_3 is larger than the slope of g_4 . Thus, we need to analyze whether the point of intersection of g_3 and g_4 , denoted by $z_{3,4}$, is smaller or larger than the intersection $z_{4,id}$ of g_4 and the identity function.

Simple calculations imply

$$z_{3,4} = \frac{1 - \beta_2 \beta_3}{\beta_1 (1 + \beta_2)} I_3, \ z_{4,\text{id}} = \frac{\beta_3 (1 + \beta_2)}{1 - \beta_1 \beta_2} I_3.$$

Hence, $z_{4,id} \leq z_{3,4}$ (using the assumption $0 < \beta_i \leq 1/2$ for all $i \in \{1, 2, 3\}$) and therefore, the fixed point is given by the maximum of m_1 and $z_{4,id}$, i.e.,

$$X_{1} = \max\left\{m_{1}, \frac{\beta_{3}(1+\beta_{2})}{1-\beta_{1}\beta_{2}}I_{3}\right\}\mathbb{1}\left\{Z_{T} \le \chi_{\alpha}\right\} + I_{1}\mathbb{1}\left\{Z_{T} > \chi_{\alpha}\right\}.$$
 (C.10)

Now assume that $\beta_1 > \beta_2$. Hence, we cannot omit $g_2(X_1)$ in the maximum in (C.8). Let us now argue that adding $g_2(X_1)$ to the maximum (C.9) does not change the fixed point. In order to do this, we compare the intersection of g_2 and the identity, which is denoted by $z_{2,id}$, with $z_{4,id}$. The intersection of g_2 and the identity is given by

$$z_{2,\mathrm{id}} \coloneqq \frac{\beta_2 \beta_3 (1+\beta_3)}{1-\beta_1 \beta_2 - \beta_1 \beta_3 - \beta_1 \beta_2 \beta_3} I_3$$

A straightforward calculation using $\beta_i \leq 1/2$ for all $i \in \{1, 2, 3\}$ yields $z_{2,id} \leq z_{4,id}$. Hence, the term $g_2(X_1)$ does not influence the fixed point and X_1 from (C.10) is the unique solution to the fixed point problem.

To summarize, the first component of the fixed point for (8.6) takes the form

$$X_1 = \max\left\{I_1, \frac{1+\beta_3}{1-\beta_1\beta_3}\beta_2 I_2, \frac{1+\beta_2}{1-\beta_1\beta_2}\beta_3 I_3, \beta_2 I_2 + \beta_3 I_3\right\}.$$

Now the assertion follows due to the symmetry of the original fixed point problem.

Theorem C.1 shows a very compact representation of the unique Nash equilibrium in the class of wealth profiles of the form (8.6) for three agents. However, it does hide some structural properties of the triple (X_1^*, X_2^*, X_3^*) . Table C.1 shows all possible combinations of the wealth profiles for agents 1, 2, and 3, given that $Z_T \leq \chi_{\alpha}$.

	X_1^*	X_2^*	X_3^*
a)	$I_1(\lambda_1 Z_T)$	$I_2(\lambda_2 Z_T)$	$I_3(\lambda_3 Z_T)$
b)	$\beta_2 I_2(\lambda_2 Z_T) + \beta_3 I_3(\lambda_3 Z_T)$	$I_2(\lambda_2 Z_T)$	$I_3(\lambda_3 Z_T)$
c)	$I_1(\lambda_1 Z_T)$	$\beta_1 I_1(\lambda_1 Z_T) + \beta_3 I_3(\lambda_3 Z_T)$	$I_3(\lambda_3 Z_T)$
d)	$I_1(\lambda_1 Z_T)$	$I_2(\lambda_2 Z_T)$	$\beta_1 I_1(\lambda_1 Z_T) + \beta_2 I_2(\lambda_2 Z_T)$
e)	$I_1(\lambda_1 Z_T)$	$\frac{1+\beta_3}{1-\beta_2\beta_3}\beta_1 I_1(\lambda_1 Z_T)$	$\frac{1+\beta_2}{1-\beta_2\beta_3}\beta_1 I_1(\lambda_1 Z_T)$
f)	$rac{1+eta_3}{1-eta_1eta_3}eta_2 I_2(\lambda_2 Z_T)$	$I_2(\lambda_2 Z_T)$	$rac{1+eta_1}{1-eta_1eta_3}eta_2 I_2(\lambda_2 Z_T)$
g)	$rac{1+eta_2}{1-eta_1eta_2}eta_3I_3(\lambda_3Z_T)$	$rac{1+eta_1}{1-eta_1eta_2}eta_3I_3(\lambda_3Z_T)$	$I_3(\lambda_3 Z_T)$

Table C.1.: Possible values of the triple (X_1^*, X_2^*, X_3^*) from Theorem C.1 for $Z_T \leq \chi_{\alpha}$.

We can divide the behavior of (X_1^*, X_2^*, X_3^*) into three different cases. First, it is possible that all three investors perform equally well in the sense that their terminal wealth in the Nash equilibrium is given by the solution $I_i(\lambda_i Z_T)$, i = 1, 2, 3, to the standard utility maximization problem (case a)). Moreover, it is also possible that two of the three investors perform equally well while the third investor performs worse, so that her wealth in the Nash equilibrium consists of a linear combination of the other two agent's wealth (cases b)-d)). Finally, if one investor outperforms the other two agents, their terminal wealth is a multiple of the classical solution for the "winning" investor (cases e)-g)).

The previous observations become more clearly visible at the end of this section where we consider the special case of logarithmic utility functions. The numerical results presented there also show that the initial capital of the investors mainly determines which of the three cases is present.

Proof (Table C.1). In the following, we prove that Table C.1 contains every possible combination of values of X_1^* , X_2^* , and X_3^* from Theorem C.1. Again, we abbreviate $I_i = I_i(\lambda_i Z_T)$, i = 1, 2, 3.

First, it is obviously possible that $X_1^* = I_1$, $X_2^* = I_2$, and $X_3^* = I_3$ occur simultaneously (case a)). Next, assume that $X_1^* = \beta_2 I_2 + \beta_3 I_3$. Then

$$I_3 \ge \frac{1}{\beta_3} \left(\frac{1+\beta_3}{1-\beta_1\beta_3} \beta_2 I_2 - \beta_2 I_2 \right) = \frac{1+\beta_1}{1-\beta_1\beta_3} \beta_2 I_2, \tag{C.11}$$

$$I_2 \ge \frac{1}{\beta_2} \left(\frac{1+\beta_2}{1-\beta_1\beta_2} \beta_3 I_3 - \beta_3 I_3 \right) = \frac{1+\beta_1}{1-\beta_1\beta_2} \beta_3 I_3, \tag{C.12}$$

since $\beta_2 I_2 + \beta_3 I_3 \ge \frac{1+\beta_3}{1-\beta_1\beta_3}\beta_2 I_2 \lor \frac{1+\beta_2}{1-\beta_1\beta_2}\beta_3 I_3$. Further, we obtain

$$\beta_{1}I_{1} + \beta_{2}I_{2} \leq \beta_{1}(\beta_{2}I_{2} + \beta_{3}I_{3}) + \beta_{2}I_{2} = \beta_{2}(1 + \beta_{1})I_{2} + \beta_{1}\beta_{3}I_{3}$$
$$\leq \beta_{2}(1 + \beta_{1})\frac{1 - \beta_{1}\beta_{3}}{1 + \beta_{1}}\frac{1}{\beta_{2}}I_{3} + \beta_{1}\beta_{3}I_{3} = I_{3}, \qquad (C.13)$$

where we used $\beta_2 I_2 + \beta_3 I_3 \ge I_1$ in the first and (C.11) in the second inequality. Analogously, we can deduce

$$\beta_1 I_1 + \beta_3 I_3 \le I_2 \tag{C.14}$$

using $\beta_2 I_2 + \beta_3 I_3 \ge I_1$ and (C.12). Finally, it follows,

$$I_3 \ge \beta_1 I_1 + \beta_2 I_2 \ge \beta_1 I_1 + \beta_2 (\beta_1 I_1 + \beta_3 I_3) = \beta_1 (1 + \beta_2) I_1 + \beta_2 \beta_3 I_3,$$

using (C.13) in the first and (C.14) in the second inequality. This implies further that

$$I_3 \ge \frac{1+\beta_2}{1-\beta_2\beta_3}\beta_1 I_1.$$

Analogously,

$$I_2 \ge \frac{1+\beta_3}{1-\beta_2\beta_3}\beta_1 I_1.$$

In summary, $X_1^* = \beta_2 I_2 + \beta_3 I_3$ implies $X_2^* = I_2$ and $X_3^* = I_3$.

Using very similar arguments, we can deduce the following implications

$$\begin{aligned} X_2^* &= \beta_1 I_1 + \beta_3 I_3 \implies X_1^* = I_1, \ X_3^* = I_3, \\ X_3^* &= \beta_1 I_1 + \beta_2 I_2 \implies X_1^* = I_1, \ X_2^* = I_2. \end{aligned}$$

So far, we justified lines a)-d) of Table C.1. Now assume that $X_1^* = \frac{1+\beta_3}{1-\beta_1\beta_3}\beta_2 I_2$. Then it follows

$$\beta_1 I_1 + \beta_2 I_2 \le \frac{1+\beta_3}{1-\beta_1\beta_3} \beta_1 \beta_2 I_2 + \beta_2 I_2 = \frac{1+\beta_1}{1-\beta_1\beta_3} \beta_2 I_2,$$

since $I_1 \leq \frac{1+\beta_3}{1-\beta_1\beta_3}\beta_2 I_2$, using the representation of X_1^* in Theorem C.1 and the assumption $X_1^* = \frac{1+\beta_3}{1-\beta_1\beta_3}\beta_2 I_2$. Moreover, we obtain

$$\frac{1+\beta_2}{1-\beta_2\beta_3}\beta_1 I_1 \le \frac{1+\beta_2}{1-\beta_2\beta_3} \cdot \frac{1+\beta_3}{1-\beta_1\beta_3}\beta_1\beta_2 I_2 \le \frac{1+\beta_1}{1-\beta_1\beta_3}\beta_2 I_2.$$

Again, we used Theorem C.1 and the assumption on X_1^* for the first inequality. The second

inequality is valid since

$$\frac{(1+\beta_2)(1+\beta_3)}{1-\beta_2\beta_3}\beta_1 \le 3\beta_1 \le 1+\beta_1,$$

which holds due to the assumption $\beta_j \leq \frac{1}{2}$, j = 1, 2, 3. Finally, $\frac{1+\beta_3}{1-\beta_1\beta_3}\beta_2I_2 \geq \beta_2I_2 + \beta_3I_3$ implies

$$I_3 \le \frac{1}{\beta_3} \left(\frac{1+\beta_3}{1-\beta_1\beta_3} \beta_2 I_2 - \beta_2 I_2 \right) = \frac{1+\beta_1}{1-\beta_1\beta_3} \beta_2 I_2.$$
(C.15)

Thus, $X_3^* = \frac{1+\beta_1}{1-\beta_1\beta_3}\beta_2 I_2$ holds. For X_2^* , we obtain

$$I_2 \ge \underbrace{\frac{1+\beta_1}{1-\beta_1\beta_3}\beta_2}_{\le 1} I_2 \ge I_3 \ge \underbrace{\frac{1+\beta_1}{1-\beta_1\beta_2}\beta_3}_{\le 1} I_3$$

from the assumption $\beta_j \leq 1/2$ for all $j \in \{1, 2, 3\}$, and the previous representation of X_3^* . Analogously,

$$I_2 \ge \frac{1+\beta_3}{1-\beta_2\beta_3}\beta_1 I_1.$$

Finally, using (C.15) and $I_1 \leq \frac{1+\beta_3}{1-\beta_1\beta_3}\beta_2 I_2 = X_1^*$,

$$I_{2} \ge \beta_{1}I_{2} + \beta_{3}I_{2} \ge \beta_{1} \cdot \underbrace{\frac{1 - \beta_{1}\beta_{3}}{1 + \beta_{3}}\frac{1}{\beta_{2}}}_{>1}I_{1} + \beta_{3} \cdot \underbrace{\frac{1 - \beta_{1}\beta_{3}}{1 + \beta_{1}}\frac{1}{\beta_{2}}}_{>1}I_{3} \ge \beta_{1}I_{1} + \beta_{3}I_{3}$$

Therefore, it follows that $X_2^* = I_2$.

Using similar arguments, we can also prove the following implications

$$\begin{split} X_1^* &= \frac{1+\beta_2}{1-\beta_1\beta_2}\beta_3 I_3 \implies X_2^* = \frac{1+\beta_1}{1-\beta_1\beta_2}\beta_3 I_3, X_3^* = I_3, \\ X_2^* &= \frac{1+\beta_3}{1-\beta_2\beta_3}\beta_1 I_1 \implies X_1^* = I_1, X_3^* = \frac{1+\beta_2}{1-\beta_2\beta_3}\beta_1 I_1, \\ X_2^* &= \frac{1+\beta_1}{1-\beta_1\beta_2}\beta_3 I_3 \implies X_1^* = \frac{1+\beta_2}{1-\beta_1\beta_2}\beta_3 I_3, X_3^* = I_3, \\ X_3^* &= \frac{1+\beta_2}{1-\beta_2\beta_3}\beta_1 I_1 \implies X_1^* = I_1, X_2^* = \frac{1+\beta_3}{1-\beta_2\beta_3}\beta_1 I_1, \\ X_3^* &= \frac{1+\beta_1}{1-\beta_1\beta_3}\beta_2 I_2 \implies X_1^* = \frac{1+\beta_3}{1-\beta_1\beta_3}\beta_2 I_2, X_2^* = I_2. \end{split}$$

Hence, lines e)-g) in Table C.1 are valid as well. The line of arguments also implies that lines a)-g) of Table C.1 contain all possible combinations of X_1^* , X_2^* , and X_3^* .

EXAMPLE: LOGARITHMIC UTILITY

If all investors use the natural logarithm as their utility function, we can determine the unique Lagrange multipliers λ_i , i = 1, 2, 3, mentioned in Theorem C.1 explicitly. Using the cases a)-g) introduced in Table C.1, we obtain the following values of X_i^* , i = 1, 2, 3.

Proposition C.3. Let n = 3 and assume that $U_i(x) = \log(x)$, x > 0, i = 1, 2, 3. Further, let $\alpha_i = \alpha \in [0, 1]$ and $\beta_i \in (0, 1/2]$, i = 1, 2, 3, and let

$$x_0^i > \alpha \max\left\{\beta_j x_0^j + \beta_k x_0^k, \, \frac{1+\beta_j}{1-\beta_i\beta_j} x_0^k, \, \frac{1+\beta_k}{1-\beta_i\beta_k}\beta_j x_0^j\right\}$$

for all pairwise distinct $i, j, k \in \{1, 2, 3\}$. Then the triples (X_1^*, X_2^*, X_3^*) displayed in Table C.1 take the following form

a) $X_i^* = \frac{x_0^i}{Z_T}, i = 1, 2, 3,$

 $b)-d) \ X_i^* = \frac{\beta_j x_0^j + \beta_k x_0^k}{Z_T} \mathbb{1} \left\{ Z_T \le \chi_\alpha \right\} + \frac{x_0^i - \alpha(\beta_j x_0^j + \beta_k x_0^k)}{(1-\alpha)Z_T} \mathbb{1} \left\{ Z_T > \chi_\alpha \right\}, \ X_j^* = \frac{x_0^j}{Z_T}, \ X_k^* = \frac{x_0^k}{Z_T} \text{ for pairwise distinct } i, j, k \in \{1, 2, 3\},$

$$\begin{aligned} e)-g) \ X_i^* &= \frac{1+\beta_j}{1-\beta_i\beta_j} \beta_k x_0^k Z_T^{-1} \mathbb{1}\left\{Z_T \le \chi_\alpha\right\} + \frac{1}{1-\alpha} \left(x_0^i - \alpha \frac{1+\beta_j}{1-\beta_i\beta_j} \beta_k x_0^k\right) Z_T^{-1} \mathbb{1}\left\{Z_T > \chi_\alpha\right\}, \\ X_j^* &= \frac{1+\beta_i}{1-\beta_i\beta_j} \beta_k x_0^k Z_T^{-1} \mathbb{1}\left\{Z_T \le \chi_\alpha\right\} + \frac{1}{1-\alpha} \left(x_0^j - \alpha \frac{1+\beta_i}{1-\beta_i\beta_j} \beta_k x_0^k\right) Z_T^{-1} \mathbb{1}\left\{Z_T > \chi_\alpha\right\}, \\ X_k &= \frac{x_0^k}{Z_T} \text{ for pairwise distinct } i, j, k \in \{1, 2, 3\}. \end{aligned}$$

Proof. Since $U_i(x) = \log(x)$, $I_i(x) = x^{-1}$ holds for all $i \in \{1, 2, 3\}$ and x > 0. In the following, we determine the Lagrange multipliers λ_i , i = 1, 2, 3, in the cases a)-g) displayed in Table C.1. As already stated in Theorem C.1, λ_i is determined via the budget constraint $\mathbb{E}[X_i^*Z_T] = x_0^i$, i = 1, 2, 3.

a) In case a), the wealth profile of agent *i* is given by $X_i^* = (\lambda_i Z_T)^{-1}$ for all $i \in \{1, 2, 3\}$, i.e., by the solution to the classical problem without the VaR-constraint. Hence,

$$\mathbb{E}[X_i^* Z_T] = \frac{1}{\lambda_i} \stackrel{!}{=} x_0^i$$

and thus, $\lambda_i = (x_0^i)^{-1}$, i = 1, 2, 3.

b) In case b), we obtain $X_1^* = (\frac{\beta_2}{\lambda_2 Z_T} + \frac{\beta_3}{\lambda_3 Z_T}) \mathbb{1} \{Z_T \le \chi_\alpha\} + (\lambda_1 Z_T)^{-1} \mathbb{1} \{Z_T > \chi_\alpha\}, X_2^* = (\lambda_2 Z_T)^{-1}, X_3^* = (\lambda_3 Z_T)^{-1}$. Analogously to a), we obtain $\lambda_2 = (x_0^2)^{-1}, \lambda_3 = (x_0^3)^{-1}$. Hence, it follows

$$\mathbb{E}[X_1^* Z_T] = (\beta_2 x_0^2 + \beta_3 x_0^3) \mathbb{E}[\mathbb{1} \{ Z_T \le \chi_\alpha \}] + \frac{1}{\lambda_1} \mathbb{E}[\mathbb{1} \{ Z_T > \chi_\alpha \}]$$
$$= \alpha (\beta_2 x_0^2 + \beta_3 x_0^3) + \frac{1 - \alpha}{\lambda_1} = x_0^1,$$

which implies $\lambda_1 = \frac{1-\alpha}{x_0^1 - \alpha(\beta_2 x_0^2 + \beta_3 x_0^3)}$. Note that we used that χ_{α} denotes the α -quantile of Z_T .

- c) Analogously to case b), we obtain $\lambda_1 = (x_0^1)^{-1}$, $\lambda_2 = \frac{1-\alpha}{x_0^2 \alpha(\beta_1 x_0^1 + \beta_3 x_0^3)}$, and $\lambda_3 = (x_0^3)^{-1}$.
- d) Analogously to case b), we obtain $\lambda_1 = (x_0^1)^{-1}$, $\lambda_2 = (x_0^2)^{-1}$, and $\lambda_3 = \frac{1-\alpha}{x_0^3 \alpha(\beta_1 x_0^1 + \beta_2 x_0^2)}$.
- e) In case e), we have $X_1^* = (\lambda_1 Z_T)^{-1}$, so that $\lambda_1 = (x_0^1)^{-1}$ follows. Moreover, X_2^* and X_3^* are given by

$$X_{2}^{*} = \frac{1+\beta_{3}}{1-\beta_{2}\beta_{3}} \frac{\beta_{1}x_{0}^{1}}{Z_{T}} \mathbb{1}\left\{Z_{T} \le \chi_{\alpha}\right\} + \frac{1}{\lambda_{2}Z_{T}} \mathbb{1}\left\{Z_{T} > \chi_{\alpha}\right\},$$

$$X_{3}^{*} = \frac{1+\beta_{2}}{1-\beta_{2}\beta_{3}} \frac{\beta_{1}x_{0}^{1}}{Z_{T}} \mathbb{1}\left\{Z_{T} \le \chi_{\alpha}\right\} + \frac{1}{\lambda_{3}Z_{T}} \mathbb{1}\left\{Z_{T} > \chi_{\alpha}\right\},$$

and therefore

$$\mathbb{E}[X_2^* Z_T] = \alpha \frac{1+\beta_3}{1-\beta_2\beta_3} \beta_1 x_0^1 + \frac{1-\alpha}{\lambda_2} \stackrel{!}{=} x_0^2 x_0^2 x_0^2 \\ \mathbb{E}[X_3^* Z_T] = \alpha \frac{1+\beta_2}{1-\beta_2\beta_3} \beta_1 x_0^1 + \frac{1-\alpha}{\lambda_3} \stackrel{!}{=} x_0^3 x_0^2 x_0^2 \\ \mathbb{E}[X_3^* Z_T] = \alpha \frac{1+\beta_2}{1-\beta_2\beta_3} \beta_1 x_0^1 + \frac{1-\alpha}{\lambda_3} \stackrel{!}{=} x_0^3 x_0^2 x_0^2 x_0^2 \\ \mathbb{E}[X_3^* Z_T] = \alpha \frac{1+\beta_2}{1-\beta_2\beta_3} \beta_1 x_0^1 + \frac{1-\alpha}{\lambda_3} \stackrel{!}{=} x_0^3 x_0^2 x_$$

which imply that

$$\lambda_2 = \frac{1 - \alpha}{x_0^2 - \alpha \frac{1 + \beta_3}{1 - \beta_2 \beta_3} \beta_1 x_0^1}, \lambda_3 = \frac{1 - \alpha}{x_0^3 - \alpha \frac{1 + \beta_2}{1 - \beta_2 \beta_3} \beta_1 x_0^1}$$

f) Analogously to case e), we obtain $\lambda_2 = (x_0^2)^{-1}$ and

$$\lambda_1 = \frac{1 - \alpha}{x_0^1 - \alpha \frac{1 + \beta_3}{1 - \beta_1 \beta_3} \beta_2 x_0^2}, \ \lambda_3 = \frac{1 - \alpha}{x_0^3 - \alpha \frac{1 + \beta_1}{1 - \beta_1 \beta_3} \beta_2 x_0^2}$$

g) Analogously to case e), we obtain $\lambda_3 = (x_0^3)^{-1}$ and

$$\lambda_1 = \frac{1 - \alpha}{x_0^1 - \alpha \frac{1 + \beta_2}{1 - \beta_1 \beta_2} \beta_3 x_0^3}, \ \lambda_2 = \frac{1 - \alpha}{x_0^2 - \alpha \frac{1 + \beta_1}{1 - \beta_1 \beta_2} \beta_3 x_0^3}$$

Finally, inserting λ_i^* into X_i^* from Table C.1 for all $i \in \{1, 2, 3\}$ yields the asserted representation of the triple (X_1^*, X_2^*, X_3^*) .

Illustration of Proposition C.3

The following figures show an application of Proposition C.3. The displayed results are based on a classical Black-Scholes market with one stock, drift $\mu = 0.03$, and volatility $\sigma = 0.2$, and one riskless bond with zero interest rate. Moreover, we chose $\alpha = 0.5$ and T = 4 (similar to the setting used in Section 8.3).

Figures C.3.1 - C.3.3 display X_i^* , i = 1, 2, 3, for varying values of the state price density Z_T . Figure C.3.1 shows case a) of Table C.1 in which the parameters of the problem are chosen so that the terminal wealth in the Nash equilibrium is the classical solution for all three agents. In Figure C.3.2, the parameters are chosen so that case b) is present. The figure displays, as expected, that agents 2 and 3 use the classical solution while agent 1 uses a linear combination of X_2^* and X_3^* for small values of Z_T (i.e., $Z_T \leq \chi_{\alpha}$) and the "classical solution" $I_1(\lambda_1 Z_T)$ for large values of Z_T . Again, we have to be careful with the term "classical solution" since the Lagrange multiplier λ_1 is not the same as in the classical problem due to the discontinuity at χ_{α} . The last of the three figures, Figure C.3.3, shows case e) in which the parameters of the problem are chosen so that agent 1 uses the classical solution while the other two agents use a multiple of the first agents optimal wealth for small values of Z_T and their "classical solution" $I_i(\lambda_i Z_T)$, i = 2, 3, for large values of Z_T . Figure C.3.4 shows the behavior of X_1^* in terms of Z_T for different choices of β_2 , while β_1 and β_3 are fixed. For the smallest value of β_2 , i.e., $\beta_2 = 0.2$, case a) is present and we can observe that X_1^* is simply given by the solution to the classical problem. For the other three values of β_2 , case f) of Table C.1 is present, which explains the discontinuity of X_1^* at $Z_T = \chi_{\alpha}$. As the value of β_2 increases, we can observe that the value of X_1^* increases for small values of Z_T ($Z_T \leq \chi_{\alpha}$) and decreases for large values of Z_T ($Z_T > \chi_{\alpha}$).



Figure C.3.1.: X_i^* , i = 1, 2, 3, from Proposition C.3 in terms of Z_T for $\alpha = 0.5$, $\beta_1 = 0.4$, $\beta_2 = 0.3$, $\beta_3 = 0.4$, $x_0^1 = 3$, $x_0^2 = 2.5$, $x_0^3 = 3$. The parameter choice implies that case a) of Table C.1 is present. The market parameters are d = 1, r = 0, $\mu = 0.03$, $\sigma = 0.2$, T = 4.



Figure C.3.2.: X_i^* , i = 1, 2, 3, from Proposition C.3 in terms of Z_T for $\alpha = 0.5$, $\beta_1 = 0.2$, $\beta_2 = 0.2$, $\beta_3 = 0.4$, $x_0^1 = 1$, $x_0^2 = 4$, $x_0^3 = 5$. The parameter choice implies that case b) of Table C.1 is present. The market parameters are d = 1, r = 0, $\mu = 0.03$, $\sigma = 0.2$, T = 4.



Figure C.3.3.: X_i^* , i = 1, 2, 3, from Proposition C.3 in terms of Z_T for $\alpha = 0.5$, $\beta_1 = 0.4$, $\beta_2 = 0.2$, $\beta_3 = 0.4$, $x_0^1 = 5$, $x_0^2 = 1$, $x_0^3 = 2$. The parameter choice implies that case e) of Table C.1 is present. The market parameters are d = 1, r = 0, $\mu = 0.03$, $\sigma = 0.2$, T = 4.



Figure C.3.4.: X_1^* from Proposition C.3 in terms of Z_T , for different choices of β_2 , while $\alpha = 0.5$, $\beta_1 = 0.3$, $\beta_3 = 0.5$, $x_0^1 = 4$, $x_0^2 = 8$, $x_0^3 = 3$. For $\beta_2 = 0.2$, case a) from Table C.1 is present; the other parameters imply that case f) is present. The market parameters are d = 1, r = 0, $\mu = 0.03$, $\sigma = 0.2$, T = 4.

C.4. PROOFS OF LEMMAS FROM SECTION 8.4

In the following, we provide the proofs of Lemmas 8.12 - 8.14, as well as Lemma 8.18, that were left out of Section 8.4. The lemmas are used to find and characterize the unique fixed point for (8.6) for general $n \in \mathbb{N}$.
Proof (Lemma 8.12). Let $L \in \{1, \ldots, n\}$ and $(j_1, \ldots, j_L) \in {\binom{[n]}{L}}$. Using $d_L(j_1, \ldots, j_L)$ defined in (8.13) and the Binomial theorem (Harris et al., 2008, p. 139) yields

$$d_{L}(j_{1},...,j_{L}) = 1 - \sum_{k=2}^{L} (k-1) \sum_{(i_{1},...,i_{k}) \in \binom{[L]}{k}} \prod_{\ell=1}^{k} \beta_{j_{i_{\ell}}}$$

$$\geq 1 - \sum_{k=2}^{L} (k-1) \sum_{(i_{1},...,i_{k}) \in \binom{[L]}{k}} \left(\frac{1}{n-1}\right)^{k} \qquad (C.16)$$

$$= 1 - \sum_{k=2}^{L} k \binom{L}{k} \left(\frac{1}{n-1}\right)^{k} + \sum_{k=2}^{L} \binom{L}{k} \left(\frac{1}{n-1}\right)^{k}$$

$$= 1 - \frac{L}{n-1} \sum_{k=1}^{L-1} \binom{L-1}{k} \left(\frac{1}{n-1}\right)^{k} + \sum_{k=2}^{L} \binom{L}{k} \left(\frac{1}{n-1}\right)^{k}$$

$$= 1 - \frac{L}{n-1} \left(1 + \frac{1}{n-1}\right)^{L-1} + \frac{L}{n-1} + \left(1 + \frac{1}{n-1}\right)^{L} - 1 - \frac{L}{n-1}$$

$$= \left(\frac{n}{n-1}\right)^{L-1} \frac{n-L}{n-1}. \qquad (C.17)$$

If L = n and $\beta_j = \frac{1}{n-1}$ for all $j \in \{1, \ldots, n\}$, we obtain equality in (C.16) and the expression in the last line (C.17) is equal to 0. Otherwise, we obtain either a strict inequality in (C.16) or a strictly positive lower bound in (C.17). This concludes our proof.

Proof (Lemma 8.13). First, let L = n and $\beta_j = \frac{1}{n-1}$ for all $j \in \{1, \ldots, n\}$. Then the entries on the diagonal of $A_n(1, \ldots, n)$ are equal to 1 while all other entries are equal to $-\frac{1}{n-1}$. Thus, the rows of $A_n(1, \ldots, n)$ sum up to the *n*-dimensional vector $\mathbf{0}_n$ of zeros and $A_n(1, \ldots, n)$ is not regular. Now assume that either $L \in \{1, \ldots, n-1\}$, or L = n and $\prod_{j=1}^n \beta_j < \left(\frac{1}{n-1}\right)^n$. Let $(j_1, \ldots, j_L) \in {[n] \choose L}$ be arbitrary but fixed. Throughout this proof, we denote $A_L(j_1, \ldots, j_L)$ by A and the claimed inverse by A^{-1} . Moreover, we set $B \coloneqq A^{-1}A$ and $d_L \coloneqq d_L(j_1, \ldots, j_L)$. Note that Lemma 8.12 implies $d_L > 0$. We show that A^{-1} is the left inverse of A. The proof that it is also the right inverse of A proceeds analogously. First, we consider the elements on the diagonal of B. Let $i \in \{1, \ldots, L\}$. Then

$$\begin{split} B[i,i] &= \sum_{k=1}^{L} A^{-1}[i,k] A[k,i] \\ &= \frac{1}{d_L} \left(1 - \sum_{k=2}^{L-1} (k-1) \sum_{\substack{(i_1,\dots,i_k) \in \left(\{j_1,\dots,j_L\} \setminus \{j_i\} \right) \\ (i_1,\dots,i_k) \in \left(\{j_1,\dots,j_L\} \setminus \{j_i\} \right)}} \prod_{\ell=1}^{k} \beta_{i_\ell} + \sum_{\substack{k=1 \\ k \neq i}}^{L} \beta_{j_k} \prod_{\substack{\ell=1 \\ \ell \neq i,k}}^{L} (1 + \beta_{j_\ell}) \cdot (-\beta_{j_i}) \right) \\ &= \frac{1}{d_L} \left(1 - \sum_{k=2}^{L-1} (k-1) \sum_{\substack{(i_1,\dots,i_k) \in \left(\{j_1,\dots,j_L\} \setminus \{j_i\} \right) \\ (i_1,\dots,i_k) \in \left(\{j_1,\dots,j_L\} \setminus \{j_i,j_k\} \right)}} \prod_{\ell=1}^{k} \beta_{i_\ell} \right) \\ &- \beta_{j_i} \sum_{\substack{k=1 \\ k \neq i}}^{L} \beta_{j_k} \sum_{m=0}^{L-2} \sum_{\substack{(i_1,\dots,i_m) \in \left(\{j_1,\dots,j_L\} \setminus \{j_i,j_k\} \right) \\ m}} \prod_{\ell=1}^{m} \beta_{i_\ell}} \beta_{i_\ell} \end{split}$$

$$\begin{split} &= \frac{1}{d_L} \left(1 - \sum_{k=2}^{L-1} (k-1) \sum_{\substack{(i_1, \dots, i_k) \in \binom{\{j_1, \dots, j_L\} \setminus \{j_i\}}{k}} \prod_{\ell=1}^k \beta_{i_\ell} \\ &\quad -\beta_{j_i} \sum_{m=2}^L (m-1) \sum_{\substack{(i_1, \dots, i_{m-1}) \in \binom{\{j_1, \dots, j_L\} \setminus \{j_i\}}{m-1} \prod_{\ell=1}^m \beta_{i_\ell}} \prod_{\ell=1}^{m-1} \beta_{i_\ell} \right) \\ &= \frac{1}{d_L} \left(1 - \sum_{k=2}^L (k-1) \sum_{\substack{(i_1, \dots, i_k) \in \binom{\{j_1, \dots, j_L\}}{k} \prod_{\ell=1}^m \beta_{\ell}} \prod_{\ell=1}^k \beta_{i_\ell} - \sum_{m=2}^L (m-1) \sum_{\substack{(i_1, \dots, i_m) \in \binom{\{j_1, \dots, j_L\}}{m} \prod_{\ell=1}^m \beta_{i_\ell}} \prod_{\ell=1}^m \beta_{i_\ell} \right) \\ &= \frac{1}{d_L} \left(1 - \sum_{k=2}^L (k-1) \sum_{\substack{(i_1, \dots, i_k) \in \binom{\{j_1, \dots, j_L\}}{k} \prod_{\ell=1}^m \beta_{\ell}} \prod_{\ell=1}^k \beta_{i_\ell} \right) = 1. \end{split}$$

Now let $i, k \in \{1, \ldots, L\}, i \neq k$. Then

$$\begin{split} B[i,k] &= \sum_{p=1}^{L} A^{-1}[i,p] \cdot A[p,k] \\ &= \frac{1}{d_{L}} \bigg(\bigg(1 - \sum_{m=2}^{L-1} (m-1) \sum_{\substack{(i_{1},...,i_{m}) \in \binom{(j_{1},...,j_{m}}{m} \setminus (j_{i})}} \prod_{\ell=1}^{m} \beta_{i_{\ell}} \bigg) \cdot (-\beta_{j_{k}}) + \beta_{j_{k}} \prod_{\ell=1}^{L} (1+\beta_{j_{\ell}}) \cdot 1 \\ &+ \sum_{\substack{p=1\\p \neq i,k}}^{L} \beta_{j_{p}} \prod_{\ell=1}^{L} (1+\beta_{j_{p}}) \cdot (-\beta_{j_{k}}) \bigg) \\ &= -\frac{\beta_{j_{k}}}{d_{L}} \bigg(1 - \sum_{m=2}^{L-1} (m-1) \sum_{\substack{(i_{1},...,i_{m}) \in \binom{(j_{1},...,j_{L}}{m} \setminus (j_{i})}} \prod_{\ell=1}^{m} \beta_{i_{\ell}} - \sum_{m=2}^{L-1} (m-1) \sum_{\substack{(i_{1},...,i_{m}) \in \binom{(j_{1},...,j_{L}}{m} \setminus (j_{i},j_{p})}} \prod_{\ell=1}^{m} \beta_{i_{\ell}} - \sum_{m=2}^{L-1} (m-1) \sum_{\substack{(i_{1},...,i_{m}) \in \binom{(j_{1},...,j_{L}}{m} \setminus (j_{i},j_{p})}} \prod_{\ell=1}^{m} \beta_{i_{\ell}} - \sum_{m=2}^{L-1} (m-1) \sum_{\substack{(i_{1},...,i_{m}) \in \binom{(j_{1},...,j_{L}}{m} \setminus (j_{i},j_{p})}} \prod_{\ell=1}^{m} \beta_{i_{\ell}} - \sum_{m=2}^{L-1} (m-1) \sum_{\substack{(i_{1},...,i_{m}) \in \binom{(j_{1},...,j_{L}}{m} \setminus (j_{i},j_{p})}} \prod_{\ell=1}^{m} \beta_{i_{\ell}} \bigg) \\ &= -\frac{\beta_{j_{k}}}}{d_{L}} \bigg(\sum_{\substack{p=1\\p \neq i,k}} \beta_{j_{p}} \sum_{m=0}^{L-2} \sum_{\substack{(i_{1},...,i_{m}) \in \binom{(j_{1},...,j_{L}}{m} \setminus (j_{i},j_{p})}} \prod_{\ell=1}^{m} \beta_{i_{\ell}} - \sum_{m=2}^{L-1} (m-1) \sum_{\substack{(i_{1},...,i_{m}}) \in \binom{(j_{1},...,j_{L}}{m} \setminus (j_{i},j_{p})}} \prod_{\ell=1}^{m} \beta_{i_{\ell}} \bigg) \\ &= -\frac{\beta_{j_{k}}}}{d_{L}} \bigg(\sum_{\substack{p=1\\p \neq i,k}} \beta_{j_{p}} \sum_{m=0}^{L-2} \sum_{\substack{(i_{1},...,i_{m}) \in \binom{(j_{1},...,j_{L}}{m} \setminus (j_{i},j_{p})}} \prod_{\ell=1}^{m} \beta_{i_{\ell}} + \sum_{m=0}^{L-2} \beta_{j_{k}} \sum_{\substack{(i_{1},...,i_{m}) \in \binom{(j_{1},...,j_{L}}{m} \setminus (j_{i},j_{p})}} \prod_{\ell=1}^{m} \beta_{i_{\ell}} \bigg) \\ &= -\frac{\beta_{j_{k}}}}{d_{L}} \bigg(\sum_{p=1}^{L-1} \beta_{j_{p}} \sum_{m=0}^{L-2} \sum_{\substack{(i_{1},...,i_{m}) \in \binom{(j_{1},...,j_{L}}{m} \setminus (j_{i},j_{p})}} \prod_{\ell=1}^{m} \beta_{i_{\ell}} - \sum_{m=1}^{L-1} m \sum_{\substack{(i_{1},...,i_{m}) \in \binom{(j_{1},...,j_{L}}{m} \setminus (j_{i},j_{p})}} \prod_{\ell=1}^{m} \beta_{i_{\ell}} - \sum_{m=1}^{L-1} m \sum_{\substack{(i_{1},...,i_{m}) \in \binom{(j_{1},...,j_{L}}{m} \setminus (j_{i},j_{p})} \prod_{\ell=1}^{m} \beta_{i_{\ell}}} \bigg) \\ &= -\frac{\beta_{j_{k}}}}{d_{L}} \bigg(\sum_{m=1}^{L-1} m \sum_{\substack{(i_{1},...,i_{m}) \in \binom{(j_{1},...,j_{L}}{m} \setminus (j_{i},j_{p})} \prod_{\ell=1}^{m} \beta_{i_{\ell}} - \sum_{m=1}^{L-1} m \sum_{\substack{(i_{1},...,i_{m}) \in \binom{(j_{1},...,j_{L}}{m} \setminus (j_{i},j_{p})} \prod_{\ell=1}^{m$$

Hence, B equals the identity matrix in $\mathbb{R}^{L \times L}$.

Proof (Lemma 8.14). Let $L \in \{1, \ldots, n-1\}$ or L = n and $\prod_{j=1}^{n} \beta_j < \left(\frac{1}{n-1}\right)^n$. Further, let $(j_1, \ldots, j_L) \in \binom{[n]}{L}$ be arbitrary but fixed. Throughout this proof, we denote $A_L(j_1, \ldots, j_L)^{-1}$ by A^{-1} and $d_L(j_1, \ldots, j_L)$ by d_L . Let $i \in \{1, \ldots, L\}$. Then, using Lemma 8.13,

$$\begin{split} d_{L} \cdot \sum_{k=1}^{L} A^{-1}[i,k] &= 1 - \sum_{m=2}^{L-1} (m-1) \sum_{\substack{(i_{1},\dots,i_{m}) \in \left(\{j_{1},\dots,j_{L}\} \setminus \{j_{i}\}\right) \\ (i_{1},\dots,i_{m}) \in \left(\{j_{1},\dots,j_{L}\} \setminus \{j_{i}\}\right)} \sum_{\ell=1}^{m} \beta_{i_{\ell}} + \sum_{\substack{k=1 \\ k \neq i}}^{L} \beta_{j_{k}} \sum_{m=0}^{L-2} \sum_{\substack{(i_{1},\dots,i_{m}) \in \left(\{j_{1},\dots,j_{L}\} \setminus \{j_{i}\}\right) \\ (i_{1},\dots,i_{m}) \in \left(\{j_{1},\dots,j_{L}\} \setminus \{j_{i}\}\right)} \prod_{\ell=1}^{m} \beta_{i_{\ell}} + \sum_{\substack{k=1 \\ k \neq i}}^{L} \beta_{j_{k}} \sum_{m=0}^{L-2} \sum_{\substack{(i_{1},\dots,i_{m}) \in \left(\{j_{1},\dots,j_{L}\} \setminus \{j_{i}\}\right) \\ (i_{1},\dots,i_{m}) \in \left(\{j_{1},\dots,j_{L}\} \setminus \{j_{i}\}\right)} \prod_{\ell=1}^{m} \beta_{i_{\ell}} + \sum_{m=1}^{L-1} m \sum_{\substack{(i_{1},\dots,i_{m}) \in \left(\{j_{1},\dots,j_{L}\} \setminus \{j_{i}\}\right) \\ (i_{1},\dots,i_{m}) \in \left(\{j_{1},\dots,j_{L}\} \setminus \{j_{i}\}\right)} \prod_{\ell=1}^{m} \beta_{i_{\ell}} + \sum_{m=1}^{L-1} m \sum_{\substack{(i_{1},\dots,i_{m}) \in \left(\{j_{1},\dots,j_{L}\} \setminus \{j_{i}\}\right) \\ m} \beta_{i_{\ell}} = 1 + \sum_{m=1}^{L-1} \sum_{\substack{(i_{1},\dots,i_{m}) \in \left(\{j_{1},\dots,j_{L}\} \setminus \{j_{i}\}\right) \\ m} \prod_{\ell=1}^{m} \beta_{i_{\ell}}} \prod_{\ell=1}^{m} \beta_{i_{\ell}} + \sum_{m=1}^{L-1} m \sum_{\substack{(i_{1},\dots,i_{m}) \in \left(\{j_{1},\dots,j_{L}\} \setminus \{j_{i}\}\right) \\ m} \beta_{i_{\ell}} = \prod_{\substack{\ell=1 \\ \ell \neq i}}^{L} (1 + \beta_{j_{\ell}}). \end{split}$$

Proof (Lemma 8.18). Let $L \in \{1, ..., n-1\}$ and $(j_1, ..., j_{L+1}) \in {[n] \choose L+1}$. Then, by definition of $d_{L+1}(j_1, ..., j_{L+1})$,

$$\begin{split} d_{L+1}(j_1, \dots, j_{L+1}) &= 1 - \sum_{k=2}^{L+1} (k-1) \sum_{(i_1, \dots, i_k) \in \binom{[L+1]}{k}} \prod_{\ell=1}^{k} \beta_{j_{i_\ell}} \\ &= 1 - \beta_{j_{L+1}} \sum_{k=2}^{L+1} (k-1) \sum_{(i_1, \dots, i_{k-1}) \in \binom{[L]}{k-1}} \prod_{\ell=1}^{k-1} \beta_{j_{i_\ell}} - \sum_{k=2}^{L} (k-1) \sum_{(i_1, \dots, i_k) \in \binom{[L]}{k}} \prod_{\ell=1}^{k} \beta_{j_{i_\ell}} \\ &= d_L(j_1, \dots, j_L) - \beta_{j_{L+1}} \sum_{k=2}^{L} (k-1) \sum_{(i_1, \dots, i_k) \in \binom{[L]}{k-1}} \prod_{\ell=1}^{k-1} \beta_{j_{i_\ell}} \\ &= d_L(j_1, \dots, j_L) - \beta_{j_{L+1}} \sum_{k=2}^{L} k \sum_{(i_1, \dots, i_k) \in \binom{[L]}{k}} \prod_{\ell=1}^{k} \beta_{j_{i_\ell}} \\ &= d_L(j_1, \dots, j_L) - \beta_{j_{L+1}} \sum_{k=2}^{L} (k-1) \sum_{(i_1, \dots, i_k) \in \binom{[L]}{k}} \prod_{\ell=1}^{k} \beta_{j_{i_\ell}} - \beta_{j_{L+1}} \sum_{k=1}^{L} \sum_{(i_1, \dots, i_k) \in \binom{[L]}{k}} \prod_{\ell=1}^{k} \beta_{j_{i_\ell}} \\ &= d_L(j_1, \dots, j_L) - \beta_{j_{L+1}} \sum_{k=2}^{L} (k-1) \sum_{(i_1, \dots, i_k) \in \binom{[L]}{k}} \prod_{\ell=1}^{k} \beta_{j_{i_\ell}} - \beta_{j_{L+1}} \sum_{k=1}^{L} \sum_{(i_1, \dots, i_k) \in \binom{[L]}{k}} \prod_{\ell=1}^{k} \beta_{j_{\ell_\ell}} \\ &= (1 + \beta_{j_{L+1}}) d_L(j_1, \dots, j_L) - \beta_{j_{L+1}} \prod_{\ell=1}^{L} (1 + \beta_{j_k}). \end{split}$$